United States Bankruptcy Court Northern District of Texas		PROOF OF CLAIM
Name of Debtor: Opus West Corporation	Case Number 09-34356	R-hdh11
NOTE: This form should not be used to make a claim for an administrative expense arising after the commencement	of the case. A re	equest for payment of an
Name of Creditor (the person or other entity to whom the debtor owes money or property): Bank of America, N.A., as agent for itself and the other Lenders	. Check thi	is box to indicate that this
Name and address where notices should be sent:		ends a previously filed
Thompson & Knight LLP Bank of America, N.A.		
c/o John S. Brannon c/o Casey Carpenter 1722 Routh St., Suite 1500 Vice-President, Real Estate Special Assets	Court Clain	Number:
Dallas, Texas 75201 333 S. Hope St., 11th Floor Felephone number: Los Angeles, California 90071-1406	(1, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7, 7,	
(214) 969-1700	Filed on:	
Name and address where payment should be sent (if different from above):	11 Check this	s box if you are aware that
Bank of America, N.A.	anyone els	se has filed a proof of clain
c/o Casey Carpenter Vice-President, Real Estate Special Assets	relating to	your claim. Attach copy of giving particulars.
333 S. Hope St., 11th Floor		
(213) 621-3604		s box if you are the debtor in this case.
. Amount of Claim as of Date Case Filed: \$_106,807,380.87	5. Amount o	of Claim Entitled to
If all or part of your claim is secured, complete item 4 below; however, if all of your claim is unsecured, do not complete item 4.	any porti	mder 11 U.S.C. §507(a). I ion of your claim falls in
	one of the	e following categories,
fall or part of your claim is entitled to priority, complete item 5.	amount.	box and state the
Check this box if claim includes interest or other charges in addition to the principal amount of claim. Attach itemized statement of interest or charges.	Specify the pr	riority of the claim.
. Basis for Claim: *See Addendum	Domestic support obligations under 11 U.S.C. §507(a)(1)(A) or (a)(1)(B).	
(See instruction #2 on reverse side.)		
Last four digits of any number by which creditor identifies debtor:	10 Wages, sal	laries, or commissions (up 0*) carned within 180 days
3a. Debtor may have scheduled account as:	before fili	ng of the bankruptcy
. Secured Claim (See instruction #4 on reverse side.)	business, y	cessation of the debtor's whichever is earlier - 11
Check the appropriate box if your claim is secured by a lien on property or a right of setoff and provide the requested information.	U.S.C. §50	
Nature of property or right of setoff:		ons to an employee benefit
Describe:	1	J.S.C. §507 (a)(5).
Value of Property:S Annual Interest Rate%	purchase, le	25* of deposits toward case, or rental of property
Amount of arrearage and other charges as of time case filed included in secured claim,	or services	for personal, family, or use - 11 U.S.C. §507
	(a)(7).	11 O.D.C. 9507
if any: \$ Basis for perfection:		enalties owed to
Amount of Secured Claim: \$ *See Addendum Amount Unsecured: \$ *See Addendum	governmen (a)(8).	tal units - 11 U.S.C. §507
Credits: The amount of all payments on this claim has been credited for the purpose of making this proof of claim.	1	ecify applicable paragraph
Documents: Attach reducted copies of any documents that support the claim, such as promissory notes, purchase	of H U.S.C	C. §507 (a)().
ders, invoices, itemized statements of running accounts, contracts, judgments, mortgages, and security agreements, ou may also attach a summary. Attach redacted copies of documents providing evidence of perfection of	Amount	entitled to priority:
security interest. You may also attach a summary. (See instruction 7 and definition of "redacted" on reverse side.)		
O NOT SEND ORIGINAL DOCUMENTS. ATTACHED DOCUMENTS MAY BE DESTROYED AFTER CANNING.	*Amounts are.	subject to adjustment on
the documents are not available, please explain:	4/1/10 and eve.	ry 3 years thereafter with is commenced on or after
Date: Signature: The person filing this claim must sign it. Sign and print came and title if our of the		FOR COURT USE ONLY
11/09/2009 other person authorized to file this claim and state address and telephone number if different from the	reditor or	OPUS WE
address above. Attach copy of power of attorney, if any. DEAN M. LYONS	ne nonce	
Cland Gon AVTHORIZED AGENT		00619

Penalty for presenting fraudulent claim: Fine of up to \$500 per in Franch pr up to 5 years, or both. 18 U.S.C. §§ 152 and 3571.

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

IN RE:	§	Chapter 11
	§	
OPUS WEST CORPORATION,	§	Case No. 09-34356-hdh11
	§	
DEBTOR.	§	

EXHIBIT A – ADDENDUM TO PROOF OF CLAIM (SHOPPES AT CHINO HILLS) OF BANK OF AMERICA, N.A., AGENT BANK

- 1. This proof of claim is made by Bank of America, N.A. ("Bank of America"), as Agent for itself and for three (3) other Lenders (as such terms are hereinafter defined).
- 2. Pursuant to that certain Construction Loan Agreement dated as of May 16, 2007, (as amended, the "Loan Agreement"), Bank of America, as the agent on behalf of itself, Union Bank of California, N.A., KeyBank National Association, and U.S. Bank, National Association (collectively, the "Lenders"), agreed to make a loan to Shoppes at Chino Hills, Inc. ("Shoppes at Chino Hills"), as Borrower, in the original principal amount of One Hundred Ten Million and No/100 Dollars (\$110,000,000.00) (the "Loan"). The purpose of the Loan was to provide financing for the development and construction of improvements on the Property (as defined herein).
- 3. The Loan is secured by perfected mortgages, security interests, and liens in and to the Property (as defined herein) as described and granted in the Loan Agreement and in the following additional documents (the following described documents, together with any and all other or additional agreements, instruments, or other documents evidencing, securing, or otherwise relating to the Loans are hereinafter referred to collectively as the "Loan Documents"):
- a. Construction Deed of Trust Notes made by Shoppes at Chino Hills to each of the four Lenders in the amount of \$27,500,000.00 each, which is an aggregate amount of \$110,000,000.00 (as amended, the "Notes").
- b. That certain Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing executed by Shoppes at Chino Hills to Bank of America, as Trustee, for the benefit of Agent dated as of May 16, 2007 and recorded in the Official Records of San Bernardino County, California (the "Official Records") on May 18, 2007, as Document No. 2007-0302264 (as amended, "Deed of Trust"). The Deed of Trust encumbers, among other things, certain real property located in San Bernardino, California, as more particularly described therein (the "Property"). The Deed of Trust, inclusive of all assignment and security provisions contained therein, gives Agent a security interest in the Property, improvements, land, proceeds, leases, rents and all other interests (as more fully described therein) derived from and related to the Property (collectively, the "Collateral").
 - c. Those certain consents, collectively referred to herein as the "Consents":

- ii. That certain Engineer's Consent and Certificate, dated as of May 17, 2007 in favor of Agent.
- d. On March 21, 2008, Agent, Shoppes at Chino Hills, and certain other parties entered into a letter agreement that amended certain provisions of the Loan Agreement (the "Modification Agreement").
- e. A form UCC-1 financing statement was filed in the Minnesota Secretary of State's Office on May 22, 2007, as Filing No. 200716885438 naming Shoppes at Chino Hills as debtor (the "Financing Statement"). The Financing Statement further perfected Agent's security interests in the Collateral.
- 4. Opus West Corporation, a Minnesota corporation, executed a Guaranty of Payment and Completion on May 16, 2007, wherein Opus West Corporation irrevocably and unconditionally guaranteed the obligations of Shoppes at Chino Hills to Agent and Lenders under the Loan.
- 5. Prior to the July 6, 2009 petition date of Opus West Corporation (the "Petition Date"), Shoppes at Chino Hills defaulted under the terms of the Loan Documents. As a result of the events of default, Agent sent Shoppes at Chino Hills default letters dated April 6, April 9, May 7, and June 16, 2009, notifying it of the various events of default that had occurred and demanding that it pay all past due amounts in accordance with the terms of the Loan Documents. Shoppes at Chino Hills failed to cure the defaults under the Loan.
- 6. As of the Petition Date, the following liquidated, uncontested, and non-contingent amount was and is due and owing to Agent under the Loan Documents in an amount not less than \$106,807,380.87 (the "Liquidated Agent Claim Amount"), which consists of:

Principal	\$105,964,121.14
Interest	\$643,879.21
Prepetition Costs and Fees of Bank of America	
Attorney's Fees	\$41,253.04
Consultant Fees	\$61,198.48
Appraisal	\$14,000.00
Trustee Fees	<u>\$82,929.00</u>
	\$199,380.52
TOTAL:	\$106,807,380.87

Case 09-34356-hdh11 Claim 82-1 Filed 11/09/09 Desc Main Document Page 4 of 6

7. Each of the Lenders has a pro rata share in the Liquidated Agent Claim Amount made herein, as follows:

	Pro-Rata Share
Bank of America, N.A.	25.00%
Union Bank of California	25.00%
Key Bank National Association	25.00%
U.S. Bank National Association	25.00%

8. The following other Lenders may have additional prepetition costs and attorney's fees for their own accounts (which shall be referred to herein as the "Other Lender Costs"):

a.	Union Bank of California	\$ as-yet undetermined
b.	Key Bank National Association	\$ as-yet undetermined
c.	U.S. Bank National Association	\$ as-yet undetermined

- 9. Additionally, the following are due to Agent under the Loan Documents:
 - a. <u>Post-Petition Interest</u>. The amount of interest accruing on the outstanding principal amount of the Loan from the Petition Date to the date of payment of the Loan, at the rates and in the manner set forth in applicable Loan Documents;
 - b. Other Post-Petition Interest. The amount of interest accruing from the Petition Date on past-due installments of interest on the Loan to the date of payment of such installments, which accrues at the rates and in the manner set forth in the applicable Loan Documents; and
 - c. Other Post-Petition Indebtedness, Charges, Costs. All other interest, charges, penalties, premiums, advances, and other sums that may be due or become due as more fully detailed in or made reference to in any or all of the Loan Documents, including, without limitation, reasonable compensation, expenses, disbursements and advances (including, without limitation, reasonable fees and disbursement of counsel and other professionals) of Agent.
- 10. As the amount of the claims asserted in paragraph 9 hereof cannot, at this time, be reasonably calculated or estimated, the total amount of all claims set forth in this proof of claim

- 11. The Agent, on behalf of itself and the other Lenders, reserves the right to amend and/or supplement this proof of claim and to assert any and all other claims of whatever kind or nature that it/they has/have, or it/they may have, against Opus West Corporation that may come to the attention of Agent or any other Lender or arise after the filing of this proof of claim. The filing of this proof of claim shall not be deemed a waiver of any such claims or amounts.
 - 12. All notices concerning this proof of claim should be sent to:

THOMPSON & KNIGHT LLP c/o John S. Brannon 1722 Routh Street, Suite 1500 Dallas, Texas 75201

BANK OF AMERICA, N.A. c/o Casey Carpenter Vice President Real Estate Special Assets 333 S. Hope St., 11th Floor Los Angeles, CA 90071-1406

4

¹ The Agent, on behalf of itself and the other Lenders, reserves the right to assert a secured claim to the extent that any prepetition transfer of Collateral is set aside for any reason.

Date: November 9, 2009

Respectfully submitted,

/s/ Katharine B. Richter
David M. Bennett
State Bar No. 2139600
John S. Brannon
State Bar No. 02895500

THOMPSON & KNIGHT LLP

1722 Routh Street, Suite 1500 Dallas, Texas 75201 Telephone: 214/969-1700 Facsimile: 214/969-1751

E-mail: <u>david.bennett@tklaw.com</u> E-mail: <u>john.brannon@tklaw.com</u>

Katharine Battaia Richter Texas Bar No. 24046712 THOMPSON & KNIGHT LLP

98 San Jacinto Boulevard, Suite 1900 Austin, Texas 78701

Telephone: 512/469-6100 Facsimile: 512/482-5076

E-mail: katie.richter@tklaw.com

ATTORNEYS FOR BANK OF AMERICA, N.A., AGENT BANK

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS DALLAS DIVISION

IN RE:	§	Chapter 11
	§	
OPUS WEST CORPORATION,	§	Case No. 09-34356-hdh11
	§	
DEBTOR.	§	

TABLE OF CONTENTS TO EXHIBIT A – ADDENDUM TO PROOF OF CLAIM (SHOPPES AT CHINO HILLS) OF BANK OF AMERICA, N.A., AGENT BANK

ATTACHMENT 1: Construction Loan Agreement dated May 16, 2007 by and among Shoppes at Chino Hills, Inc., Bank of America, N.A., Union Bank of California, N.A., KeyBank National Association, and U.S. Bank, National Association in the maximum principal amount of \$110,000,000.00.

ATTACHMENT 2: Construction Deed of Trust Notes in the amount of \$27,500,000.00 to each of the four Syndicate Lenders.

ATTACHMENT 3: Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated May 16, 2007.

ATTACHMENT 4: Architect's Consent and Certificate, dated as of May 16, 2007.

ATTACHMENT 5: Engineer's Consent and Certificate, dated as of May 17, 2007.

ATTACHMENT 6: Modification Agreement dated March 21, 2008 by and among Bank of America, Shoppes at Chino Hills, and certain other parties.

ATTACHMENT 7: UCC-1 Financing Statement filed in the Minnesota Secretary of State's Office on May 22, 2007.

ATTACHMENT 8: Guaranty of Payment and Completion dated May 16, 2007.

ATTACHMENT 9: Default Notices dated April 6, April 9, May 7, and June 16, 2009.

Case 09-34356-hdh11 Claim 82-1 Part 2 Filed 11/09/09 Desc Attachments to Addendum Part 1 of 7 Page 2 of 32

ATTACHMENT 1

Construction Loan Agreement

among

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

and

BANK OF AMERICA, N.A. as Administrative Agent

and

The Other Financial Institutions
Party Hereto

Dated as of May 16, 2007

Banc of America Securities LLC, as Sole Arranger and Sole Book Manager



Table of Contents

		Page
ARTICLE 1	THE LOAN	1
1.1	General Information and Exhibits	1
1.2	Purpose	1
1.3	Commitment to Lend	2
1.4	Budget	2
1.5	Borrower's Deposit	2
1.6	Evidence of Debt	
ARTICLE 2	INTEREST RATES, ADVANCES AND PAYMENTS	3
2.1	Interest Rates	
2.2	Interest Rate Elections	3
2.3	General Conditions Precedent to LIBOR Rate Election	4
2.4	Authorized Persons	4
2.5	Computations and Determinations	5
2.6	Unavailability of Rate	5
2.7	Increased Cost and Reduced Return	5
2.8	Past Due Rate	
2.9	Prepayment	6
2.10	Late Charges	7
2.11	Payment Schedule and Maturity Date	8
2.12	Taxes	10
2.13	Advances and Payments	11
2.14	Administrative Agent Advances	13
2.15	Defaulting Lender	13
2.16	Several Obligations; No Liability; No Release	15
2.17	Replacement of Lenders	16
ARTICLE 3	ADDITIONAL COVENANTS AND AGREEMENTS	16
3.1	Construction of the Improvements	16
3.2	Plans and Changes	17
3.3	Contracts	17
3.4	Assignment of Plans and Contracts	18

Case 09-34356-hdh11 Claim 82-1 Part 2 Filed 11/09/09 Desc Attachments to Addendum Part 1 of 7 Page 5 of 32

Table of Contents (continued)

		Page
3.5	Storage of Materials	18
3.6	Construction Consultant	18
3.7	Inspection	19
3.8	Notice to Lenders	19
3.9	Financial Statements, Records and Reports	19
3.10	Other Information	20
3.11	Reports and Testing	20
3.12	Advertising by Lenders	21
3.13	Appraisal	21
3.14	Loan-to-Value Ratio	21
3.15	Loan-to-Cost Amount	21
3.16	Debt Service Coverage Ratio	21
3.17	[Intentionally Left Blank]	21
3.18	Payment of Withholding Taxes	21
3.19	ERISA and Prohibited Transaction Taxes	21
3.20	Change in Interest in Borrower	22
ARTICLE 4	REPRESENTATIONS AND WARRANTIES	22
4.1	Organization	22
4.2	Authorization; No Conflict	23
4.3	Enforceability	23
4.4	No Violation; No Litigation	23
4.5	Taxes	23
4.6	Plans	23
4.7	Requirements; Zoning	23
4.8	Separate Tax Lot	23
4.9	No Conveyance	23
4.10	Construction Matters	24
4.11	Utilities	24
4.12	Financial Matters	24
4.13	Borrower Not a Foreign Person	24

Table of Contents (continued)

		Page
4.14	Business Loan	24
4.15	No Work Prior to Recordation	24
ARTICLE 5	DEFAULT AND REMEDIES	25
5.1	Events of Default	25
5.2	Remedies	26
ARTICLE 6	ADMINISTRATIVE AGENT	28
6.1	Appointment and Authorization of Administrative Agent	28
6.2	Delegation of Duties	29
6.3	Liability of Administrative Agent	29
6.4	Reliance by Administrative Agent	30
6.5	Notice of Default	30
6.6	Credit Decision; Disclosure of Information by Administrative Agent	30
6.7	Indemnification of Administrative Agent	31
6.8	Administrative Agent in Individual Capacity	32
6.9	Successor Administrative Agent.	32
6.10	Releases; Acquisition and Transfers of Collateral	33
6.11	Application of Payments	34
6.12	Benefit	35
6.13	Co Agents; Lead Managers	35
ARTICLE 7	GENERAL TERMS AND CONDITIONS	35
7.1	Consents	35
7.2	Borrower's Indemnity	36
7.3	Miscellaneous	37
7.4	Notices	37
7.5	Payments Set Aside	38
7.6	Successors and Assigns	39
7.7	Confidentiality	41
7.8	No Set-off	42
7.9	Sharing of Payments	42
7.10	Amendments; Survival	42

Case 09-34356-hdh11 Claim 82-1 Part 2 Filed 11/09/09 Desc Attachments to Addendum Part 1 of 7 Page 7 of 32

Table of Contents (continued)

		Page
7.11	Costs and Expenses	44
7.12	Tax Forms	45
7.13	Further Assurances	46
7.14	Inducement to Lenders	47
7.15	Forum	47
7.16	Interpretation	47
7.17	No Partnership, etc	48
7.18	Records	48
7.19	Dispute Resolution Provision	48
7.20	Service of Process	50
7.21	USA Patriot Act Notice	51
7.22	Entire Agreement	51

EXHIBITS:

EXHIBIT "A" EXHIBIT "B"	LEGAL DESCRIPTION OF THE LAND DEFINITIONS AND FINANCIAL STATEMENTS
EXHIBIT "C"	CONDITIONS PRECEDENT TO THE FIRST ADVANCE
EXHIBIT "D"	BUDGET
EXHIBIT "E"	PLANS
EXHIBIT "F"	ADVANCES
EXHIBIT "F-1"	DRAW REQUEST
EXHIBIT "G"	SURVEY REQUIREMENTS
EXHIBIT "H"	RELEASE PROVISIONS
EXHIBIT "I"	LEASING AND TENANT MATTERS
EXHIBIT "J"	[INTENTIONALLY LEFT BLANK]
EXHIBIT "K"	[INTENTIONALLY LEFT BLANK]
EXHIBIT "L"	ASSIGNMENT AND ASSUMPTION
EXHIBIT "M"	DEED OF TRUST NOTE
EXHIBIT "N"	SCHEDULE OF LENDERS

CONSTRUCTION LOAN AGREEMENT

(The Shoppes at Chino Hills, CA - Construction Loan)

This Construction Loan Agreement ("Agreement") dated as of May 16, 2007 is made by and among each lender from time to time a party hereto (individually, a "Lender" and collectively, the "Lenders") and BANK OF AMERICA, N.A., a national banking association, as Administrative Agent, and SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower"), who agree as follows:

ARTICLE 1

THE LOAN

1.1 General Information and Exhibits. This Agreement includes the Exhibits listed below which are marked by an "X," all of which Exhibits are attached hereto and made a part hereof for all purposes. Borrower and Lenders agree that if any Exhibit to be attached to this Agreement contains blanks, the same shall be completed correctly and in accordance with this Agreement prior to or at the time of the execution and delivery thereof.

<u>X</u>	Exhibit "A"		Legal Description of the Land
<u>X</u>	Exhibit "B"	_	Definitions
<u>X</u>	Exhibit "C"		Conditions Precedent to First Advance
X	Exhibit "D"	-	Budget
_X	Exhibit "E"	_	Plans
<u>X</u>	Exhibit "F"	-	Advances
<u>X</u>	Exhibit "F-1"	_	Draw Request
<u>X</u>	Exhibit "G"	_	Survey Requirements
<u>X</u>	Exhibit "H"		Release Provisions
<u>X</u>	Exhibit "I"	_	Leasing and Tenant Matters
	Exhibit "J"	_	[Intentionally left blank]
	Exhibit "K"	_	[Intentionally left blank]
<u>X</u>	Exhibit "L"	-	Assignment and Assumption
<u>X</u>	Exhibit "M"	-	Deed of Trust Note
<u>X</u>	Exhibit "N"	-	Schedule of Lenders

The Exhibits contain other terms, provisions and conditions applicable to the Loan. Capitalized terms used in this Agreement shall have the meanings assigned to them in the Definitions set forth in Exhibit "B". This Agreement, the other Loan Documents and the Environmental Agreement, which must be in form, detail and substance satisfactory to Lenders, evidence the agreements of Borrower and Lenders with respect to the Loan. Borrower shall comply with all of the Loan Documents and the Environmental Agreement.

1.2 <u>Purpose</u>. The proceeds of the Loan shall be used by Borrower to pay (a) the cost of the construction of Improvements on the Land, and (b) other fees, costs and expenses relating to the Property if and to the extent that such costs are specifically provided for in the Loan Allocation column in the Budget.

- 1.3 <u>Commitment to Lend</u>. Borrower agrees to borrow from each Lender, and each Lender severally agrees to make advances of its Pro Rata Share of the Loan proceeds to Borrower in amounts at any one time outstanding not to exceed such Lender's Pro Rata Share of the Loan and (except for Administrative Agent with respect to Administrative Agent Advances), on the terms and subject to the conditions set forth in this Agreement and <u>Exhibit "F"</u> attached to this Agreement. Lenders' commitment to lend shall expire and terminate automatically (a) if the Loan is prepaid in full, and (b) on the Maturity Date. The Loan is not revolving. Any amount repaid may not be reborrowed.
- Budget. Upfront Equity and Loan funds are allocated to payment of the costs of the Property as described in the Budget attached to this Agreement as Exhibit "D". Borrower shall not amend the Budget or otherwise reallocate funds from one Budget line item to another without the prior written approval of Administrative Agent not to be unreasonably withheld. The Budget has been prepared by Borrower and Borrower represents to Administrative Agent and Lenders that the Budget includes all costs incident to the Loan and the Project through the maturity date of the Loan (collectively, the "Aggregate Cost") after taking into account the requirements of this Agreement, including "hard" and "soft" costs, fees and expenses. Lenders shall not be required to make any advance for any cost which is not set forth in the Budget. Lenders shall not be obligated to make any advance which, when added to all prior advances from the same line item, would exceed Borrower's actual cost incurred for that line item or the sum shown in the Budget for that line item, whichever is the lesser. Lenders shall have no obligation to make any advance from any contingency line item unless Administrative Agent consents to such advance in its reasonable discretion. After the Improvements are placed in operation, Lenders shall not be required to make any advance to pay interest on the Loan or taxes or reserves related to the Loan if and to the extent that there is sufficient net operating income from the Property to pay such interest, taxes or reserves, subject to the provisions of Exhibit "I". Lenders may make advances from any line item in the Budget for purposes other than those for which such amounts were originally allocated or in different proportions, as Administrative Agent in its sole discretion deems necessary or advisable.
- Borrower's Deposit. If at any time Administrative Agent determines that the sum 1.5 of: (i) any unadvanced portion of the Loan to which Borrower is entitled, plus (ii) the portions of the Aggregate Cost that are to be paid by Borrower from other funds that, to Administrative Agent's satisfaction, are available, set aside and committed, is or will be insufficient to pay the actual unpaid Aggregate Cost, Borrower shall, within fifteen (15) days after written notice from Administrative Agent, deposit ("Borrower's Deposit") with Administrative Agent the amount of the deficiency ("Budget Shortfall") in an interest-bearing account of Administrative Agent's selection with interest earned thereon to be part of Borrower's Deposit. Such Borrower's Deposit is hereby pledged to Administrative Agent and Lenders as additional security for the Loan, and Borrower hereby grants and conveys to Administrative Agent for the ratable benefit of Administrative Agent and Lenders a security interest in all funds so deposited with Administrative Agent, as additional security for the Loan. Administrative Agent may advance all or a portion of the Borrower's Deposit prior to the Loan proceeds. After an Event of Default, Administrative Agent may (but shall have no obligation to) apply all or any part of Borrower's Deposit against the unpaid Indebtedness in such order as Lender determines. Each Borrower's Deposit shall be accompanied by a revised proposed Budget for approval by Administrative Agent, which approval shall not be unreasonably withheld.

1.6 Evidence of Debt. Amounts of the Loan funded by each Lender shall be evidenced by one or more accounts or records maintained by such Lender and by Administrative Agent in the ordinary course of business. Any failure to record such amounts, interest or payments or any error in doing so shall not, however, limit or otherwise affect the obligation of Borrower hereunder to pay any amount owing with respect to the Indebtedness. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of Administrative Agent in respect of such matters, the accounts and records of Administrative Agent shall control in the absence of manifest error. Each Lender may attach schedules to its Note(s) and endorse thereon the date, amount and maturity of the applicable Note and payments with respect thereto.

ARTICLE 2

INTEREST RATES, ADVANCES AND PAYMENTS

- 2.1 <u>Interest Rates</u>. The Principal Debt from day to day outstanding which is not past due shall bear interest at a rate per annum equal to the following (computed as provided in Section 2.5 hereof) as applicable:
 - (a) On Base Rate Principal, on any day, the Base Rate; and
- (b) On LIBOR Rate Principal, for the applicable Interest Period, the applicable LIBOR Rate.

2.2 Interest Rate Elections.

- (a) Subject to the conditions and limitations in this Agreement, Borrower may by written notice to Administrative Agent in the form specified by Administrative Agent (a "Rate Election Notice"):
 - (i) Elect, for a new advance of funds, that such Principal Debt will be Base Rate Principal, LIBOR Rate Principal or a combination thereof;
 - (ii) Elect to convert, on a LIBOR Business Day, all or part of Base Rate Principal into LIBOR Rate Principal;
 - (iii) Elect to convert, on the last day of the Interest Period applicable thereto, all or part of any LIBOR Rate Principal into Base Rate Principal; or
 - (iv) Elect to continue, commencing on the last day of the Interest Period applicable thereto, any LIBOR Rate Principal.
- If, for any reason, an effective election is not made in accordance with the terms and conditions hereof for any principal advance or for any LIBOR Rate Principal for which the corresponding Interest Period is expiring or to convert Base Rate Principal to LIBOR Rate Principal, then the sums in question will be Base Rate Principal until an effective LIBOR Rate Election is thereafter made for such sums.

- (b) Each Rate Election Notice must be received by Administrative Agent not later than 10:00 a.m. Administrative Agent's Time on the applicable date as follows:
 - (i) With respect to an advance of or conversion to Base Rate Principal, one (1) Business Day prior to the proposed date of advance or conversion; and
 - (ii) With respect to an advance of, conversion to or continuation of LIBOR Rate Principal, three (3) LIBOR Business Days prior to the proposed date of advance, conversion or continuation.

Unless otherwise specified herein, no conversion from LIBOR Rate Principal may be made other than at the end of the corresponding Interest Period. Each Rate Election Notice shall stipulate: (A) the amount of the advance or of the Principal Debt to be converted or continued; (B) the nature of the proposed advance, conversion or continuation, which shall be either Base Rate Principal, LIBOR Rate Principal or a combination thereof, and in the case of a conversion or continuation, the nature of the Principal Debt to be converted or continued; and (C) in the case of LIBOR Rate Principal, the proposed commencement date and duration of the Interest Period. All such notices shall be irrevocable once given, and shall be deemed to have been given only when actually received by Administrative Agent in writing in form specified by Administrative Agent.

- (c) Administrative Agent shall promptly notify Borrower and Lenders of the interest rate applicable to each portion of the Principal Debt other than Base Rate Principal upon determination of the same.
- 2.3 <u>General Conditions Precedent to LIBOR Rate Election</u>. In addition to any other conditions herein, a LIBOR Rate Election shall not be permitted if:
- (a) A Default has occurred and has not been waived by Administrative Agent or a Potential Default has occurred and is continuing; or
- (b) After giving effect to the requested LIBOR Rate Election, the sum of all LIBOR Rate Principal plus all Base Rate Principal would exceed the combined Commitments; or
- (c) The requested LIBOR Rate Election would cause more than five (5) LIBOR Rate Elections by Borrower to be in effect at any one time; or
- (d) The amount of LIBOR Rate Principal requested in the LIBOR Rate Election is other than \$1,000,000.00 or more; or
- (e) The requested interest period does not conform to the definition of Interest Period herein; or
- (f) Any of the circumstances referred to in <u>Section 2.6</u> hereof shall apply with respect to the requested LIBOR Rate Election or the requested LIBOR Rate Principal.
- 2.4 <u>Authorized Persons</u>. Borrower hereby designates Chuck Vogel or Vickie Sixta as being authorized to give Rate Election Notices on behalf of Borrower. Administrative Agent

shall be entitled to rely on written or oral directions from such persons until this authorization is revoked by Borrower in writing.

- 2.5 <u>Computations and Determinations</u>. All interest shall be computed on the basis of a year of 360 days and paid for the actual number of days elapsed (including the first day but excluding the last day). Administrative Agent shall determine each interest rate applicable to the Principal Debt in accordance with this Agreement and its determination thereof shall be conclusive in the absence of manifest error. The books and records of Administrative Agent shall be <u>prima facie</u> evidence of all sums owing to Lenders from time to time under the Loan, but the failure to record any such information shall not limit or affect the obligations of Borrower under the Loan Documents.
- Unavailability of Rate. If, with respect to any LIBOR Rate Election or any 2.6 LIBOR Rate Principal outstanding hereunder, the Required Lenders determine that no adequate basis exists for determining the LIBOR Rate or that the LIBOR Rate will not adequately and fairly reflect the cost to Lenders of funding or maintaining the applicable LIBOR Rate Principal for such Interest Period, and Administrative Agent so notifies Borrower and Lenders, then, until Administrative Agent notifies Borrower and Lenders that the circumstances giving rise to such determination no longer exist, the obligation of Lenders to permit such LIBOR Rate Election shall be suspended and all existing affected LIBOR Rate Principal shall automatically become Base Rate Principal on the last day of the corresponding Interest Period. In addition, with respect to any LIBOR Rate Election or any LIBOR Rate Principal outstanding hereunder, if any Lender determines, and notifies Administrative Agent and Borrower, that any applicable Law or any request or directive (whether or not having the force of Law) of any Tribunal or compliance therewith by the Lender prohibits or restricts or makes impossible the making or maintaining of such LIBOR Rate Election or LIBOR Rate Principal or the charging of interest on such LIBOR Rate Principal, then until the Lender notifies Administrative Agent and Borrower that the circumstances giving rise to such suspension no longer exist, (a) the obligation of the Lender to permit such LIBOR Rate Election shall be suspended, and (b) all existing affected LIBOR Rate Principal of the Lender shall automatically become Base Rate Principal, either (i) on the last day of the corresponding Interest Period (if the Lender determines that it may lawfully continue to fund and maintain the affected LIBOR Rate Principal to such day); or (ii) immediately (if the Lender determines that it may not lawfully continue to fund and maintain the affected LIBOR Rate Principal to such day) and in such case Borrower shall pay to such Lender the Consequential Loss, if any, pursuant to Section 2.9 hereof. Each Lender agrees to designate a different Lending Office if such designation will avoid the need for such notice and will not, in the good faith judgment of the Lender, otherwise be materially disadvantageous to the Lender.
- 2.7 <u>Increased Cost and Reduced Return</u>. If at any time after the date hereof, any Lender (which shall include, for purposes of this <u>Section 2.7</u>, any corporation controlling any Lender) determines that the adoption or modification of any applicable Law regarding taxation, the Lender's required levels of reserves, deposits, insurance or capital (including any allocation of capital requirements or conditions), or similar requirements, or any interpretation or administration thereof by any Tribunal or compliance by the Lender with any of such requirements, has or would have the effect of (a) increasing the Lender's costs relating to the Indebtedness, or (b) reducing the yield or rate of return of the Lender on the Indebtedness, to a level below that which the Lender could have achieved but for the adoption or modification of

any such requirements, Borrower shall, within fifteen (15) days of any request by the Lender, pay to the Lender such additional amounts as (in the Lender's sole judgment, after good faith and reasonable computation) will compensate the Lender for such increase in costs or reduction in yield or rate of return of the Lender. No failure by any Lender to immediately demand payment of any additional amounts payable hereunder shall constitute a waiver of the Lender's right to demand payment of such amounts at any subsequent time. Nothing herein contained shall be construed or shall so operate as to require Borrower to pay any interest, fees, costs or charges greater than is permitted by applicable Law.

2.8 Past Due Rate. Any principal of, and to the extent permitted by applicable law, any interest, and any other sum payable under any Loan Document, which is not paid within fifteen (15) days after it becomes due shall bear interest, from the date due and payable until paid, at a rate per annum (the "Past Due Rate") equal to the lesser of (i) the maximum non-insurance rate of interest allowed by applicable law or (ii) the higher of (a) the Base Rate, or (b) the LIBOR Rate plus three percent (3%).

2.9 Prepayment.

- Borrower may prepay the principal balance of this Note, in full at any time (a) or in part from time to time, provided that: (i) Administrative Agent shall have actually received from Borrower prior written notice of Borrower's intent to prepay, the amount of principal that will be prepaid (the "Prepaid Principal"), and the date on which the prepayment will be made; (ii) each prepayment shall be in the amount of \$1,000 or larger integral multiple of \$1,000 (unless the prepayment retires the outstanding balance of the Loan in full); (iii) each prepayment shall be in the amount of 100% of the Prepaid Principal, plus accrued unpaid interest thereon to the date of prepayment, plus any other sums which have become due to Administrative Agent and Lenders under the Loan Documents on or before the date of prepayment but which have not been paid; and (iv) no portion of LIBOR Rate Principal may be prepaid except on the last day of the Interest Period applicable thereto, unless (x) the prior written consent of Administrative Agent is obtained which consent, if given, shall provide, without limitation, the manner and order in which the prepayment is to be applied to the Indebtedness, and (y) Borrower pays any Consequential Loss as a result thereof in accordance with this Section 2.9. Borrower expressly waives any right to prepay any LIBOR Rate Principal except on the express terms of this Section 2.9.
- (b) Within fifteen (15) days after request by any Lender (or at the time of any prepayment), Borrower shall pay to such Lender such amount or amounts as will compensate such Lender for any loss, cost, expense, penalty, claim or liability, including any loss incurred in obtaining, prepaying, liquidating or employing deposits or other funds from third parties and any loss of revenue, profit or yield, as determined by such Lender [on a present value basis] in its judgment reasonably exercised (together, "Consequential Loss") incurred by such Lender with respect to any LIBOR Rate, including any LIBOR Rate Election or LIBOR Rate Principal as a result of: (a) the failure of Borrower to make payments on the date specified under this Agreement or in any notice from Borrower to Administrative Agent; (b) the failure of Borrower to borrow, continue or convert into LIBOR Rate Principal on the date or in the amount specified in a notice given by Borrower to Administrative Agent pursuant to this Agreement or the Loan Agreement; (c) the early termination of any Interest Period for any reason; or (d) the payment or

Case 09-34356-hdh11 Claim 82-1 Part 2 Filed 11/09/09 Desc Attachments to Addendum Part 1 of 7 Page 14 of 32

prepayment of any amount on a date other than the date such amount is required or permitted to be paid or prepaid, whether voluntarily or by reason of acceleration, including, but not limited to, acceleration upon any transfer or conveyance of any right, title or interest in the Property giving Administrative Agent on behalf of Lenders the right to accelerate the maturity of the Loan as provided in the Deed of Trust.

- (c) Notwithstanding the foregoing, the amount of the Consequential Loss shall never be less than zero or greater than is permitted by applicable Law. If any Consequential Loss will be due, the applicable Lender shall deliver to Borrower a certificate as to the amount of the Consequential Loss, which certificate shall be conclusive in the absence of manifest error. If the Loan is prepaid in full, any commitment of Lenders for further advances shall automatically terminate. Neither Administrative Agent nor any Lender shall have any obligation to purchase, sell and/or match funds in connection with the funding or maintaining of the Loan or any portion thereof. The obligations of Borrower under this Section 2.9 shall survive any termination of the Loan Documents and payment of the Loan and shall not be waived by any delay by Administrative Agent in seeking such compensation.
- (d) By its signature below, Borrower waives any right under California Civil Code Section 2954.10 or otherwise to prepay the Loan, in whole or in part, without payment of any and all Consequential Loss as described above. Borrower acknowledges that prepayment of the Loan may result in Lenders incurring additional losses, costs, expenses and liabilities, including lost revenues and lost profits. Borrower therefore agrees to pay any and all Consequential Loss if any LIBOR Rate Principal is prepaid, whether voluntarily or by reason of acceleration, including acceleration upon any transfer or conveyance of any right, title or interest in the Property giving Administrative Agent on behalf of Lenders the right to accelerate the maturity of the Loan as provided in the Deed of Trust. Borrower agrees that Lenders' willingness to offer the LIBOR Rate to Borrower is sufficient and independent consideration, given individual weight by Lenders, for this waiver. Borrower understands that Lenders would not offer the LIBOR Rate to Borrower absent this waiver.

Dated: May 16, 2007 SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Name: Charles Vogel

Title: Vice President

2.10 <u>Late Charges</u>. If Borrower fails to make any payment under the terms of this Agreement within fifteen (15) days after the date such payment is due, Borrower shall pay to the applicable Lender or Lenders on demand a late charge equal to four percent (4%) of such payment. Such fifteen (15) day period shall not be construed as in any way extending the due date of any payment. The "late charge" is imposed for the purpose of defraying the expenses of a Lender incident to handling such delinquent payment. This charge shall be in addition to, and not in lieu of, any other remedy Lenders may have and is in addition to any fees and charges of

any agents or attorneys which Administrative Agent or Lenders may employ upon the occurrence of a Default, whether authorized herein or by Law. Borrower will pay this late charge only once on any late payment.

2.11 Payment Schedule and Maturity Date.

- (a) The entire principal balance of the Loan then unpaid and all accrued interest then unpaid shall be due and payable in full on the Maturity Date. Accrued unpaid interest shall be due and payable on the 1st day of the first calendar month after the date of this Agreement and on the same day of each succeeding calendar month thereafter until all principal and accrued interest owing on this Loan shall have been fully paid and satisfied.
- (b) Borrower may elect to extend the Maturity Date to June 1, 2010 (the "Extended Maturity Date"), upon and subject to the following terms and conditions; unless otherwise agreed by Administrative Agent with the consent of all Lenders in writing:
 - (i) Borrower shall request the extension, if at all, by written notice to Administrative Agent not less than sixty (60) days prior to the Maturity Date.
 - (ii) At the time of the request, the construction of the Improvements shall have been completed in accordance with the requirements of the Loan Documents, the temporary certificate of occupancy or its equivalent for all the Improvements shall have been issued, and all conditions to the final disbursement shall have been satisfied.
 - (iii) At the time of the request, and at the time of the extension, there shall not exist any Default, nor any condition or state of facts which after notice and/or lapse of time would constitute a Default under any Loan Document.
 - (iv) If required by Administrative Agent, current financial statements regarding Guarantor (dated not earlier than thirty (30) days prior to the request for extension) and all other financial statements and other information as may be required under the Loan Documents regarding Borrower, Guarantor and the Property, shall have been submitted promptly to Administrative Agent, and there shall not have occurred, in the opinion of Administrative Agent, any material adverse change in the business or financial condition of Borrower or Guarantor or any tenant of the Property, or in the Property or in any other state of facts submitted to Administrative Agent in connection with the Loan Documents, from that which existed on the date of this Agreement.
 - (v) Whether or not the extension becomes effective, Borrower shall pay all out-of-pocket costs and expenses incurred by Administrative Agent and Lenders in connection with the proposed extension (pre- and post-closing), including, without limitation, appraisal fees, environmental audit and legal fees; all such costs and expenses incurred up to the time of Lenders' written agreement to the extension shall be due and payable prior to Lenders' execution of that agreement (or if the proposed extension does not become effective, then upon demand by Administrative Agent), and any future failure to pay such amounts shall constitute a default under the Loan Documents.

- (vi) All applicable regulatory requirements, including appraisal requirements, shall have been satisfied with respect to the extension.
- (vii) Not later than the Maturity Date, (A) the extension shall have been consented to and documented to Administrative Agent and Lenders' satisfaction by Borrower, Guarantor, Lenders, and all other parties deemed necessary by Administrative Agent (such as any permitted subordinate lienholders, tenants of the Property and permanent lenders (if any)); (B) Administrative Agent shall have been provided with an updated title report and judgment and lien searches, and appropriate title insurance endorsements shall have been issued as required by Administrative Agent; and (C) Borrower shall have paid to Administrative Agent for the pro rata benefit of Lenders a non-refundable extension fee an amount equal to one quarter of one percent (0.25%) of the then outstanding principal balance hereunder.
- (viii) At the time of such extension, the Net Commitment Amount shall not exceed eighty percent (80.0%) of the appraised Discounted Stabilized Value (as hereinafter defined) of the Property. As used herein, "Net Commitment Amount" means, as of any date, the total obtained by adding the amount of the outstanding principal amount of the Loan to the amount of the undisbursed Loan proceeds. As used herein, "Discounted Stabilized Value" means the discounted value of the Property at stabilized occupancy, based on an appraisal meeting all applicable regulatory requirements, taking into account current market conditions, including vacancy factors, estimated date of stabilization, discount rates, and rental rates and concessions, as determined by Administrative Agent in its reasonable discretion. Administrative Agent may determine the Discounted Stabilized Value of the Property based on a current appraisal or the original appraisal obtained in connection with the origination of the Loan, as Administrative Agent in its reasonable discretion may elect. Any appraisal used to determine the Discounted Stabilized Value of the Property shall meet all applicable regulatory requirements, shall be satisfactory to Administrative Agent in all respects, and shall be obtained at the sole cost and expense of Borrower. In the event this Loan to Value Ratio is not met, Borrower may satisfy this Loan to Value Ratio prior to the extension date by either (A) making a principal curtailment on the Loan in an amount sufficient to bring this Loan to Value Ratio into compliance and/or (B) providing additional collateral acceptable to Administrative Agent, which shall have value (as determined by Administrative Agent) which when added to the Property value is sufficient to satisfy this Loan to Value Ratio.
- (ix) At the time of such extension, Borrower shall satisfy a Debt Service Coverage Ratio (as hereinafter defined) of at least 1.20 to 1.00, where for the purposes of this subsection (ix), "Debt Service Coverage Ratio" means, for any period, the ratio of Stabilized Net Operating Income from the rental portion of the Property to the Debt Service, where "Stabilized Net Operating Income" means actual net operating income for the preceding three (3) months (on an annualized basis) and where "Debt Service" means the amount of the regularly scheduled debt service payments payable for the corresponding period based on the then outstanding balance of the Loan, assuming that interest accrues at the rate of the greater of (a) the ten (10) year Treasury Rate plus

one and three-quarters percent (1.75%) or (b) seven percent (7.00%), and that interest and principal payments are based on a thirty (30) year amortization schedule.

(x) At the time of such extension, at least eighty-five percent (85.0%) of the rental Improvements shall be leased.

2.12 Taxes.

- Any and all payments by Borrower to or for the account of Administrative (a) Agent or any Lender under any Loan Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and all liabilities with respect thereto, excluding, in the case of Administrative Agent and any Lender, taxes imposed on or measured by its net income, and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the Laws of which Administrative Agent or such Lender, as the case may be, is organized or maintains a lending office (all such non-excluded taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges and liabilities being included within the definition of "Taxes" herein). If Borrower shall be required by any Laws to deduct any Taxes from or in respect of any sum payable under any Loan Document to Administrative Agent or any Lender, (i) the sum payable shall be increased as necessary so that. after making all required deductions (including deductions applicable to additional sums payable under this Section 2.12), Administrative Agent or such Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) Borrower shall make such deductions, (iii) Borrower shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable Laws, and (iv) within thirty (30) days after the date of such payment, Borrower shall furnish to Administrative Agent (who shall forward the same to such Lender) the original or a certified copy of a receipt evidencing payment thereof.
- (b) In addition, Borrower agrees to pay any and all present or future stamp, court or documentary taxes and any other excise or property taxes or charges or similar levies which arise from any payment made under any Loan Document or from the execution, delivery, performance, enforcement or registration of, or otherwise with respect to, any Loan Document (hereinafter referred to as "Other Taxes").
- (c) If Borrower shall be required by the Laws of any jurisdiction outside the United States to deduct any Taxes or Other Taxes from or in respect of any sum payable under any Loan Document to Administrative Agent or any Lender, Borrower shall also pay to such Lender or Administrative Agent (for the account of such Lender), at the time interest is paid, such additional amount that such Lender specifies is necessary to preserve the after-tax yield (after factoring in United States (federal and state) taxes imposed on or measured by net income) the Lender would have received if such deductions (including deductions applicable to additional sums payable under this Section 2.12) had not been made.
- (d) Borrower agrees to indemnify Administrative Agent and each Lender for the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this <u>Section 2.12</u>) paid by Administrative Agent and such Lender and any liability (including penalties, interest and expenses) arising

therefrom or with respect thereto, in each case whether or not such Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Tribunal. Payment under this clause (d) shall be made within thirty (30) days after Administrative Agent or the Lender makes a demand therefor.

(e) Without prejudice to the survival of any other agreement of Borrower hereunder, the agreements and obligations of Borrower contained in this Section 2.12 shall survive the termination of the Commitments and the payment in full of all the other Indebtedness.

2.13 Advances and Payments.

- Following receipt of a Draw Request, Administrative Agent shall (a) promptly provide each Lender with a copy of the Draw Request in the form of Exhibit "F 1", the related AIA Documents G-702 and G-703, the related written certification by Borrower's Architect and, if available, the related written certification of the Construction Consultant. Administrative Agent shall notify each Lender telephonically (with confirmation by facsimile) or by facsimile (with confirmation by telephone) not later than 1:00 p.m. Administrative Agent's Time two (2) Business Days prior to the advance Funding Date for LIBOR Rate Principal advances, and one (1) Business Day prior to the advance Funding Date for all other advances, of its Pro Rata Share of the amount Administrative Agent has determined shall be advanced in connection therewith ("Advance Amount"). In the case of an advance of the Loan, each Lender shall make the funds for its Pro Rata Share of the Advance Amount available to Administrative Agent not later than 11:00 a.m. Administrative Agent's Time on the funding date thereof. After Administrative Agent's receipt of the Advance Amount from Lenders, Administrative Agent shall make proceeds of the Loan in an amount equal to the Advance Amount (or, if less, such portion of the Advance Amount that shall have been paid to Administrative Agent by Lenders in accordance with the terms hereof) available to Borrower on the applicable Funding Date by advancing such funds to Borrower in accordance with the provisions of Exhibit "F".
- (b) All payments by Borrower shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by Borrower hereunder shall be made to Administrative Agent not later than 12:00 p.m. (Administrative Agent's Time) on the date specified herein. Administrative Agent shall distribute to each Lender, such funds as such Lender may be entitled to receive hereunder, (i) on or before 3:00 p.m. (Administrative Agent's Time) on the day Administrative Agent receives such funds, if Administrative Agent has received such funds on or before 12:00 p.m. (Administrative Agent's Time) on the Business Day following the day Administrative Agent receives such funds, if Administrative Agent receives such funds after 12:00 p.m. (Administrative Agent's Time). If Administrative Agent fails to timely pay any amount to any Lender in accordance with this clause (b), Administrative Agent shall pay to such Lender interest at the Federal Funds Rate on such amount, for each day from the day such amount was to be paid until it is paid to such Lender.
- (c) Except as otherwise expressly provided herein, all payments by Borrower or any Lender shall be made to Administrative Agent at Administrative Agent's Office not later than the time for such type of payment specified in this Agreement. All payments received after

such time shall be deemed received on the next succeeding Business Day. All payments shall be made in immediately available funds in lawful money of the United States of America. Whenever any payment falls due on a day which is not a Business Day, such payment may be made on the next succeeding Business Day.

- (d) Upon satisfaction of any applicable terms and conditions set forth herein, Administrative Agent shall promptly make any amounts received in accordance with clause (c) above available in like funds received as follows: (i) if payable to Borrower, in accordance with Exhibit "F", except as otherwise specified herein, and (ii) if payable to any Lender, by wire transfer to such Lender at the address specified in the Schedule of Lenders.
- (e) Unless Borrower or any Lender has notified Administrative Agent, prior to the date any payment is required to be made by it to Administrative Agent, that Borrower or such Lender, as the case may be, will not make such payment, Administrative Agent may assume that Borrower or such Lender, as the case may be, has timely made such payment and may (but shall not be required to do so) in reliance thereon, make available a corresponding amount to the person or entity entitled thereto. If and to the extent that such payment was not in fact made to Administrative Agent in immediately available funds, then:
 - (i) if Borrower failed to make such payment, each Lender shall forthwith on demand repay to Administrative Agent the portion of such assumed payment that was made available to such Lender in immediately available funds, together with interest thereon in respect of each day from and including the date such amount was made available by Administrative Agent to such Lender to the date such amount is repaid to Administrative Agent in immediately available funds at the Federal Funds Rate from time to time in effect; and
 - if any Lender failed to make such payment, such Lender or, if applicable, Electing Lender or Lenders shall forthwith on demand pay to Administrative Agent the amount thereof in immediately available funds, together with interest thereon for the period from the date such amount was made available by Administrative Agent to Borrower to the date such amount is recovered by Administrative Agent (the "Compensation Period") at a rate per annum equal to the interest rate applicable to such amount under the Loan. If such Lender pays such amount to Administrative Agent, then such amount shall constitute such Lender's Pro Rata Share, included in the applicable Loan advance. If such Lender does not pay such amount forthwith upon Administrative Agent's demand therefor, Administrative Agent may make a demand therefor upon Borrower, and Borrower shall pay such amount to Administrative Agent, together with interest thereon for the Compensation Period at a rate per annum equal to the rate of interest applicable to such amount under the Loan. Nothing herein shall be deemed to relieve any Lender from its obligation to fulfill its Commitment or to prejudice any rights which Administrative Agent or Borrower may have against any Lender as a result of any default by such Lender hereunder.

A notice of Administrative Agent to any Lender or Borrower with respect to any amount owing under this clause (e) shall be conclusive, absent any manifest error.

- (f) If any Lender makes available to Administrative Agent funds for any Loan advance to be made by such Lender as provided in the foregoing provisions of this Section 2.13, and the funds are not advanced to Borrower or otherwise used to satisfy any Obligations of such Lender hereunder, Administrative Agent shall return such funds (in like funds as received from such Lender) to such Lender, without interest.
- (g) Nothing herein shall be deemed to obligate any Lender to obtain the funds for any Loan advance in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Loan advance in any particular place or manner.

2.14 Administrative Agent Advances.

- (a) Administrative Agent is authorized, from time to time, in Administrative Agent's sole discretion to make, authorize or determine advances of the Loan, or otherwise expend funds, on behalf of Lenders ("Administrative Agent Advances"), (i) to pay any costs, fees and expenses as described in Section 7.11 herein, (ii) when the applicable conditions precedent set forth in Exhibit "C" and Exhibit "F" have been satisfied to the extent required by Administrative Agent, and (iii) when Administrative Agent deems necessary or desirable to preserve or protect the Loan collateral or any portion thereof (including those with respect to property taxes, insurance premiums, completion of construction, operation management, improvements, maintenance, repair, sale and disposition) (A) subject to Section 6.5, after the occurrence of a Default, and (B) subject to Section 6.10, after acquisition of all or a portion of the Loan collateral by foreclosure or otherwise.
- Lenders under this Agreement, shall be repayable on demand and secured by the Loan collateral, and if unpaid by Lenders as set forth below, shall bear interest at the rate applicable to such amount under the Loan or if no longer applicable, at the Base Rate. Administrative Agent shall notify each Lender in writing of each Administrative Agent Advance. Upon receipt of notice from Administrative Agent of its making of an Administrative Agent Advance, each Lender shall make the amount of such Lender's Pro Rata Share of the outstanding principal amount of the Administrative Agent Advance available to Administrative Agent, in same day funds, to such account of Administrative Agent as Administrative Agent may designate, (i) on or before 3:00 p.m. (Administrative Agent's Time) on the day Administrative Agent provides Lenders with notice of the making of such Administrative Agent Advance if Administrative Agent provides such notice on or before 12:00 p.m. (Administrative Agent's Time), or (ii) on or before 12:00 p.m. (Administrative Agent provides Lenders with notice of the making of such advance if Administrative Agent provides Lenders with notice of the making of such advance if Administrative Agent provides Lenders with notice of the making of such advance if Administrative Agent provides Lenders with notice of the making of such advance if Administrative Agent provides notice after 12:00 p.m. (Administrative Agent's Time).

2.15 Defaulting Lender.

(a) Notice and Cure of Lender Default; Election Period; Electing Lenders. Administrative Agent shall notify (such notice being referred to as the "Default Notice") Borrower (for Loan advances) and each non-Defaulting Lender if any Lender is a Defaulting Lender. Each non-Defaulting Lender shall have the right, but in no event and under no

circumstance the obligation, to fund such Defaulting Lender Amount, provided that within twenty (20) days after the date of the Default Notice (the "Election Period"), the non-Defaulting Lender or Lenders (each such Lender, an "Electing Lender") irrevocably commit(s) by notice in writing (an "Election Notice") to Administrative Agent, the other Lenders and Borrower to fund the Defaulting Lender Amount and to assume the Defaulting Lender's obligations with respect to the advancing of the entire undisbursed portion of the Defaulting Lender's principal obligations under this Agreement (such entire undisbursed portion of the Defaulting Lender's principal obligations under this Agreement, including its portion of the Payment Amount that is the subject of the default, is hereinafter referred to as the "Defaulting Lender Obligation"). If Administrative Agent receives more than one Election Notice within the Election Period, then the commitment to fund the Defaulting Lender Amount and the Defaulting Lender Obligation shall be apportioned pro rata among the Electing Lenders in the proportion that the amount of each such Electing Lender's Commitment bears to the total Commitments of all Electing Lenders. If the Defaulting Lender fails to pay the Defaulting Lender Payment Amount within the Election Period, the Electing Lender or Lenders, as applicable, shall be automatically obligated to fund the Defaulting Lender Amount and the Defaulting Lender Obligation (and the Defaulting Lender shall no longer be entitled to fund such Defaulting Lender Amount or such Defaulting Lender Obligation) within three (3) Business Days following the expiration of the Election Period to reimburse Administrative Agent or make payment to Borrower, as applicable. Notwithstanding anything to the contrary contained herein, if Administrative Agent has funded the Defaulting Lender Amount, Administrative Agent shall be entitled to reimbursement for its portion of the Defaulting Lender Payment Amount pursuant to Section 6.11.

Removal of Rights; Indemnity. Administrative Agent shall not be obligated to transfer to a Defaulting Lender any payments made by or on behalf of Borrower to Administrative Agent for the Defaulting Lender's benefit; nor shall a Defaulting Lender be entitled to the sharing of any payments hereunder or under any Note until all Defaulting Lender Payment Amounts are paid in full. Amounts payable to a Defaulting Lender shall be paid by Administrative Agent to reimburse Administrative Agent and any Electing Lender pro rata for all Defaulting Lender Payment Amounts. Solely for the purposes of voting or consenting to matters with respect to the Loan Documents or the Environmental Agreement, a Defaulting Lender shall be deemed not to be a "Lender" and such Defaulting Lender's Commitment shall be deemed to be zero. A Defaulting Lender shall have no right to participate in any discussions among and/or decisions by Lenders hereunder and/or under the other Loan Documents or the Environmental Agreement. Further, any Defaulting Lender shall be bound by any amendment to, or waiver of, any provision of, or any action taken or omitted to be taken by Administrative Agent and/or the non-Defaulting Lenders under, any Loan Document or the Environmental Agreement which is made subsequent to the Defaulting Lender's becoming a Defaulting Lender. This Section 2.15 shall remain effective with respect to a Defaulting Lender until such time as the Defaulting Lender shall no longer be in default of any of its obligations under this Agreement by curing such default by payment of all Defaulting Lender Payment Amounts (i) within the Election Period, or (ii) after the Election Period with the consent of the non-Defaulting Lenders. Such Defaulting Lender nonetheless shall be bound by any amendment to or waiver of any provision of, or any action taken or omitted to be taken by Administrative Agent and/or the non-Defaulting Lenders under any Loan Document or the Environmental Agreement which is made subsequent to that Lender's becoming a Defaulting Lender and prior to such cure or waiver. The operation of this clause (b) or clause (a) above alone shall not be construed to increase or otherwise affect

the Commitment of any non-Defaulting Lender, or relieve or excuse the performance by Borrower of its duties and obligations hereunder or under any of the other Loan Documents or the Environmental Agreement. Furthermore, nothing contained in this Section 2.15 shall release or in any way limit a Defaulting Lender's obligations as a Lender hereunder and/or under any of the other Loan Documents or the Environmental Agreement. Further, a Defaulting Lender shall indemnify and hold harmless Administrative Agent and each of the non-Defaulting Lenders from any claim, loss, or costs incurred by Administrative Agent and/or the non-Defaulting Lenders as a result of a Defaulting Lender's failure to comply with the requirements of this Agreement, including any and all additional losses, damages, costs and expenses (including attorneys' fees) incurred by Administrative Agent and any non-Defaulting Lender as a result of and/or in connection with (i) a non-Defaulting Lender's acting as an Electing Lender, (ii) any enforcement action brought by Administrative Agent against a Defaulting Lender, and (iii) any action brought against Administrative Agent and/or Lenders. The indemnification provided above shall survive any termination of this Agreement.

- (c) Commitment Adjustments. In connection with the adjustment of the amounts of the Loan Commitments of the Defaulting Lender and Electing Lender(s) upon the expiration of the Election Period as aforesaid, Borrower, Administrative Agent and Lenders shall execute such modifications to the Loan Documents and the Environmental Agreement as shall, in the reasonable judgment of Administrative Agent, be necessary or desirable in connection with the adjustment of the amounts of Commitments in accordance with the foregoing provisions of this Section. For the purpose of voting or consenting to matters with respect to the Loan Documents or the Environmental Agreement, such modifications shall also reflect the removal of voting rights of the Defaulting Lender and increase in voting rights of Electing Lenders to the extent an Electing Lender has funded the Defaulting Lender Amount and assumed the Defaulting Lender Obligation. In connection with such adjustments, the Defaulting Lender shall execute and deliver an Assignment and Assumption covering that Lender's Commitment and otherwise comply with Section 7.6. If a Lender refuses to execute and deliver such Assignment and Assumption or otherwise comply with Section 7.6, such Lender hereby appoints Administrative Agent to do so on such Lender's behalf. Administrative Agent shall distribute an amended Schedule of Lenders, which shall thereafter be incorporated into this Agreement, to reflect such adjustments. However, all such Defaulting Lender Amounts and such Defaulting Lender Obligation funded by Administrative Agent or Electing Lenders shall continue to be Defaulting Lender Amounts of the Defaulting Lender pursuant to its obligations under this Agreement.
- (d) <u>No Election</u>. In the event that no Lender elects to commit to fund the Defaulting Lender Amount and Defaulting Lender Obligation within the Election Period, Administrative Agent shall, upon the expiration of the Election Period, so notify Borrower and each Lender.
- 2.16 Several Obligations; No Liability; No Release. Notwithstanding that certain of the Loan Documents and/or the Environmental Agreement now or hereafter may have been or will be executed only by or in favor of Administrative Agent in its capacity as such, and not by or in favor of Lenders, any and all obligations on the part of Administrative Agent (if any) to make any advances of the Loan or reimbursements for other Payment Amounts shall constitute the several (and not joint) obligations of the respective Lenders on a ratable basis, according to their respective Pro Rata Shares. Except as may be specifically provided in this Agreement, no

Lender shall have any liability for the acts of any other Lender. No Lender shall be responsible to Borrower or any other person for any failure by any other Lender to fulfill its obligations to make advances of the Loan or reimbursements for other Payment Amounts, or to take any other action on its behalf hereunder or in connection with the financing contemplated herein. The failure of any Lender to pay to Administrative Agent its Pro Rata Share of a Payment Amount shall not relieve any other Lender of any obligation hereunder to pay to Administrative Agent its Pro Rata Share of such Payment Amounts as and when required herein, but no Lender shall be responsible for the failure of any other Lender to so fund its Pro Rata Share of the Payment Amount. In furtherance of the foregoing, Lenders shall comply with their obligation to pay Administrative Agent their Pro Rata Shares of such Payment Amounts regardless of (i) the occurrence of any Default hereunder or under any Loan Document or the Environmental Agreement; (ii) any failure of consideration, absence of consideration, misrepresentation, fraud, or any other event, failure, deficiency, breach or irregularity of any nature whatsoever in the Loan Documents or the Environmental Agreement; (iii) any bankruptcy, insolvency or other like event with regard to Borrower or Guarantor. The obligation of Lenders to pay such Payment Amounts are in all regards independent of any claims between Administrative Agent and any Lender.

2.17 Replacement of Lenders. If any Lender is a Defaulting Lender, Borrower may, upon notice to such Lender and Administrative Agent, replace such Lender by causing such Lender to assign its Commitment with the payment of any assignment fee by the replaced Lender to one or more other lenders or Eligible Assignees acceptable to Borrower, and the Administrative Agent. Borrower shall or shall cause the replacement lender to (subject to the provisions of Sections 2.14 and 2.15 providing for payment of all Defaulting Lender Payment Amounts to Administrative Agent and/or Electing Lenders, as applicable, prior to payment of amounts due to a Defaulting Lender), (a) pay in full all principal, interest, fees and other amounts owing to such Lender through the date of replacement, (b) provide appropriate assurances and indemnities (which may include letters of credit) as such Lender may reasonably require with respect to such replaced Lender's obligation to fund its participation interest in any letters of credit then outstanding, and (c) provide a release of such Lender from its obligations under the Loan Documents and the Environmental Agreement. Any Lender being replaced shall execute and deliver an Assignment and Assumption covering that Lender's Commitment and otherwise comply with Section 7.6. If a Lender being replaced refuses to execute and deliver such Assignment and Assumption or otherwise comply with Section 7.6, such Lender hereby appoints Administrative Agent to do so on such Lender's behalf. Administrative Agent shall distribute an amended Schedule of Lenders, which shall thereafter be incorporated into this Agreement, to reflect adjustments to Lenders and their Commitments.

ARTICLE 3

ADDITIONAL COVENANTS AND AGREEMENTS

3.1 Construction of the Improvements. Borrower shall commence construction of the Improvements as soon as practicable after the recording of the Deed of Trust in the official records of the county in which the Property is located, and shall prosecute the construction of the Improvements with diligence and continuity, in a good and workmanlike manner, and in accordance with sound building and engineering practices, all applicable Laws and governmental

requirements, the Plans and the Loan Documents. Borrower shall not permit the cessation of work for a period in excess of ten (10) consecutive Business Days, except for Excusable Delays. Borrower shall complete construction of the Improvements free and clear of all liens (except liens created by the Loan Documents), and shall obtain a certificate of occupancy and all other permits, licenses and approvals from all applicable governmental authorities required for the occupancy, use and operation of the Improvements, in each case satisfactory to Administrative Agent, on or before the Completion Date. Borrower shall promptly correct (a) any material defect in the Improvements, (b) any material departure from the Plans, Law or governmental requirements, or (c) any encroachment by any Improvements or structure on any building setback line, easement, property line or restricted area.

- Plans and Changes. No construction shall be undertaken on the Land except as 3.2 shown in the Plans. Borrower assumes full responsibility for the compliance of the Plans and the Property with all Laws, governmental requirements and sound building and engineering practices. No plans or specifications or any changes thereto shall be included as part of the Plans until approved by Administrative Agent, Construction Consultant, all applicable governmental authorities, and all other parties required under the Loan Documents. Without Administrative Agent's prior written consent, Borrower shall not change or modify the Plans, agree to any change order, or allow any extras to any contractor or any subcontractor; provided, however, that Borrower may make Permitted Changes if (a) Borrower notifies Administrative Agent in writing of the change or extra with appropriate supporting documentation and information, (b) Borrower obtains the approval of the applicable contractor, Borrower's architect and all sureties, (c) the structural integrity, quality and standard of workmanship of the Improvements will not be impaired by such change or extra, (d) no substantial change in architectural appearance will be effected by such change or extra, (e) no default in any obligation to any person or violation of any Law or governmental requirement will result from such change or extra, (f) Borrower complies with Section 1.5 of this Agreement to cover any excess cost resulting from the change or extra, and (g) completion of the Improvements by the Completion Date will not be affected. Administrative Agent shall not be obligated to review a proposed change unless it has received all documents necessary to review such change, including the change order, cost estimates, plans and specifications, and evidence that all required approvals other than that of Administrative Agent have been obtained.
- 2.3 Contracts. Without Administrative Agent's prior written approval as to parties, terms and all other matters, Borrower shall not (a) enter into any Material Contract for the performance of any work or the supplying of any labor, materials or services for the design or construction of the Improvements, or (b) modify, amend or terminate any such Material Contracts. All such Material Contracts shall provide that no change order shall be effective without the prior written consent of Administrative Agent except for change orders which implement Permitted Changes. Borrower may enter into any management, leasing, maintenance or other contract pertaining to the Property which is not a Material Contract so long as the contract is unconditionally terminable by Borrower or any successor owner without penalty or payment on not more than thirty (30) days' notice to the other party. Borrower shall not default under any contract or permit any contract to terminate by reason of any failure of Borrower to perform thereunder, and Borrower shall promptly notify Administrative Agent of any default thereunder.

- Assignment of Plans and Contracts. As additional security for the Obligations, 3.4 Borrower hereby transfers and assigns to Administrative Agent for the ratable benefit of Administrative Agent and Lenders all of Borrower's right, title and interest, but not its liability, in, under and to the Plans and all construction, architectural and design contracts, and agrees that all of the same are covered by the security agreement provisions of the Deed of Trust. Borrower agrees to deliver to Administrative Agent from time to time upon Administrative Agent's request such consents to the foregoing assignment from parties contracting with Borrower as Administrative Agent may require. Neither this assignment nor any action by Administrative Agent or Lenders shall constitute an assumption by Administrative Agent or Lenders of any obligation with respect to the Plans or under any contract. Borrower hereby agrees to perform all of its obligations under any contract, and Borrower shall continue to be liable for all obligations of Borrower with respect thereto. Administrative Agent shall have the right at any time (but shall have no obligation) to take in its name or in the name of Borrower such action as Administrative Agent may determine to be necessary to cure any default with respect to the Plans or under any contract or to protect the rights of Borrower, Administrative Agent or Lenders with respect thereto. Borrower irrevocably constitutes and appoints Administrative Agent as Borrower's attorney in fact, which power of attorney is coupled with an interest and irrevocable, to enforce in Borrower's name or in the names of Administrative Agent and Lenders all rights of Borrower with respect to the Plans or under any contract. Administrative Agent shall incur no liability if any action so taken by it or on its behalf shall prove to be inadequate or invalid. Administrative Agent may use the Plans for any purpose relating to the Improvements. Borrower indemnifies and holds Administrative Agent and Lenders harmless from and against any loss, cost, liability or expense (including consultants' fees and expenses and attorneys' fees and expenses) incurred in connection with Borrower's failure to perform such contracts or any action taken by Administrative Agent or Lenders. Borrower represents and warrants to Administrative Agent and Lenders that the copies of the Plans delivered to Administrative Agent are and shall be true and complete copies thereof, that the copy of any contract furnished or to be furnished to Administrative Agent is and shall be a true and complete copy thereof, that there have been no modifications thereof which are not fully set forth in the copies delivered, and that Borrower's interest therein is not subject to any claim, setoff or encumbrance.
- 3.5 Storage of Materials. Borrower shall cause all materials supplied for or intended to be utilized in the construction of the Improvements, but not yet affixed to or incorporated into the Improvements or the Land, to be stored on the Land or at such other site as Administrative Agent may approve, in each case with adequate safeguards to prevent loss, theft, damage or commingling with materials for other projects. Borrower shall not purchase or order materials for delivery more than forty-five (45) days prior to the scheduled incorporation of such materials into the Improvements.
- 3.6 <u>Construction Consultant</u>. Administrative Agent may retain the services of a Construction Consultant, whose duties may include, among others, reviewing the Plans and any proposed changes to the Plans, performing construction cost analyses, observing work in place and reviewing Draw Requests. The duties of Construction Consultant run solely to Administrative Agent for the ratable benefit of Lenders, and Construction Consultant shall have no obligations or responsibilities whatsoever to Borrower, Borrower's architect, engineer, contractor or any of their agents or employees. Unless prohibited by applicable Law, all reasonable fees, costs, and expenses of Construction Consultant shall be paid by Borrower.

Borrower shall cooperate with Construction Consultant and will furnish to Construction Consultant such information and other material as Construction Consultant may consider necessary or useful in performing its duties. Administrative Agent will provide Borrower with a good faith estimate of costs for the initial review of the Plans and on site inspection.

- 3.7 <u>Inspection</u>. Administrative Agent and its agents, including Construction Consultant, may enter upon the Property to inspect the Property, the Project and any materials at any reasonable time, unless Administrative Agent deems such inspection is of an emergency nature, in which event Borrower shall provide Administrative Agent with immediate access to the Property. Borrower will also permit Administrative Agent and its agents, including Construction Consultant, to photograph the Property during normal business hours and at any other reasonable time. Borrower will furnish to Administrative Agent and its agents, including Construction Consultant, for inspection and copying, all Plans, shop drawings, specifications, books and records, and other documents and information that Administrative Agent may request from time to time.
- Notice to Lenders. Borrower shall, within ten (10) days after the occurrence of 3.8 any of the following events, notify each Lender in writing thereof, specifying in each case the action Borrower has taken or will take with respect thereto: (a) any violation of any Law or governmental requirement; (b) any litigation, arbitration or governmental investigation or proceeding instituted or threatened against Borrower or any Guarantor or the Property where the amount claimed is \$50,000.00 or more, and any material development therein; (c) any actual or threatened condemnation of any portion of the Property, any negotiations with respect to any such taking, or any loss of or substantial damage to the Property; (d) any labor controversy pending or threatened against Borrower or any contractor, and any material development in any labor controversy where the amount claimed is \$50,000.00 or more; (e) any notice received by Borrower with respect to the cancellation, alteration or non renewal of any insurance coverage maintained with respect to the Property; (f) any failure by Borrower or any contractor, subcontractor or supplier to perform any material obligation under any construction contract, any event or condition which would permit termination of a construction contract or suspension of work thereunder, or any notice given by Borrower or any contractor with respect to any of the foregoing; (g) any lien filed against the Property or any stop notice served on Borrower in connection with construction of the Improvements; or (h) the lapse of any required permit, license, certificate or approval with respect to the Property or the failure of any of the foregoing to be in full force and effect.
- 3.9 <u>Financial Statements, Records and Reports.</u> If reasonably required by Administrative Agent, Borrower shall provide or cause to be provided to Administrative Agent all of the following:
- (a) Financial Statements of Guarantor for each fiscal year of such Reporting Party, as soon as reasonably practicable and in any event within one hundred twenty (120) days after the close of each fiscal year, unqualified audited certified financial statements of Guarantor, and for each fiscal quarter, as soon as reasonably practicable and in any event within sixty (60) days after the close of each such reporting period company-prepared financial statements of Guarantor.

- (b) Commencing the first month after the issuance of a certificate of occupancy or its equivalent, provided the Property is operating, within forty-five (45) days after the close of each fiscal quarter, operating statements of the Project, rent rolls and leasing reports of the Project.
- (c) From time to time promptly after Administrative Agent's reasonable request, such additional information, reports and statements respecting the Property and the Improvements, or the business operations and financial condition of each Reporting Party, as Administrative Agent may reasonably request.

Borrower will make all of its books, records and accounts available to Administrative Agent and its representatives at the Property upon request and will permit them to review and copy the same. All Financial Statements shall be in form and detail satisfactory to Administrative Agent and shall contain or be attached to the signed and dated written certification of the Reporting Party in form specified by Administrative Agent to certify that the Financial Statements are furnished to Administrative Agent in connection with the extension of credit by Lenders and constitute a true and correct statement of the Reporting Party's financial position. All certifications and signatures on behalf of corporations, partnerships, limited liability companies or other entities shall be by a representative of the Reporting Party satisfactory to Administrative Agent. Borrower shall promptly notify Administrative Agent of any event or condition that could reasonably be expected to have a Material Adverse Effect in the financial condition of Borrower or Guarantor (if known by Borrower), or in the construction progress of the Improvements. Administrative Agent shall provide a copy of such Financial Statements to each Lender upon receipt.

- 3.10 Other Information. Borrower shall furnish to Administrative Agent from time to time upon Administrative Agent's request: (a) copies of any or all subcontracts entered into by contractors or subcontractors and the names and addresses of all persons or entities with whom Borrower or any contractor has contracted or intends to contract for the construction of the Improvements or the furnishing of labor or materials in connection therewith; (b) copies of any or all contracts, bills of sale, statements, receipts or other documents under which Borrower claims title to any materials, fixtures or articles of personal property incorporated or to be incorporated into the Improvements or subject to the lien of the Deed of Trust; (c) a list of all unpaid bills for labor and materials with respect to construction of the Improvements and copies of all invoices therefor; (d) budgets of Borrower and revisions thereof showing the estimated costs and expenses to be incurred in connection with the completion of construction of the Improvements; (e) current or updated detailed Project schedules or construction schedules; and (f) such other information relating to Borrower, Guarantor, the Improvements, the Property, or any indemnitor or other person or party connected with Borrower, the Loan, the construction of the Improvements or any security for the Loan.
- 3.11 <u>Reports and Testing</u>. Borrower shall (a) promptly deliver to Administrative Agent copies of all reports, studies, inspections and tests made on the Land, the Improvements or any materials to be incorporated into the Improvements, and (b) make such additional tests on the Land, the Improvements or any materials to be incorporated into the Improvements as Administrative Agent reasonably requires. Borrower shall immediately notify Administrative

Agent of any report, study, inspection or test that indicates any adverse condition relating to the Land, the Improvements or any such materials.

- 3.12 <u>Advertising by Lenders</u>. At Administrative Agent's request and at Borrower's expense, Borrower shall erect and maintain on the Property one or more advertising signs furnished by Administrative Agent indicating that the construction financing for the Property has been provided by Lenders.
- 3.13 Appraisal. From time to time, Administrative Agent may obtain an appraisal of all or any part of the Property prepared in accordance with written instructions from Administrative Agent by a third party appraiser engaged directly by Administrative Agent. Each such appraiser and appraisal shall be satisfactory to Administrative Agent (including satisfaction of applicable regulatory requirements). The cost of any such appraisal shall be borne by Borrower if such appraisal is the first appraisal in any calendar year and in all events if Administrative Agent obtains such appraisal after the occurrence of a Default. Whenever Borrower is obligated to pay the cost of an appraisal hereunder, such cost is due and payable by Borrower on demand and shall be secured by the Loan Documents. Administrative Agent shall provide a copy of such appraisal to each Lender upon receipt.
- 3.14 <u>Loan-to-Value Ratio</u>. The Property_shall have at all times a "<u>Loan-to-Value Ratio</u>" of not greater than eighty percent (80%), which Loan-to-Value Ratio shall be calculated as follows: the Net Commitment Amount (as defined in Section 2.11(b)(viii) hereof) as of the date of the determination of the ratio shall be divided by the appraised Discounted Stabilized Value of the Property (as defined in <u>Section 2.11(b)(viii)</u> hereof).
- 3.15 <u>Loan-to-Cost Amount</u>. The Property shall have at all times a "<u>Loan-to-Cost Ratio</u>" of not greater than ninety percent (90%), which "<u>Loan to Cost Ratio</u>" shall be calculated as follows: Net Commitment Amount (as defined in Section 2.11(b)(viii) hereof) divided by total Property costs as set forth in the Budget.
- 3.16 <u>Debt Service Coverage Ratio</u>. Borrower shall satisfy at all times a "<u>Debt Service Coverage Ratio</u>" of at least 1.15 to 1.00, where Debt Service Coverage Ratio shall have the meaning given in Section 2.11(b)(ix) hereof except that Stabilized Net Operating Income shall mean the appraised net operating income.

3.17 [Intentionally Left Blank].

- 3.18 Payment of Withholding Taxes. Borrower shall not use, or knowingly permit any contractor or subcontractor to use, any portion of the proceeds of any Loan advance to pay the wages of employees unless a portion of the proceeds or other funds are also used to make timely payment to or deposit with (a) the United States of all amounts of tax required to be deducted and withheld with respect to such wages under the Internal Revenue Code, and (b) any state and/or local Tribunal or agency having jurisdiction of all amounts of tax required to be deducted and withheld with respect to such wages under any applicable state and/or local Laws.
- 3.19 ERISA and Prohibited Transaction Taxes. As of the date hereof and throughout the term of this Agreement: (a) Borrower is not and will not be (i) an "employee benefit plan," as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended

("ERISA"), or (ii) a "plan" within the meaning of Section 4975(e) of the Internal Revenue Code; (b) the assets of Borrower do not and will not constitute "plan assets" within the meaning of the United States Department of Labor Regulations set forth in 29 C.F.R. § 2510.3-101; (c) Borrower is not and will not be a "governmental plan" within the meaning of Section 3(32) of ERISA; (d) transactions by or with Borrower are not and will not be subject to state statutes applicable to Borrower regulating investments of fiduciaries with respect to governmental plans; and (e) Borrower shall not engage in any transaction which would cause any obligation or action taken or to be taken hereunder (or the exercise by Administrative Agent of any of Lenders' rights under this Agreement, the Note or the other Loan Documents or the Environmental Agreement) to be a non-exempt (under a statutory or administrative class exemption) prohibited transaction under ERISA or Section 4975 of the Internal Revenue Code. Borrower further agrees to deliver to Administrative Agent such certifications or other evidence of compliance with the provisions of this Section 3.19 as Administrative Agent may from time to time request.

Change in Interest in Borrower. Without the prior written consent of Lender, no more than fifty percent (50.0%) of the voting power in the Borrower, Borrower's Manager or Guarantor shall be transferred after the date hereof to another person or entity. Notwithstanding the foregoing, transfers of any interest in Borrower, Borrower's Manager or Guarantor to any person or entity that is directly or indirectly controlled by, controlling or under common control of or with (i) Opus West Corporation, Opus, Corporation, Opus, L.L.C., or (ii) the founder of Opus Corporation, his children, his grandchildren or other members of his family, or (iii) the trustee of a trust or trusts for the benefit of the founder of Opus Corporation, his children, his grandchildren or other members of his family, shall be permitted hereunder without Lender's consent, so long as (1) Shoppes at Chino Hills, Inc. remains fully obligated for the Borrower's obligations with respect to the Loan and Opus West Corporation remains the Guarantor under the Guaranty, and (2) Opus West Corporation or one of the other entities listed in subsections (i), (ii) or (iii) immediately above, shall at all times maintain, directly or indirectly, ownership at least 10% of the voting stock in the Borrower. NOTICE - THE LOAN IS SUBJECT TO ACCELERATION IN THE EVENT OF A TRANSFER WHICH IS PROHIBITED UNDER THIS SECTION 3.20.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES

To induce Lenders to make the Loan, Borrower hereby represents and warrants to Administrative Agent and Lenders that except as otherwise disclosed to Administrative Agent in writing:

4.1 Organization. If Borrower is a corporation, partnership, limited liability company or other legal entity, Borrower is and will continue to be (a) duly organized, validly existing and in good standing under the Laws of the state of its organization, (b) authorized to do business and in good standing in each state in which the Property is located, and (c) possessed of all requisite power and authority to carry on its business, to own the Property and to develop and operate the Improvements as contemplated in this Agreement and the other Loan Documents.

- 4.2 <u>Authorization</u>; No Conflict. Each Loan Document executed by Borrower has been duly authorized, executed and delivered by Borrower, and the obligations thereunder and the performance thereof by Borrower in accordance with their terms are and will continue to be within Borrower's power and authority without the necessity of joinder or consent of any other person. No provision of the Loan Documents violates or will violate any applicable Law, any covenants or restrictions affecting the Property, any order of any court or governmental authority or any contract or agreement binding on Borrower or the Property.
- 4.3 <u>Enforceability</u>. The Loan Documents constitute legal, valid and binding obligations of Borrower enforceable in accordance with their terms, except as the enforceability thereof may be limited by Debtor Relief Laws and except as the availability of certain remedies may be limited by general principles of equity.
- 4.4 <u>No Violation; No Litigation</u>. Borrower is not in violation of any Law, regulation or ordinance, or any order of any court or Tribunal, and no provision of the Loan Documents violates any applicable Law, any covenants or restrictions affecting the Property, any order of any court or Tribunal or any contract or agreement binding on Borrower or the Property.
- 4.5 <u>Taxes</u>. To the extent required by applicable Law, Borrower and Guarantor have filed all necessary tax-returns and reports and have paid all taxes and governmental charges thereby shown to be owing.
- 4.6 <u>Plans</u>. The Plans are or will be complete in all material respects, contain all necessary detail and are or will be adequate for construction of the Improvements, are satisfactory to Borrower. The Plans have been or will be approved by all applicable governmental authorities and comply with the Loan Documents and all applicable Laws, restrictive covenants, and governmental requirements, rules and regulations. The Plans do, and the Improvements when constructed will, comply with all legal requirements regarding access and facilities for handicapped or disabled persons.
- 4.7 Requirements; Zoning. The Land and Improvements comply with all Laws and governmental requirements, including all subdivision and platting requirements, without reliance on any adjoining or neighboring property. The current and anticipated use of the Property complies with all applicable zoning ordinances, regulations and restrictive covenants affecting the Land without the existence of any variance, non-complying use, nonconforming use or other special exception, all use restrictions of any Tribunal having jurisdiction have been satisfied, and no violation of any Law or regulation exists with respect thereto.
- 4.8 <u>Separate Tax Lot</u>. The Land is not part of a larger tract of land owned by Borrower or any of its affiliates or any Guarantor, is not otherwise included under any unity of title or similar covenant with other lands not encumbered by the Deed of Trust, and constitutes a separate tax lot or lots with a separate tax assessment or assessments for the Land and Improvements, independent of those for any other lands or improvements.
- 4.9 <u>No Conveyance</u>. Borrower has not directly or indirectly conveyed, assigned or otherwise disposed of or transferred (or agreed to do so) any development rights, air rights or

other similar rights, privileges or attributes with respect to the Property, including those arising under any zoning or land use ordinance or other Law or governmental requirement.

- 4.10 <u>Construction Matters</u>. The construction schedule for the Project is realistic and the Completion Date is a reasonable estimate of the time required to complete the Project.
- 4.11 <u>Utilities</u>. All utility services necessary for the development of the Land and the construction of the Improvements and the operation thereof for their intended purpose are, or will be, available at the boundaries of the Land, including electric and natural gas facilities, telephone service, water supply, storm and sanitary sewer facilities.
- 4.12 <u>Financial Matters</u>. Borrower is solvent after giving effect to all borrowings contemplated by the Loan Documents and no proceeding under any Debtor Relief Law is pending (or to Borrower's knowledge threatened) by or against Borrower as a debtor. All Financial Statements (collectively, "<u>Submissions</u>") heretofore furnished or hereafter to be furnished by or on behalf of Borrower to Administrative Agent are true, correct and complete in all material respects as of their respective dates. There has been no event or condition that could reasonably be expected to have a Material Adverse Effect on Borrower's or Guarantor's financial condition from the financial condition of Borrower or Guarantor (as the case may be) indicated in such Submissions.
- 4.13 <u>Borrower Not a Foreign Person</u>. Borrower is not a "foreign person" within the meaning of the Internal Revenue Code, Sections 1445 and 7701 (i.e., Borrower is not a non-resident alien, foreign corporation, foreign partnership, foreign trust or foreign estate as those terms are defined therein and in any regulations promulgated thereunder).
- 4.14 <u>Business Loan</u>. The Loan is solely for business and/or investment purposes, and is not intended for personal, family, household or agricultural purposes. Borrower warrants that the proceeds of the Loan shall be used for commercial purposes and stipulates that the Loan shall be construed for all purposes as a commercial loan.
- 4.15 No Work Prior to Recordation. Prior to recordation of the Deed of Trust, except as disclosed to Administrative Agent in writing or as endorsed against by the Title Insurer, no work of any kind (including destruction or removal of any existing improvements, site work, clearing, grading, grubbing, draining or fencing of the Land) has been or will be commenced or performed on the Land, no equipment or material has been or will be delivered to or placed upon the Land for any purpose whatsoever, and no contract (or memorandum or affidavit thereof) for the supplying of labor, materials or services for the design or construction of the Improvements, or the surveying of the Land or Improvements, and no affidavit or notice of commencement of construction of the Improvements, has been or will be executed or recorded, which could cause a mechanic's or materialmen's lien or similar lien to achieve priority over the Deed of Trust or the rights of Administrative Agent and Lenders thereunder.

ARTICLE 5

DEFAULT AND REMEDIES

Events of Default. The occurrence of any one of the following shall be a default 5.1 under this Agreement ("Default"): (a) any of the Indebtedness is not paid within five (5) days after written notice from Administrative Agent; provided, however, no notice shall be required at maturity; (b) any covenant, agreement, condition, representation or warranty in this Agreement (other than covenants to pay the Indebtedness and other than Defaults expressly listed in this Section) is not fully and timely performed, observed or kept and such failure is not cured within fifteen (15) days after written notice from Administrative Agent; (c) the occurrence of a Default under any other Loan Document (taking into account any applicable notice and cure period set forth in such Loan Document); (d) construction of the Improvements ceases for more than fifteen (15) consecutive days except for Excusable Delays; (e) the construction of the Improvements, or any materials for which an advance has been requested, materially fails to comply with the Plans, the Loan Documents, any laws or governmental requirements, or any applicable restrictive covenants and such failure is not cured within fifteen (15) days after written notice from Administrative Agent; (f) construction of the Improvements is abandoned, Administrative Agent reasonably determines that construction of the Improvements in accordance with this Agreement will not be completed on or before the Completion Date, or Borrower fails to complete construction of the Improvements (and obtain all applicable permits, licenses, certificates and approvals) in accordance with this Agreement on or before the Completion Date; (g) any required permit, license, certificate or approval with respect to the Property lapses or ceases to be in full force and effect and such lapse has a material adverse effect on the operations of the Property (as determined by Administrative Agent in its reasonable discretion), and is not cured within fifteen (15) days after written notice from Administrative Agent after such determination; (h) a Borrower's Deposit is not made with Administrative Agent within fifteen (15) days after Administrative Agent's request therefor in accordance with Section 1.5; (i) construction is enjoined or Borrower or Administrative Agent is enjoined or prohibited from performing any of its respective obligations under any of the Loan Documents and such prohibition is not cured or construction is not continued within fifteen (15) days after written notice from Administrative Agent; (i) the owner of the Property enters into any lease of part or all of the Property which does not comply with the Loan Documents; (k) a lien for the performance of work or the supply of materials which is established against the Property, or any stop notice served on Borrower, the general contractor, or Administrative Agent, remains unsatisfied or unbonded for a period of fifteen (15) days after written notice to the Borrower thereof or service, whichever is earlier; (1) the occurrence of any condition or situation which, in the reasonable determination of Administrative Agent, constitutes a danger to or impairment of the Property or the lien of the Deed of Trust, if such condition or situation is not remedied within ten (10) days after written notice to the Borrower thereof; (m) the entry of a judgment against Borrower or any Guarantor that materially affects its ability to perform under the Loan Documents or the issuance of any attachment, sequestration, or similar writ levied upon any of its property which is not discharged or bonded against within a period of thirty (30) days; (n) Administrative Agent reasonably determines that a material adverse change has occurred in the financial condition of Borrower, any Guarantor or in the condition of the Property; (o) the dissolution or insolvency of Borrower or any Guarantor; (p) Borrower or Guarantor files a bankruptcy petition or makes a general assignment for the benefit of creditors, or a bankruptcy petition is filed against Borrower or

Guarantor and such involuntary bankruptcy petition continues undismissed for a period of ninety (90) days after the filing thereof; or (q) without the prior written consent of Administrative Agent, more than fifty percent (50.0%) of the voting power in Guarantor is transferred to another Person. Notwithstanding the foregoing, transfers of any interest in Guarantor to any person or entity that is directly or indirectly controlled by, controlling or under common control of or with (a) Opus West Corporation, Opus Corporation, Opus, L.L.C., (b) the founder of Opus Corporation, his children, his grandchildren or other members of his family, or (c) the trustee of a trust or trust for the benefit of the founder of Opus Corporation, his children, his grandchildren or other members of his family, shall be permitted hereunder without Administrative Agent's consent so long as Opus West Corporation remains fully obligated for the payment of the Obligations as a guarantor. NOTICE – THE LOAN IS SUBJECT TO ACCELERATION IN THE EVENT OF A TRANSFER WHICH IS PROHIBITED UNDER THIS SECTION 5.1.

Remedies. Upon a Default, Administrative Agent may with the consent of, and 5.2 shall at the direction of, the Required Lenders, without notice, exercise any and all rights and remedies afforded by this Agreement, the other Loan Documents, the Environmental Agreement, Law, equity or otherwise, including (a) declaring any and all Indebtedness immediately due and payable, (b) reducing any claim to judgment, or (c) obtaining appointment of a receiver (to which Borrower hereby consents) and/or judicial or nonjudicial foreclosure under the Deed of Trust. Provided, however, upon a Default, Administrative Agent at its election may (but shall not be obligated to) without the consent of, and shall at the direction of, the Required Lenders. without notice, do any one or more of the following: (a) terminate Lenders' Commitment to lend and any obligation to disburse any Borrower's Deposit hereunder; (b) in its own name on behalf of Lenders or in the name of Borrower, enter into possession of the Property, perform all work necessary to complete construction of the Improvements substantially in accordance with the Plans (as modified as deemed necessary by Administrative Agent), the Loan Documents, and all applicable Laws, governmental requirements and restrictive covenants, and continue to employ Borrower's architect, engineer and any contractor pursuant to the applicable contracts or otherwise; or (c) set off and apply against any Indebtedness, to the extent thereof and to the maximum extent permitted by Law, any and all deposits, funds, or assets in which Administrative Agent has been granted a security interest on behalf of Lenders pursuant to any Loan Document to or for the credit or account of Borrower.

Borrower hereby appoints Administrative Agent as Borrower's attorney in fact, which power of attorney is irrevocable and coupled with an interest, with full power of substitution if Administrative Agent so elects, to do any of the following in Borrower's name upon the occurrence of a Default: (i) use such sums as are necessary, including any proceeds of the Loan and any Borrower's Deposit, make such changes or corrections in the Plans, and employ such architects, engineers, and contractors as may be required, or as Lenders may otherwise consider desirable, for the purpose of completing construction of the Improvements substantially in accordance with the Plans (as modified as deemed necessary by Administrative Agent), the Loan Documents, and all applicable Laws, governmental requirements and restrictive covenants; (ii) execute all applications and certificates in the name of Borrower which may be required for completion of construction of the Improvements; (iii) endorse the name of Borrower on any checks or drafts representing proceeds of any insurance policies, or other checks or instruments payable to Borrower with respect to the Property; (iv) do every act with respect to the construction of the Improvements that Borrower may do; (v) prosecute or defend any action or

proceeding incident to the Property; (vi) pay, settle, or compromise all bills and claims so as to clear title to the Property; and (vii) take over and use all or any part of the labor, materials, supplies and equipment contracted for, owned by, or under the control of Borrower, whether or not previously incorporated into the Improvements. Any amounts expended by Administrative Agent on its own behalf or on behalf of Lenders to construct or complete the Improvements or in connection with the exercise of its remedies herein shall be deemed to have been advanced to Borrower hereunder as a demand obligation owing by Borrower to Administrative Agent or Lenders as applicable and shall constitute a portion of the Indebtedness, regardless of whether such amounts exceed any limits for Indebtedness otherwise set forth herein. Neither Administrative Agent nor any Lender shall have any liability to Borrower for the sufficiency or adequacy of any such actions taken by Administrative Agent.

No delay or omission of Administrative Agent or Lenders to exercise any right, power or remedy accruing upon the happening of a Default shall impair such right, power or remedy or be construed to be a waiver of such Default or any acquiescence therein. No delay or omission on the part of Administrative Agent or Lenders to exercise any option for acceleration of the maturity of the Indebtedness or for foreclosure of the Deed of Trust following any Default as aforesaid, or any other option granted to Administrative Agent and Lenders hereunder in any one or more instances, and no acceptance by Administrative Agent or Lenders of any partial payment on account of the Indebtedness, shall constitute a waiver of such Default, and each such option shall remain continuously in full force and effect. No remedy herein conferred upon or reserved to Administrative Agent and/or Lenders is intended to be exclusive of any other remedies provided for in any Note, any of the other Loan Documents or the Environmental Agreement, and each and every such remedy shall be cumulative, and shall be in addition to every other remedy given hereunder, or under any Note, any of the other Loan Documents or the Environmental Agreement, or now or hereafter existing at Law or in equity or by statute. Every right, power and remedy given to Administrative Agent and Lenders by this Agreement, any Note, any of the other Loan Documents or the Environmental Agreement shall be concurrent, and may be exercised from time to time as often as may be deemed expedient by the Required Lenders. Every right, power and remedy given to Administrative Agent and Lenders by this Agreement, any Note or any of the Loan Documents may be pursued separately, successively or together against Borrower, or the Property or any part thereof, or any personal property granted as security under the Loan Documents.

Regardless of how a Lender may treat payments received from the exercise of remedies under the Loan Documents or the Environmental Agreement for the purpose of its own accounting, for the purpose of computing the Indebtedness, payments shall be applied as elected by Lenders. No application of payments will cure any Event of Default or prevent acceleration or continued acceleration of amounts payable under the Loan Documents or the Environmental Agreement or prevent the exercise or continued exercise of rights or remedies of Administrative Agent and Lenders hereunder or thereunder or at Law or in equity.

ARTICLE 6

ADMINISTRATIVE AGENT

6.1 Appointment and Authorization of Administrative Agent.

Each Lender hereby irrevocably (subject to Section 6.9) appoints, (a) designates and authorizes Administrative Agent to take such action on its behalf under the provisions of this Agreement, each other Loan Document and the Environmental Agreement and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement, any other Loan Document or the Environmental Agreement, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Loan Document or the Environmental Agreement, Administrative Agent shall not have any duties or responsibilities except those expressly set forth herein, nor shall Administrative Agent have or be deemed to have any fiduciary relationship with any Lender. No implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or the Environmental Agreement or shall otherwise exist against Administrative Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Loan Documents and the Environmental Agreement with reference to Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

(b) [Intentionally omitted].

No individual Lender or group of Lenders shall have any right to amend or (c) waive, or consent to the departure of any party from any provision of any Loan Document or the Environmental Agreement, or secure or enforce the obligations of Borrower or any other party pursuant to the Loan Documents or the Environmental Agreement, or otherwise. All such rights, on behalf of Administrative Agent or any Lender or Lenders, shall be held and exercised solely by and at the option of Administrative Agent for the pro rata benefit of Lenders. Such rights, however, are subject to the rights of a Lender or Lenders, as expressly set forth in this Agreement, to approve matters or direct Administrative Agent to take or refrain from taking action as set forth in this Agreement. Except as expressly otherwise provided in this Agreement, the other Loan Documents or the Environmental Agreement, Administrative Agent shall have and may use its sole discretion with respect to exercising or refraining from exercising any discretionary rights or taking or refraining from taking any actions which Administrative Agent is expressly entitled to exercise or take under this Agreement, the other Loan Documents and the Environmental Agreement, including (i) the determination if and to what extent matters or items subject to Administrative Agent's satisfaction are acceptable or otherwise within its discretion, (ii) the making of Administrative Agent Advances, and (iii) the exercise of remedies pursuant to, but subject to, Article 5 or pursuant to any other Loan Document or the Environmental Agreement, and any action so taken or not taken shall be deemed consented to by Lenders.

- (d) In case of the pendency of any receivership, insolvency, liquidation, bankruptcy, reorganization, arrangement, adjustment, composition or other judicial proceeding relative to Borrower or Guarantor, no individual Lender or group of Lenders shall have the right, and Administrative Agent (irrespective of whether the principal of the Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether Administrative Agent shall have made any demand on Borrower) shall be exclusively entitled and empowered on behalf of itself and Lenders, by intervention in such proceeding or otherwise:
 - (i) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loan and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of Lenders and Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of Lenders and Administrative Agent and their respective agents and counsel and all other amounts due Lenders and Administrative Agent under Section 7.11 allowed in such judicial proceeding; and
 - (ii) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender to make such payments to Administrative Agent and, in the event that Administrative Agent shall consent to the making of such payments directly to Lenders, to pay to Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of Administrative Agent and its agents and counsel, and any other amounts due Administrative Agent under Section 7.11.

Nothing contained herein shall be deemed to authorize Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of Lenders except as approved by the Required Lenders or to authorize Administrative Agent to vote in respect of the claims of Lenders except as approved by the Required Lenders in any such proceeding.

- 6.2 <u>Delegation of Duties</u>. Administrative Agent may execute any of its duties under this Agreement, any other Loan Document or the Environmental Agreement by or through agents, employees or attorneys in fact and shall be entitled to advice of counsel and other consultant experts concerning all matters pertaining to such duties. Administrative Agent shall not be responsible for the negligence or misconduct of any agent or attorney in fact that it selects with reasonable care.
- 6.3 <u>Liability of Administrative Agent</u>. No Agent Related Persons shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement, any other Loan Document or the Environmental Agreement or the transactions contemplated hereby (except for its own gross negligence or willful misconduct), or (b) be responsible in any manner to any Lender for any recital, statement, representation or warranty made by Borrower or any subsidiary or Affiliate of Borrower, or any officer thereof, contained herein, or in any other Loan Document or the Environmental Agreement, or in any certificate,

report, statement or other document referred to or provided for in, or received by Administrative Agent under or in connection with, this Agreement, any other Loan Document or the Environmental Agreement, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement, any other Loan Document or the Environmental Agreement, or for any failure of Borrower or any other party to any Loan Document or the Environmental Agreement to perform its obligations hereunder or thereunder. No Agent Related Person shall be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document or the Environmental Agreement, or to inspect the properties, books or records of Borrower, Guarantor, or any of their Affiliates.

- Reliance by Administrative Agent. Administrative Agent shall be entitled to rely, 6.4 and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper person or persons, and upon advice and statements of legal counsel (including counsel to any party to the Loan Documents or the Environmental Agreement), independent accountants and other experts selected by Administrative Agent. Administrative Agent shall be fully justified in failing or refusing to take any action under this Agreement, any other Loan Document or the Environmental Agreement unless it shall first receive such advice or concurrence of the Required Lenders or all Lenders if required hereunder as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement, any other Loan Document or the Environmental Agreement in accordance with a request or consent of the Required Lenders or such greater number of Lenders as may be expressly required hereby in any instance, and such request and any action taken or failure to act pursuant thereto shall be binding upon all Lenders. In the absence of written instructions from the Required Lenders or such greater number of Lenders, as expressly required hereunder, Administrative Agent may take or not take any action, at its discretion, unless this Agreement specifically requires the consent of the Required Lenders or such greater number of Lenders.
- 6.5 Notice of Default. Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default, unless Administrative Agent shall have received written notice from a Lender or Borrower referring to this Agreement, describing such Default that Administrative Agent determines will have a Material Adverse Effect. Administrative Agent will notify Lenders of its receipt of any such notice. Administrative Agent shall take such action with respect to such Default as may be requested by the Required Lenders in accordance with Article 5; provided, however, that unless and until Administrative Agent has received any such request, Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default as it shall deem advisable or in the best interest of Lenders.

6.6 Credit Decision; Disclosure of Information by Administrative Agent.

(a) Each Lender acknowledges that none of Agent Related Persons has made any representation or warranty to it, and that no act by Administrative Agent hereafter taken,

including any consent to and acceptance of any assignment or review of the affairs of Borrower and/or Guarantor, shall be deemed to constitute any representation or warranty by any Agent-Related Person to Lenders or any of them as to any matter, including whether Agent Related Persons have disclosed material information in their possession. Each Lender represents to Administrative Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of Borrower and Guarantor, and all applicable bank or other regulatory Laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to Borrower hereunder. Each Lender also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents and the Environmental Agreement, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of Borrower and Guarantor.

- Administrative Agent upon its receipt shall provide each Lender such notices, reports and other documents expressly required to be furnished to Lenders by Administrative Agent herein. To the extent not already available to a Lender, Administrative Agent shall also provide the Lender and/or make available for the Lender's inspection during reasonable business hours and at the Lender's expense, upon the Lender's written request therefor: (i) copies of the Loan Documents and the Environmental Agreement; (ii) such information as is then in Administrative Agent's possession in respect of the current status of principal and interest payments and accruals in respect of the Loan; (iii) copies of all current financial statements in respect of Borrower, or any Guarantor or other person liable for payment or performance by Borrower of any obligations under the Loan Documents or the Environmental Agreement, then in Administrative Agent's possession with respect to the Loan; and (iv) other current factual information then in Administrative Agent's possession with respect to the Loan and bearing on the continuing creditworthiness of Borrower or any Guarantor, or any of their respective Affiliates; provided that nothing contained in this Section 6.6 shall impose any liability upon Administrative Agent for its failure to provide a Lender any of such Loan Documents, the Environmental Agreement, information, or financial statements, unless such failure constitutes willful misconduct or gross negligence on Administrative Agent's part; and provided, further, that Administrative Agent shall not be obligated to provide any Lender with any information in violation of Law or any contractual restrictions on the disclosure thereof (provided such contractual restrictions shall not apply to distributing to a Lender factual and financial information expressly required to be provided herein). Except as set forth above, Administrative Agent shall not have any duty or responsibility to provide Lenders or any of them with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of Borrower or Guarantor or any of their respective Affiliates which may come into the possession of any of Agent Related Persons.
- 6.7 <u>Indemnification of Administrative Agent</u>. Whether or not the transactions contemplated hereby are consummated, Lenders shall indemnify upon demand each Agent Related Person (to the extent not reimbursed by or on behalf of Borrower and without limiting the obligation of Borrower to do so), pro rata, and hold harmless each Agent Related Person

from and against any and all Indemnified Liabilities incurred by it; provided, however, that no Lender shall be liable for the payment to any Agent Related Person of any portion of such Indemnified Liabilities to the extent determined in a final, non-appealable judgment by a court of competent jurisdiction to have resulted from such Agent-Related Person's own gross negligence or willful misconduct; provided, further, that no action taken in accordance with the directions of the Required Lenders shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section 6.7. Without limitation of the foregoing, to the extent that Administrative Agent is not reimbursed by or on behalf of Borrower, each Lender shall reimburse Administrative Agent upon demand for its ratable share of any costs or out of pocket expenses (including attorneys' fees) incurred by Administrative Agent as described in Section 7.11. The undertaking in this Section 6.7 shall survive the payment of all Indebtedness hereunder and the resignation or replacement of Administrative Agent.

- Administrative Agent in Individual Capacity. Administrative Agent, in its 6.8 individual capacity, and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with any party to the Loan Documents or the Environmental Agreement and their respective Affiliates as though Administrative Agent were not Administrative Agent hereunder and without notice to or consent of Lenders. Lenders acknowledge that Borrower and Bank of America, N.A. or its Affiliate have entered or may enter into Swap Transactions. A portion of the Loan may be funded to honor Borrower's payment obligations under the terms of such Swap Transactions, and Lenders shall have no right to share in any portion of such payments. Lenders acknowledge that, pursuant to such activities, Bank of America, N.A. or its Affiliates may receive information regarding any party to the Loan Documents or the Environmental Agreement or their respective Affiliates (including information that may be subject to confidentiality obligations in favor of such parties or such parties' Affiliates) and acknowledge that Administrative Agent shall be under no obligation to provide such information to them. With respect to its Pro Rata Share of the Loan, Bank of America, N.A. shall have the same rights and powers under this Agreement as any other Lenders and may exercise such rights and powers as though it were not Administrative Agent or party to Swap Transactions, and the terms "Lender" and "Lenders" include Bank of America, N.A. in its individual capacity.
- 6.9 Successor Administrative Agent. Administrative Agent may, and at the request of the Required Lenders as a result of Administrative Agent's gross negligence or willful misconduct in performing its duties under this Agreement shall, resign as Administrative Agent upon thirty (30) days' notice to Lenders. If Administrative Agent resigns under this Agreement, the Required Lenders shall appoint from among Lenders a successor administrative agent for Lenders, which successor administrative agent shall be consented to by Borrower at all times other than during the existence of a Default (which consent of Borrower shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of Administrative Agent, Administrative Agent may appoint, after consulting with Lenders and Borrower, a successor administrative agent from among Lenders. Upon the acceptance of its appointment as successor administrative agent hereunder, such successor administrative agent and the term "Administrative Agent" shall mean such successor administrative agent, and the retiring Administrative Agent's appointment, powers and duties as

Administrative Agent shall be terminated. After any retiring Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article 6 and other applicable Sections of this Agreement shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement. If no successor administrative agent has accepted appointment as Administrative Agent by the date which is thirty (30) days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent's resignation shall nevertheless thereupon become effective and Lenders shall perform all of the duties of Administrative Agent hereunder until such time, if any, as the Required Lenders appoint a successor agent as provided for above.

6.10 Releases; Acquisition and Transfers of Collateral.

- (a) Lenders hereby irrevocably authorize Administrative Agent to transfer or release any lien on, or after foreclosure or other acquisition of title by Administrative Agent on behalf of the Lenders to transfer or sell, any Loan collateral (i) upon the termination of the Commitments and payment and satisfaction in full of all Indebtedness, (ii) in connection with a release, transfer or sale of a lien or Loan collateral if Borrower will certify to Administrative Agent that the release, transfer or sale is permitted under this Agreement or the other Loan Documents (and Administrative Agent may rely conclusively on any such certificate without further inquiry), or (iii) after foreclosure if approved by the Required Lenders.
- If all or any portion of the Loan collateral is acquired by foreclosure or by deed in lieu of foreclosure, Administrative Agent shall take title to the collateral in its name or by an Affiliate of Administrative Agent, but for the benefit of all Lenders in their Pro Rata Shares on the date of the foreclosure sale or recordation of the deed in lieu of foreclosure (the Administrative Agent and all Lenders hereby expressly waive and "Acquisition Date"). relinquish any right of partition with respect to any collateral so acquired. After any collateral is acquired, Administrative Agent shall appoint and retain one or more persons (individually and collectively, the "Property Manager") experienced in the management, leasing, sale and/or disposition of similar properties. After consulting with the Property Manager, Administrative Agent shall prepare a written plan for completion of construction (if required), operation, management, improvement, maintenance, repair, sale and disposition of the Loan collateral and a budget for the aforesaid, which may include a reasonable management fee payable to Administrative Agent (the "Business Plan"). Administrative Agent will deliver the Business Plan not later than the sixtieth (60th) day after the Acquisition Date to each Lender with a written request for approval of the Business Plan. If the Business Plan is approved by the Required Lenders, Administrative Agent and the Property Manager shall adhere to the Business Plan until a different Business Plan is approved by the Required Lenders. Administrative Agent may propose an amendment to the Business Plan as it deems appropriate, which shall also be subject to Required Lender approval. If the Business Plan (as may be amended) proposed by Administrative Agent is not approved by the Required Lenders, or if sixty (60) days have elapsed following the Acquisition Date without a Business Plan being proposed by Administrative Agent, any Lender may propose an alternative Business Plan, which Administrative Agent shall submit to all Lenders for their approval. If an alternative Business Plan is approved by the Required Lenders, Administrative Agent may appoint one of the approving Lenders to implement the alternative Business Plan. Notwithstanding any other provision of this Agreement, unless in violation of an approved Business Plan or otherwise in an emergency

situation, Administrative Agent shall, subject to clause (a) of this <u>Section 6.10</u>, have the right but not the obligation to take any action in connection with the Loan collateral (including those with respect to property taxes, insurance premiums, completion of construction, operation, management, improvement, maintenance, repair, sale and disposition), or any portion thereof.

- (c) Upon request by Administrative Agent or Borrower at any time, Lenders will confirm in writing Administrative Agent's authority to sell, transfer or release any such liens of particular types or items of Loan collateral pursuant to this Section 6.10; provided, however, that (i) Administrative Agent shall not be required to execute any document necessary to evidence such release, transfer or sale on terms that, in Administrative Agent's opinion, would expose Administrative Agent to liability or create any obligation or entail any consequence other than the transfer, release or sale without recourse, representation or warranty, and (ii) such transfer, release or sale shall not in any manner discharge, affect or impair the obligations of Borrower other than those expressly being released.
- (d) If only two (2) Lenders exist at the time Administrative Agent receives a purchase offer for Loan collateral for which one of the Lenders does not consent within ten (10) Business Days after notification from Administrative Agent, the consenting Lender may offer ("Purchase Offer") to purchase all of the non-consenting Lender's right, title and interest in the collateral for a purchase price equal to the non-consenting Lender's Pro Rata Share of the net proceeds anticipated from such sale of such collateral (as reasonably determined by Administrative Agent, including the undiscounted face principal amount of any purchase money obligation not payable at closing) ("Anticipated Net Proceeds"). Within ten (10) Business Days thereafter the non-consenting Lender shall be deemed to have accepted such Purchase Offer unless the non-consenting Lender notifies Administrative Agent that it elects to purchase all of the consenting Lender's right, title and interest in the collateral for a purchase price payable by the non-consenting Lender in an amount equal to the consenting Lender's Pro Rata Share of the Anticipated Net Proceeds. Any amount payable hereunder by a Lender shall be due on the earlier to occur of the closing of the sale of the collateral or ninety (90) days after the Purchase Offer, regardless of whether the collateral has been sold.
- Application of Payments. Except as otherwise provided below with respect to Defaulting Lenders, aggregate principal and interest payments, payments for Indemnified Liabilities, and/or foreclosure or sale of the collateral, and net operating income from the collateral during any period it is owned by Administrative Agent on behalf of Lenders ("Payments") shall be apportioned pro rata among Lenders and payments of any fees (other than fees designated for Administrative Agent's separate account) shall, as applicable, be apportioned pro rata among Lenders. Notwithstanding anything to the contrary in this Agreement, all Payments due and payable to Defaulting Lenders shall be due and payable to and be apportioned pro rata among Administrative Agent and Electing Lenders. Such apportionment shall be in the proportion that the Defaulting Lender Payment Amounts paid by them bears to the total Defaulting Lender Payment Amounts of such Defaulting Lender. Such apportionment shall be made until Administrative Agent and Lenders have been paid in full for the Defaulting Lender Payment Amounts. All pro rata Payments shall be remitted to Administrative Agent and all such payments not constituting payment of specific fees, and all proceeds of the Loan collateral received by Administrative Agent, shall be applied first, to pay any fees, indemnities, costs, expenses (including those in Section 6.7) and reimbursements then due to Administrative Agent

from Borrower; second, to pay any fees, costs, expenses and reimbursements then due to Lenders from Borrower; third, to pay pro rata interest and late charges due in respect of the Indebtedness and Administrative Agent Advances; fourth, to pay or prepay pro rata principal of the Indebtedness and Administrative Agent Advances; fifth, to pay any indebtedness of Borrower under Swap Transactions; and last, to Borrower, if required by Law, or Lenders in Pro Rata Share percentages equal to their percentages at the termination of the Aggregate Commitments.

- 6.12 <u>Benefit</u>. The terms and conditions of this <u>Article 6</u> are inserted for the sole benefit of Administrative Agent and Lenders; the same may be waived in whole or in part, with or without terms or conditions, without prejudicing Administrative Agent's or Lenders' rights to later assert them in whole or in part.
- 6.13 Co Agents; Lead Managers. None of the Lenders or other persons identified on the facing page or signature pages of this Agreement as a "syndication agent," "documentation agent," "co agent," "book manager," or "lead manager," "arranger," "lead arranger" or "coarranger" shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than, in the case of such lenders, those applicable to all Lenders as such. Without limiting the foregoing, none of the Lenders or other persons so identified as a "syndication agent," "documentation agent," "co-agent" or "lead manager" shall have or be deemed to have any fiduciary relationship- with Lenders or any of them. Each Lender acknowledges that it has not relied, and will not rely, on any of Lenders or other persons so identified in deciding to enter into this Agreement or in taking or not taking action hereunder.

ARTICLE 7

GENERAL TERMS AND CONDITIONS

Consents. Except where otherwise expressly provided in the Loan Documents or 7.1 the Environmental Agreement, in any instance where the approval or consent of Administrative Agent or Lenders or the exercise of judgment by Administrative Agent or Lenders is required, the granting or denial of such approval or consent and the exercise of such judgment shall be (a) within the sole discretion of Administrative Agent or Lenders, and (b) deemed to have been given only by a specific writing intended for the purpose given and executed by Administrative Agent or Lenders. Notwithstanding any approvals or consents by Administrative Agent or Lenders, neither Administrative Agent nor any Lender has any obligation or responsibility whatsoever for the adequacy, form or content of the Plans, the Budget, any appraisal, any contract, any change order, any lease, or any other matter incident to the Property or the construction of the Improvements. Administrative Agent's or Lenders' acceptance of an assignment of the Plans for the benefit of Administrative Agent and Lenders shall not constitute approval of the Plans. Any inspection, appraisal or audit of the Property or the books and records of Borrower, or the procuring of documents and financial and other information by or on behalf of Administrative Agent, shall be for the protection of Administrative Agent and Lenders only, and shall not constitute an assumption of responsibility to Borrower or anyone else with regard to the condition, value, construction, maintenance or operation of the Property, or relieve Borrower of any of Borrower's obligations. Borrower has selected all surveyors, architects, engineers, contractors, materialmen and all other persons or entities furnishing services or materials to the Project. Neither Administrative Agent nor any Lender has any duty to supervise

or to inspect the Property or the construction of the Improvements, or has any duty of care to Borrower or any other person to protect against, or to inform Borrower or any other person of the existence of, any negligent, faulty, inadequate or defective design or construction of the Improvements.

Borrower's Indemnity. Neither Administrative Agent nor any Lender shall be 7.2 liable or responsible for, and Borrower shall indemnify each Agent-Related Person and each Lender and their respective Affiliates, directors, officers, agents, attorneys and employees (collectively, the "Indemnitees") from and against: (a) any claim, action, loss or cost (including attorneys' fees and costs) arising from or relating to (i) any defect in the Property or the Improvements, (ii) the performance or default of Borrower, Borrower's surveyors, architects, engineers or contractors, the Construction Consultant or any other person, (iii) any failure to construct, complete, protect or insure the Improvements, (iv) the payment of costs of labor, materials or services supplied for the construction of the Improvements, (v) in connection with the protection and preservation of the Loan collateral (including those with respect to property taxes, insurance premiums, completion of construction, operation, management, improvements, maintenance, repair, sale and disposition), or (vi) the performance of any obligation of Borrower whatsoever; (b) any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including attorneys' fees and costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (i) the execution, delivery, enforcement, performance or administration of any Loan Document, the Environmental Agreement or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, (ii) any Commitment or Loan; or (iii) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether any Indemnitee is a party thereto; (c) any and all claims, demands, actions or causes of action arising out of or relating to the use of Information (as defined in Section 7.7) or other materials obtained through internet, Intralinks or other similar information transmission systems in connection with this Agreement; and (d) any and all liabilities, losses, costs or expenses (including attorneys' fees and costs) that any Indemnitee suffers or incurs as a result of the assertion of any of the foregoing claims, demands, actions, causes of action or proceedings, or as a result of the preparation of any defense in connection with any of the foregoing claims, demands, actions, causes of action or proceedings, in all cases, whether or not an Indemnitee is a party to such claim, demand, action, cause of action or proceeding and whether it is defeated, successful or withdrawn (all of the foregoing, collectively, the "Indemnified Liabilities"); provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. Nothing, including any advance or acceptance of any document or instrument, shall be construed as a representation or warranty, express or implied, to any party by Inspection shall not constitute an acknowledgment or Administrative Agent or Lenders. representation by Administrative Agent or any Lender or Construction Consultant that there has been or will be compliance with the Plans, the Loan Documents, the Environmental Agreement

or applicable Laws, governmental requirements or restrictive covenants, or that the construction is free from defective materials or workmanship. Inspection, whether or not followed by notice of Default, shall not constitute a waiver of any Default then existing, or a waiver of Administrative Agent's and Lenders' right thereafter to insist that the Improvements be constructed in accordance with the Plans, the Loan Documents, and all applicable Laws, governmental requirements and restrictive covenants. Administrative Agent's failure to inspect shall not constitute a waiver of any rights of Administrative Agent or Lenders under the Loan Documents or the Environmental Agreement or at Law or in equity.

Miscellaneous. This Agreement may be executed in several counterparts, all of which are identical, and all of which counterparts together shall constitute one and the same instrument. The Loan Documents and the Environmental Agreement are for the sole benefit of Administrative Agent, Lenders and Borrower and are not for the benefit of any third party. A determination that any provision of this Agreement is unenforceable or invalid shall not affect the enforceability or validity of any other provision and a determination that the application of any provision of this Agreement to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons, entities or circumstances. Time shall be of the essence with respect to Borrower's obligations under the Loan Documents and the Environmental Agreement. This Agreement and its validity, enforcement and interpretation shall be governed by Arizona Law (without regard to any conflict of Laws principles) and applicable United States federal Law.

7.4 Notices.

Modes of Delivery; Changes. Except as otherwise provided herein, all (a) notices and other communications required or which any party desires to give under this Agreement, any other Loan Document or the Environmental Agreement shall be in writing. Unless otherwise specifically provided in such other Loan Document or the Environmental Agreement, all such notices and other communications shall be deemed sufficiently given or furnished if delivered by personal delivery, by courier, by registered or certified United States mail, postage prepaid, or by facsimile (with, subject to clause (b) below, a confirmatory duplicate copy sent by first class United States mail), addressed to the party to whom directed or by (subject to clause (c) below) electronic mail address to Borrower, at the addresses set forth at the end of this Agreement or to Administrative Agent or Lenders at the addresses specified for notices on the Schedule of Lenders (unless changed by similar notice in writing given by the particular party whose address is to be changed). Any such notice or communication shall be deemed to have been given and received either at the time of personal delivery, or in the case of courier or mail, as of the date of first attempted delivery at the address and in the manner provided herein, or in the case of facsimile, upon receipt; provided, however, that service of a notice required by any applicable statute shall be considered complete when the requirements of that statute are met. Notwithstanding the foregoing, no notice of change of address shall be effective except upon actual receipt. This Section 7.4 shall not be construed in any way to affect or impair any waiver of notice or demand provided in any Loan Document or the Environmental Agreement or to require giving of notice or demand to or upon any person in any situation or for any reason.

- (b) <u>Effectiveness of Facsimile Signatures</u>. The Loan Documents and the Environmental Agreement may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on all parties to the Loan Documents and the Environmental Agreement. Administrative Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; <u>provided</u>, <u>however</u>, that the failure to request or deliver the same shall not affect the effectiveness of any facsimile document or signature.
- (c) <u>Limited Use of Electronic Mail</u>. Electronic mail and internet and intranet websites may be used only to distribute routine communications, such as financial statements and other information, and to distribute the Loan Documents and the Environmental Agreement for execution by the parties thereto, and may not be used for any other purpose.
- Reliance by Administrative Agent and Lenders. Administrative Agent and Lenders shall be entitled to rely and act upon any notices (including telephonic Loan advance notices) purportedly given by or on behalf of Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. Borrower shall indemnify each Agent-Related Person and each Lender from all losses, costs, expenses and liabilities resulting from the reliance by such person on each notice purportedly given by or on behalf of Borrower. All telephonic notices to and other communications with Administrative Agent may be recorded by Administrative Agent, and each of the parties hereto hereby consents to such recording. If a Lender does not notify or inform Administrative Agent as to whether or not it consents to, or approves of, or agrees to any matter of any nature whatsoever with respect to which its consent, approval or agreement is required under the express provisions of this Agreement or with respect to which its consent, approval or agreement is otherwise requested by Administrative Agent, in connection with the Loan or any matter pertaining to the Loan, within ten (10) Business Days (or such longer period as may be specified by Administrative Agent) after such consent, approval or agreement is requested by Administrative Agent, Lender shall be deemed to have given its consent, approval or agreement, as the case may be, with respect to the matter in question.
- 7.5 Payments Set Aside. To the extent that any payment by or on behalf of Borrower is made to Administrative Agent or any Lender, or Administrative Agent or any Lender exercises any right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by Administrative Agent or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law, to a depository (including Administrative Agent, any Lender or its or their Affiliates) for returned items or insufficient collected funds or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force as if such payment had not been made or such set-off had not occurred, and (b) each Lender severally agrees to pay to Administrative Agent upon demand its applicable share of any amount so recovered from or repaid by Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect.

Notwithstanding the foregoing, no Lender Party (as defined in <u>Section 7.8</u>) may exercise any right of set-off except in accordance with <u>Section 7.8</u>.

7.6 Successors and Assigns.

- (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender, and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of clause (b) below, (ii) by way of participation in accordance with the provisions of clause (d) below, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of clause (f) below (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any person (other than the parties hereto, their respective successors and assigns permitted hereby and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) Any Lender may assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and Pro Rata Share of the Loan at the time owing to it); provided that:
 - (i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and Pro Rata Share of the Loan at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender or an Approved Fund as defined in Exhibit "B" with respect to a Lender, the aggregate amount of the Commitment (which for this purpose includes its Pro Rata Share of the Loan outstanding) subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to Administrative Agent, shall not be less than \$10,000,000.00 unless each of Administrative Agent and, so long as no Default has occurred and is continuing, Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed);
 - (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to its Pro Rata Share of the Loan and the Commitment assigned; and
 - (iii) any assignment of a Commitment must be approved by Administrative Agent, unless the person that is the proposed assignee is itself a Lender (whether or not the proposed assignee would otherwise qualify as an Eligible Assignee); and
 - (iv) the parties to each assignment shall execute and deliver to Administrative Agent an Assignment and Assumption, together with a processing and recordation fee of \$3,500.00 plus the cost of any applicable endorsement to the Title Insurance or new Title Insurance.

Subject to acceptance and recording thereof by Administrative Agent pursuant to clause (c) of this Section 7.6, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Assumption, shall have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of this Agreement with respect to Borrower's obligations surviving termination of this Agreement). Upon request, Administrative Agent shall prepare and Borrower shall execute and deliver a Note ("Replacement Note") to the assignee Lender evidencing the assignee Lender's Pro Rata Share of the Loan. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this clause (b) shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with clause (d) of this Section 7.6.

- (c) Administrative Agent, acting solely for this purpose as an agent of Borrower, shall forward the Assignment and Assumption, and the Replacement Note to the Title Company for issuance of an applicable endorsement to the Title Insurance or new Title Insurance, and shall maintain at Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of Lenders, and the Commitments of, and principal amount of each Lender's Pro Rata Share of the Loan owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and Borrower, Administrative Agent and Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) Any Lender may, without the consent of, but with prior notice to Administrative Agent, sell participations to one or more banks or other entities (a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or its Pro Rata Share of the Loan owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) Borrower, Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement, and (iv) except to the extent consented to by Administrative Agent in its sole discretion with respect to each participation, any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement.
- (e) A Participant shall not be entitled to receive any greater payment under Sections 2.1 through 2.10, inclusive, than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant.

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release a Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) If the consent of Borrower to an assignment or to an assignee is required hereunder (including a consent to an assignment which does not meet the minimum assignment threshold specified in clause (i) of the provision to the first sentence of clause (b) above), Borrower shall be deemed to have given its consent five (5) Business Days after the date notice thereof has been delivered by the assigning Lender (through Administrative Agent) unless such consent is expressly refused by Borrower prior to such fifth Business Day.

(h) [Intentionally omitted].

Confidentiality. Administrative Agent and each Lender agrees to maintain the 7.7 confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its and its Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the persons to whom such disclosure is made... will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (b) to the extent requested by any regulatory authority; (c) to the extent required by applicable Laws or regulations or by any subpoena or similar legal process; (d) to any other party to this Agreement; (e) in connection with the exercise of any remedies hereunder or any suit, action or proceeding relating to this Agreement or the enforcement of rights hereunder; (f) subject to an agreement containing provisions substantially the same as those of this Section 7.7, to (i) any assignee of or participant in, any of its rights or obligations under this Agreement, or (ii) any direct or indirect contractual counterparty or prospective counterparty (or such contractual counterparty's or prospective counterparty's professional advisor) to any Swap Contract or credit derivative transaction relating to obligations to Borrower or Guarantor; (g) with the consent of Borrower; (h) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section 7.7, or (ii) becomes available to Administrative Agent or any Lender on a nonconfidential basis from a source other than Borrower; or (i) to the National Association of Insurance Commissioners or any other similar organization. For the purposes of this Section 7.7, "Information" means all information received from Borrower or Guarantor relating to Borrower or Guarantor or their business, other than any such information that is available to Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by Borrower or Guarantor; provided that, in the case of information received from Borrower or Guarantor after the date hereof, such information is clearly identified in writing at the time of delivery as confidential. Any person required to maintain the confidentiality of Information as provided in this Section 7.7 shall be considered to have complied with its obligation to do so if such person has exercised the same degree of care to maintain the confidentiality of such Information as such person would accord to its own confidential information. Administrative Agent and Lenders may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry, and service providers to Administrative Agent and Lenders in

connection with the administration and management of this Agreement, the Loan and the Loan Documents.

- 7.8 No Set-off. Neither Administrative Agent nor any Lender nor any assignee, Participant or Affiliate thereof (each, a "Lender Party") shall proceed directly, by right of set-off, banker's lien, counterclaim or otherwise, against any assets of Borrower or Guarantor (including any general or special, time or demand, provisional or other deposits or other indebtedness owing by such Lender Party to or for the credit or the account of Borrower or Guarantor) for the purpose of applying such assets against the Indebtedness, without the prior written consent of all Lenders.
- Sharing of Payments. If, other than as expressly provided elsewhere herein, any 7.9 Lender shall obtain on account of the portions of the Loan advanced by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off subject to Section 7.8 or otherwise) in excess of its ratable share (or other share contemplated hereunder) thereof, such Lender shall immediately (a) notify Administrative Agent of such fact, and (b) purchase from the other Lenders such participations in the portions of the Loan made by them as shall be necessary to cause such purchasing Lender to share the excess payment in respect of such portions of the Loan or such participations, as the case may be, pro rata with each of them; provided, however, that if all or any portion of such excess payment is thereafter recovered from the purchasing Lender under any of the circumstances described in Section 7.5 (including pursuant to any settlement entered into by the purchasing Lender in its discretion), such purchase shall to that extent be rescinded and each other Lender shall repay to the purchasing Lender the purchase price paid therefor, together with an amount equal to such paying Lender's ratable share (according to the proportion of (i) the amount of such paying Lender's required payment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered without further interest thereon. Borrower agrees that any Lender so purchasing a participation from another Lender may, to the fullest extent permitted by Law, exercise all of its rights of payment (including the right of set-off, but subject to Section 7.8) with respect to such participation as fully as if such Lender were the direct creditor of Borrower in the amount of such participation. Administrative Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of participations purchased under this Section 7.9 and will in each case notify Lenders following any such purchases or repayments. Each Lender that purchases a participation interest pursuant to this Section 7.9 shall from and after such purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Obligations purchased.
- 7.10 Amendments; Survival. Administrative Agent and Lenders shall be entitled to amend (whether pursuant to a separate intercreditor agreement or otherwise) any of the terms, conditions or agreements set forth in Article 6 or as to any other matter in the Loan Documents or the Environmental Agreement respecting payments to Administrative Agent or Lenders or the required number of Lenders to approve or disapprove any matter or to take or refrain from taking any action, without the consent of Borrower or any other Person or the execution by Borrower or any other Person of any such amendment or intercreditor agreement. Subject to the foregoing, Administrative Agent may amend or waive any provision of this Agreement, any other Loan

Document or the Environmental Agreement, or consent to any departure by any party to the Loan Documents or the Environmental Agreement therefrom which amendment, waiver or consent is intended to be within Administrative Agent's discretion or determination, or otherwise in Administrative Agent's reasonable determination shall not have a Material Adverse Effect; provided, however, otherwise no such amendment, waiver or consent shall be effective unless in writing, signed by the Required Lenders and Borrower or the applicable party to the Loan Documents or the Environmental Agreement, as the case may be, and acknowledged by Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided further, however, no such amendment, waiver or consent shall:

- (a) extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 5.2</u>), without the written consent of such Lender (it being understood that a waiver of a Default shall not constitute an extension or increase in any Lender's Commitment);
- (b) postpone any date fixed by this Agreement, any other Loan Document or the Environmental Agreement for any payment of principal, interest, fees or other amounts due to Lenders (or any of them) hereunder or under any other Loan Document or the Environmental Agreement, without the written consent of each Lender directly affected thereby;
- (c) reduce the principal of, or the rate of interest specified herein on, any portion of the Loan, or any fees or other amounts payable hereunder or any other Loan Document or the Environmental Agreement, without the written consent of each Lender directly affected thereby; provided, however, that Administrative Agent may waive any obligation of Borrower to pay interest at the Past Due Rate and/or late charges for periods of up to thirty (30) days, and only the consent of the Required Lenders shall be necessary to waive any obligation of Borrower to pay interest at the Past Due Rate or late charges thereafter, or to amend the definition of "Past Due Rate" or "late charges";
- (d) change the percentage of the combined Commitments or of the aggregate unpaid principal amount of the Loan which is required for Lenders or any of them to take any action hereunder, without the written consent of each Lender;
- (e) change the definition of "Pro Rata Share" or "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender;
- (f) amend Sections 3.14, 3.15, 3.16, 7.8, 7.9 or this Section 7.10, without the written consent of each Lender;
- (g) release the liability of Borrower or any existing Guarantor without the written consent of each Lender;
- (h) permit the sale, transfer, pledge, mortgage or assignment of any Loan collateral or any direct or indirect interest in Borrower, except as expressly permitted under the Loan Documents, without the written consent of each Lender; or

(i) transfer or release any lien on, or after foreclosure or other acquisition of title by Administrative Agent on behalf of Lenders transfer or sell, any Loan collateral except as permitted in Section 6.10, without the written consent of each Lender;

and, provided, further, that no amendment, waiver or consent shall, unless in writing and signed by Administrative Agent in addition to the Lenders required above, affect the rights or duties of Administrative Agent under this Agreement, any other Loan Document or the Environmental Agreement. Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder, except that the Commitment of such Lender may not be increased without the consent of such Lender.

This Agreement shall continue in full force and effect until the Indebtedness is paid in full and all of Administrative Agent's and Lenders' obligations under this Agreement are terminated; and all representations and warranties and all provisions herein for indemnity of the Indemnitees, Administrative Agent and Lenders (and any other provisions herein specified to survive) shall survive payment in full, satisfaction or discharge of the Indebtedness, the resignation or removal of Administrative Agent or replacement of any Lender, and any release or termination of this Agreement or of any other Loan Documents.

Costs and Expenses. Without limiting any Loan Document or the Environmental 7.11 Agreement and to the extent not prohibited by applicable Laws, Borrower shall pay when due, shall reimburse to Administrative Agent for the benefit of itself and Lenders on demand and shall indemnify Administrative Agent and Lenders from, all out-of-pocket fees, costs and expenses paid or incurred by Administrative Agent in connection with the negotiation, preparation and execution of this Agreement, the other Loan Documents and the Environmental Agreement (and any amendments, approvals, consents, waivers and releases requested, required, proposed or done from time to time), or in connection with the disbursement, administration or collection of the Loan or the enforcement of the obligations of Borrower or the exercise of any right or remedy of Administrative Agent, including: (a) all fees and expenses of Administrative Agent's counsel (including the market value of services of in-house counsel); (b) fees and charges of each Construction Consultant, inspector and engineer; (c) appraisal, re-appraisal and survey costs; (d) title insurance charges and premiums; (e) title search or examination costs, including abstracts, abstractors' certificates and Uniform Commercial Code searches; (f) judgment and tax lien searches for Borrower and each Guarantor; (g) escrow fees; (h) fees and costs of environmental investigations, site assessments and remediations; (i) recordation taxes, documentary taxes, transfer taxes and mortgage taxes; (i) filing and recording fees; and (k) loan brokerage fees. Borrower shall pay all costs and expenses incurred by Administrative Agent, including attorneys' fees (including the market value of services of in-house counsel), if the obligations or any part thereof are sought to be collected by or through an attorney at law, whether or not involving probate, appellate or administrative proceedings or proceedings under any Debtor Relief Law. Borrower shall pay all costs and expenses of complying with the Loan Documents and the Environmental Agreement, whether or not such costs and expenses are included in the Budget. Borrower's obligations under this Section 7.11 shall survive the delivery of the Loan Documents and the Environmental Agreement, the making of advances, the payment in full of the Indebtedness, the release or reconveyance of any of the Loan Documents, the foreclosure of the Deed of Trust or conveyance in lieu of foreclosure, any proceeding under any Debtor Relief Law, and any other event whatsoever.

7.12 Tax Forms.

- Each Lender and each holder of a participation interest herein that (a) (i) is not a "United States person" (a "Foreign Lender") within the meaning of Section 7701(a)(30) of the Internal Revenue Code of 1986, as amended from time to time (the "Code") shall deliver to Administrative Agent, prior to receipt of any payment subject to withholding (or after accepting an assignment or receiving a participation interest herein), two duly-signed completed copies of either Form W-8BEN or any successor thereto (relating to such Foreign Lender and entitling it to a complete exemption from withholding on all payments to be made to such Foreign Lender by Borrower pursuant to this Agreement) or Form W-8ECI or any successor thereto (relating to all payments to be made to such Foreign Lender by Borrower pursuant to this Agreement) of the United States Internal Revenue Service or such other evidence satisfactory to Borrower and Administrative Agent that such Foreign Lender is entitled to an exemption from or reduction of, United States withholding tax, including any exemption pursuant to Section 881(c) of the Code. Thereafter and from time to time, each such Foreign Lender shall (A) promptly submit to Administrative Agent such additional duly-completed and signed copies of one of such forms (or such successor forms as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then-current-United States Laws and regulations to avoid, or such evidence as .- issatisfactory to Borrower and Administrative Agent of any available exemption from or reduction of, United States withholding taxes in respect of all payments to be made to such Foreign Lender by Borrower pursuant to the Loan Documents or the Environmental Agreement, (B) promptly notify Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (C) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Lenders, and as may be reasonably necessary (including the re-designation of its lending office, if any) to avoid any requirement of applicable Laws that Borrower make any deduction or withholding for taxes from amounts payable to such Foreign Lender.
- Each Foreign Lender, to the extent it does not act or ceases to act (ii) for its own account with respect to any portion of any sums paid or payable to such Lender under any of the Loan Documents or the Environmental Agreement (for example, in the case of a typical participation by such Lender), shall deliver to Administrative Agent on the date when such Foreign Lender ceases to act for its own account with respect to any portion of any such sums paid or payable, and at such other times as may be necessary in the determination of Administrative Agent (in the reasonable exercise of its discretion), (A) two duly signed completed copies of the forms or statements required to be provided by such Lender as set forth above, to establish the portion of any such sums paid or payable with respect to which such Lender acts for its own account that is not subject to U.S. withholding tax, and (B) two duly signed completed copies of United States Internal Revenue Service Form W-8IMY (or any successor thereto), together with any information such Lender chooses to transmit with such form, and any other certificate or statement of exemption required under the Code, to establish that such Lender is not acting for its own account with respect to a portion of any such sums payable to such Lender.

- Borrower shall not be required to pay any additional amount to any (iii) Foreign Lender under Section 2.12(a) with respect to any Taxes required to be deducted or withheld on the basis of the information, certificates or statements of exemption such Lender transmits with a United States Internal Revenue Service Form W-8IMY pursuant to this subsection (a) if such Lender shall have failed to satisfy the foregoing provisions of this subsection (a); provided that if such Lender shall have satisfied the requirements of this subsection (a) on the date such Lender became a Lender or ceased to act for its own account with respect to any payment under any of the Loan Documents or the Environmental Agreement, nothing in this subsection (a) shall relieve Borrower of its obligation to pay any amounts pursuant to Section 2.12 in the event that, as a result of any change in any applicable law, treaty or governmental rule, regulation or order, or any change in the interpretation, administration or application thereof, such Lender is no longer properly entitled to deliver forms, certificates or other evidence at a subsequent date establishing the fact that such Lender or other person for the account of which such Lender receives any sums payable under any of the Loan Documents or the Environmental Agreement is not subject to withholding or is subject to withholding at a reduced rate.
- (iv) Administrative Agent may, without reduction, withhold any Taxes required to be deducted and withheld from any payment under any of the Loan Documents or the Environmental Agreement with respect to which Borrower is not required to pay additional amounts under this subsection (a).
- (b) Upon the request of Administrative Agent, each Lender that is a "United States person" within the meaning of Section 7701(a)(3) of the Code shall deliver to Administrative Agent two duly-signed completed copies of United States Internal Revenue Service Form W-9. If such Lender fails to deliver such forms, then Administrative Agent may withhold from any interest payment to such Lender an amount equivalent to the applicable back-up withholding tax imposed by the Code, without reduction.
- (c) If any Tribunal asserts that Administrative Agent did not properly withhold or back-up withhold, as the case may be, any tax or other amount from payments made to or for the account of any Lender, such Lender shall indemnify Administrative Agent therefor, including all penalties and interest and costs and expenses (including attorneys' fees) of Administrative Agent. The obligation of Lenders under this clause (c) shall survive the removal or replacement of a Lender, the payment of all Indebtedness and the resignation or replacement of Administrative Agent.
- 7.13 Further Assurances. Borrower will, upon Administrative Agent's request (a) promptly correct any defect, error or omission in any Loan Document or the Environmental Agreement, (b) execute, acknowledge, deliver, procure, record or file such further instruments and do such further acts as Administrative Agent deems necessary, desirable or proper to carry out the purposes of the Loan Documents and the Environmental Agreement and to identify and subject to the liens and security interest of the Loan Documents any property intended to be covered thereby, including any renewals, additions, substitutions, replacements or appurtenances to the Property, (c) execute, acknowledge, deliver, procure, file or record any document or instrument Administrative Agent deems necessary, desirable or proper to protect the liens or the

security interest under the Loan Documents against the rights or interests of third persons, and (d) provide such certificates, documents, reports, information, affidavits and other instruments and do such further acts deemed necessary, desirable or proper by Administrative Agent to comply with the requirements of any agency having jurisdiction over Administrative Agent. In addition, at any time, and from time to time, upon request by Administrative Agent or any Lender, Borrower will, at Borrower's expense, provide any and all further instruments, certificates and other documents as may, in the opinion of Administrative Agent or such Lender, be necessary or desirable in order to verify Borrower's identity and background in a manner satisfactory to Administrative Agent or such Lender.

- 7.14 <u>Inducement to Lenders</u>. The representations and warranties contained in this Agreement, the other Loan Documents and the Environmental Agreement (a) are made to induce Lenders to make the Loan and extend any other credit to or for the account of Borrower pursuant hereto, and Administrative Agent and Lenders are relying thereon and will continue to rely thereon, and (b) shall survive any foreclosure, any conveyance in lieu of foreclosure, or any proceedings under any Debtor Relief Law involving Borrower, Guarantor or the Property.
- Forum. Each party to this Agreement hereby irrevocably submits generally and unconditionally for itself and in respect of its property to the jurisdiction of any state court or any United States federal court sitting in the State specified in-Section 7.3 of this Agreement and to the jurisdiction of any state court or any United States federal court sitting in the state in which any of the Property is located, over any suit, action or proceeding arising out of or relating to this Agreement or the Indebtedness. Each party to this Agreement hereby irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of venue in any such court and any claim that any such court is an inconvenient forum. Each party to this Agreement hereby agrees and consents that, in addition to any methods of service of process provided for under applicable Law, all service of process in any such suit, action or proceeding in any state court or any United States federal court sitting in the state specified in Section 7.3 may be made by certified or registered mail, return receipt requested, directed to such party at its address for notice stated in the Loan Documents, or at a subsequent address of which Administrative Agent received actual notice from such party in accordance with the Loan Documents, and service so made shall be complete five (5) days after the same shall have been so mailed. Nothing herein shall affect the right of Administrative Agent to serve process in any manner permitted by Law or limit the right of Administrative Agent to bring proceedings against any party in any other court or jurisdiction.
- 7.16 Interpretation. References to "Dollars," "\$," "money," "payments" or other similar financial or monetary terms are references to lawful money of the United States of America. References to Articles, Sections and Exhibits are, unless specified otherwise, references to articles, sections and exhibits of this Agreement. Words of any gender shall include each other gender. Words in the singular shall include the plural and words in the plural shall include the singular. References to Borrower or Guarantor shall mean each person comprising the same, jointly and severally. The words "herein," "hereof," "hereunder" and other similar compounds of the word "here" shall refer to this entire Agreement (including the attached exhibits) and not to any particular Article, Section, paragraph or provision. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." Captions and headings in the Loan Documents are for convenience only and shall not affect the

construction of the Loan Documents. The term "person" shall include firms, associations, partnerships (including limited partnerships and limited liability partnerships), joint ventures, trusts, corporations, limited liability companies and other legal entities, including public or governmental bodies, agencies or instrumentalities, as well as natural persons.

- Agent) and Borrower is solely that of lender and borrower. Neither Administrative Agent nor any Lender has any fiduciary or other special relationship with or duty to Borrower and none is created by the Loan Documents or the Environmental Agreement. Nothing contained in the Loan Documents, and no action taken or omitted pursuant to the Loan Documents, is intended or shall be construed to create any partnership, joint venture, association or special relationship between Borrower and Administrative Agent or any Lender or in any way to make Administrative Agent or any Lender a co principal with Borrower with reference to the Project, the Property or otherwise. In no event shall the rights and interests of Administrative Agent or Lenders under the Loan Documents or the Environmental Agreement be construed to give Administrative Agent or any Lender the right to control, or be deemed to indicate that Administrative Agent or any Lender is in control of, the business, properties, management or operations of Borrower.
- 7.18 Records. The unpaid amount of the Loan and the amount of any other credit extended by Administrative Agent or Lenders to or for the account of Borrower set forth on the books and records of Administrative Agent shall be presumptive evidence of the amount thereof owing and unpaid, but failure to record any such amount on Administrative Agent's books and records shall not limit or affect the obligations of Borrower under the Loan Documents to make payments on the Loan when due.
- 7.19 <u>Dispute Resolution Provision</u>. This Section, including the subsections below, is referred to as the "<u>Dispute Resolution Provision</u>." This Dispute Resolution Provision is a material inducement for the parties entering into this Agreement.
- (a) This Dispute Resolution Provision concerns the resolution of any controversies or claims between or among the parties, whether arising in contract, tort or by statute, including controversies or claims that arise out of or relate to: (i) this Agreement (including any renewals, extensions or modifications); or (ii) any Loan Document, the Environmental Agreement or any other document related to this Agreement (collectively, a "Claim"). For the purposes of this Dispute Resolution Provision only, the term "parties" shall include any parent corporation, subsidiary or affiliate of Administrative Agent involved in the servicing, management or administration of any obligation described or evidenced by this Agreement.
- (b) Except to the extent expressly provided below, any Claim shall, upon the mutual agreement of the parties, acting in their sole and absolute discretion, be resolved by binding arbitration in accordance with the Federal Arbitration Act (Title 9, U.S. Code) (the "Federal Arbitration Act"). The Federal Arbitration Act will apply even though this Agreement provides that it is governed by Arizona law.

- Arbitration Act, the then-current rules and procedures for the arbitration of financial services disputes of the AAA and the terms of this Dispute Resolution Provision. In the event of any inconsistency, the terms of this Dispute Resolution Provision shall control. If AAA is unwilling or unable to (i) serve as the provider of arbitration or (ii) enforce any provision of this arbitration clause, Administrative Agent may designate another arbitration organization with similar procedures to serve as the provider of arbitration.
- otherwise required by law, in any U.S. state where real or tangible personal property for the Loan is located or if there is no such collateral, in the state specified in the governing law section of this Agreement. All Claims shall be determined by one arbitrator; however, if Claims exceed Five Million Dollars (\$5,000,000.00), upon the request of any party, the Claims shall be decided by three arbitrators. All arbitration hearings shall commence within ninety (90) days of the demand for arbitration and close within ninety (90) days of commencement and the award of the arbitrator(s) shall be issued within thirty (30) days of the close of the hearing. However, the arbitrator(s), upon a showing of good cause, may extend the commencement of the hearing for up to an additional sixty (60) days. The arbitrator(s) shall provide a concise written statement of reasons for the award. The arbitration award may be submitted to any court having jurisdiction to be confirmed and have judgment entered and enforced.
- (e) The arbitrator(s) will give effect to statutes of limitation in determining any Claim and may dismiss the arbitration on the basis that the Claim is barred. For purposes of the application of any statutes of limitation, the service on AAA under applicable AAA rules of a notice of Claim is the equivalent of the filing of a lawsuit. Any dispute concerning this arbitration provision or whether a Claim is arbitrable shall be determined by the arbitrator(s), except as set forth at clause (j) of this Dispute Resolution Provision. The arbitrator(s) shall have the power to award legal fees pursuant to the terms of this Agreement.

- (f) The procedure described above will not apply if the Claim, at the time of the proposed submission to arbitration, arises from or relates to an obligation to Administrative Agent and/or Lenders secured by real property. In this case, all of the parties to this Agreement, in their sole and absolute discretion, must consent to submission of the Claim to arbitration.
- (g) To the extent any Claims are not arbitrated, to the extent permitted by law the Claims shall be resolved in court by a judge without a jury, except any Claims which are brought in California state court shall be determined by judicial reference as described below.
- (h) Any Claim which is not arbitrated and which is brought in California state court will be resolved by a general reference to a referee (or a panel of referees) as provided in California Code of Civil Procedure ("CCP") Section 638. The referee (or presiding referee of the panel) shall be a retired Judge or Justice. The referee (or panel of referees) shall be selected by mutual written agreement of the parties. If the parties do not agree, the referee shall be selected by the Presiding Judge of the Court (or his or her representative) as provided in CCP Section 638 and the following related sections. The referee shall determine all issues in accordance with existing California law and the California rules of evidence and civil procedure. The referee shall be empowered to enter equitable as well as legal relief, provide all temporary or provisional

remedies, enter equitable orders that will be binding on the parties and rule on any motion which would be authorized in a trial, including motions for summary judgment or summary adjudication. The award that results from the decision of the referee(s) will be entered as a judgment in the court that appointed the referee, in accordance with the provisions of CCP Sections 644(a) and 645. The parties reserve the right to seek appellate review of any judgment or order, including orders pertaining to class certification, to the same extent permitted in a court of law.

- (i) This Dispute Resolution Provision does not limit the right of any party to: (i) exercise self-help remedies, such as but not limited to, setoff; (ii) initiate judicial or non-judicial foreclosure against any real or personal property collateral; (iii) exercise any judicial or power of sale rights; or (iv) act in a court of law to obtain an interim remedy, such as but not limited to, injunctive relief, writ of possession or appointment of a receiver, or additional or supplementary remedies. The filing of a court action is not intended to constitute a waiver of the right of any party, including the suing party, thereafter to request or require submittal of the Claim to arbitration or judicial reference as provided herein.
- (j) Any arbitration, judicial reference or trial by a judge of any Claim will take place on an individual basis without resort to any form of class or representative action (the "Class Action Waiver"). Regardless of anything else in this Dispute Resolution Provision, the validity and effect of the Class Action Waiver may be determined only by a court or referee and not by an arbitrator. The parties to this Agreement acknowledge that the Class Action Waiver is material and essential to the arbitration of any disputes between the parties and is nonseverable from the agreement to arbitrate Claims. If the Class Action Waiver is limited, voided or found unenforceable, then the parties' agreement to arbitrate shall be null and void with respect to such proceeding, subject to the right to appeal the limitation or invalidation of the Class Action Waiver. The parties acknowledge and agree that under no circumstances will a class action be arbitrated.
- (k) By agreeing to binding arbitration or judicial reference, the parties irrevocably and voluntarily waive any right they may have to a trial by jury as permitted by law in respect of any Claim. Furthermore, without intending in any way to limit this Dispute Resolution Provision, to the extent any Claim is not arbitrated or submitted to judicial reference, the parties irrevocably and voluntarily waive any right they may have to a trial by jury to the extent permitted by law in respect of such Claim. This waiver of jury trial shall remain in effect even if the Class Action Waiver is limited, voided or found unenforceable. WHETHER THE CLAIM IS DECIDED BY ARBITRATION, BY JUDICIAL REFERENCE, OR BY TRIAL BY A JUDGE, THE PARTIES AGREE AND UNDERSTAND THAT THE EFFECT OF THIS AGREEMENT IS THAT THEY ARE GIVING UP THE RIGHT TO TRIAL BY JURY TO THE EXTENT PERMITTED BY LAW.
- 7.20 Service of Process. Borrower hereby irrevocably designates and appoints Gregory L. Mast of Gallagher & Kennedy, P.A., 2575 East Camelback Road, Phoenix, Arizona 85016, as Borrower's authorized agent to accept and acknowledge on Borrower's behalf service of any and all process that may be served in any suit, action or proceeding instituted in connection with this Loan in any state or federal court sitting in the State of Arizona. If such agent shall cease so to act, Borrower shall irrevocably designate and appoint without delay

another such agent in the State of Arizona satisfactory to Administrative Agent and shall promptly deliver to Administrative Agent evidence in writing of such agent's acceptance of such appointment and its agreement that such appointment shall be irrevocable. Borrower hereby consents to process being served in any suit, action or proceeding instituted in connection with this Loan by (a) the mailing of a copy thereof by certified mail, postage prepaid, return receipt requested, to Borrower and (b) serving a copy thereof upon Gregory L. Mast of Gallagher & Kennedy, P.A., 2575 East Camelback Road, Phoenix, Arizona 85016, the agent hereby designated and appointed by Borrower as Borrower's agent for service of process. Borrower irrevocably agrees that such service shall be deemed to be service of process upon Borrower in any such suit, action or proceeding. Nothing in any Note shall affect the right of Administrative Agent to serve process in any manner otherwise permitted by Law and nothing in any Note will limit the right of Administrative Agent on behalf of Lenders otherwise to bring proceedings against Borrower in the courts of any jurisdiction or jurisdictions.

- 7.21 <u>USA Patriot Act Notice</u>. Each Lender and Administrative Agent (for itself and not on behalf of any Lender) hereby notifies Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act"), it is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow such Lender or Administrative Agent, as applicable, to identify Borrower in accordance with the Act.
- Entire Agreement. The Loan Documents and the Environmental Agreement 7.22 constitute the entire understanding and agreement between and among Borrower, Administrative Agent and Lenders with respect to the transactions arising in connection with the Loan, and supersede all prior written or oral understandings and agreements between and among Borrower. Administrative Agent and Lenders with respect to the matters addressed in the Loan Documents or the Environmental Agreement. In particular, and without limitation, the terms of any commitment letter, letter of intent or quote letter by Administrative Agent or any Lender to make the Loan are merged into the Loan Documents and/or the Environmental Agreement, as applicable. Neither Administrative Agent nor any Lender has made any commitments to extend the term of the Loan past the stated Maturity Date or to provide Borrower with financing except as set forth in the Loan Documents. Except as incorporated in writing into the Loan Documents or the Environmental Agreement, there are not, and were not, and no persons are or were authorized by Administrative Agent or any Lender to make, any representations, understandings, stipulations, agreements or promises, oral or written, with respect to the matters addressed in the Loan Documents or the Environmental Agreement.

(Signatures appear on the following page)

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

Borrower's Address for Notices: 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Senior Vice President, Real Estate Finance and Sales with a copy to: 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, as Borrower By: Luck Forward, Hamilton & Scripps LLP 600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender By: Name: Edgardo E. Martinez	written above.	
Phoenix, Arizona 85016-9267 Attn: Senior Vice President, Real Estate Finance and Sales with a copy to: 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		
Attn: Senior Vice President, Real Estate Finance and Sales with a copy to: 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By: Charles Voge Title: Vice President Vice P		
with a copy to: 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	Attn: Senior Vice President, Real Estate	$\alpha / \alpha / \alpha $
with a copy to: Title: Vice President 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		By: Wules Voyal
2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		
2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	with a copy to:	
Phoenix, Arizona 85016-9267 Attn: Legal Department with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	2555 East Camelback, Suite 800	
with a copy to: Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		[Sign page 7
Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	Attn: Legal Department	
Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	with a copy to:	
Suite 2600 San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	Luce, Forward, Hamilton & Scripps LLP	
San Diego, California 92101 Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	•	
Attn: David M. Hymer, Esq. Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Bank OF AMERICA, N.A., individually as Administrative Agent, and a Lender		
Telephone: (619) 699-2518 Fax: (619) 645-5334 Lender's Address for Notices: Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Bank OF AMERICA, N.A., individually as Administrative Agent, and a Lender Bank OF AMERICA, N.A., individually as Administrative Agent, and a Lender		
Fax: (619) 645-5334 Lender's Address for Notices: BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		
as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	-	
as Administrative Agent, and a Lender Mail Code: AZ1-200-22-17 201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		
201 East Washington Street, 22 nd Floor Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	Lender's Address for Notices:	•
Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:		
Phoenix, Arizona 85004-2343 Attn: N. Alonzo By:	201 East Washington Street, 22nd Floor	
	Phoenix, Arizona 85004-2343	Den
	Attn: N. Alonzo	

Title: Assistant Vice President

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

Borrower's Address for Notices:	SHOPPES AT CHINO HILLS Minnesota corporation, as Bor	
2555 East Camelback, Suite 800		
Phoenix, Arizona 85016-9267		
Attn: Senior Vice President, Real Estate		
Finance and Sales	By:	
	Name:	
with a copy to:	Title:	
2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department	[Sign page 7	. 1
with a copy to:	·	
Luce, Forward, Hamilton & Scripps LLP 600 West Broadway Suite 2600		
San Diego, California 92101		

Lender's Address for Notices:

Attn: David M. Hymer, Esq. Telephone: (619) 699-2518

Fax: (619) 645-5334

Mail Code: AZ1-200-22-17

201 East Washington Street, 22nd Floor

Phoenix, Arizona 85004-2343

Attn: N. Alonzo

BANK OF AMERICA, N.A., individually as Administrative Agent, and a Lender

Name: Edgardo E. Martinez
Title: Assistant Vice President

Case 09-34356-hdh11 Claim 82-1 Part 3 Filed 11/09/09 Desc Attachments to Addendum Part 2 of 7 Page 29 of 32

Lender's Address for Notices:	UNION BANK OF CALIFORNIA, N.A., as a Lender
Union Bank of California, N.A. San Francisco Corporate Office 350 California Street, 7th Floor San Francisco, CA 94104	By: You Hoheme
Attn: Karen Kokame	Name: Karen Kokame
Atti. Maron Rokamo	Title: Vice President
Lender's Address for Notices:	KEY BANK NATIONAL
man and a second second second	ASSOCIATION, as a Lender
Key Bank National Association 2390 East Camelback Road, Suite 220 Phoenix, AZ 85016	
Attn: Azona M. Labostrie	By:
Atti. 720ia IV. Subostio	Name: Azona M. Labostrie Title: Assistant Vice President
Lender's Address for Notices:	U.S. BANK, NATIONAL ASSOCIATION, as a Lender
U.S. Bank, National Association	
Mail Code: LM-AZ-X16E	
101 North First Avenue, Suite 1600	_
Phoenix, AZ 85003	By:
Attn: Barbara Richards	Name: Theresa McKinley
	Title: Vice President

Case 09-34356-hdh11 Claim 82-1 Part 3 Filed 11/09/09 Desc Attachments to Addendum Part 2 of 7 Page 30 of 32

Lender's Address for Notices:	UNION BANK OF CALIFORNIA, N.A., as a Lender
Union Bank of California, N.A. San Francisco Corporate Office	
350 California Street, 7th Floor	
San Francisco, CA 94104	By:
Attn: Karen Kokame	Name: Karen Kokame
	Title: Vice President
Lender's Address for Notices:	KEYBANK NATIONAL ASSOCIATION,
	as a Lender
KeyBank National Association 2390 East Camelback Road, Suite 220 Phoenix, AZ 85016 Attn: Azona M. LaBostrie	By: Open D. LaBot
Attii. Azolia W. Bazosalo	Name: Azona M. LaBostrie
	Title: Assistant Vice President
Lender's Address for Notices:	U.S. BANK, NATIONAL
	ASSOCIATION, as a Lender
U.S. Bank, National Association	
Mail Code: LM-AZ-X16E	
101 North First Avenue, Suite 1600	
Phoenix, AZ 85003	By:
Attn: Barbara Richards	Name: Theresa McKinley
	Title: Vice President

Case 09-34356-hdh11 Claim 82-1 Part 3 Filed 11/09/09 Desc Attachments to Addendum Part 2 of 7 Page 31 of 32

Lender's Address for Notices: Union Bank of California, N.A. San Francisco Corporate Office	UNION BANK OF CALIFORNIA, N.A., as a Lender
350 California Street, 7th Floor	Ву:
San Francisco, CA 94104 Attn: Karen Kokame	Name: Karen Kokame
Attii. Raion Rokamo	Title: Vice President
Lender's Address for Notices:	KEY BANK NATIONAL
Tr. D. 107 d. 1 Americalism	ASSOCIATION, as a Lender
Key Bank National Association 2390 East Camelback Road, Suite 220 Phoenix, AZ 85016	
Attn: Azona M. Labostrie	Ву:
	Name: Azona M. Labostrie
	Title: Assistant Vice President
Lender's Address for Notices:	U.S. BANK, NATIONAL
	ASSOCIATION, as a Lender
U.S. Bank, National Association	
Mail Code: LM-AZ-X16E	as harding
101 North First Avenue, Suite 1600	By Sheresa McKirley
Phoenix, AZ 85003	<i>D</i> _J

Name: Theresa McKinley
Title: Vice President

Attn: Barbara Richards

Exhibit "A"

PARCEL A:

THE NORTH 30 ACRES OF THAT PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1 PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

BEGINNING AT A POINT IN THE WEST LINE OF SECTION 16, SOUTH 00 40' 39" 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION; THENCE SOUTH ALONG THE WEST LINE OF SAID SECTION 16 TO A POINT 330.3 FEET SOUTH OF THE NORTHWEST CORNER OF THE SOUTHWEST ONE-QUARTER OF SAID SECTIONS; THENCE EAST PARALLEL TO THE NORTH LINE OF SECTION 16 TO THE WEST LINE OF LAND CONVEYED TO KATE FOWLER MERLE-SMITH, FORMERLY KATE FOWLER, BY THE CHINO LAND AND WATER COMPANY BY DEED RECORDED APRIL 6,1925 IN BOOK 895 OF DEEDS, PAGE 87; THENCE NORTHERLY ALONG THE WEST LINE OF THE ABOVE DESCRIBED LAND TO A POINT 660.7 FEET SOUTH OF THE NORTH LINE OF SECTION 16; THENCE WEST PARALLEL TO THE NORTH OF SECTION 16 TO THE POINT OF BEGINNING.

EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT A POINT ON THE WEST LINE OF THE ABOVE DESCRIBED LAND, 280 FEET NORTH FROM THE SOUTHWEST CORNER OF SAID NORTH 30 ACRES; THENCE EAST 38 FEET; THENCE NORTH 17 FEET, 4 INCHES; THENCE WEST 38 FEET TO THE WEST LINE OF SAID SECTION 16, AND THENCE SOUTH ALONG SAID LINE 17 FEET, 4 INCHES TO THE POINT OF BEGINNING

ALSO EXCEPTING THEREFROM THE NORTH 30 FEET AND THE WEST 30 FEET OF THE ABOVE DESCRIBED PROPERTY DEEDED FOR COUNTY ROADS.

ALSO EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT THE INTERSECTION OF THE EAST LINE OF THAT CERTAIN PARCEL OF LAND, AS DESCRIBED BY DEED TO LELAND C.

LAUNDER, RECORDED JANUARY 21, 1957 IN BOOK 4135 OF OFFICIAL RECORDS, PAGE 105, WITH THE SOUTHERLY LINE OF PEYTON DRIVE, 60 FEET WIDE, AS DESCRIBED IN DEED RECORDED IN BOOK 359 OF OFFICIAL RECORDS, PAGE 205; THENCE ALONG SAID SOUTHERLY LINE, BEING A NON-TANGENT CURVE CONCAVE NORTHERLY WITH A RADIUS OF 377 FEET, FROM A TANGENT BEARING SOUTH 76 41' 27" WEST, THROUGH AN ANGLE OF 12 51' 28" A DISTANCE OF 84.60 FEET;

THENCE NORTH 89 32' 55" EAST 83.90 FEET TO SAID EAST LINE, DISTANT ALONG SAID EAST LINE SOUTH 0 28' 26" EAST 9.45 FEET FROM THE POINT OF BEGINNING; THENCE NORTH 0 28' 26" WEST 9.45 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THAT PORTION AS CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM THAT PORTION THEREOF LYING EASTERLY, NORTHERLY AND WESTERLY OF THE FOLLOWING DESCRIBED LINE:

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED LAND; THENCE ALONG THE SOUTHERLY LINE OF SAID LAND SOUTH 89 24' 44" WEST 21.61 FEET TO TRUE POINT OF BEGINNING, BETING THE BEGINNING OF A NON TANGENT CURVE CONCAVE SOUTHEASTERLY HAVING A RADIUS OF 388.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS SOUTH 89 14' 57" WEST: THENCE NORTHEASTERLY 93.49 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 13 48' 22"; THENCE NORTH 13 03' 19" EAST 10.00 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 45.00 FEET; THENCE NORTHWESTERLY 60.79 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 77 21' 57" TO THE SOUTH LINE OF THE LAND AS DESCRIBED IN THE OUITCLAIM DEED TO THE CITY OF CHINO HILLS RECORDED AUGUST 11, 2003 AS DOCUMENT NO.2003-0591428, OFFICIAL RECORDS, COUNTY OF SAN BERNARDINO, BEING A POINT ON A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET; THENCE NORTHWESTERLY 660.12 FEET ALONG SAID CURVE AND SOUTHERLY LINE THROUGH A CENTRAL ANGLE OF 26 15' 56"; THENCE SOUTH 89 25' 26" WEST 293.36 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1740.00 FEET; THENCE WESTERLY 153.03 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 10 03' 09" TO THE BEGINNING OF A COMPOUND CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 40.00 FEET: THENCE SOUTHWESTERLY 63.89 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 91 31' 07" TO THE BEGINNING OF A REVERSE CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1740.00 FEET; THENCE SOUTHERLY 146.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 05 43' 08" TO THE NORTHERLY LINE OF THE PROPERTY CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OF OFFICIAL RECORDS: THENCE ALONG SAID NORTHERLY LINE SOUTH 89 25' 26" WEST 15.29 FEET TO THE EAST LINE OF THE LAND DESCRIBED IN THE DECLARATION OF ROAD RIGHT OF WAY, RECORDED MARCH 15, 1988 AS DOCUMENT NO. 88-077222 OFFICIAL RECORDS. SAID POINT BEING ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE AND ALONG SAID EAST LINE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE SOUTH 00 37' 46" EAST 590.81 FEET TO THE SOUTH LINE OF SAID NORTH 30 ACRES.

PARCEL B:

THAT PORTION OF THE NORTH 30 ACRES DESCRIBED IN THAT CERTAIN DEED TO THE COUNTY OF SAN BERNARDINO AS PER DEED RECORDED ON DECEMBER 16, 1983, AS INSTRUMENT NO. 83-296886, OFFICIAL RECORDS, AND BEING A PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1, PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

COMMENCING AT THE CENTERLINE INTERSECTION OF GRAND AVENUE AND PEYTON DRIVE, AS SHOWN ON MAP OF TRACT NO. 12581-12, AS PER MAP RECORDED IN BOOK 188 OF MAPS, PAGES 98 THROUGH 100, INCLUSIVE, RECORDS OF SAID COUNTY, SAID INTERSECTION BEING SOUTH 00 40' 39" EAST, A DISTANCE OF 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION 16 AS SHOWN ON THAT RECORD OF SURVEY MAP RECORDED IN BOOK 54 OF RECORDS OF SURVEY, PAGES 58 AND 59, RECORDS OF SAID COUNTY; THENCE SOUTH 00 37' 46" EAST, ALONG THE CENTERLINE OF SAID PEYTON DRIVE, A DISTANCE OF 987.02 FEET TO THE SOUTH LINE OF AFORESAID NORTH 30 ACRES; THENCE NORTH 89 24' 51" EAST ALONG SAID SOUTH LINE, A DISTANCE OF 60.00 FEET TO A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SAID POINT BEING THE TRUE POINT OF BEGINNING;

Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Case 09-34356-hdh11 Addendum Part 3 of 7 Page 2 of 32 THENCE CONTINUING ALONG SAID SOUTH LINE NORTH 89 24 51 EAST, A DISTANCE OF 1263.78 FEET TO THE SOUTHEAST CORNER OF SAID NORTH 30 ACRES; THENCE ALONG THE EAST LINE OF SAID NORTH 30 ACRES, NORTH 00 35' 16" WEST, DISTANCE OF 624.82 FEET; THENCE AT RIGHT ANGLES TO THE EAST LINE OF SAID NORTH 30 ACRES, SOUTH 89 24' 44" WEST, A DISTANCE OF 66.00 FEET TO A LINE THAT IS PARALLEL WITH AND 66.00 FEET WEST OF THE EAST LINE OF SAID NORTH 30 ACRES: THENCE ALONG SAID PARALLEL LINE, NORTH 00 35' 16" WEST, A DISTANCE OF 20.00 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHEAST AND HAVING A RADIUS OF 166.00 FEET; THENCE NORTHEASTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 28 06' 13", A DISTANCE OF 81.42 FEET; THENCE TANGENT TO SAID LAST CURVE NORTH 27 30' 57" EAST A DISTANCE OF 27.82 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 27.00 FEET; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 92 25' 57" A DISTANCE OF 43.55 FEET TO A POINT OF COMPOUND CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 1440.00 FEET, SAID LAST CURVE BEING THE SOUTHERLY LINE OF GRAND AVENUE; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 25 40' 34", A DISTANCE OF 645.31 FEET TO A POINT OF TANGENCY WITH A LINE THAT IS PARALLEL WITH AND 60.00 FEET SOUTH OF THE CENTERLINE OF PEYTON DRIVE AND THE NORTH LINE OF SAID NORTH 30 ACRES AS SHOWN ON AFORESAID RECORD OF SURVEY MAP; THENCE ALONG SAID PARALLEL LINE AND THE SOUTHERLY LINE OF GRAND AVENUE AS DESCRIBED IN THAT DECLARATION OF ROAD RIGHT-OF-WAY BY THE COUNTY OF SAN BERNARDINO AS RESOLUTION NO. 88-88 AND RECORDED AS INSTRUMENT NO. 88-077222, OFFICIAL RECORDS OF SAID COUNTY, SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET; THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 155.00 FEET; THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 90.0 FEET;

THENCE SOUTH 00 37' 46" WEST, A DISTANCE OF 12.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 92.00 FEET;

THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 91.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 380.00 FEET TO THE BEGINNING OF A NON-TANGENT CURVE CONCAVE TO THE WEST HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING OF SAID CURVE THROUGH SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06", A DISTANCE OF 78.23 FEET; THENCE TANGENT TO SAID LAST

CURVE AND ALONG A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SOUTH 00 37' 46" EAST, A DISTANCE OF 590.81 FEET TO THE TRUE POINT OF BEGINNING.

EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE SOUTHWEST CORNER OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG THE WESTERLY LINE NORTH 00 37' 76" WEST 221.18 FEET TO THE TRUE POINT OF BEGINNING; THENCE NORTH 89 22' 13" EAST 1.99 FEET; THENCE NORTH 00 48' 09" EAST 108.60 FEET; THENCE NORTH 45 24' 04" EAST 35.04 FEET; THENCE NORTH 01 28' 29" EAST 221.44 FEET TO THE NORTHERLY LINE OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 25' 26" WEST 15.29 FEET TO THE WESTERLY LINE OF SAID PARCEL BEING A POINT ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE CONTINUING ALONG SAID WESTERLY LINE SOUTH 00 37' 46" EAST 369.63 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

BEGINNING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE OF SAID PARCEL NORTH 00 35' 16" WEST 26.89 FEET; THENCE SOUTH 41 26' 12" WEST 30.53 FEET; THENCE SOUTH 00 39' 11" EAST 4.21 FEET TO THE SOUTHERLY LINE OF SAID PARCEL; THENCE ALONG SAID SOUTHERLY LINE NORTH 89 24' 51" EAST 20.43 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 3 of 32 COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE NORTH 00 35' 16" WEST 453.32 FEET TO THE TRUE POINT OF BEGINNING, BEING THE BEGINNING OF A TANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 312.00 FEET; THENCE NORTHWESTERLY 77.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 16' 30" TO THE BEGINNING OF A REVERSE CURVE CONCAVE NORTHEASTERLY HAVING A RADIUS OF 388.00 FEET; THENCE NORTHERLY 95.57 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 06' 44" TO A POINT ON THE NORTHERLY LINE OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 24' 44" EAST 21.61 FEET TO THE EASTERLY LINE OF SAID PARCEL; THENCE ALONG SAID EASTERLY LINE SOUTH 00 35' 21" EAST 171.50 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE EASTERLY TERMINUS OF THE COURSE SHOWN AS "SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET" IN THE NORTHERLY LINE OF THE ABOVE PARCEL, SAID POINT BEING THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1440.00; THENCE SOUTHEASTERLY 44.43 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 01 46' 04" TO THE TRUE POINT OF BEGINNING; THENCE NONMTANGENT SOUTH 44 24' 14" EAST 44.95 FEET; THENCE SOUTH 86 16' 02" EAST 64.80 FEET; THENCE NORTH 48 08' 12" EAST 41.99 FEET TO THE NORTHERLY LINE OF SAID PARCEL, BEING A POINT ON A NONTANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS NORTH 06 16' 21" EAST; THENCE NORTHWESTERLY 127.70 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 05 04' 51" TO THE TRUE POINT OF BEGINNING.

PARCEL C:

A NON-EXCLUSIVE EASEMENT FOR THE PURPOSE OF INSTALLING, MAINTAINING, OPERATING, REPAIRING, AND REPLACING A PYLON SIGN AS DESCRIBED IN THAT CERTAIN PYLON SIGN EASEMENT AND MAINTENANCE AGREEMENT RECORDED JUNE 2, 2006 AS INSTRUMENT NO. 06-380081 OF OFFICIAL RECORDS.

EXHIBIT "B"

DEFINITIONS

As used in this Agreement and the attached exhibits, the following terms shall have the following meanings:

"AAA" means the American Arbitration Association or any successor thereof.

"Adjusted LIBOR Rate" means the quotient obtained by dividing (a) the applicable London Interbank Offered Rate by (b) 1.00 minus the LIBOR Reserve Percentage, where,

"London Interbank Offered Rate" means, with respect to any applicable Interest Period, the rate per annum equal to the British Bankers Association LIBOR Rate ("BBA LIBOR"), as published by Reuters (or other commercially available source providing quotations of BBA LIBOR as selected by Administrative Agent from time to time) at approximately 11:00 a.m. London time two (2) London Banking Days before the commencement of the Interest Period, for deposits in U.S. Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period. If such rate is not available at such time for any reason, then the rate for that Interest Period will be determined by such alternate method as reasonably selected by Administrative Agent; and

"LIBOR Reserve Percentage" means, with respect to any applicable Interest Period, for any day that percentage (expressed as a decimal) which is in effect on such day, as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement (including basic, supplemental, emergency, special and marginal reserves) generally applicable to financial institutions regulated by the Federal Reserve Board, whether or not applicable to any Lender, in respect of "Eurocurrency liabilities" (or in respect of any other category of liabilities which includes deposits by reference to which the interest rate on LIBOR Rate Principal is determined), whether or not any Lender has any Eurocurrency liabilities. The LIBOR Rate shall be adjusted automatically as of the effective date of each change in the LIBOR Reserve Percentage.

"Administrative Agent" means Bank of America, N.A., in its capacity as administrative agent under any of the Loan Documents or the Environmental Agreement, or any successor administrative agent.

"Administrative Agent Advances" means advances as set forth in Section 2.14 of this Agreement.

"Administrative Agent's Office" means Administrative Agent's address and, as appropriate, account as set forth on the Schedule of Lenders, or such other address or account as Administrative Agent hereafter may designate by written notice to Borrower and Lenders.

"Administrative Agent's Time" means the time of day observed in the city where Administrative Agent's Office is located.

"Advance Termination Date" means the Maturity Date.

"Affiliate" means any person, directly or indirectly, through one or more intermediaries, controlling, controlled by, or under direct or indirect common control with, another person. A person shall be deemed to be "controlled by" any other person if such other person possesses, directly or indirectly, power (a) to vote 10% or more of the securities (on a fully diluted basis) having ordinary voting power for the election of directors or managing general partners or the equivalent; or (b) to direct or cause the direction of the management and policies of such person whether by contract or otherwise.

"Agent-Related Persons" means Administrative Agent, together with its Affiliates, and the officers, directors, employees, agents and attorneys-in-fact of such persons and Affiliates.

"Aggregate Commitments" means the Commitments of all Lenders.

"Aggregate Cost" has the meaning set forth in Section 1.4 of this Agreement.

"Agreement" has the meaning set forth in the introductory paragraph of this Agreement, and includes all exhibits attached hereto and referenced in Section 1.1.

"Appraised Value" means \$174,900,000.00.

"Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender, or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Arranger" means Banc of America Securities LLC, in its capacity as sole arranger and sole book manager.

"Assignment and Assumption" means an Assignment and Assumption substantially in the form of Exhibit "L".

"Base Rate" means, on any day, a simple rate per annum equal to the sum of the Prime Rate for that day plus the Base Rate Margin. Without notice to Borrower or anyone else, the Base Rate shall automatically fluctuate upward and downward as and in the amount by which the Prime Rate fluctuates.

"Base Rate Margin" means zero percent (0%) per annum.

"Base Rate Principal" means, at any time, the Principal Debt minus the portion, if any, of such Principal Debt which is LIBOR Rate Principal.

"Borrower" has the meaning set forth in the introductory paragraph of this Agreement.

"Borrower's Deposit" has the meaning set forth in Section 1.5 of this Agreement.

"Budget" means the budget and cost itemization for the Project attached as Exhibit "D".

"Budget Shortfall" has the meaning set forth in Section 1.5 of this Agreement.

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 6 of 32

"Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, the state where Administrative Agent's Office is located.

"Claim" has the meaning set forth in Section 7.19 of this Agreement.

"Closing Checklist" means that certain Closing Requirements and Checklist setting forth the conditions for closing the Loan and recording the Deed of Trust.

"Commitment" means, as to each Lender, its obligation to advance its Pro Rata Share of the Loan in an aggregate principal amount not exceeding the amount set forth opposite such Lender's name on the Schedule of Lenders at any one time outstanding, as such amount may be adjusted from time to time in accordance with this Agreement.

"Completion Date" means September 1, 2008.

"Construction Consultant" means the construction consultant, if any, engaged by Administrative Agent with respect to the Project.

"<u>Debt Service Coverage Ratio</u>" has the meaning set forth in <u>Section 2.11</u> of this Agreement.

"Debtor Relief Law" means the Bankruptcy Code of the United States of America, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Deed of Trust" means the Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing dated May 16, 2007, granted by the Borrower to Administrative Agent, as Trustee, for the benefit of Administrative Agent and Lenders, securing repayment of the Indebtedness and Borrower's performance of its other obligations to Administrative Agent and Lenders under the Loan Documents, as amended, modified, supplemented, restated and replaced from time to time.

"Default" has the meaning set forth in Section 5.1 of this Agreement.

"Defaulting Lender" means a Lender that fails to pay its Pro Rata Share of a Payment Amount within five (5) Business Days after notice from Administrative Agent, until such Lender cures such failure as permitted in this Agreement.

"<u>Defaulting Lender Amount</u>" means the Defaulting Lender's Pro Rata Share of a Payment Amount.

"<u>Defaulting Lender Payment Amount</u>" means a Defaulting Lender Amount plus interest from the date such Defaulting Lender Amount was funded by Administrative Agent and/or an Electing Lender, as applicable, to the date such amount is repaid to Administrative Agent and/or

such Electing Lender, as applicable, at the rate per annum applicable to such Defaulting Lender Amount under the Loan or otherwise at the Base Rate.

"Draw Request" has the meaning set forth in Section 1 of Exhibit "F".

"Eligible Assignee" means (a) a Lender, (b) an Affiliate of a Lender, (c) an Approved Fund, and (d) any other person (other than a natural person) approved by Administrative Agent and, unless a Default has occurred and is continuing, Borrower (each such approval not to be unreasonably withheld or delayed).

"Environmental Agreement" means the Environmental Indemnity Agreement of even date herewith by and among Borrower, Guarantor and Administrative Agent for the benefit of Lenders.

"Environmental Assessment" means a written report (including all drafts thereof) of an environmental assessment of the Property of such scope as may be requested by Administrative Agent, including the taking of soil borings and air and groundwater samples and other above-and below-ground testing, by a consulting firm acceptable to Administrative Agent and made in accordance with Administrative Agent's established guidelines.

"ERISA" has the meaning set forth in Section 3.19 of this Agreement.

"Excusable Delay" means a delay, not to exceed a total of ten (10) consecutive Business Days, caused by unusually adverse weather conditions which have not been taken into account in the construction schedule, fire, earthquake or other acts of God, strikes, lockouts, acts of public enemy, riots or insurrections or any other unforeseen circumstances or events beyond the control of Borrower (except financial circumstances or events or matters which may be resolved by the payment of money), and as to which Borrower notifies Administrative Agent in writing within five (5) days after such occurrence; provided, however, no Excusable Delay shall extend the Completion Date or suspend or abate any obligation of Borrower or any Guarantor or any other person to pay any money.

"Federal Arbitration Act" has the meaning set forth in Section 7.19 of this Agreement.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for that day shall be the rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on the next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upwards to the next higher 1/100 of 1%) charged to Bank of America, N.A. on such day on such transactions as determined by Administrative Agent.

"<u>Financial Statements</u>" means (a) for each Reporting Party other than an individual, a balance sheet, income statement, statements of cash flow and amounts and sources of contingent liabilities, a reconciliation of changes in equity and liquidity verification, and unless Administrative Agent otherwise consents, consolidated and consolidating statements if the

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 8 of 32

Reporting Party is a holding company or a parent of a subsidiary entity, and (b) for each Reporting Party who is an individual, a balance sheet, statements of amounts and sources of contingent liabilities, sources and uses of cash and liquidity verification, and unless Administrative Agent otherwise consents, Financial Statements for each entity owned or jointly owned by the Reporting Party.

"<u>First Advance</u>" means the first advance of Loan proceeds after the recording of the Deed of Trust and after the conditions set forth in <u>Exhibit "C"</u> and <u>Section 3</u> of <u>Exhibit "F"</u> to this Agreement have been satisfied.

"Fund" means any person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial real estate loans and similar extensions of credit in the ordinary course of its business.

"Funding Date" means the date on which an advance of Loan proceeds or Borrower's Deposit shall occur.

"Guarantor" means OPUS WEST CORPORATION, a Minnesota corporation, whether one or more, and if more than one, each one individually or all collectively.

"Improvements" means all on-site and off-site improvements to the Land for an approximately 370,446 square foot mixed use retail and office center to be constructed on the Land, together with all fixtures, tenant improvements and appurtenances now or later to be located on the Land and/or in such improvements.

"Indebtedness" means any and all indebtedness to Administrative Agent and/or Lenders evidenced, governed or secured by, or arising under, any of the Loan Documents, including the Loan.

"Indemnified Liabilities" has the meaning set forth in Section 7.2.

"Interest Period" means with respect to any LIBOR Rate Principal, the period commencing on the date such LIBOR Rate Principal is disbursed or on the date on which the Principal Debt or any portion thereof is converted into or continued as such LIBOR Rate Principal, and ending on the date one (1), two (2), or three (3) months thereafter, as elected by Borrower in the applicable Rate Election Notice; provided that:

- (a) Each Interest Period must commence on a LIBOR Business Day;
- (b) In the case of the continuation of LIBOR Rate Principal, the Interest Period applicable after the continuation of such LIBOR Rate Principal shall commence on the last day of the preceding Interest Period;
- (c) The last day for each Interest Period and the actual number of days during the Interest Period shall be determined by Administrative Agent using the practices of the London interbank eurodollar market; and

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 9 of 32

(d) No Interest Period shall extend beyond the Maturity Date, and any Interest Period which begins before the Maturity Date and would otherwise end after the Maturity Date shall instead end on the Maturity Date.

"Internal Revenue Code" means the Internal Revenue Code of 1986, as amended.

"Land" means the real property described in Exhibit "A", consisting of approximately 25.563 acres of land located in the southeast corner of Grand Avenue and Peyton Drive, Chino Hills, California."

"Laws" means all constitutions, treaties, statutes, laws, ordinances, regulations, rules, orders, writs, injunctions, or decrees of the United States of America, any state or commonwealth, any municipality, any foreign country, any territory or possession, or any Tribunal.

"Lender" means each lender from time to time party to this Agreement.

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such on the Schedule of Lenders, or such other office or offices as such Lender may from time to time notify Borrower and Administrative Agent.

"LIBOR Business Day" means a Business Day which is also a London Banking Day.

"LIBOR Margin" means one and three-fourths percent (1.75%) per annum.

"<u>LIBOR Rate</u>" means for any applicable Interest Period for any LIBOR Rate Principal, a simple rate per annum equal to the sum of the LIBOR Margin plus the Adjusted LIBOR Rate.

"<u>LIBOR Rate Election</u>" means an election by Borrower of an applicable LIBOR Rate in accordance with this Agreement.

"<u>LIBOR Rate Principal</u>" means any portion of the Principal Debt which bears interest at an applicable LIBOR Rate at the time in question.

"Loan" means the loan by Lenders to Borrower, in the maximum amount of \$110,000,000.00, but not to exceed, in the aggregate, an amount equal to the lesser of (i) the payment of 90% of the costs incident to the Project as specified in the Budget, (ii) an amount complying with Section 3.14, and (iii) an amount complying with Section 3.16. In the event the aggregate amount of the actual costs incident to the Project are less than the aggregate amount specified in the Budget, the maximum amount described above shall be reduced by the difference between the aggregate amount specified in the Budget and the aggregate amount of such actual costs.

"Loan Documents" means this Agreement (including all exhibits), the Construction Deed of Trust, the Note, any guaranty, financing statements, the Budget, each Draw Request, and such other documents evidencing, securing or pertaining to the Loan as shall, from time to time, be executed and/or delivered by Borrower, Guarantor, or any other party to Administrative Agent or any Lender pursuant to this Agreement (but not including the Environmental Agreement), as they may be amended, modified, restated, replaced or supplemented from time to time.

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 10 of 32

"Loan-to-Cost Ratio" has the meaning set forth in Section 3.15 of this Agreement.

"Loan-to-Value Ratio" has the meaning set forth in Section 3.14 of this Agreement.

"London Banking Day" means a day on which dealings in dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Major Lease" means any tenant lease of space in the Improvements in excess of 5,000 square feet of such Improvements.

"Material Adverse Effect" means: (a) a material adverse change in, or a material adverse effect upon, the Project, or the operations, business, properties, liabilities (actual or contingent), condition (financial or otherwise) or prospects of Borrower or Borrower and its Subsidiaries taken as a whole; (b) a material impairment of the ability of any party to the Loan Documents or the Environmental Agreement to perform its obligations under any such Loan Document or Environmental Agreement to which it is a party; or (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any party to the Loan Documents or the Environmental Agreement of any such document or agreement to which it is a party.

"Material Contract" means a contract for the performance of any work or the supplying of any labor, materials or services that exceeds \$500,000.00 in total price.

"Maturity Date" means June 1, 2009, as it may be earlier terminated or extended in accordance with the terms hereof.

"Notes" means the Construction Deed of Trust Notes each dated May 16, 2007 executed by Borrower and payable to the order of each Lender in the amount of the Lender's Commitment and collectively in the maximum principal amount of the Loan, substantially in the form of Exhibit "M", as amended, modified, replaced, restated, extended or renewed from time to time.

"Obligations" means all liabilities, obligations, covenants and duties of, any party to a Loan Document arising under or otherwise with respect to any Loan Document, whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any party to a Loan Document or any Affiliate thereof of any proceeding under any Debtor Relief Law naming such person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceedings.

"Past Due Rate" has the meaning set forth in Section 2.8 of this Agreement.

"Payment Amount" means an Advance of the Loan, an unreimbursed Administrative Agent Advance, an unreimbursed Indemnified Liability, or any other amount that a Lender is required to fund under this Agreement.

"Permitted Changes" means changes to the Plans or Improvements, provided the cost of any single change or extra does not exceed \$250,000.00 and the aggregate amount of all such changes and extras (whether positive or negative) does not exceed \$750,000.00.

"Plans" means the plans and specifications listed in <u>Exhibit "E"</u> and all modifications thereof and additions thereto that are included as part of the Plans as the same shall be approved by Administrative Agent in the exercise of its sole discretion in accordance with the terms of this Agreement.

"Potential Default" means any condition or event which with the giving of notice or lapse of time or both would, unless cured or waived, become a Default.

"Prime Rate" means, on any day, the rate of interest per annum then most recently established by Administrative Agent as its "prime rate," it being understood and agreed that such rate is set by Administrative Agent as a general reference rate of interest, taking into account such factors as Administrative Agent may deem appropriate, that it is not necessarily the lowest or best rate actually charged to any customer or a favored rate, that it may not correspond with future increases or decreases in interest rates charged by other lenders or market rates in general, and that Administrative Agent may make various business or other loans at rates of interest having no relationship to such rate. If Administrative Agent (including any subsequent Administrative Agent) ceases to exist or to establish or publish a prime rate from which the Prime Rate is then determined, the applicable variable rate from which the Prime Rate is determined thereafter shall be instead the prime rate reported in The Wall Street Journal (or the average prime rate if a high and a low prime rate are therein reported), and the Prime Rate shall change without notice with each change in such prime rate as of the date such change is reported.

"Principal Debt" means the aggregate unpaid principal balance of this Loan at the time in question.

"Pro Rata Share" means, with respect to each Lender at any time, a fraction expressed as a percentage, the numerator of which is the amount of the Commitment of such Lender at such time and the denominator of which is the amount of the Aggregate Commitments at such time or, if the Aggregate Commitments have been terminated, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the total outstanding amount of all Indebtedness held by such Lender at such time and the denominator of which is the total outstanding amount of all Indebtedness at such time. The initial Pro Rata Share of each Lender named on the signature pages hereto is set forth opposite the name of that Lender on the Schedule of Lenders.

"Project" means the acquisition of the Land, the construction of the Improvements, and if applicable, the leasing and operation of the Improvements.

"Property" means the Land, the Improvements and all other property constituting the "Property" as described and defined in the Deed of Trust, or subject to a right, lien or security interest to secure the Loan pursuant to any other Loan Document, less any property released pursuant to the provisions of Exhibit "H".

"Reporting Party" means each party for whom Financial Statements are required to be delivered to Lender.

"Reporting Period" means a specified period to which Financial Statements relate.

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 12 of 32

"Required Lenders" means as of any date of determination at least two Lenders having more than 66 3% of the Aggregate Commitments or, if the Aggregate Commitments have been terminated, at least two Lenders holding in the aggregate more than 66 3% of the total outstanding amount of all Indebtedness; provided that the Commitment of, and the portion of the total outstanding amount of all Indebtedness held by, any Defaulting Lender shall be excluded for purposes of making a determination of the Required Lenders.

"Schedule of Lenders" means the schedule of Lenders party to this Agreement as set forth on Exhibit "N", as it may be modified from time to time in accordance with this Agreement.

"Stored Materials" means building materials or furnishings that are not yet incorporated into the Improvements.

"Stored Materials Advance Limit" means \$500,000.00.

"Subsidiary" means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries.

"Survey" means a survey prepared in accordance with Exhibit "G" or as otherwise approved by Administrative Agent in its sole discretion.

"Swap Transactions" means (a) any and all rate swap transactions, basis swaps, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, or (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., or any other master agreement (any such master agreement, together with any related schedules, as amended, restated, extended, supplemented or otherwise modified in writing from time to time, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"<u>Taxes</u>" means all taxes, assessments, fees, levies, imposts, duties, deductions, withholdings, or other charges of any nature whatscever from time to time or at any time imposed by any Law or Tribunal, including any such Taxes as defined and described in Section 2.12.

"<u>Title Company</u>" means First American Title Insurance Company, a California corporation.

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 13 of 32

"<u>Title Insurance</u>" means the loan policy or policies of title insurance issued to Administrative Agent for the benefit of Lenders by Title Company, in an amount equal to the maximum principal amount of the Loan, insuring the validity and priority of the Deed of Trust encumbering the Land and Improvements for the benefit of Administrative Agent and Lenders, as further described in <u>Exhibit "C"</u> hereto.

"<u>Tribunal</u>" means any state, commonwealth, federal, foreign, territorial or other court or governmental department, commission, board, bureau, district, authority, agency, central bank, or instrumentality, or any arbitration authority.

EXHIBIT "C"

CONDITIONS PRECEDENT TO FIRST ADVANCE

As conditions precedent to the First Advance, if and to the extent required by Administrative Agent, Administrative Agent shall have received and approved the following:

- 1. <u>Fees and Expenses</u>. Any and all required commitment and other fees and evidence satisfactory to Administrative Agent that Borrower has paid all other fees, costs and expenses (including the fees and costs of Administrative Agent's counsel) then required to be paid pursuant to this Agreement and all other Loan Documents, including all fees, costs and expenses that Borrower is required to pay pursuant to any loan application or commitment.
- 2. <u>Financial Statements</u>. The Financial Statements of Borrower and Guarantor or any other party required by any loan application or commitment or otherwise required by Administrative Agent.
- 3. Appraisal. A market value appraisal of the Property made within one hundred eighty (180) days prior to the date of this Agreement, which appraises the Property on a "completed value" basis at not less than the Appraised Value. The appraiser, appraisal and appraised value of the Property must be satisfactory to Administrative Agent (including satisfaction of applicable regulatory requirements) and the appraiser must be engaged directly by Administrative Agent.
- 4. <u>Draw Schedule and Budget</u>. Borrower's proposed cash flow, draw schedule and construction schedule for the Project, and Administrative Agent shall be satisfied, in its sole discretion, that the Improvements may be completed in accordance with the construction schedule and for costs not exceeding those set forth in the Budget.
- 5. <u>Authorization</u>. Such evidence as Administrative Agent may require of the existence, good standing, authority and capacity of Borrower, each Guarantor, and their respective constituent partners, members, managers and owners (however remote) to execute, deliver and perform their respective obligations to Administrative Agent and Lenders under the Loan Documents and the Environmental Agreement, including:
- (a) For each partnership (including a joint venture or limited partnership):
 (i) a true and complete copy of an executed partnership agreement or limited partnership agreement and all amendments thereto; (ii) for each limited partnership, a copy of the certificate of limited partnership and all amendments thereto accompanied by a certificate issued by the appropriate governmental official of the jurisdiction of formation that the copy is true and complete, and such evidence as Administrative Agent may require of registration or qualification to do business in the state where Borrower's principal place of business is located and if different, the state in which the Property is located; and (iii) a partnership affidavit certifying who will be authorized to execute or attest any of the Loan Documents and the Environmental Agreement, and a true and complete copy of the partnership resolutions approving the Loan Documents and the Environmental Agreement and authorizing the transactions contemplated in this Agreement, the other Loan Documents and the Environmental Agreement.

- (b) For each corporation: (i) a true and complete copy of its articles of incorporation and by-laws and all amendments thereto, a certificate of incumbency of all of its officers who are authorized to execute or attest to any of the Loan Documents and the Environmental Agreement, and a true and complete copy of resolutions approving the Loan Documents and the Environmental Agreement and authorizing the transactions contemplated in this Agreement, the other Loan Documents and the Environmental Agreement; and (ii) certificates of existence, good standing and qualification to do business issued by the appropriate governmental officials in the state of its formation and if different, the state in which the Property is located.
- (c) For each limited liability company or limited liability partnership: (i) a true and complete copy of the articles of organization and operating agreement and all amendments thereto, a certificate of incumbency of all of its members who are authorized to execute or attest to any of the Loan Documents and the Environmental Agreement, and a true and complete copy of resolutions approving the Loan Documents and the Environmental Agreement and authorizing the transactions contemplated in this Agreement, the other Loan Documents and the Environmental Agreement; and (ii) certificates of existence, good standing and qualification to do business issued by appropriate governmental officials in the state of its formation and if different, the state in which the Property is located.
- (d) For each entity or organization that is not a corporation, partnership, limited partnership, joint venture, limited liability company or limited liability partnership, a copy of each document creating it or governing the existence, operation, power or authority of it or its representatives.
- (e) All certificates, resolutions and consents required by Administrative Agent applicable to the foregoing.
- (f) All due diligence materials deemed necessary by Administrative Agent and each Lender with respect to verifying Borrower's identity and background information in a manner satisfactory to Administrative Agent and each Lender.
- 6. Loan Documents and Environmental Agreement. From Borrower, Guarantor and each other person required by Administrative Agent, all Loan Documents then required by Administrative Agent and the Environmental Agreement (with a copy for each Lender), each duly executed, acknowledged and/or sworn to as required, each dated concurrently herewith and each in form and content satisfactory to Administrative Agent, and such evidence as Administrative Agent may require that the Deed of Trust has been recorded in the official records of the city or county in which the Property is located and UCC-1 financing statements have been filed in all filing offices that Administrative Agent may require.
 - 7. [Intentionally Left Blank].
- 8. Survey; No Special Flood Hazard Area. (a) Two (2) prints of an original survey (with a copy for each Lender) of the Land and improvements thereon dated not more than sixty (60) days prior to the date of this Agreement (or dated such earlier date, if any, as is satisfactory to Title Company, but in any event not more than one hundred eighty (180) days prior to the date

of this Agreement) satisfactory to Administrative Agent and Title Company and otherwise complying with <u>Exhibit "G"</u>; and (b) a flood insurance policy (with a copy for each Lender) in an amount sufficient to meet the requirements of applicable law and the Federal Emergency Management Agency, as such requirements may from time to time be in effect, and otherwise in compliance with the requirements of the Loan Documents, or evidence satisfactory to Administrative Agent that no part of the Improvements is or will be located in a Special Flood Hazard Area (SFHA).

- Title Insurance. An ALTA extended coverage lender's title insurance policy, issued by Title Company (which shall be approved by Administrative Agent) in the maximum amount of the Loan plus any other amount secured by the Deed of Trust, on a coinsurance and/or reinsurance basis if and as required by Administrative Agent: (a) insuring without exclusion or exception for creditors' rights that the Deed of Trust constitutes a valid lien covering the Land and all Improvements thereon, having the priority required by Administrative Agent and subject only to those exceptions and encumbrances (regardless of rank or priority) as Administrative Agent may approve, in a form acceptable to Administrative Agent, and with all "standard" exceptions which can be deleted, including the exception for matters which a current survey would show, deleted to the fullest extent authorized under applicable title insurance rules, and Borrower shall satisfy all requirements therefor; (b) containing no exception for standby fees or real estate taxes or assessments other than those for the year in which the closing occurs to the extent the same are not then due and payable and endorsed "not yet due and payable" and no exception for subsequent assessments for prior years; (c) providing full coverage against mechanic's, materialmen's and other similar liens to the extent authorized under applicable title insurance rules, and Borrower shall satisfy all requirements therefor; (d) insuring that no restrictive covenants shown in the Title Insurance have been violated, and that no violation of the restrictions will result in a reversion or forfeiture of title; (e) insuring all appurtenant easements; insuring that indefeasible or marketable (as coverage is available) title to the Land and Improvements is vested in Borrower; (f) containing such affirmative coverage and endorsements as Administrative Agent may require and are available under applicable title insurance rules, and Borrower shall satisfy all requirements therefor; (g) insuring any easements, leasehold estates or other matters appurtenant to or benefiting the Land and/or the Improvements as part of the insured estate; (h) insuring the right of access to the Land to the extent authorized under applicable title insurance rules, and Borrower shall satisfy all requirements therefor; and (i) containing provisions acceptable to Administrative Agent regarding advances and/or readvances of Loan funds after closing. Borrower and Borrower's counsel shall not have any interest, direct or indirect, in Title Company (or its agent) or any portion of the premium paid for the Title Insurance.
- 10. <u>Plans</u>. Two (2) true and correct copies of all existing Plans (including the site plan), together with evidence satisfactory to Administrative Agent that all applicable governmental authorities, Borrower, Borrower's architect, engineer and contractors and Construction Consultant have approved the same.
- 11. <u>Contracts</u>. (a) A list containing the names and addresses of all existing contractors, architects, engineers and other suppliers of services and materials who are parties to Material Contracts for the Project, their respective contract amounts, and a copy of their contracts, and (b) duly executed, acknowledged and delivered originals, from each contractor,

architect, engineer, subcontractor or supplier of services or materials required by Administrative Agent, of consents or other agreements satisfactory to Administrative Agent.

- 12. <u>Insurance Policies</u>. The insurance policies initially required by Administrative Agent pursuant to the Loan Documents, together with evidence satisfactory to Administrative Agent that all premiums therefor have been paid for a period of not less than one (1) year from the date of this Agreement and that the policies are in full force and effect.
- 13. <u>Leases</u>. If <u>Exhibit "I"</u> is attached hereto, (a) true and correct copies of all leases and subleases and guaranties thereof; (b) such estoppel certificates, subordination agreements and/or subordination, nondisturbance and attornment agreements (each, an "<u>SNDA</u>") as Administrative Agent in its discretion may require, dated within thirty (30) days prior to this Agreement and in the respective forms attached as exhibits to the Closing Checklist, or otherwise in form and content satisfactory to Administrative Agent, from such tenants, subtenants and guarantors as Administrative Agent may require; (c) evidence satisfactory to Administrative Agent of Borrower's compliance with the leases; and (d) evidence satisfactory to Administrative Agent of the tenants' approval of all matters requiring their approval. Notwithstanding anything herein to the contrary, an SNDA shall not be required by Administrative Agent as to any lease that is not a Major Lease unless required by the Title Company.
- Environmental Compliance/Report. Evidence satisfactory to Administrative 14. Agent that no portion of the Land is "wetlands" under any applicable Law and that the Land does not contain and is not within or near any area designated as a hazardous waste site by any governmental authority, that neither the Property nor any adjoining property contains or has ever contained any substance classified as hazardous or toxic (or otherwise regulated, such as, without limitation, asbestos, radon and/or petroleum products) under any Law or governmental requirement pertaining to health or the environment, and that neither the Property nor any use or activity thereon violates or is or could be subject to any response, remediation, clean-up or other obligation under any Law or governmental requirement pertaining to health or the environment, including an Environmental Assessment of the Property made within twelve (12) months prior to the date of this Agreement by a consulting firm acceptable to Administrative Agent, showing that there is no evidence of any such substance which has been generated, treated, stored, released or disposed of in the Property and such additional evidence as may be required by Administrative Agent. All reports, drafts of reports and recommendations, whether written or oral, from such consulting firm shall be made available and communicated to Administrative Agent.
- 15. Soil Reports. A soil composition and test boring report and a foundation report satisfactory to Administrative Agent regarding the Land, made within three (3) years prior to the date of this Agreement by a licensed professional engineer satisfactory to Lenders.
- 16. Access, Utilities and Laws. (a) Evidence satisfactory to Administrative Agent that the Property abuts and has fully adequate direct and free access to one or more public streets dedicated to public use, fully installed and accepted by the appropriate governmental authority, that all fees, costs and expenses of the installation and acceptance thereof have been paid in full, and that there are no restrictions on the use and enjoyment of such streets which would adversely affect the Project; (b) letters from the applicable utility companies or governmental authorities

confirming that all utilities necessary for the Improvements are available at the Land in sufficient capacity, together with evidence satisfactory to Administrative Agent of paid impact fees, utility reservation deposits and connection fees required to assure the availability of such services; (c) evidence satisfactory to Administrative Agent that all applicable zoning ordinances, restrictive covenants and governmental requirements affecting the Property permit the use for which the Property is intended and have been or will be complied with without the existence of any variance, non-complying use, nonconforming use or other special exception; (d) evidence satisfactory to Administrative Agent that the Land and Improvements comply and will comply with all Laws and governmental requirements regarding subdivision and platting and would so comply if the Land and the Improvements thereon were conveyed as a separate parcel; (e) a true and correct copy of a valid building permit for the Improvements, together with all other consents, licenses, permits and approvals necessary for construction of the Improvements, all in assignable form (to the extent appropriate) and in full force and effect; (f) evidence satisfactory to Administrative Agent of compliance by Borrower and the Property, and the proposed construction, use and occupancy of the Improvements, with such other applicable Laws and governmental requirements as Administrative Agent may request, including all Laws and governmental requirements regarding access and facilities for handicapped or disabled persons, including, to the extent applicable, The Federal Architectural Barriers Act (42 U.S.C. § 4151 et seq.), The Fair Housing Amendments Act of 1988 (42 U.S.C. § 3601 et seq.), The Americans With Disabilities Act of 1990 (42 U.S.C. § 12101 et seq.), The Rehabilitation Act of 1973 (29 U.S.C. § 794), and any applicable state requirements; and (g) written evidence satisfactory to Administrative Agent that construction of the Improvements on the Land is permissible under all federal, state and local statutes, regulations and rulings protecting tidal and non-tidal wetlands and other environmentally protected areas.

- 17. Priority. (a) Evidence satisfactory to Administrative Agent that prior to and as of the time the Deed of Trust was filed for record (i) no activity or circumstance was visible on or near the Land which would constitute inception of a mechanic's, materialmen's or other similar lien against the Property, and (ii) no mechanic's, materialmen's or other similar lien claim or notice, lis pendens, judgment, or other claim or encumbrance against the Property has been filed for record in the county where the Property is located or in any other public record which by Law provides notice of claims or encumbrances regarding the Property; (b) a certificate or certificates of a reporting service acceptable to Administrative Agent, reflecting the results of searches made not earlier than ten (10) days prior to the date of this Agreement, (i) of the central and local Uniform Commercial Code records, showing no filings against any of the collateral for the Loan or against Borrower otherwise except as consented to by Administrative Agent, and (ii) if required by Administrative Agent, of the appropriate judgment and tax lien records, showing no outstanding judgment or tax lien against Borrower or any Guarantor.
- 18. <u>Construction Analysis</u>. Administrative Agent shall have received and approved, in its sole discretion, a construction analysis of the Project.
- 19. Tax and Standby Fee Certificates. Evidence satisfactory to Administrative Agent (a) of the identity of all taxing authorities and utility districts (or similar authorities) having jurisdiction over the Property or any portion thereof, (b) that all taxes, standby fees and any other similar charges have been paid, including copies of receipts or statements marked "paid" by the appropriate authority, and (c) that the Land is a separate tax lot or lots with separate assessment

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 19 of 32

or assessments of the Land and Improvements independent of any other land or improvements and that the Land is a separate legally subdivided parcel.

- 20. <u>Upfront Equity</u>. Evidence satisfactory to Administrative Agent that the Upfront Equity (as defined in <u>Section 1.4</u>) has been fully paid and funded.
- 21. <u>Pre-Leasing</u>. Executed leases for at least 55,000 square feet and letters of intent for at least 27,000 square feet of the retail space.
- 22. Other Documents. Such other documents and certificates as Administrative Agent may reasonably request from Borrower, any Guarantor, and any other person or entity, in form and content satisfactory to Administrative Agent.

المسامين

EXHIBIT "D"

BUDGET

Borrower Name: To Be Determined

Relationship GCI #: 901180422

Project Name: Shoppes at Chino Hills

Borrower GCI #: Obligor #:

Location: Chino Hills, CA

BAC Portion: 100%

Loan Amount: 110,000,000

Date of Borrower's Budget:

Total Sq. Ft.: 370,446 # of Units: 1

Date of RU's Adjusted Budget: 2/15/2007

BUDGET

		DODOLI			<u> </u>
LINE ITEMS	\$ Per Sq. Ft. of Land Area	% of Total Costs	\$ Per Sq. Ft. of Bldg Area	\$ Per Unit	Total Costs
LAND					
Land-Acquisition (Cost)	6.29	5.7%	18.90	7,000,008	7,000,008.00
SUB-TOTAL LAND	6.29	5.7%	18.90	7,000,008.00	7,000,008.00
HARD COSTS		Y			· · · · · <u>- · · · · · · · · · · · · · ·</u>
Construction Contract Tenant Improvement Costs Offsite Improvements Onsite Improvements Offsite Enhancements Promenade & Paseos Tenant Shell Package		30.3% 24.5% 0.1% 15.5% 0.7% 1.1% 2.9%	100.17 80.91 0.19 51.02 2.29 3.78 9.72	37,109,344 29,974,270 70,300 18,899,297 850,000 1,400,000 3,600,000	37,109,344.00 29,974,270.00 70,300.00 18,899,297.00 850,000.00 1,400,000.00 3,600,000.00
			-		
Hard Costs Contingency SUB-TOTAL HARD COSTS	1.1%	<u>0.8%</u> 76.0%	<u>2,70</u> 250.79	1,000,000 92,903,211.00	<u>1,000,000,00</u> 92,903,211.00
SOFT COSTS					
Testing (ESA, soil, other) Real Estate Taxes Construction Management Fee Marketing Government Charges Sheriff Relocation Leasing Commissions Tenant Rep. Commissions Legal Interim Loan Costs		0.0% 0.1% 1.8% 0.1% 1.7% 0.3% 3.7% 0.2% 0.7% 1.0%	0.13 0.37 6.01 0.27 5.61 1.08 12.33 0.67 2.29 3.27	50,000 136,932 2,226,561 100,000 2,077,751 400,000 4,568,329 250,000 850,000 1,210,627	50,000.00 136,932.00 2,226,561.00 100,000.00 2,077,751.00 400,000.00 4,568,329.00 250,000.00 850,000.00 1,210,627.00
,					

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 21 of 32

LOAN PROCEEDS		90.0%	296.94	110,000,000	110,000,000.00
SUB-TOTAL EQUITY/OTHER SOURCES OF FUNDS		10.0%	33.13	12,273,317	12,273,317.00
Deferred Fees	Deferred	4.7%	15.62 - -	5,787,920	5,787,920.00
Cash	Upfront	5.3%	17.51 -	6,485,397	6,485,397.00
EQUITY / OTHER SOURCES OF FUND	TIMING		i		
TOTAL BUDGET		100.0%	330.07	122,273,317	122,273,317.00
SUB-TOTAL SOFT COSTS		18.3%	60.39	22,370,098	22,370,098.00
Soft Cost Contingency	12.6%	_2,0%	6.75	2,500,000	2,500,000.00
Interest Reserve Operating Deficit		3.6%	11.98	4,438,539	4,438,539.00
Developer's Fee		2.9%	9.61	3,561,359	3,561,359.00
Fees			-		

1. Total Interest Reserve using interest rate of: 7.00%

(See attached Interest Reserve Calculation)

Comparison to budgeted Interest Reserve (above) of \$4,438,539

4,438,539 **FAVORABLE**

Previous \$9,100,000 A&D loan contemplated \$2,400,000 in deferred equity to be disbursed pari passu with bank funding (on a 70%-30% basis). Deferred fees include GC and Development Fees.

Prepared By:

Martinez, Edgardo - RU

EXHIBIT "E"

PLANS

EXHIBIT "F"

ADVANCES

- Draw Request. A "Draw Request" means a properly completed and executed written application by Borrower to Administrative Agent in the form of Exhibit "F-1" (or in another form approved by Administrative Agent) setting forth the amount of Upfront or Deferred Equity and/or Loan proceeds desired, together with the related AIA Documents G-702 and G-703 and such schedules, affidavits, releases, waivers, statements, invoices, bills, and other documents, certificates and information satisfactory to Administrative Agent. At least five (5) Business Days before the requested date of each advance made under the Budget from Upfront [or Deferred] Equity or the Loan, Borrower shall deliver a Draw Request to Administrative Agent. Borrower shall be entitled to an advance only in an amount approved by Administrative Agent in accordance with the terms of this Agreement and the Loan Documents. Lenders shall not be required to make advances more frequently than once each calendar month. Lenders shall, only upon the satisfaction, as determined by Administrative Agent in its sole discretion, of all applicable conditions of this Agreement and the Loan Documents, be required to make the requested advance to Borrower on a Funding Date which is a Business Day within five (5) Business Days after such satisfaction. Each Draw Request, and Borrower's acceptance of any advance, shall be deemed to ratify and confirm, as of the date of the Draw Request and the advance, respectively, except as specified in the Draw Request, all representations and warranties in the Loan Documents remain true and correct.
- 2. Advances. Borrower shall use and apply all advances made to Borrower for payment of the costs and expenses specified in the Budget for which the advances were made, and for no other purpose. Following receipt and approval of a Draw Request, all supporting documentation and information required by Administrative Agent, and receipt and approval of a written report (or an oral report followed by written confirmation) from Construction Consultant satisfactory to Administrative Agent, Administrative Agent will determine the amount of the advance Lenders will make in accordance with this Agreement, the Loan Documents, the Budget and, if and to the extent required by Administrative Agent, to Administrative Agent's satisfaction, the following standards:
- (a) An initial advance in an amount sufficient to pay off the existing financing related to the acquisition of the Land.
- (b) For construction work other than tenant improvement work, advances on the basis of ninety-five percent (95%) of the costs shown on the application for payment from the contractor reviewed and approved by Administrative Agent of the work or material in place on the Improvements that complies with the terms of the Loan Documents, minus all previous advances and all amounts required to be paid by Borrower, as described in the Budget.
- (c) For tenant improvement work, advances on the basis of one hundred percent (100%) of the costs shown for each lease in the application for payment from the contractor reviewed and approved by Administrative Agent for the work or material in place that complies with the terms of the Loan Documents, provided that if required by Administrative

- Agent: (i) an application for payment may be submitted only after all applicable tenant improvements have been completed for a given lease; (ii) the amount of the requested advance does not exceed \$33,574,270.00 in the aggregate as provided in the Budget for Tenant Improvement Costs and Tenant Shell Package; (iii) all provisions of the Loan Documents, including Sections 3 and 4 of this Exhibit "F", have been satisfied; (iv) the term of the applicable lease has commenced; (v) Administrative Agent has received from the applicable tenant a tenant estoppel certificate and, with respect to each Major Lease, a subordination agreement or a subordination, nondisturbance and attornment agreement, as Administrative Agent may require, in the respective forms attached as exhibits to the Closing Checklist, or otherwise in form and content satisfactory to Administrative Agent; (vi) Administrative Agent has received two (2) sets of as-built plans for the applicable tenant improvements; and (vii) Administrative Agent has received evidence of satisfaction of all applicable legal requirements, including applicable certificates of occupancy and evidence that the plans comply with all legal requirements regarding access and facilities for handicapped or disabled persons.
- (d) No advances will be made for building materials or furnishings that are not yet incorporated into the Improvements ("stored materials") unless the stored materials are in Borrower's possession and satisfactorily stored on the Land and the aggregate of advances for stored materials that have not yet been incorporated into the Improvements does not exceed the Strated Materials Advance Limit.
- 3. <u>Conditions to All Advances</u>. As conditions precedent to each advance made pursuant to a Draw Request, in addition to all other requirements contained in this Agreement, if and to the extent required by Administrative Agent, the Advance Termination Date shall not have passed, and Administrative Agent shall have received and approved the following:
- (a) Evidence satisfactory to Administrative Agent of the continued satisfaction of all conditions to the recording of the Deed of Trust, as set forth in Exhibit "C".
 - (b) A Draw Request.
- (c) Evidence satisfactory to Administrative Agent that no Default or any event which, with the giving of notice or the lapse of time or both, could become a Default, exists.
- (d) Evidence satisfactory to Administrative Agent that the representations and warranties made in the Loan Documents and the Environmental Agreement are true and correct on and as of the date of each advance and no event has occurred or condition or circumstance exists which, if known to Borrower, would render any such representation or warranty incorrect or misleading.
- (e) Evidence satisfactory to Administrative Agent that each subcontract or other contract for labor, materials, services and/or other work included in a Draw Request has been duly executed and delivered by all parties thereto and is effective, and a true and complete copy of a fully executed copy of each such subcontract or other contract as Administrative Agent may have requested, together with performance and payment bonds securing such contracts and subcontracts, to the extent required by Administrative Agent, in form and substance satisfactory to Administrative Agent.

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 25 of 32

- (f) Evidence satisfactory to Administrative Agent that no mechanic's, materialmen's or other similar lien or other encumbrance has been filed and remains in effect against the Property, no stop notices have been served on Lenders that have not been bonded by Borrower in a manner and amount satisfactory to Administrative Agent, and releases or waivers of mechanic's liens and receipted bills showing payment of all amounts due to all parties who have furnished materials or services or performed labor of any kind in connection with the Property.
- (g) If required by Administrative Agent, evidence satisfactory to Administrative Agent that the Title Insurance has been endorsed and brought to date in a manner satisfactory to Administrative Agent through the date of each advance with no additional title change or exception not approved by Administrative Agent.
- (h) Certification by Construction Consultant and if required by Administrative Agent, by Borrower's architect, that to the best of such party's knowledge, information and belief, construction is in accordance with the Plans, the quality of the work for which the advance is requested is in accordance with the applicable contract, the amount of the advance requested represents work in place based on on-site observations and the data comprising the Draw Request, the work has progressed in accordance with the construction contract and schedule, and the applicable contractor is entitled to payment of the amount certified.
- (i) If the advance is requested within thirty (30) days after the laying of the foundation of any building or structure of the Improvements, (i) a foundation survey satisfactory to Administrative Agent and to the extent required by Administrative Agent complying with Exhibit "G", (ii) a certificate of Borrower's architect stating that based on personal inspection the foundations have been completed in accordance with the Plans and are satisfactory in all respects, and (iii) a bearing capacity test report with respect to the excavated footings and foundations, reviewed and approved by Construction Consultant and Borrower's architect.
- (j) If required by Administrative Agent, if the advance is requested within thirty (30) days after the pouring of concrete for any Improvements, a report satisfactory to Construction Consultant of the results of concrete tests made at the time the concrete is poured.
- (k) If required by Administrative Agent, if the advance is requested within thirty (30) days after the compaction of any soil for construction, a report satisfactory to Construction Consultant of the results of soil tests.
- (l) Evidence satisfactory to Administrative Agent that as of the date of making such advance, no event has occurred, nor does any condition exist, that could impair the ability of Borrower or any Guarantor to fulfill its material obligations under the Loan Documents or the Environmental Agreement or could have a Material Adverse Effect on (i) the enforceability of the Loan Documents or the Environmental Agreement, (ii) the financial condition of Borrower or any Guarantor, or (iii) the Property whatsoever.
 - (m) [Intentionally left blank].

Case 09-34356-hdh11 Claim 82-1 Part 4 Filed 11/09/09 Desc Attachments to Addendum Part 3 of 7 Page 26 of 32

- (n) Evidence satisfactory to Administrative Agent that the Improvements have not been damaged without being repaired and are not the subject of any pending or threatened condemnation or adverse zoning proceeding.
- (o) Evidence satisfactory to Administrative Agent that Borrower has paid all amounts then required to be paid by Borrower under the Budget.
 - (p) The Borrower's Deposit if required under <u>Section 1.5</u> of this Agreement.
- (q) With respect to any advance to pay a contractor, original applications for payments in form approved by Administrative Agent, containing a breakdown by trade and/or other categories acceptable to Administrative Agent, executed and certified by each contractor and Borrower's architect, accompanied by invoices, and approved by Construction Consultant.
- (r) Copies of notarized partial lien waiver forms executed by each contractor and each appropriate subcontractor, supplier and materialman, including such partial lien waivers from all parties sending statutory notices to contractors, notices to owners, or notices of nonpayment, specifying in such partial lien waivers the amount paid in consideration of such partial releases.
- (s) Such other information, documents and supplemental legal opinions as may be required by Administrative Agent.
- 4. <u>Final Advance for Improvements</u>. If and to the extent required by Administrative Agent, the final advance for the Improvements (including retainage) shall not be made until thirty (30) days after the later of (i) the date on which the Improvements have been "completed" as defined by applicable state Law, and (ii) if required by Administrative Agent, the date on which an affidavit of completion has been recorded. In the case of each such Draw Request, if and to the extent required by Administrative Agent, Administrative Agent shall have received the following as additional conditions precedent to the requested advance:
- (a) Certificates from Borrower's architect, engineer, contractor and if required by Administrative Agent, from Construction Consultant, certifying that the Improvements (including any off-site improvements) have been completed in accordance (and as completed comply) with the Plans and all Laws and governmental requirements; and Administrative Agent shall have received two (2) sets of detailed "as-built" Plans approved in writing by Borrower, Borrower's architect and each contractor.
- (b) Final affidavits (in a form approved by Administrative Agent) from Borrower's architect and engineer and each contractor certifying that each of them and each of their subcontractors, laborers and materialmen has been paid in full for all labor and materials for construction of the Improvements; and final lien releases or waivers (in a form approved by Administrative Agent) by Borrower's architect, engineer, contractor, and all subcontractors, materialmen, and other parties who have supplied labor, materials or services for the construction of the Improvements, or who otherwise might be entitled to claim a contractual, statutory or constitutional lien against the Property.

- (c) The Title Insurance shall be endorsed to remove any exception for mechanic's, materialmen's or similar liens or pending disbursements, with no additional title change or exception objectionable to Administrative Agent, and with such other endorsements required by Administrative Agent.
- (d) Evidence satisfactory to Administrative Agent that all Laws and governmental requirements have been satisfied, including receipt by Borrower of all necessary governmental licenses, certificates and permits (including certificates of occupancy) with respect to the completion, use, occupancy and operation of the Improvements, together with evidence satisfactory to Administrative Agent that all such licenses, certificates and permits are in full force and effect and have not been revoked, canceled or modified.
- (e) Three (3) copies of a final as-built survey satisfactory to Administrative Agent and, to the extent required by Administrative Agent, complying with <u>Exhibit "G"</u>.
 - (f) [Intentionally left blank].
- (g) A written confirmation by each tenant having the right to do so that such tenant has approved the completed Improvements.
- 5. Direct Advances. Borrower hereby irrevocably authorizes Administrative Agent (but Administrative Agent shall have no obligation) to (a) advance Loan funds directly to Lenders to pay interest due on the Loan, and (b) advance and directly apply the proceeds of any advance to the satisfaction of any of Borrower's obligations under any of the Loan Documents or the Environmental Agreement, even though Borrower did not include that amount in a Draw Request and/or no Default exists. Each such direct advance (except for application of a Borrower's Deposit) shall be added to the outstanding principal balance of the Loan and shall be secured by the Loan Documents. Unless Borrower pays such interest from other resources, Administrative Agent may advance Loan funds pursuant to this Section for interest payments as and when due. Nothing contained in this Agreement shall be construed to permit Borrower to defer payment of interest on the Loan beyond the date(s) due. The allocation of Loan funds in the Budget for interest shall not affect Borrower's absolute obligation to pay the same in accordance with the Loan Documents. Administrative Agent may hold, use, disburse and apply the Loan and Borrower's Deposit for payment of any obligation of Borrower under the Loan Documents or the Environmental Agreement. Borrower hereby assigns and pledges the proceeds of the Loan and any Borrower's Deposit to Administrative Agent for itself and for the benefit of Lenders for such purposes. Administrative Agent on behalf of Lenders may advance and incur such expenses as Administrative Agent deems necessary for the completion of the Improvements and to preserve the Property, and any other security for the Loan, and such expenses, even though in excess of the amount of the Loan, shall be secured by the Loan Documents and shall be payable to Administrative Agent on behalf of Lenders on demand. Administrative Agent on behalf of Lenders may disburse any portion of any advance at any time, and from time to time, to persons other than Borrower for the purposes specified in this Section and the amount of advances to which Borrower shall thereafter be entitled shall be correspondingly reduced.
- 6. <u>Conditions and Waivers</u>. All conditions precedent to the obligation of Lenders to make any advance are imposed hereby solely for the benefit of Administrative Agent and

Lenders, and no other party may require satisfaction of any such condition precedent or be entitled to assume that Lenders will refuse to make any advance in the absence of strict compliance with such conditions precedent. Any requirement of this Agreement may be waived, in whole or in part, in a specific written waiver intended for that purpose and signed by Administrative Agent. Administrative Agent shall have the right to approve and verify the periodic progress, costs incurred by Borrower, and the estimated costs remaining to be incurred, after consultation with Construction Consultant. No advance shall constitute an approval or acceptance by Administrative Agent of any construction work, or a waiver of any condition precedent to any further advance, or preclude Administrative Agent from thereafter declaring the failure of Borrower to satisfy such condition precedent to be a Default. No waiver by Administrative Agent of any condition precedent or obligation shall preclude Administrative Agent from requiring such condition or obligation to be met prior to making any other advance or from thereafter declaring the failure to satisfy such condition or obligation to be a Default.

7. Funding. Borrower shall establish and maintain a special account with Administrative Agent into which advances funded directly to Borrower shall be deposited by Borrower, and against which checks shall be drawn only for the payment of costs specified in the Budget, but which special account shall not be used for any other purpose. Borrower hereby irrevocably authorizes Administrative Agent to deposit each advance requested by Borrower to the credit of Borrower in that account, by wire transfer or other deposit. Advances may also be made, in addition to other methods contemplated herein, at Administrative Agent's option, by direct or joint check payment to any or all persons entitled to payment for work or services performed or material furnished in connection with the Project or the Loan, or by having the proceeds thereof made available to Title Company (or its agent) for disbursement. Neither Administrative Agent nor any Lender shall be required or have any responsibility to supervise the proper application or distribution of funds to third parties.

EXHIBIT "F-1"

DRAW REQUEST

[BORROWER'S LETTERHEAD]

	DRAW REQUEST NO	- 		
TO:	BANK OF AMERICA, N.A. ("Administrative Agent")			
LOAN NO.				
PROJECT	The Shoppes at Chino Hills			
LOCATION	Chino Hills, CA			
BORROWER	SHOPPES AT CHINO HILLS, IN	C., a Minnesota corporation		
FOR PERIOD ENDING				
dated May 16, 2007, amor	a the Construction Loan Agreement in the Borrower, Administrative Agent and be advanced from Borrower's Deposit, \$	the Lenders as defined therein		
at	Account No			
	REQUEST FOR HARD COSTS	\$		
CURRENT DRAW REQUEST FOR SOFT COSTS		\$		
3. TOTAL DRAW RE	TOTAL DRAW REQUEST			
AUTHORIZED SIGNER:				
	Dated:			

EXHIBIT "G"

SURVEY REQUIREMENTS

- 1. Requirements. The Survey shall be made in accordance with, and meet the requirements of, the certification below by a registered professional engineer or registered professional land surveyor. The description shall be a single metes and bounds perimeter description of the entire Land, and a separate metes and bounds description of the perimeter of each constituent tract or parcel out of the Land. The total acreage and square footage of the Land and each constituent tract or parcel of the Land shall be certified. If the Land has been recorded on a map or plat as part of an abstract or subdivision, all survey lines must be shown, and all lot and block lines (with distances and bearings) and numbers, must be shown. The date of any revisions subsequent to the initial survey prepared pursuant to these requirements must also be shown.
- 2. <u>Certification</u>. The certification for the property description and the map or plat shall be addressed to Administrative Agent for the Lenders, Borrower and the Title Company, signed by the surveyor (a registered professional land surveyor or registered professional engineer), bearing current date, registration number and seal, and shall be in the following form or its substantial equivalent:

This is to certify to Bank of America, N.A., as Administrative Agent for certain Lenders, Shoppes at Chino Hills, Inc., a Minnesota corporation, as Borrower, and First American Title Insurance Company, a California corporation, as the Title Company that this map or plat and the survey on which it is based were made in accordance with the "Minimum Standard Detail Requirements for ALTA/ACSM Land Title Surveys" jointly established and adopted by ALTA and NSPS in 2005, and include optional items 1, 2, 4 (in square feet or acres), 6, 8, 10, 11(b), 16, 17, 18, and if buildings are located on the land, optional items 7(a), 7(b)(1), 7(b)(2) and 9 of Table A thereof. Pursuant to the Accuracy Standards as adopted by ALTA and NSPS and in effect on the date of this certification, the undersigned further certifies that in my professional opinion, as a land surveyor registered in the State of California, the maximum Relative Positional Accuracy resulting from the measurements made on the survey does not exceed the Allowable Relative Positional Accuracy for Measurements Controlling Land Boundaries on ALTA/ACSM Land Title Surveys (0.07 feet or 20mm + 50 ppm). The undersigned additionally certifies that: (a) this survey was made on the ground under my supervision; (b) I have received and examined a copy of the Title Insurance Commitment No. NCS-284385-PHX1 issued by the Title Company as well as a copy of each instrument listed therein, and the subject land and each tract or parcel thereof described in this survey is the same land as described in the Title Commitment; (c) if the subject land consists of two or more tracts or parcels having common boundaries, those tracts and parcels are contiguous along the common boundaries; (d) the subject land and each tract or parcel thereof has a tax map designation separate and distinct from that of any other land and the subject land and each tract or parcel thereof is a separate, legally subdivided parcel; (e) this survey correctly shows all matters of record (and to the extent they can be located, their location and dimensions) of which I have been advised affecting the subject land according to the legal description in

such matters (with instrument, book, and page number indicated); (f) except as shown on this survey, no part of the subject land is located in a 100-year Flood Plain or in an identified "flood prone area," as defined pursuant to the Flood Disaster Protection Act of 1973, as amended, as reflected by Flood Insurance Rate Map Panel #06071C 8615F dated March 18, 1996, which such map panel covers the area in which the subject land is situated and this survey correctly indicates the zone designation of any area as being in the 100-year Flood Plain or "flood prone area"; (g) to the best of my knowledge, this survey shows the relation of and distance of all substantial, visible buildings, sidewalks and other improvements to easements and setback lines; and (h) to the best of my knowledge, except as shown on this survey, neither the subject land nor any tract or parcel thereof serves any adjoining land for drainage, utilities, or ingress or egress.

EXHIBIT "H"

RELEASE PROVISIONS

Provided that no Event of Default beyond all applicable cure periods provided in this Agreement exists, Borrower shall be entitled to the release of the lien of the Deed of Trust and of any other instrument securing the Note from a portion of the Property (each, a "Parcel") upon the terms and conditions contained herein, including the payment of the applicable Release Price (as hereinafter defined) for each Parcel, and subject to the following terms and conditions:

- (a) All state, county, municipal or other commercially reasonable requirements regarding the sale of that Parcel shall have been satisfied and evidence thereof provided to Administrative Agent.
- (b) Each partial release shall be in connection with the sale of the Parcel to a bona fide purchaser for value.
- (c) Administrative Agent shall have received such endorsements to its policy of title insurance insuring the lien of the Deed of Trust as Lender may require in order to confirm the priority of Lenders' first lien on the Property remaining subject to the Deed of Trust.
- (d) Borrower shall have subjected the Parcel to an agreement providing for the development of that Parcel and the remainder of the Improvements under a common plan and for reciprocal ingress/egress, parking and other easements. That agreement and all other related documents and instruments shall be reasonably satisfactory to Administrative Agent in both form and substance.
- (e) Administrative Agent shall have received a written request for the partial release together with such documents and information as Administrative Agent may reasonably request to verify that the conditions for such release have been satisfied. After receipt of such notice and the satisfaction of all conditions precedent for the partial release, Administrative Agent shall deliver to Trustee a standard form "Request for Partial Release and Reconveyance," for the property to be released, executed by Administrative Agent and, if requested by Borrower, a Deed of Partial Release and Reconveyance covering the property to be released, to be executed by Administrative Agent.
- (f) All reasonable costs and expenses of Administrative Agent relating to all releases shall be paid by Borrower, including but not limited to reconveyance fees, title fees, recording fees and legal expenses.
- (g) No partial release shall impair or adversely affect Administrative Agent's security in the Property remaining subject to the Deed of Trust or any term or provision of the Deed of Trust as it pertains to the Property remaining subject to the Deed of Trust.

- (h) Each Release Price received in connection with the Loan shall be applied as a prepayment of the outstanding principal balance under the Note in the manner provided therein for prepayments of principal. Regular payments of interest or principal due under the Note shall not apply toward any Release Price. "Release Price" means as to a Parcel an amount equal to the gross sale proceeds minus the closing costs for the Parcel.
- (i) After giving effect to such release, Borrower shall be in compliance with the requirements of Sections 3.14 and 3.16.

EXHIBIT "I"

LEASING AND TENANT MATTERS

Borrower and Lenders agree as follows:

- Approved Leases. Borrower shall not enter into any tenant lease of space in the Improvements unless approved or deemed approved by Administrative Agent prior to execution. Borrower's standard form of tenant lease, and any revisions thereto, must have the prior written approval of Administrative Agent. Administrative Agent shall be "deemed" to have approved any tenant lease that: (a) is on the standard form lease approved by Administrative Agent with no deviations except as approved by Administrative Agent; (b) is entered into in the ordinary course of business with a bona fide unrelated third party tenant, and Borrower, acting in good faith and exercising due diligence, has determined that the tenant is financially capable of performing its obligations under the lease; (c) is received by Administrative Agent (together with each guaranty thereof (if any) and financial information regarding the tenant and each guarantor (if any) received by Borrower) within fifteen (15) days after execution; (d) reflects an arm's length transaction at then-current market rate for comparable space; (e) contains no right to purchase the Property, or any present or future interest therein; (f) does not require Borrower to provide funds for tenant improvements in excess of the per square foot allowance provided in the Budget; and (g) is expressly subordinate to the Deed of Trust. Borrower shall provide to Administrative Agent a correct and complete copy of each tenant lease, including any exhibits, and each guaranty thereof (if any), prior to execution unless the lease in question meets the foregoing requirements for "deemed" approval by Administrative Agent. Borrower shall, throughout the term of this Agreement, pay all reasonable costs incurred by Administrative Agent in connection with Administrative Agent's review and approval of tenant leases and each guaranty thereof (if any), including reasonable attorneys' fees and costs.
- 2. <u>Effect of Lease Approval</u>. No approval of any lease by Administrative Agent shall be for any purpose other than to protect Lenders' security and to preserve Lenders' rights under the Loan Documents. No approval by Administrative Agent shall result in a waiver of any default of Borrower. In no event shall any approval by Administrative Agent of a lease be a representation of any kind with regard to the lease or its adequacy or enforceability, or the financial capacity of any tenant or guarantor.
- 3. Representations Concerning Leases. Borrower represents and warrants to Administrative Agent and Lenders that Borrower has delivered to Administrative Agent a true and correct copy of all tenant leases and each guaranty thereof (if any), affecting any part of the Improvements, together with an accurate and complete rent roll for the Project, and no such lease or guaranty contains any option to purchase all or any portion of the Property or any interest therein or contains any right of first refusal relating to any sale of the Property or any portion thereof or interest therein.
- 4. <u>Delivery of Leasing Information and Documents</u>. Borrower shall promptly (a) deliver to Lender such monthly rent rolls, leasing schedules and reports, operating statements, financial statements for all tenants other than tenants with a lease term for less than one year and

other information regarding tenants and prospective tenants or other leasing information as Lender from time to time may request, and (b) with respect to all Major Leases, obtain and deliver to Lender such estoppel certificates and subordination and attornment agreements executed by such tenants (and guarantors, if any) in such forms as Lender from time to time may require.

- 5. <u>Income from the Property</u>. Borrower shall first apply all income from leases and all other income derived from the Property to pay costs and expenses associated with the ownership, maintenance, development, operation and marketing of the Land and Improvements, including all amounts then required to be paid under the Loan Documents and the Environmental Agreement, before using or applying such income for any other purpose.
- 6. Compliance and Default. As additional conditions to Lenders' obligations under this Agreement, all tenants having the right to do so must approve all Plans and all changes thereto, the construction of the Improvements, and all other aspects of the Project requiring tenants' approval. A default by Borrower under or any failure by Borrower to satisfy any of the conditions of a lease shall constitute a Default under this Agreement. Borrower shall promptly notify Administrative Agent in writing of any failure by any party to perform any material obligation under any lease, any event or condition which would permit a tenant to terminate or cancel a lease, or any notice given by a tenant with respect to the foregoing, specifying in each case the action Borrower has taken or will take with respect thereto.

EXHIBIT "J"

[INTENTIONALLY LEFT BLANK]

EXHIBIT "K"

[INTENTIONALLY LEFT BLANK]

EXHIBIT "L"

ASSIGNMENT AND ASSUMPTION

the me	th below and is eanings given to ment"), receipt and Conditions	entered ("As them in of a cop s set fort	into by and betweensignee"). Capitalized tendent the Construction Loan A y of which is hereby acknown in Annex 1 attached he	ment") is dated as of the ms used but not defined. Agreement identified belonowledged by Assignee. reto are hereby agreed to t as if set forth herein in the	"Assignor") and herein shall have bw (the "Loan The Standard and incorporated
accord Date i obligated deliver below identification (in its connect the retornal contraction and the retornal rights the rein Assign	ssignee hereby in lance with the Sonserted by Admittions as a Lendered pursuant the formal of all of such of fied below (included under application with the Lender or in any way not claims, tort or, related to the rand obligations a collectively as	irrevocal Standard ninistrativer under er under ereto to to outstandin uding, we ender) appeared on can Agr based on claims, me rights and sold and the "Ass as expre	bly purchases and assume Terms and Conditions are ve Agent as contemplated the Loan Agreement and the extent related to the are rights and obligations without limitation Guarant w, all claims, suits, causes gainst any person, whether element, any other document or related to any of the halpractice claims, statuted obligations sold and assigned pursuant to classigned Interest"). Such says	rrevocably sells and assigned from Assignor, subjected the Loan Agreement, and below, (i) all of Assignor any other documents or mount and percentage into of Assignor under the resties, and (ii) to the extent as of action and any other er known or unknown, are the er known or unknown, are the er to instruments deliver foregoing, including, but the ory claims and all other claims and all other claims and all other claims (i) and (ii) above be alle and assignment is with ignment, without represent	to and in as of the Effective or's rights and instruments erest identified spective facilities permitted to be right of Assignor ising under or in ered pursuant not limited to, aims at Law or in (i) above (the sing referred to hout recourse to
1.	Assignor:				
2.	Assignee:]	[is an Affiliate/Ap	proved Fund of
3.	Borrower(s):	SHOPP	PES AT CHINO HILLS,	INC., a Minnesota corpo	ration
4.	Administrative Agent: BANK OF AMERICA, N.A., a national banking association, as the administrative agent under the Loan Agreement				
5.	Loan Agreeme		among Shoppes at Chino	Agreement, dated as of Months, Inc., a Minnesota and Bank of America, N.	corporation, the

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 7 of 64

6.	Assigned Interest:			
	Aggregate Amount of Commitment/Loans for all Lenders	Amount of Commitment/L Assigned	oans	Percentage Assigned of Commitment/Loans
	\$	\$		%
AGE TRA	NT AND WHICH SHAL NSFER IN THE REGIST	L BE THE EFF TER THEREFO	ECTIVI R.]	SERTED BY ADMINISTRATIVE E DATE OF RECORDATION OF
The to	erms set forth in this Assig	nment are hereby	agreed t	o:
			ASSI	GNOR:
		To when .		
			By: _	Title:
				·
			ASSI	GNEE:
			By:	
			-) .	Title:
[Cons	sented to and] Accepted:			
	······································			
as Ad	ministrative Agent			
By: _				
,	Title:			
[Cons	sented to:]			
Ву: _				
ر	Title:	4		

ANNEX 1 TO ASSIGNMENT AND ASSUMPTION

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

1. Representations and Warranties.

- 1.1. Assignor. Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim, and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Loan Agreement, any other Loan Document or the Environmental Agreement, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents, or any collateral thereunder, or the Environmental Agreement, (iii) the financial condition of Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or the Environmental Agreement, or (iv) the performance or observance by Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document or the Environmental Agreement.
- Assignee. Assignee (a) represents and warrants that (i) it has full power and 1.2. authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby and to become a Lender under the Loan Agreement, (ii) it meets all requirements of an Eligible Assignee under the Loan Agreement (subject to receipt of such consents as may be required under the Loan Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Loan Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Loan Agreement, together with copies of the most recent financial statements delivered pursuant to Section 3.9 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision independently and without reliance on Administrative Agent or any other Lender to enter into this Assignment and to purchase the Assigned Interest on the basis of which it has made such analysis and decision, and (v) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Loan Agreement, duly completed and executed by Assignee; and (b) agrees that (i) it will, independently and without reliance on Administrative Agent, Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents and the Environmental Agreement, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents and the Environmental Agreement are required to be performed by it as a Lender.
- 1.3 <u>Assignee's Address for Notices, etc.</u> Attached hereto as <u>Schedule 1</u> is all contact information, address, account and other administrative information relating to Assignee.

- 2. Payments. From and after the Effective Date, Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to Assignee whether such amounts have accrued prior to or on or after the Effective Date. Assignor and Assignee shall make all appropriate adjustments in payments by Administrative Agent for periods prior to the Effective Date or with respect to the making of this Assignment directly between themselves.
- 3. General Provisions. This Assignment shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment. This Assignment shall be governed by, and construed in accordance with, the Law of the State of Arizona.

SCHEDULE 1 TO ASSIGNMENT AND ASSUMPTION

ADMINISTRATIVE DETAILS

(Assignee to list names of credit contacts, addresses, phone and facsimile numbers, electronic mail addresses and account and payment information)

(a)	LIBOR Lending Office:
	Assignee Name:
	Address:
	Mail Code:
	Attn.:
	Telephone:
	Facsimile:
	Electronic Mail:
(b)	Domestic Lending Office:
	Assignee Name:
	Address:
	Mail Code:
	Mail Code:
	Attn.: Telephone:
	T ' '1
	Electronic Mail:

(c)	Notices:
	Assignee Name:
	Address:
	Mail Code:
	Attn ·
	Attn.: Telephone:
	Facsimile:
	Electronic Mail:
(d)	Payment Instructions:
	Account No.:
	Attn.:
	Reference:

EXHIBIT "M"

CONSTRUCTION DEED OF TRUST NOTE

FOR VALUE RECEIVED, SHOPPES AT CHINO HILLS, INC., a Minnesota
rporation ("Borrower," whether one or more), hereby promises to pay to the order of
("Lender"), under that certain Loan Agreement (defined below) among
orrower, Bank of America, N.A., a national banking association (together with any and all of
s successors and assigns, "Administrative Agent") as agent for the benefit of the lenders
ollectively, "Lenders") from time to time a party to that certain Construction Loan Agreement
he "Loan Agreement") dated May 16, 2007 of even date herewith, without offset, in
nmediately available funds in lawful money of the United States of America, at Administrative
gent's Office as defined in the Loan Agreement, the principal sum of
DOLLARS (\$) (or the
npaid balance of all principal advanced against this Note, if that amount is less), together with
terest on the unpaid principal balance of this Note from day to day outstanding as hereinafter rovided.

- 1. Note; Interest; Payment Schedule and Maturity Date. This Note is one of the Notes referred to in the Loan Agreement and is entitled to the benefits thereof and subject to prepayment in whole or in part as provided therein. The entire principal balance of this Note then unpaid shall be due and payable at the times set forth in the Loan Agreement. Accrued unpaid interest shall be due and payable at the times and at the interest rate(s) set forth in the Loan Agreement until all principal and accrued interest owing on this Note shall have been fully paid and satisfied. Any amount not paid when due and payable hereunder shall, to the extent permitted by applicable Law, bear interest and, if applicable, a late charge as set forth in the Loan Agreement.
- 2. Security; Loan Documents. The security for this Note includes a Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing (which, as it may have been or may be amended, restated, modified or supplemented from time to time, is herein called the "Deed of Trust") dated May 16, 2007 from SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, to Administrative Agent, as Trustee, covering certain property in Chino Hills, San Bernardino County, California described therein (the "Property"). This Note, the Deed of Trust, the Loan Agreement and all other documents now or hereafter securing, guaranteeing or executed in connection with the loan evidenced by this Note (the "Loan"), as the same have been or may be amended, restated, modified or supplemented from time to time, are herein sometimes called individually a "Loan Document" and together the "Loan Documents." Notwithstanding the foregoing, the Environmental Indemnity Agreement of even date herewith ("Environmental Indemnity Agreement") is not a Loan Document.

3. Defaults.

- (a) It shall be a default ("<u>Default</u>") under this Note and each of the other Loan Documents if (i) any principal, interest or other amount of money due under this Note is not paid in full within five (5) days of the date when due, regardless of how such amount may have become due, (ii) any covenant, agreement, condition, representation or warranty herein or in any other Loan Document is not fully and timely performed, observed or kept, subject to any applicable grace or cure periods, or (iii) there shall occur any default or event of default under the Deed of Trust or any other Loan Document, subject to any applicable grace or cure periods. Upon the occurrence of a Default, Administrative Agent on behalf of Lenders shall have the rights to declare the unpaid principal balance and accrued but unpaid interest on this Note, and all other amounts due hereunder and under the other Loan Documents, at once due and payable (and upon such declaration, the same shall be at once due and payable), to foreclose any liens and security interests securing payment hereof and to exercise any of its other rights, powers and remedies under this Note, under any other Loan Document, or at Law or in equity.
- All of the rights, remedies, powers and privileges (together, "Rights") of Administrative Agent on behalf of Lenders provided for in this Note and in any other Loan Document are cumulative of each other and of any and all other Rights at Law or in equity. The resort to any Right shall not prevent the concurrent or subsequent employment of any other appropriate Right. No single or partial exercise of any Right shall exhaust it or preclude any other or further exercise thereof, and every Right may be exercised at any time and from time to time. No failure by Administrative Agent or Lenders to exercise, and no delay in exercising any Right, including the right to accelerate the maturity of this Note, shall be construed as a waiver of any Default or as a waiver of any Right. Without limiting the generality of the foregoing provisions. the acceptance by Lender from time to time of any payment under this Note which is past due or which is less than the payment in full of all amounts due and payable at the time of such payment, shall not (i) constitute a waiver of or impair or extinguish the right of Administrative Agent or Lenders to accelerate the maturity of this Note or to exercise any other Right at the time or at any subsequent time, or nullify any prior exercise of any such Right, (ii) constitute a waiver of the requirement of punctual payment and performance or a novation in any respect, or (iii) in any way excuse the existence of a Default.
- (c) If any holder of this Note retains an attorney in connection with any Default or at maturity or to collect, enforce or defend this Note or any other Loan Document in any lawsuit or in any probate, reorganization, bankruptcy, arbitration or other proceeding, or if Borrower sues any holder in connection with this Note or any other Loan Document and does not prevail, then Borrower agrees to pay to each such holder, in addition to principal, interest and any other sums owing to Lender[s] hereunder and under the other Loan Documents, all costs and expenses incurred by such holder in trying to collect this Note or in any such suit or proceeding, including, without limitation, attorneys' fees and expenses, investigation costs and all court costs, whether or not suit is filed hereon, whether before or after the Maturity Date, or whether in connection with bankruptcy, insolvency or appeal, or whether collection is made against Borrower or any guarantor or endorser or any other person primarily or secondarily liable hereunder.

- Heirs, Successors and Assigns. The terms of this Note and of the other Loan 4. Documents shall bind and inure to the benefit of the heirs, devisees, representatives, successors and assigns of the parties. The foregoing sentence shall not be construed to permit Borrower to assign the Loan except as otherwise permitted under the Loan Documents. As further provided in the Loan Agreement, a Lender may, at any time, sell, transfer, or assign all or a portion of its interest in this Note, the Deed of Trust and the other Loan Documents, as set forth in the Loan Agreement.
- General Provisions. Time is of the essence with respect to Borrower's obligations 5. under this Note. If more than one person or entity executes this Note as Borrower, all of said parties shall be jointly and severally liable for payment of the indebtedness evidenced hereby. Borrower and all sureties, endorsers, guarantors and any other party now or hereafter liable for the payment of this Note in whole or in part, hereby severally (a) waive demand, presentment for payment, notice of dishonor and of nonpayment, protest, notice of protest, notice of intent to accelerate, notice of acceleration and all other notices (except any notices which are specifically required by this Note or any other Loan Document), filing of suit and diligence in collecting this Note or enforcing any of the security herefor; (b) agree to any substitution, subordination, exchange or release of any such security or the release of any party primarily or secondarily liable hereon; (c) agree that neither Administrative Agent nor any Lender shall be required first to institute suit or exhaust its remedies hereon against Borrower or others liable or to become liable hereon or to perfect or enforce its rights against them or any security herefor; (d) consent to any extensions or postponements of time of payment of this Note for any period or periods of time and to any partial payments, before or after maturity, and to any other indulgences with respect hereto, without notice thereof to any of them; and (e) submit (and waive all rights to object) to non-exclusive personal jurisdiction of any state or federal court sitting in the city and county, and venue in the city or county, in which payment is to be made as specified in Section 1 of this Note, for the enforcement of any and all obligations under this Note and the Loan Documents; (f) waive the benefit of all homestead and similar exemptions as to this Note; (g) agree that their liability under this Note shall not be affected or impaired by any determination that any security interest or lien taken by Lender to secure this Note is invalid or unperfected; and (h) hereby subordinate any and all rights against Borrower and any of the security for the payment of this Note, whether by subrogation, agreement or otherwise, until this Note is paid in full. A determination that any provision of this Note is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Note to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances. This Note may not be amended except in a writing specifically intended for such purpose and executed by the party against whom enforcement of the amendment is sought. Captions and headings in this Note are for convenience only and shall be disregarded in construing it. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." THIS NOTE, AND ITS VALIDITY, ENFORCEMENT AND INTERPRETATION, SHALL BE GOVERNED BY ARIZONA LAW (WITHOUT REGARD TO ANY CONFLICT OF LAWS PRINCIPLES) AND APPLICABLE UNITED STATES FEDERAL LAW.
- Notices. Any notice, request, or demand to or upon Borrower or Lender shall be deemed to have been properly given or made when delivered in accordance with the Loan Agreement.

- M-3 -

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 14 of 64

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN AND AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the date first above written.

BORROWER:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Ву:	 _		
Name:	 		
Title:	 -		

EXHIBIT "N"

SCHEDULE OF LENDERS

BANK OF AMERICA, N.A., as Administrative Agent

Notices: 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Mail Code: AZ1-200-22-17 Attn.: Edgardo E. Martinez Telephone: 602-523-1250 Facsimile: 602-523-4396 Electronic Mail: Edgardo.e.martinez@bankofamerica.com Payment Instructions: Account No.: 1366211723001 ABA #: 026009593 Reference: Shoppes at Chino Hills – Loan #761025 BANK OF AMERICA, N.A., as Lender Domestic and LIBOR Lending Office: Commitment Amount: \$27,500,000.00 Bank of America, N.A. Pro Rata Share: 25.0% Commercial Real Estate Banking Group, AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Notices: Bank of America, N.A. Commercial Real Estate Banking Group, AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343 Payment Instructions: Account No.: ___ Attention: Nancy Alonzo

Reference:

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 16 of 64

U.S. BANK, NATIONAL ASSOCIATION

Domestic and LIBOR Lending Office:

Commitment Amount: \$27,500,000.00

Commitment Amount: \$27,500,000.00

Pro Rata Share: 25.0%

Pro Rata Share: 25.0%

U.S. Bank, National Association

Mail Code: LM-AZ-X16E

101 North First Avenue, Suite 1600

Phoenix, Arizona 85003 Attn.: Barbara Richards

Telephone: (602) 257-5421 Facsimile: (602) 257-5422

Electronic Mail: barbara.richards@usbank.com

Notices:

U.S. Bank, National Association Mail Code: LM-AZ-X16E

101 North First Avenue, Suite 1600

Phoenix, Arizona 85003 Attn.: Barbara Richards

Telephone: (602) 257-5421 Facsimile: (602) 257-5422

Electronic Mail: barbara.richards@usbank.com

Payment Instructions:

Account No.: 00340012160600

ABA#: 123000220

Attn.: Commercial Loan Servicing West Reference: Shoppes at Chino Hills, Inc.

KEYBANK NATIONAL ASSOCIATION

Domestic and LIBOR Lending Office:

KeyBank National Association 2350 East Camelback Road, Suite 220

Phoenix, Arizona 850016 Attn.: James Endrizzi

Telephone: 602-778-4090 Facsimile: 602-778-4098

Electronic Mail: james a endrizzi@keybank.com

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 17 of 64

Commitment Amount: \$27,500,000.00

Pro Rata Share: 25.0%

Notices:

KeyBank National Association 2350 East Camelback Road, Suite 220 Phoenix, Arizona 850016

Attn.: James Endrizzi

Telephone: 602-778-4090 Facsimile: 602-778-4098

Electronic Mail: james a endrizzi@keybank.com

Payment Instructions:

Account No.: 1140228209012

Attn.: Leslie Marquez ABA #: 041001039

Reference: Shoppes at Chino Hills

UNION BANK OF CALIFORNIA, N.A.

Domestic and LIBOR Lending Office:

Union Bank of California, N.A. San Francisco Corporate Office 350 California Street, 7th Floor

San Francisco, California 94104

Mail Code: H-710 Attn.: Karen Kokame

Telephone: 415-705-7116 Facsimile: 415-433-7438

Electronic Mail: karen.kokame@uboc.com

Notices:

Union Bank of California, N.A. San Francisco Corporate Office 350 California Street, 7th Floor San Francisco, California 94104

Mail Code: H-710 Attn.: Karen Kokame

Telephone: 415-705-7116 Facsimile: 415-433-7438

Electronic Mail: karen.kokame@uboc.com

With a copy to:

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 18 of 64

Union Bank of California, N.A. Commercial Real Estate Loan Administration 18300 Von Karman Avenue Irvine, California 92612 Mail Code: 4-455-495 Attn.: Rosalind Johnson

Telephone: 949-553-7154 Facsimile: 949-553-7123

Electronic Mail: rosalind.johnson@uboc.com

Payment Instructions:

Account No.: 77070196431

ABA#: 122000496

Attn.: RE-96406 Commercial Loan Ops Reference: Shoppes at Chino Hills, Inc.

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 19 of 64

ATTACHMENT 2

CONSTRUCTION DEED OF TRUST NOTE

\$27,500,000.00 May 16, 2007

FOR VALUE RECEIVED, SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower," whether one or more), hereby promises to pay to the order of BANK OF AMERICA, N.A. ("Lender"), under that certain Loan Agreement (defined below) among Borrower, Bank of America, N.A., a national banking association (together with any and all of its successors and assigns, "Administrative Agent") as agent for the benefit of the lenders (collectively, "Lenders") from time to time a party to that certain Construction Loan Agreement (the "Loan Agreement") dated May 16, 2007 of even date herewith, without offset, in immediately available funds in lawful money of the United States of America, at Administrative Agent's Office as defined in the Loan Agreement, the principal sum of TWENTY SEVEN MILLION FIVE HUNDRED THOUSAND AND NO/100 DOLLARS (\$27,500,000.00) (or the unpaid balance of all principal advanced against this Note, if that amount is less), together with interest on the unpaid principal balance of this Note from day to day outstanding as hereinafter provided.

- Note: Interest; Payment Schedule and Maturity Date. This Note is one of the Notes referred to in the Loan Agreement and is entitled to the benefits thereof and subject to prepayment in whole or in part as provided therein. The entire principal balance of this Note then unpaid shall be due and payable at the times set forth in the Loan Agreement. Accrued unpaid interest shall be due and payable at the times and at the interest rate(s) set forth in the Loan Agreement until all principal and accrued interest owing on this Note shall have been fully paid and satisfied. Any amount not paid when due and payable hereunder shall, to the extent permitted by applicable Law, bear interest and, if applicable, a late charge as set forth in the Loan Agreement.
- 2. <u>Security; Loan Documents</u>. The security for this Note includes a Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing (which, as it may have been or may be amended, restated, modified or supplemented from time to time, is herein called the "<u>Deed of Trust</u>") dated May 16, 2007 from SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, to Administrative Agent, as Trustee, covering certain property in Chino Hills, San Bernardino County, California described therein (the "<u>Property</u>"). This Note, the Deed of Trust, the Loan Agreement and all other documents now or hereafter securing, guaranteeing or executed in connection with the loan evidenced by this Note (the "<u>Loan</u>"), as the same have been or may be amended, restated, modified or supplemented from time to time, are herein sometimes called individually a "<u>Loan Document</u>" and together the "<u>Loan Documents</u>." Notwithstanding the foregoing, the Environmental Indemnity Agreement of even date herewith ("<u>Environmental Indemnity Agreement</u>") is not a Loan Document.

3. Defaults.

(a) It shall be a default ("Default") under this Note and each of the other Loan Documents if (i) any principal, interest or other amount of money due under this Note is not paid in full within five (5) days of the date when due, regardless of how such amount may

have become due, (ii) any covenant, agreement, condition, representation or warranty herein or in any other Loan Document is not fully and timely performed, observed or kept, subject to any applicable grace or cure periods, or (iii) there shall occur any default or event of default under the Deed of Trust or any other Loan Document, subject to any applicable grace or cure periods. Upon the occurrence of a Default, Administrative Agent on behalf of Lenders shall have the rights to declare the unpaid principal balance and accrued but unpaid interest on this Note, and all other amounts due hereunder and under the other Loan Documents, at once due and payable (and upon such declaration, the same shall be at once due and payable), to foreclose any liens and security interests securing payment hereof and to exercise any of its other rights, powers and remedies under this Note, under any other Loan Document, or at Law or in equity.

- All of the rights, remedies, powers and privileges (together, "Rights") of Administrative Agent on behalf of Lenders provided for in this Note and in any other Loan Document are cumulative of each other and of any and all other Rights at Law or in equity. The resort to any Right shall not prevent the concurrent or subsequent employment of any other appropriate Right. No single or partial exercise of any Right shall exhaust it or preclude any other or further exercise thereof, and every Right may be exercised at any time and from time to time. No failure by Administrative Agent or Lenders to exercise, and no delay in exercising any Right, including the right to accelerate the maturity of this Note, shall be construed as a waiver of any Default or as a waiver of any Right. Without limiting the generality of the foregoing provisions, the acceptance by Lender from time to time of any payment under this Note which is past due or which is less than the payment in full of all amounts due and payable at the time of such payment, shall not (i) constitute a waiver of or impair or extinguish the right of Administrative Agent or Lenders to accelerate the maturity of this Note or to exercise any other Right at the time or at any subsequent time, or nullify any prior exercise of any such Right, (ii) constitute a waiver of the requirement of punctual payment and performance or a novation in any respect, or (iii) in any way excuse the existence of a Default.
- (c) If any holder of this Note retains an attorney in connection with any Default or at maturity or to collect, enforce or defend this Note or any other Loan Document in any lawsuit or in any probate, reorganization, bankruptcy, arbitration or other proceeding, or if Borrower sues any holder in connection with this Note or any other Loan Document and does not prevail, then Borrower agrees to pay to each such holder, in addition to principal, interest and any other sums owing to Lender[s] hereunder and under the other Loan Documents, all costs and expenses incurred by such holder in trying to collect this Note or in any such suit or proceeding, including, without limitation, attorneys' fees and expenses, investigation costs and all court costs, whether or not suit is filed hereon, whether before or after the Maturity Date, or whether in connection with bankruptcy, insolvency or appeal, or whether collection is made against Borrower or any guarantor or endorser or any other person primarily or secondarily liable hereunder.
- 4. <u>Heirs, Successors and Assigns</u>. The terms of this Note and of the other Loan Documents shall bind and inure to the benefit of the heirs, devisees, representatives, successors and assigns of the parties. The foregoing sentence shall not be construed to permit Borrower to assign the Loan except as otherwise permitted under the Loan

Documents. As further provided in the Loan Agreement, a Lender may, at any time, sell, transfer, or assign all or a portion of its interest in this Note, the Deed of Trust and the other Loan Documents, as set forth in the Loan Agreement.

- 5. General Provisions. Time is of the essence with respect to Borrower's obligations under this Note. If more than one person or entity executes this Note as Borrower, all of said parties shall be jointly and severally liable for payment of the indebtedness evidenced hereby. Borrower and all sureties, endorsers, guarantors and any other party now or hereafter liable for the payment of this Note in whole or in part, hereby severally (a) waive demand, presentment for payment, notice of dishonor and of nonpayment, protest, notice of protest, notice of intent to accelerate, notice of acceleration and all other notices (except any notices which are specifically required by this Note or any other Loan Document), filing of suit and diligence in collecting this Note or enforcing any of the security herefor; (b) agree to any substitution, subordination, exchange or release of any such security or the release of any party primarily or secondarily liable hereon; (c) agree that neither Administrative Agent nor any Lender shall be required first to institute suit or exhaust its remedies hereon against Borrower or others liable or to become liable hereon or to perfect or enforce its rights against them or any security herefor; (d) consent to any extensions or postponements of time of payment of this Note for any period or periods of time and to any partial payments, before or after maturity. and to any other indulgences with respect hereto, without notice thereof to any of them; and (e) submit (and waive all rights to object) to non-exclusive personal jurisdiction of any state or federal court sitting in the city and county, and venue in the city or county, in which payment is to be made as specified in Section 1 of this Note, for the enforcement of any and all obligations under this Note and the Loan Documents; (f) waive the benefit of all homestead and similar exemptions as to this Note; (g) agree that their liability under this Note shall not be affected or impaired by any determination that any security interest or lien taken by Lender to secure this Note is invalid or unperfected; and (h) hereby subordinate any and all rights against Borrower and any of the security for the payment of this Note, whether by subrogation, agreement or otherwise, until this Note is paid in full. A determination that any provision of this Note is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Note to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances. This Note may not be amended except in a writing specifically intended for such purpose and executed by the party against whom enforcement of the amendment is sought. Captions and headings in this Note are for convenience only and shall be disregarded in construing it. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." THIS NOTE, AND ITS VALIDITY, ENFORCEMENT AND INTERPRETATION, SHALL BE GOVERNED BY ARIZONA LAW (WITHOUT REGARD TO ANY CONFLICT OF LAWS PRINCIPLES) AND APPLICABLE UNITED STATES FEDERAL LAW.
- 6. <u>Notices</u>. Any notice, request, or demand to or upon Borrower or Lender shall be deemed to have been properly given or made when delivered in accordance with the Loan Agreement.

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN AND AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 23 of 64

EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the date first above written.

BORROWER:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Name:

Title:

Charles Vog

CONSTRUCTION DEED OF TRUST NOTE

\$27,500,000.00 May 16, 2007

FOR VALUE RECEIVED, SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower," whether one or more), hereby promises to pay to the order of KEYBANK NATIONAL ASSOCIATION ("Lender"), under that certain Loan Agreement (defined below) among Borrower, Bank of America, N.A., a national banking association (together with any and all of its successors and assigns, "Administrative Agent") as agent for the benefit of the lenders (collectively, "Lenders") from time to time a party to that certain Construction Loan Agreement (the "Loan Agreement") dated May 16, 2007 of even date herewith, without offset, in immediately available funds in lawful money of the United States of America, at Administrative Agent's Office as defined in the Loan Agreement, the principal sum of TWENTY SEVEN MILLION FIVE HUNDRED THOUSAND AND NO/100 DOLLARS (\$27,500,000.00) (or the unpaid balance of all principal advanced against this Note, if that amount is less), together with interest on the unpaid principal balance of this Note from day to day outstanding as hereinafter provided.

- Note; Interest; Payment Schedule and Maturity Date. This Note is one of the Notes referred to in the Loan Agreement and is entitled to the benefits thereof and subject to prepayment in whole or in part as provided therein. The entire principal balance of this Note then unpaid shall be due and payable at the times set forth in the Loan Agreement. Accrued unpaid interest shall be due and payable at the times and at the interest rate(s) set forth in the Loan Agreement until all principal and accrued interest owing on this Note shall have been fully paid and satisfied. Any amount not paid when due and payable hereunder shall, to the extent permitted by applicable Law, bear interest and, if applicable, a late charge as set forth in the Loan Agreement.
- 2. <u>Security; Loan Documents</u>. The security for this Note includes a Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing (which, as it may have been or may be amended, restated, modified or supplemented from time to time, is herein called the "<u>Deed of Trust</u>") dated May 16, 2007 from SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, to Administrative Agent, as Trustee, covering certain property in Chino Hills, San Bernardino County, California described therein (the "<u>Property</u>"). This Note, the Deed of Trust, the Loan Agreement and all other documents now or hereafter securing, guaranteeing or executed in connection with the loan evidenced by this Note (the "<u>Loan</u>"), as the same have been or may be amended, restated, modified or supplemented from time to time, are herein sometimes called individually a "<u>Loan Document</u>" and together the "<u>Loan Documents</u>." Notwithstanding the foregoing, the Environmental Indemnity Agreement of even date herewith ("Environmental Indemnity Agreement") is not a Loan Document.

3. Defaults.

(a) It shall be a default ("<u>Default</u>") under this Note and each of the other Loan Documents if (i) any principal, interest or other amount of money due under this Note is not paid in full within five (5) days of the date when due, regardless of how such amount may

have become due, (ii) any covenant, agreement, condition, representation or warranty herein or in any other Loan Document is not fully and timely performed, observed or kept, subject to any applicable grace or cure periods, or (iii) there shall occur any default or event of default under the Deed of Trust or any other Loan Document, subject to any applicable grace or cure periods. Upon the occurrence of a Default, Administrative Agent on behalf of Lenders shall have the rights to declare the unpaid principal balance and accrued but unpaid interest on this Note, and all other amounts due hereunder and under the other Loan Documents, at once due and payable (and upon such declaration, the same shall be at once due and payable), to foreclose any liens and security interests securing payment hereof and to exercise any of its other rights, powers and remedies under this Note, under any other Loan Document, or at Law or in equity.

- All of the rights, remedies, powers and privileges (together, "Rights") of (b) Administrative Agent on behalf of Lenders provided for in this Note and in any other Loan Document are cumulative of each other and of any and all other Rights at Law or in equity. The resort to any Right shall not prevent the concurrent or subsequent employment of any other appropriate Right. No single or partial exercise of any Right shall exhaust it or preclude any other or further exercise thereof, and every Right may be exercised at any time and from time to time. No failure by Administrative Agent or Lenders to exercise, and no delay in exercising any Right, including the right to accelerate the maturity of this Note, shall be construed as a waiver of any Default or as a waiver of any Right. Without limiting the generality of the foregoing provisions, the acceptance by Lender from time to time of any payment under this Note which is past due or which is less than the payment in full of all amounts due and payable at the time of such payment, shall not (i) constitute a waiver of or impair or extinguish the right of Administrative Agent or Lenders to accelerate the maturity of this Note or to exercise any other Right at the time or at any subsequent time, or nullify any prior exercise of any such Right, (ii) constitute a waiver of the requirement of punctual payment and performance or a novation in any respect, or (iii) in any way excuse the existence of a Default.
- (c) If any holder of this Note retains an attorney in connection with any Default or at maturity or to collect, enforce or defend this Note or any other Loan Document in any lawsuit or in any probate, reorganization, bankruptcy, arbitration or other proceeding, or if Borrower sues any holder in connection with this Note or any other Loan Document and does not prevail, then Borrower agrees to pay to each such holder, in addition to principal, interest and any other sums owing to Lender[s] hereunder and under the other Loan Documents, all costs and expenses incurred by such holder in trying to collect this Note or in any such suit or proceeding, including, without limitation, attorneys' fees and expenses, investigation costs and all court costs, whether or not suit is filed hereon, whether before or after the Maturity Date, or whether in connection with bankruptcy, insolvency or appeal, or whether collection is made against Borrower or any guarantor or endorser or any other person primarily or secondarily liable hereunder.
- 4. <u>Heirs, Successors and Assigns</u>. The terms of this Note and of the other Loan Documents shall bind and inure to the benefit of the heirs, devisees, representatives, successors and assigns of the parties. The foregoing sentence shall not be construed to permit Borrower to assign the Loan except as otherwise permitted under the Loan

Documents. As further provided in the Loan Agreement, a Lender may, at any time, sell, transfer, or assign all or a portion of its interest in this Note, the Deed of Trust and the other Loan Documents, as set forth in the Loan Agreement.

- 5. General Provisions. Time is of the essence with respect to Borrower's obligations under this Note. If more than one person or entity executes this Note as Borrower, all of said parties shall be jointly and severally liable for payment of the indebtedness evidenced hereby. Borrower and all sureties, endorsers, guarantors and any other party now or hereafter liable for the payment of this Note in whole or in part, hereby severally (a) waive demand, presentment for payment, notice of dishonor and of nonpayment, protest, notice of protest, notice of intent to accelerate, notice of acceleration and all other notices (except any notices which are specifically required by this Note or any other Loan Document), filing of suit and diligence in collecting this Note or enforcing any of the security herefor; (b) agree to any substitution, subordination, exchange or release of any such security or the release of any party primarily or secondarily liable hereon; (c) agree that neither Administrative Agent nor any Lender shall be required first to institute suit or exhaust its remedies hereon against Borrower or others liable or to become liable hereon or to perfect or enforce its rights against them or any security herefor; (d) consent to any extensions or postponements of time of payment of this Note for any period or periods of time and to any partial payments, before or after maturity, and to any other indulgences with respect hereto, without notice thereof to any of them; and (e) submit (and waive all rights to object) to non-exclusive personal jurisdiction of any state or federal court sitting in the city and county, and venue in the city or county, in which payment is to be made as specified in Section 1 of this Note, for the enforcement of any and all obligations under this Note and the Loan Documents; (f) waive the benefit of all homestead and similar exemptions as to this Note; (g) agree that their liability under this Note shall not be affected or impaired by any determination that any security interest or lien taken by Lender to secure this Note is invalid or unperfected; and (h) hereby subordinate any and all rights against Borrower and any of the security for the payment of this Note, whether by subrogation, agreement or otherwise, until this Note is paid in full. A determination that any provision of this Note is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Note to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances. This Note may not be amended except in a writing specifically intended for such purpose and executed by the party against whom enforcement of the amendment is sought. Captions and headings in this Note are for convenience only and shall be disregarded in construing it. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." THIS NOTE, AND ITS VALIDITY, ENFORCEMENT AND INTERPRETATION, SHALL BE GOVERNED BY ARIZONA LAW (WITHOUT REGARD TO ANY CONFLICT OF LAWS PRINCIPLES) AND APPLICABLE UNITED STATES FEDERAL LAW.
- 6. <u>Notices</u>. Any notice, request, or demand to or upon Borrower or Lender shall be deemed to have been properly given or made when delivered in accordance with the Loan Agreement.

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN AND AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 27 of 64

EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the date first above written.

BORROWER:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Name:___

Title:

Charles Vogel
Vice President

CONSTRUCTION DEED OF TRUST NOTE

\$27,500,000.00 May 16, 2007

FOR VALUE RECEIVED, SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower," whether one or more), hereby promises to pay to the order of UNION BANK OF CALIFORNIA, N.A. ("Lender"), under that certain Loan Agreement (defined below) among Borrower, Bank of America, N.A., a national banking association (together with any and all of its successors and assigns, "Administrative Agent") as agent for the benefit of the lenders (collectively, "Lenders") from time to time a party to that certain Construction Loan Agreement (the "Loan Agreement") dated May 16, 2007 of even date herewith, without offset, in immediately available funds in lawful money of the United States of America, at Administrative Agent's Office as defined in the Loan Agreement, the principal sum of TWENTY SEVEN MILLION FIVE HUNDRED THOUSAND AND NO/100 DOLLARS (\$27,500,000.00) (or the unpaid balance of all principal advanced against this Note, if that amount is less), together with interest on the unpaid principal balance of this Note from day to day outstanding as hereinafter provided.

- Note; Interest; Payment Schedule and Maturity Date. This Note is one of the Notes referred to in the Loan Agreement and is entitled to the benefits thereof and subject to prepayment in whole or in part as provided therein. The entire principal balance of this Note then unpaid shall be due and payable at the times set forth in the Loan Agreement. Accrued unpaid interest shall be due and payable at the times and at the interest rate(s) set forth in the Loan Agreement until all principal and accrued interest owing on this Note shall have been fully paid and satisfied. Any amount not paid when due and payable hereunder shall, to the extent permitted by applicable Law, bear interest and, if applicable, a late charge as set forth in the Loan Agreement.
- 2. Security; Loan Documents. The security for this Note includes a Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing (which, as it may have been or may be amended, restated, modified or supplemented from time to time, is herein called the "Deed of Trust") dated May 16, 2007 from SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, to Administrative Agent, as Trustee, covering certain property in Chino Hills, San Bernardino County, California described therein (the "Property"). This Note, the Deed of Trust, the Loan Agreement and all other documents now or hereafter securing, guaranteeing or executed in connection with the loan evidenced by this Note (the "Loan"), as the same have been or may be amended, restated, modified or supplemented from time to time, are herein sometimes called individually a "Loan Document" and together the "Loan Documents." Notwithstanding the foregoing, the Environmental Indemnity Agreement of even date herewith ("Environmental Indemnity Agreement") is not a Loan Document.

3. Defaults.

(a) It shall be a default ("<u>Default</u>") under this Note and each of the other Loan Documents if (i) any principal, interest or other amount of money due under this Note is not paid in full within five (5) days of the date when due, regardless of how such amount may

have become due, (ii) any covenant, agreement, condition, representation or warranty herein or in any other Loan Document is not fully and timely performed, observed or kept, subject to any applicable grace or cure periods, or (iii) there shall occur any default or event of default under the Deed of Trust or any other Loan Document, subject to any applicable grace or cure periods. Upon the occurrence of a Default, Administrative Agent on behalf of Lenders shall have the rights to declare the unpaid principal balance and accrued but unpaid interest on this Note, and all other amounts due hereunder and under the other Loan Documents, at once due and payable (and upon such declaration, the same shall be at once due and payable), to foreclose any liens and security interests securing payment hereof and to exercise any of its other rights, powers and remedies under this Note, under any other Loan Document, or at Law or in equity.

- All of the rights, remedies, powers and privileges (together, "Rights") of Administrative Agent on behalf of Lenders provided for in this Note and in any other Loan Document are cumulative of each other and of any and all other Rights at Law or in equity. The resort to any Right shall not prevent the concurrent or subsequent employment of any other appropriate Right. No single or partial exercise of any Right shall exhaust it or preclude any other or further exercise thereof, and every Right may be exercised at any time and from time to time. No failure by Administrative Agent or Lenders to exercise, and no delay in exercising any Right, including the right to accelerate the maturity of this Note, shall be construed as a waiver of any Default or as a waiver of any Right. Without limiting the generality of the foregoing provisions, the acceptance by Lender from time to time of any payment under this Note which is past due or which is less than the payment in full of all amounts due and payable at the time of such payment, shall not (i) constitute a waiver of or impair or extinguish the right of Administrative Agent or Lenders to accelerate the maturity of this Note or to exercise any other Right at the time or at any subsequent time, or nullify any prior exercise of any such Right, (ii) constitute a waiver of the requirement of punctual payment and performance or a novation in any respect, or (iii) in any way excuse the existence of a Default.
- (c) If any holder of this Note retains an attorney in connection with any Default or at maturity or to collect, enforce or defend this Note or any other Loan Document in any lawsuit or in any probate, reorganization, bankruptcy, arbitration or other proceeding, or if Borrower sues any holder in connection with this Note or any other Loan Document and does not prevail, then Borrower agrees to pay to each such holder, in addition to principal, interest and any other sums owing to Lender[s] hereunder and under the other Loan Documents, all costs and expenses incurred by such holder in trying to collect this Note or in any such suit or proceeding, including, without limitation, attorneys' fees and expenses, investigation costs and all court costs, whether or not suit is filed hereon, whether before or after the Maturity Date, or whether in connection with bankruptcy, insolvency or appeal, or whether collection is made against Borrower or any guarantor or endorser or any other person primarily or secondarily liable hereunder.
- 4. <u>Heirs, Successors and Assigns</u>. The terms of this Note and of the other Loan Documents shall bind and inure to the benefit of the heirs, devisees, representatives, successors and assigns of the parties. The foregoing sentence shall not be construed to permit Borrower to assign the Loan except as otherwise permitted under the Loan

Documents. As further provided in the Loan Agreement, a Lender may, at any time, sell, transfer, or assign all or a portion of its interest in this Note, the Deed of Trust and the other Loan Documents, as set forth in the Loan Agreement.

- Time is of the essence with respect to Borrower's 5. General Provisions. obligations under this Note. If more than one person or entity executes this Note as Borrower, all of said parties shall be jointly and severally liable for payment of the indebtedness evidenced hereby. Borrower and all sureties, endorsers, guarantors and any other party now or hereafter liable for the payment of this Note in whole or in part, hereby severally (a) waive demand, presentment for payment, notice of dishonor and of nonpayment, protest, notice of protest, notice of intent to accelerate, notice of acceleration and all other notices (except any notices which are specifically required by this Note or any other Loan Document), filing of suit and diligence in collecting this Note or enforcing any of the security herefor; (b) agree to any substitution, subordination, exchange or release of any such security or the release of any party primarily or secondarily liable hereon; (c) agree that neither Administrative Agent nor any Lender shall be required first to institute suit or exhaust its remedies hereon against Borrower or others liable or to become liable hereon or to perfect or enforce its rights against them or any security herefor; (d) consent to any extensions or postponements of time of payment of this Note for any period or periods of time and to any partial payments, before or after maturity, and to any other indulgences with respect hereto, without notice thereof to any of them; and (e) submit (and waive all rights to object) to non-exclusive personal jurisdiction of any state or federal court sitting in the city and county, and venue in the city or county, in which payment is to be made as specified in Section 1 of this Note, for the enforcement of any and all obligations under this Note and the Loan Documents; (f) waive the benefit of all homestead and similar exemptions as to this Note; (g) agree that their liability under this Note shall not be affected or impaired by any determination that any security interest or lien taken by Lender to secure this Note is invalid or unperfected; and (h) hereby subordinate any and all rights against Borrower and any of the security for the payment of this Note, whether by subrogation, agreement or otherwise, until this Note is paid in full. A determination that any provision of this Note is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Note to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances. This Note may not be amended except in a writing specifically intended for such purpose and executed by the party against whom enforcement of the amendment is sought. Captions and headings in this Note are for convenience only and shall be disregarded in construing it. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." THIS NOTE, AND ITS VALIDITY, ENFORCEMENT AND INTERPRETATION, SHALL BE GOVERNED BY ARIZONA LAW (WITHOUT REGARD TO ANY CONFLICT OF LAWS PRINCIPLES) AND APPLICABLE UNITED STATES FEDERAL LAW.
- 6. <u>Notices</u>. Any notice, request, or demand to or upon Borrower or Lender shall be deemed to have been properly given or made when delivered in accordance with the Loan Agreement.

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN AND AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 31 of 64

EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the date first above written.

BORROWER:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Name: Charles Vogel

Title: Vice President

CONSTRUCTION DEED OF TRUST NOTE

\$27,500,000.00 May 16, 2007

FOR VALUE RECEIVED, SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower," whether one or more), hereby promises to pay to the order of U.S. BANK, NATIONAL ASSOCIATION ("Lender"), under that certain Loan Agreement (defined below) among Borrower, Bank of America, N.A., a national banking association (together with any and all of its successors and assigns, "Administrative Agent") as agent for the benefit of the lenders (collectively, "Lenders") from time to time a party to that certain Construction Loan Agreement (the "Loan Agreement") dated May 16, 2007 of even date herewith, without offset, in immediately available funds in lawful money of the United States of America, at Administrative Agent's Office as defined in the Loan Agreement, the principal sum of TWENTY SEVEN MILLION FIVE HUNDRED THOUSAND AND NO/100 DOLLARS (\$27,500,000.00) (or the unpaid balance of all principal advanced against this Note, if that amount is less), together with interest on the unpaid principal balance of this Note from day to day outstanding as hereinafter provided.

- 1. Note; Interest; Payment Schedule and Maturity Date. This Note is one of the Notes referred to in the Loan Agreement and is entitled to the benefits thereof and subject to prepayment in whole or in part as provided therein. The entire principal balance of this Note then unpaid shall be due and payable at the times set forth in the Loan Agreement. Accrued unpaid interest shall be due and payable at the times and at the interest rate(s) set forth in the Loan Agreement until all principal and accrued interest owing on this Note shall have been fully paid and satisfied. Any amount not paid when due and payable hereunder shall, to the extent permitted by applicable Law, bear interest and, if applicable, a late charge as set forth in the Loan Agreement.
- 2. Security; Loan Documents. The security for this Note includes a Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing (which, as it may have been or may be amended, restated, modified or supplemented from time to time, is herein called the "Deed of Trust") dated May 16, 2007 from SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, to Administrative Agent, as Trustee, covering certain property in Chino Hills, San Bernardino County, California described therein (the "Property"). This Note, the Deed of Trust, the Loan Agreement and all other documents now or hereafter securing, guaranteeing or executed in connection with the loan evidenced by this Note (the "Loan"), as the same have been or may be amended, restated, modified or supplemented from time to time, are herein sometimes called individually a "Loan Document" and together the "Loan Documents." Notwithstanding the foregoing, the Environmental Indemnity Agreement of even date herewith ("Environmental Indemnity Agreement") is not a Loan Document.

3. Defaults.

(a) It shall be a default ("Default") under this Note and each of the other Loan Documents if (i) any principal, interest or other amount of money due under this Note is not paid in full within five (5) days of the date when due, regardless of how such amount may

have become due, (ii) any covenant, agreement, condition, representation or warranty herein or in any other Loan Document is not fully and timely performed, observed or kept, subject to any applicable grace or cure periods, or (iii) there shall occur any default or event of default under the Deed of Trust or any other Loan Document, subject to any applicable grace or cure periods. Upon the occurrence of a Default, Administrative Agent on behalf of Lenders shall have the rights to declare the unpaid principal balance and accrued but unpaid interest on this Note, and all other amounts due hereunder and under the other Loan Documents, at once due and payable (and upon such declaration, the same shall be at once due and payable), to foreclose any liens and security interests securing payment hereof and to exercise any of its other rights, powers and remedies under this Note, under any other Loan Document, or at Law or in equity.

- All of the rights, remedies, powers and privileges (together, "Rights") of (b) Administrative Agent on behalf of Lenders provided for in this Note and in any other Loan Document are cumulative of each other and of any and all other Rights at Law or in equity. The resort to any Right shall not prevent the concurrent or subsequent employment of any other appropriate Right. No single or partial exercise of any Right shall exhaust it or preclude any other or further exercise thereof, and every Right may be exercised at any time and from time to time. No failure by Administrative Agent or Lenders to exercise, and no delay in exercising any Right, including the right to accelerate the maturity of this Note, shall be construed as a waiver of any Default or as a waiver of any Right. Without limiting the generality of the foregoing provisions, the acceptance by Lender from time to time of any payment under this Note which is past due or which is less than the payment in full of all amounts due and payable at the time of such payment, shall not (i) constitute a waiver of or impair or extinguish the right of Administrative Agent or Lenders to accelerate the maturity of this Note or to exercise any other Right at the time or at any subsequent time, or nullify any prior exercise of any such Right, (ii) constitute a waiver of the requirement of punctual payment and performance or a novation in any respect, or (iii) in any way excuse the existence of a Default.
- (c) If any holder of this Note retains an attorney in connection with any Default or at maturity or to collect, enforce or defend this Note or any other Loan Document in any lawsuit or in any probate, reorganization, bankruptcy, arbitration or other proceeding, or if Borrower sues any holder in connection with this Note or any other Loan Document and does not prevail, then Borrower agrees to pay to each such holder, in addition to principal, interest and any other sums owing to Lender[s] hereunder and under the other Loan Documents, all costs and expenses incurred by such holder in trying to collect this Note or in any such suit or proceeding, including, without limitation, attorneys' fees and expenses, investigation costs and all court costs, whether or not suit is filed hereon, whether before or after the Maturity Date, or whether in connection with bankruptcy, insolvency or appeal, or whether collection is made against Borrower or any guarantor or endorser or any other person primarily or secondarily liable hereunder.
- 4. <u>Heirs, Successors and Assigns</u>. The terms of this Note and of the other Loan Documents shall bind and inure to the benefit of the heirs, devisees, representatives, successors and assigns of the parties. The foregoing sentence shall not be construed to permit Borrower to assign the Loan except as otherwise permitted under the Loan

Documents. As further provided in the Loan Agreement, a Lender may, at any time, sell, transfer, or assign all or a portion of its interest in this Note, the Deed of Trust and the other Loan Documents, as set forth in the Loan Agreement.

- 5. General Provisions. Time is of the essence with respect to Borrower's obligations under this Note. If more than one person or entity executes this Note as Borrower, all of said parties shall be jointly and severally liable for payment of the indebtedness evidenced hereby. Borrower and all sureties, endorsers, guarantors and any other party now or hereafter liable for the payment of this Note in whole or in part, hereby severally (a) waive demand, presentment for payment, notice of dishonor and of nonpayment, protest, notice of protest, notice of intent to accelerate, notice of acceleration and all other notices (except any notices which are specifically required by this Note or any other Loan Document), filing of suit and diligence in collecting this Note or enforcing any of the security herefor; (b) agree to any substitution, subordination, exchange or release of any such security or the release of any party primarily or secondarily liable hereon; (c) agree that neither Administrative Agent nor any Lender shall be required first to institute suit or exhaust its remedies hereon against Borrower or others liable or to become liable hereon or to perfect or enforce its rights against them or any security herefor; (d) consent to any extensions or postponements of time of payment of this Note for any period or periods of time and to any partial payments, before or after maturity, and to any other indulgences with respect hereto, without notice thereof to any of them; and (e) submit (and waive all rights to object) to non-exclusive personal jurisdiction of any state or federal court sitting in the city and county, and venue in the city or county, in which payment is to be made as specified in Section 1 of this Note, for the enforcement of any and all obligations under this Note and the Loan Documents; (f) waive the benefit of all homestead and similar exemptions as to this Note; (g) agree that their liability under this Note shall not be affected or impaired by any determination that any security interest or lien taken by Lender to secure this Note is invalid or unperfected; and (h) hereby subordinate any and all rights against Borrower and any of the security for the payment of this Note, whether by subrogation, agreement or otherwise, until this Note is paid in full. A determination that any provision of this Note is unenforceable or invalid shall not affect the enforceability or validity of any other provision and the determination that the application of any provision of this Note to any person or circumstance is illegal or unenforceable shall not affect the enforceability or validity of such provision as it may apply to other persons or circumstances. This Note may not be amended except in a writing specifically intended for such purpose and executed by the party against whom enforcement of the amendment is sought. Captions and headings in this Note are for convenience only and shall be disregarded in construing it. The words "include" and "including" shall be interpreted as if followed by the words "without limitation." THIS NOTE, AND ITS VALIDITY, ENFORCEMENT AND INTERPRETATION, SHALL BE GOVERNED BY ARIZONA LAW (WITHOUT REGARD TO ANY CONFLICT OF LAWS PRINCIPLES) AND APPLICABLE UNITED STATES FEDERAL LAW.
- 6. <u>Notices</u>. Any notice, request, or demand to or upon Borrower or Lender shall be deemed to have been properly given or made when delivered in accordance with the Loan Agreement.

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN AND AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 35 of 64

EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO ORAL AGREEMENTS BETWEEN THE PARTIES.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the date first above written.

BORROWER:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Name:_____Title:

Vice President

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 36 of 64

ATTACHMENT 3

11:03 AM



LARRY WALKER Auditor/Controller - Recorder

706 First American - SR

Recording Requested By/Return To:

Bank of America, N.A. Commercial Real Estate Banking Group, AZ1-200-22-17 201 East Washington Street, 22nd Floor Phoenix, Arizona 85004-2343

Attention: N. Alonzo NCS. 284385

Doc #:	2007-0302264

Titles:	4	Pages: 31
Fees		122.00
Taxes		.00
Other		.00
PAID		122.00

CONSTRUCTION DEED OF TRUST, ASSIGNMENT, SECURITY AGREEMENT AND **FIXTURE FILING** (The Shoppes at Chino Hills, CA -Construction Loan)

by

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, as Grantor,

to and in favor of

BANK OF AMERICA, N.A., a national banking association, as Trustee,

and

BANK OF AMERICA, N.A., a national banking association, as Administrative Agent, as Beneficiary

This document serves as a Fixture Filing under the California Uniform Commercial Code.

Grantor's Organizational Identification Number is 12K-315.

Recording Requested By/Return To:

Bank of America, N.A.
Commercial Real Estate Banking
Group, AZ1-200-22-17
201 East Washington Street, 22nd Floor
Phoenix, Arizona 85004-2343
Attention: N. Alonzo
NC5. 284 385

CONSTRUCTION DEED OF TRUST, ASSIGNMENT,
SECURITY AGREEMENT AND
FIXTURE FILING
(The Shoppes at Chino Hills, CA -Construction Loan)

by

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, as Grantor,

to and in favor of

BANK OF AMERICA, N.A., a national banking association, as Trustee,

and

BANK OF AMERICA, N.A., a national banking association, as Administrative Agent, as Beneficiary

This document serves as a Fixture Filing under the California Uniform Commercial Code.

Grantor's Organizational Identification Number is 12K-315.

CONSTRUCTION DEED OF TRUST, ASSIGNMENT, SECURITY AGREEMENT AND FIXTURE FILING

(The Shoppes at Chino Hills, CA -Construction Loan)

This Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing is made as of the 16th day of May, 2007, by SHOPPES AT CHINO HILLS, INC., a Minnesota corporation (herein referred to as "Grantor"), whose address is 2555 East Camelback Road, Suite 800, Phoenix, Arizona 85016-9267, to BANK OF AMERICA, N.A., a national banking association ("Initial Trustee"), whose address is Commercial Real Estate Banking Group, AZ1-200-22-17, 201 East Washington Street, 22nd Floor, Phoenix, Arizona 85004-2343, Attention: N. Alonzo, and BANK OF AMERICA, N.A., a national banking association, as Administrative Agent for itself and other Lenders (collectively together with their successors and assigns, "Lender"), whose address is Commercial Real Estate Banking Group, AZ1-200-22-17, 201 East Washington Street, 22nd Floor, Phoenix, Arizona 85004-2343, Attention: N. Alonzo.

RECITALS

Grantor has requested that Lender make the Loan (as hereinafter defined) to Grantor. As a condition precedent to making the Loan, Lender has required that Grantor execute and deliver this Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing to Trustee and Lender.

GRANTS AND AGREEMENTS

NOW, THEREFORE, in order to induce Lender to make the Loan to Grantor, Grantor agrees as follows:

ARTICLE I DEFINITIONS

As used in this Deed of Trust, the terms defined in the Preamble hereto shall have the respective meanings specified therein, and the following additional terms shall have the meanings specified:

"Accessories" means all fixtures, equipment, systems, machinery, furniture, furnishings, appliances, inventory, goods, building and construction materials, supplies and other articles of personal property, of every kind and character, tangible and intangible (including software embedded therein), now owned or hereafter acquired by Grantor, which are now or hereafter attached to or situated in, on or about the Land or Improvements, or used in or necessary to the complete and proper planning, development, use, occupancy or operation thereof, or acquired (whether delivered to the Land or stored elsewhere) for use or installation in or on the Land or Improvements, and all Additions to the foregoing, all of which are hereby declared to be permanent accessions to the Land.

"Accounts" means all accounts of Grantor within the meaning of the Uniform Commercial Code of the State, derived from or arising out of the use, occupancy or enjoyment of the Property or for services rendered therein or thereon.

"Additions" means any and all alterations, additions, accessions and improvements to the Property, substitutions therefor, and renewals and replacements thereof.

"Beneficiary" means Lender and its successors and assigns.

"Claim" means any liability, suit, action, claim, demand, loss, expense, penalty, fine, judgment or other cost of any kind or nature whatsoever arising out of or relating to the Property, including fees, costs and expenses of attorneys, consultants, contractors and experts.

"Condemnation" means any taking of title to, use of, or any other interest in the Property under the exercise of the power of condemnation or eminent domain, whether temporarily or permanently, by any Governmental Authority or by any other Person acting under or for the benefit of a Governmental Authority.

"Condemnation Awards" means any and all judgments, awards of damages (including severance and consequential damages), payments, proceeds, settlements, amounts paid for a taking in lieu of Condemnation, or other compensation heretofore or hereafter made, including interest thereon, and the right to receive the same, as a result of, or in connection with, any Condemnation or threatened Condemnation.

"Contract of Sale" means any contract for the sale of all or any part of the Property or any interest therein, whether now in existence or hereafter executed.

"Deed of Trust" means this Construction Deed of Trust, Assignment, Security Agreement and Fixture Filing, as the same may from time to time be extended, amended, restated, supplemented or otherwise modified.

"Default" means an event or circumstance which, with the giving of Notice or lapse of time, or both, would constitute an Event of Default under the provisions of this Deed of Trust.

"Design and Construction Documents" means, collectively, (a) all contracts for services to be rendered, work to be performed or materials to be supplied in the development of the Land or the construction or repair of Improvements, including all agreements with architects, engineers or contractors for such services, work or materials; (b) all plans, drawings and specifications for the development of the Land or the construction or repair of Improvements; (c) all permits, licenses, variances and other rights or approvals issued by or obtained

from any Governmental Authority or other Person in connection with the development of the Land or the construction or repair of Improvements; and (d) all amendments of or supplements to any of the foregoing.

"Encumbrance" means any Lien, easement, right of way, roadway (public or private), condominium regime, cooperative housing regime, condition, covenant or restriction (including any CC&Rs in connection with any condominium development or cooperative housing development), Lease or other matter of any nature that would affect title to the Property.

"Environmental Agreement" means the Environmental Indemnification and Release Agreement of even date herewith by and between Grantor and Lender pertaining to the Property, as the same may from time to time be extended, amended, restated or otherwise modified. The Environmental Agreement is one of the Loan Documents, but this Deed of Trust does not secure the obligations of Grantor under the Environmental Agreement.

"Event of Default" means an event or circumstance specified in Article VI and the continuance of such event or circumstance beyond the applicable grace and/or cure periods therefor, if any, set forth in Article VI.

"Expenses" means all fees, charges, costs and expenses of any nature whatsoever incurred at any time and from time to time (whether before or after an Event of Default) by Beneficiary or Trustee in making, funding, administering or modifying the Loan, in negotiating or entering into any "workout" of the Loan, or in exercising or enforcing any rights, powers and remedies provided in this Deed of Trust or any of the other Loan Documents, including attorneys' fees, court costs, receiver's fees, management fees and costs incurred in the repair, maintenance and operation of, or taking possession of, or selling, the Property.

"Governmental Authority" means any governmental or quasigovernmental entity, including any court, department, commission, board, bureau, agency, administration, service, district or other instrumentality of any governmental entity.

"Improvements" means all buildings, structures and other improvements now or hereafter existing, erected or placed on the Land, together with any on-site improvements and off-site improvements in any way used or to be used in connection with the use, enjoyment, occupancy or operation of the Land.

"Insurance Proceeds" means the insurance claims under and the proceeds of any and all policies of insurance covering the Property or any part thereof, including all returned and unearned premiums with respect to any insurance relating to such Property, in each case whether now or hereafter existing or arising.

"Land" means the real property described in Exhibit "A" attached hereto and made a part hereof.

"<u>Laws</u>" means all federal, state and local laws, statutes, rules, ordinances, regulations, codes, licenses, authorizations, decisions, injunctions, interpretations, orders or decrees of any court or other Governmental Authority having jurisdiction as may be in effect from time to time.

"Leases" means all leases, license agreements and other occupancy or use agreements (whether oral or written), now or hereafter existing, which cover or relate to the Property or any part thereof, together with all options therefor, amendments thereto and renewals, modifications and guaranties thereof, including any cash or security deposited under the Leases to secure performance by the tenants of their obligations under the Leases, whether such cash or security is to be held until the expiration of the terms of the Leases or applied to one or more of the installments of rent coming due thereunder.

"Letter of Credit" means any letter of credit issued by Beneficiary for the account of Grantor or its nominee in connection with the development of the Land or the construction of the Improvements, together with any and all extensions, renewals or modifications thereof, substitutions therefor or replacements thereof.

"<u>Lien</u>" means any mortgage, deed of trust, pledge, security interest, assignment, judgment, lien or charge of any kind relating to the Property, including any conditional sale or other title retention agreement, any lease in the nature thereof, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code of any jurisdiction.

"Loan" means the loan from Beneficiary to Grantor, the repayment obligations in connection with which are evidenced by the Note.

"Loan Agreement" means the Loan Agreement of even date herewith between Grantor and Lender which sets forth, among other things, the terms and conditions upon which the proceeds of the Loan will be disbursed, as the same may from time to time be extended, amended, restated, supplemented or otherwise modified.

"Loan Documents" means this Deed of Trust, the Note, the Environmental Agreement, the Loan Agreement, any Swap Contract, any application or reimbursement agreement executed in connection with any Letter of Credit, and any and all other documents which Grantor or any other party or parties have executed and delivered, or may hereafter execute and deliver, to evidence, secure or guarantee the Obligations, or any part thereof, as the same may from time to time be extended, amended, restated, supplemented or otherwise modified.

"Note" means one or more Deed of Trust Notes of even date herewith in the aggregate principal face amount of ONE HUNDRED TEN MILLION AND NO/100 DOLLARS (\$110,000,000.00) made by Grantor to the order of Lender, as the same may from time to time be extended, amended, restated, supplemented or otherwise modified.

"Notice" means a notice, request, consent, demand or other communication given in accordance with the provisions of Section 9.8 of this Deed of Trust.

"Obligations" means all present and future debts, obligations and liabilities of Grantor to Beneficiary and/or Trustee arising pursuant to, and/or on account of the provisions of this Deed of Trust, the Note or any of the other Loan Documents, including the obligations: (a) to pay all principal, interest, late charges, prepayment premiums (if any) and other amounts due at any time under the Note; (b) to pay all Expenses, indemnification payments, fees and other amounts due at any time under this Deed of Trust or any of the other Loan Documents, together with interest thereon as herein or therein provided; (c) to pay and perform all obligations of Grantor under any Swap Contract; (d) to perform, observe and comply with all of the other terms, covenants and conditions, expressed or implied, which Grantor is required to perform, observe or comply with pursuant to this Deed of Trust or any of the other Loan Documents; and (e) to pay and perform all future advances and other obligations that Grantor or any successor in ownership of all or part of the Property may agree to pay and/or perform (whether as principal, surety or guarantor) for the benefit of Beneficiary, when a writing evidences the parties' agreement that the advance or obligation be secured by this Deed of Trust; excluding, however, the debts, obligations and liabilities of Grantor under the Environmental Agreement. This Deed of Trust does not secure the Environmental Agreement or any other Loan Document that is expressly stated to be unsecured.

"Permitted Encumbrances" means (a) any matters set forth in any policy of title insurance issued to Beneficiary and insuring Beneficiary's interest in the Property which are acceptable to Beneficiary as of the date hereof, (b) the Liens and interests of this Deed of Trust, and (c) any other Encumbrance that Beneficiary shall expressly approve in its sole and absolute discretion, as evidenced by a "marked-up" commitment for title insurance initialed on behalf of Beneficiary.

"Person" means an individual, a corporation, a partnership, a joint venture, a limited liability company, a trust, an unincorporated association, any Governmental Authority or any other entity.

"Personalty" means all personal property of any kind or nature whatsoever, whether tangible or intangible and whether now owned or hereafter

acquired, in which Grantor now has or hereafter acquires an interest and which is used in the construction of, or is placed upon, or is derived from or used in connection with the maintenance, use, occupancy or enjoyment of, the Property, including (a) the Accessories; (b) the Accounts; (c) all franchise, license, management or other agreements with respect to the operation of the Real Property or the business conducted therein (provided all of such agreements shall be subordinate to this Deed of Trust, and Beneficiary shall have no responsibility for the performance of Grantor's obligations thereunder) and all general intangibles (including payment intangibles, trademarks, trade names, goodwill, software and symbols) related to the Real Property or the operation thereof; (d) all sewer and water taps, appurtenant water stock or water rights, any Type 2 nonirrigation grandfathered water rights, contractual rights to water, allocations and agreements for utilities, bonds, letters of credit, permits, certificates, licenses, guaranties, warranties, causes of action, judgments, Claims, profits, security deposits, utility deposits, and all rebates or refunds of fees, Taxes, assessments, charges or deposits paid to any Governmental Authority related to the Real Property or the operation thereof; (e) all of Grantor's rights and interests under all Swap Contracts, including all rights to the payment of money from Beneficiary under any Swap Contract and all accounts, deposit accounts and general intangibles, including payment intangibles, described in any Swap Contract; (f) all insurance policies held by Grantor with respect to the Property or Grantor's operation thereof; and (g) all money, instruments and documents (whether tangible or electronic) arising from or by virtue of any transactions related to the Property, and all deposits and deposit accounts of Grantor with Beneficiary related to the Property, including any such deposit account from which Grantor may from time to time authorize Beneficiary to debit and/or credit payments due with respect to the Loan; together with all Additions to and Proceeds of all of the foregoing.

"Proceeds," when used with respect to any of the Property, means all proceeds of such Property, including all Insurance Proceeds and all other proceeds within the meaning of that term as defined in the Uniform Commercial Code of the State.

"Property" means the Real Property and the Personalty and all other rights, interests and benefits of every kind and character which Grantor now has or hereafter acquires in, to or for the benefit of the Real Property and/or the Personalty and all other property and rights used or useful in connection therewith, including all Leases, all Rents, all Condemnation Awards, all Proceeds, and all of Grantor's right, title and interest in and to all Design and Construction Contracts, all Contracts of Sale and all Refinancing Commitments.

"Property Assessments" means all Taxes, payments in lieu of taxes, water rents, sewer rents, assessments, condominium and owner's association assessments and charges, maintenance charges and other governmental or

municipal or public or private dues, charges and levies and any Liens (including federal tax liens) which are or may be levied, imposed or assessed upon the Property or any part thereof, or upon any Leases or any Rents, whether levied directly or indirectly or as excise taxes, as income taxes, or otherwise.

"Real Property" means the Land and Improvements, together with (a) all estates, title interests, title reversion rights, remainders, increases, issues, profits, rights of way or uses, additions, accretions, servitudes, strips, gaps, gores, liberties, privileges, water rights, water courses, alleys, passages, ways, vaults, licenses, tenements, franchises, hereditaments, appurtenances, easements, rights-of-way, rights of ingress or egress, parking rights, timber, crops, mineral interests and other rights, now or hereafter owned by Grantor and belonging or appertaining to the Land or Improvements; (b) all Claims whatsoever of Grantor with respect to the Land or Improvements, either in law or in equity, in possession or in expectancy; (c) all estate, right, title and interest of Grantor in and to all streets, roads and public places, opened or proposed, now or hereafter adjoining or appertaining to the Land or Improvements; and (d) all options to purchase the Land or Improvements, or any portion thereof or interest therein, and any greater estate in the Land or Improvements, and all Additions to and Proceeds of the foregoing.

"Refinancing Commitment" means any commitment from or other agreement with any Person providing for the financing of the Property, some or all of the proceeds of which are intended to be used for the repayment of all or a portion of the Loan.

"Rents" means all of the rents, royalties, issues, profits, revenues, earnings, income and other benefits of the Property, or arising from the use or enjoyment of the Property, including all such amounts paid under or arising from any of the Leases and all fees, charges, accounts or other payments for the use or occupancy of rooms or other public facilities within the Real Property.

"State" means the State of Arizona.

"Swap Contract" means any agreement, whether or not in writing, relating to any transaction that is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap or option, bond, note or bill option, interest rate option, forward foreign exchange transaction, interest cap, collar or floor transaction, currency swap, cross-currency rate swap, swap option, currency option or any other similar transaction (including any option to enter into the foregoing) or any combination of the foregoing, and, unless the context otherwise clearly requires, any form of master agreement published by the International Swaps and Derivatives Association, Inc., or any other master agreement, entered into between Beneficiary (or its affiliate) and Grantor (or its affiliate) in connection with the Loan, together with any related

schedules and confirmations, as amended, supplemented, superseded or replaced from time to time, relating to or governing any or all of the foregoing.

"Taxes" means, solely with respect to the Property, all taxes and assessments, whether general or special, ordinary or extraordinary, or foreseen or unforeseen, which at any time may be assessed, levied, confirmed or imposed by any Governmental Authority or any community facilities or other private district on Grantor or on any of its properties or assets or any part thereof or in respect of any of its franchises, businesses, income or profits.

"Transfer" means any direct or indirect sale, assignment, conveyance or transfer, including any Contract of Sale and any other contract or agreement to sell, assign, convey or transfer, whether made voluntarily or by operation of Law or otherwise, and whether made with or without consideration.

"Trustee" means the Initial Trustee or its successor in trust who may be acting under and pursuant to this Deed of Trust from time to time.

ARTICLE II GRANTING CLAUSES; CONDITION OF GRANT

Section 2.1 <u>Conveyances and Security Interests.</u>

In order to secure the prompt payment and performance of the Obligations, Grantor (a) hereby irrevocably and unconditionally grants, conveys, transfers and assigns to Trustee, in trust, for the benefit of Beneficiary, with power of sale and right of entry and possession, all estate, right, title and interest that Grantor now has or may later acquire in and to the Real Property; (b) grants to Beneficiary a security interest in the Personalty; (c) assigns to Beneficiary, and grants to Beneficiary a security interest in, all Condemnation Awards and all Insurance Proceeds; and (d) assigns to Beneficiary, and grants to Beneficiary a security interest in, all of Grantor's right, title and interest in, but not any of Grantor's obligations or liabilities under, all Design and Construction Documents, all Contracts of Sale and all Refinancing Commitments. All Persons who may have or acquire an interest in all or any part of the Property will be deemed to have notice of, and will be bound by, the terms of the Obligations and each other agreement or instrument made or entered into in connection with each of the Obligations. Such terms include any provisions in the Note, the Loan Agreement or any Swap Contract which provide that the interest rate on one or more of the Obligations may vary from time to time. The definition of "Obligations" includes future advances.

Section 2.2 Absolute Assignment of Leases and Rents.

In consideration of the making of the Loan by Beneficiary to Grantor and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Grantor absolutely and unconditionally assigns the Leases and Rents to Beneficiary. This assignment is, and is intended to be, an unconditional, absolute and present assignment from Grantor to

Beneficiary of all of Grantor's right, title and interest in and to the Leases and the Rents and not an assignment in the nature of a pledge of the Leases and Rents or the mere grant of a security interest therein. So long as no Event of Default shall exist, however, and so long as Grantor is not in default in the performance of any obligation, covenant or agreement contained in the Leases, Grantor shall have a license (which license shall terminate automatically and without notice upon the occurrence of an Event of Default or a default by Grantor under the Leases) to collect, but not prior to accrual, all Rents. Grantor agrees to collect and hold all Rents in trust for Beneficiary and to use the Rents for the payment of the cost of operating and maintaining the Property and for the payment of the other Obligations before using the Rents for any other purpose.

Section 2.3 Security Agreement, Fixture Filing and Financing Statement.

This Deed of Trust creates a security interest in the Personalty, and, to the extent the Personalty is not real property, this Deed of Trust constitutes a security agreement from Grantor to Beneficiary under the Uniform Commercial Code of the State. In addition to all of its other rights under this Deed of Trust and otherwise, Beneficiary shall have all of the rights of a secured party under the Uniform Commercial Code of the State, as in effect from time to time, or under the Uniform Commercial Code in force from time to time in any other state to the extent the same is applicable Law. This Deed of Trust shall be effective as a financing statement filed as a fixture filing with respect to all fixtures included within the Property and is to be filed for record in the real estate records of each county where any part of the Property (including such fixtures) is situated. This Deed of Trust shall also be effective as a financing statement with respect to any other Property as to which a security interest may be perfected by the filing of a financing statement and may be filed as such in any appropriate filing or recording office. The respective mailing addresses of Grantor and Beneficiary are set forth in the opening paragraph of this Deed of Trust. A carbon, photographic or other reproduction of this Deed of Trust or any other financing statement relating to this Deed of Trust shall be sufficient as a financing statement for any of the purposes referred to in this Section. Grantor hereby irrevocably authorizes Beneficiary at any time and from time to time to file any initial financing statements, amendments thereto and continuation statements as authorized by applicable Law, reasonably required by Beneficiary to establish or maintain the validity, perfection and priority of the security interests granted in this Deed of Trust.

Section 2.4 Release of Deed of Trust and Termination of Assignments and Financing Statements.

(a) Reconveyance; Termination Statements. If and when Grantor has paid and performed all of the Obligations, and no further advances are to be made under the Loan Agreement, Trustee, upon request by Beneficiary, will provide a reconveyance of the Property from the lien of this Deed of Trust and termination statements for filed financing statements, if any, to Grantor. Grantor shall be responsible for the recordation of such reconveyance and the payment of any recording and filing costs. Upon the recording of such reconveyance and the filing of such termination statements, the absolute assignments set forth in Section 2.2 shall automatically terminate and become null and void.

- (b) <u>Partial Releases; No Release in Default</u>. Partial releases of the lien of this Deed of Trust shall be made in accordance with the terms and provisions of the Loan Agreement, or in accordance with such other terms and conditions as may subsequently be agreed to by Beneficiary. No partial release shall be sought, requested or required if any Event of Default has occurred which has not been cured.
- (c) <u>Effect of Partial Release</u>. Beneficiary may, regardless of consideration, cause the release of any part of the Property from the lien of this Deed of Trust without in any manner affecting or impairing the lien or priority of this Deed of Trust as to the remainder of the Property.

ARTICLE III REPRESENTATIONS AND WARRANTIES

Grantor makes the following representations and warranties to Beneficiary:

Section 3.1 <u>Title to Real Property</u>.

Grantor (a) owns fee simple title in the Real Property, (b) owns all of the beneficial and equitable interest in and to the Real Property, and (c) is lawfully seized and possessed of the Real Property. Grantor has the right and authority to convey the Real Property and does hereby convey the Real Property. The Real Property is subject to no Encumbrances other than the Permitted Encumbrances.

Section 3.2 Title to Other Property.

Grantor has good title to the Personalty, and the Personalty is not subject to any Encumbrance other than the Permitted Encumbrances. None of the Leases, Rents, Design and Construction Documents, Contracts of Sale or Refinancing Commitments are subject to any Encumbrance other than the Permitted Encumbrances.

Section 3.3 Property Assessments.

The Real Property is assessed for purposes of Property Assessments as a separate and distinct parcel from any other property, such that the Real Property shall never become subject to the Lien of any Property Assessments levied or assessed against any property other than the Real Property.

Section 3.4 Independence of the Real Property.

No buildings or other improvements on property not covered by this Deed of Trust rely on the Real Property or any interest therein to fulfill any requirement of any Governmental Authority for the existence of such property, building or improvements; and none of the Real Property relies, or will rely, on any property not covered by this Deed of Trust or any interest therein to fulfill any requirement of any Governmental Authority. The Real Property has been

properly subdivided from all other property in accordance with the requirements of any applicable Governmental Authorities.

Section 3.5 Existing Improvements.

The existing Improvements, if any, were constructed, and are being used and maintained, in accordance with all applicable Laws, including zoning Laws.

Section 3.6 Leases and Tenants.

The Leases are valid and are in full force and effect, and Grantor is not in default under any of the terms thereof. Except as expressly permitted in the Loan Agreement, Grantor has not accepted any Rents in advance of the time the same became due under the Leases and has not forgiven, compromised or discounted any of the Rents. Grantor has title to and the right to assign the Leases and Rents to Beneficiary, and no other assignment of the Leases or Rents has been granted. To the best of Grantor's knowledge and belief, no tenant or tenants occupying, individually or in the aggregate, more than five percent (5%) of the net rentable area of the Improvements are in default under their Lease(s) or are the subject of any bankruptcy, insolvency or similar proceeding.

ARTICLE IV AFFIRMATIVE COVENANTS

Section 4.1 Obligations.

Grantor agrees to promptly pay and perform all of the Obligations, time being of the essence in each case.

Section 4.2 Property Assessments; Documentary Taxes.

Grantor (a) will promptly pay in full and discharge prior to delinquency all Property Assessments, and (b) will furnish to Beneficiary, upon demand, the receipted bills for such Property Assessments prior to the day upon which the same shall become delinquent. Property Assessments shall be considered delinquent as of the first day any interest or penalty commences to accrue thereon. Grantor will promptly pay all stamp, documentary, recordation, transfer and intangible taxes and all other taxes that may from time to time be required to be paid with respect to the Loan, the Note, this Deed of Trust or any of the other Loan Documents.

Section 4.3 Permitted Contests.

Grantor shall not be required to pay any of the Property Assessments, or to comply with any Law, so long as Grantor shall in good faith, and at its cost and expense, contest the amount or validity thereof, or take other appropriate action with respect thereto, in good faith and in an appropriate manner or by appropriate proceedings; provided that (a) such proceedings operate to prevent the collection of, or other realization upon, such Property Assessments or enforcement of the Law so contested, (b) there will be no sale, forfeiture or loss of the Property during the

contest, (c) neither Beneficiary nor Trustee is subjected to any Claim as a result of such contest, and (d) Grantor provides assurances satisfactory to Beneficiary (including the establishment of an appropriate reserve account with Beneficiary) of its ability to pay such Property Assessments or comply with such Law in the event Grantor is unsuccessful in its contest. Each such contest shall be promptly prosecuted to final conclusion or settlement, and Grantor shall indemnify and save Beneficiary and Trustee harmless for, from and against all Claims in connection therewith. Promptly after the settlement or conclusion of such contest or action, Grantor shall comply with such Law and/or pay and discharge the amounts which shall be levied, assessed or imposed or determined to be payable, together with all penalties, fines, interests, costs and expenses in connection therewith.

Section 4.4 Compliance with Laws.

Grantor will comply with and not violate, and cause to be complied with and not violated, all present and future Laws applicable to the Property and its use and operation.

Section 4.5 <u>Maintenance and Repair of the Property.</u>

Grantor, at Grantor's sole expense, will (a) keep and maintain Improvements and Accessories in good condition, working order and repair, reasonable wear and tear excepted, and (b) make all necessary or appropriate repairs and Additions to Improvements and Accessories, if any, so that each part of the Improvements and all of the Accessories shall at all times be in good condition and fit and proper for the respective purposes for which they were originally intended, erected, or installed.

Section 4.6 Additions to Security.

All right, title and interest of Grantor in and to all Improvements and Additions hereafter constructed or placed on the Property and in and to any Accessories hereafter acquired shall, without any further deed of trust, conveyance, assignment or other act by Grantor, become subject to the Lien of this Deed of Trust as fully and completely, and with the same effect, as though now owned by Grantor and specifically described in the granting clauses hereof. Grantor agrees, however, to execute and deliver to Trustee and/or Beneficiary such further documents as may be required by the terms of the Loan Agreement and the other Loan Documents.

Section 4.7 <u>Subrogation</u>.

To the extent permitted by Law, Beneficiary shall be subrogated, notwithstanding its release of record, to any Lien now or hereafter existing on the Property to the extent that such Lien is paid or discharged by Beneficiary whether or not from the proceeds of the Loan. This Section shall not be deemed or construed, however, to obligate Beneficiary to pay or discharge any Lien.

Section 4.8 Leases.

- (a) Except as expressly permitted in the Loan Agreement, Grantor shall not enter into any Lease with respect to all or any portion of the Property without the prior written consent of Beneficiary; provided, however, Beneficiary has approved Grantor's form lease agreement for the Property as of the date hereof, and any such consent is hereby deemed given by the Beneficiary.
- (b) Neither Trustee nor Beneficiary shall be obligated to perform or discharge any obligation of Grantor under any Lease. The assignment of Leases provided for in this Deed of Trust in no manner places on Beneficiary or Trustee any responsibility for (i) the control, care, management or repair of the Property, (ii) the carrying out of any of the terms and conditions of the Leases, (iii) any waste committed on the Property, or (iv) any dangerous or defective condition on the Property (whether known or unknown).
- (c) No approval of any Lease by Beneficiary shall be for any purpose other than to protect Beneficiary's security and to preserve Beneficiary's rights under the Loan Documents, and no such approval shall result in a waiver of a Default or Event of Default.

ARTICLE V NEGATIVE COVENANTS

Section 5.1 Encumbrances.

Grantor will not permit any of the Property to become subject to any Encumbrance other than the Permitted Encumbrances and the liens created by the Loan Documents. Within thirty (30) days after the filing of any mechanic's lien or other Lien or Encumbrance against the Property, Grantor will promptly discharge the same by payment or filing a bond or otherwise as permitted by Law. So long as Beneficiary's security has been protected by the filing of a bond or otherwise in a manner satisfactory to Beneficiary in its sole and absolute discretion, Grantor shall have the right to contest in good faith any Claim, Lien or Encumbrance, provided that Grantor does so diligently and without prejudice to Beneficiary or delay in completing construction of the Improvements. Grantor shall give Beneficiary Notice of any default under any Lien and Notice of any foreclosure or threat of foreclosure with respect to any of the Property.

Section 5.2 <u>Transfer of the Property.</u>

Grantor will not Transfer, or contract to Transfer, all or any part of the Property or any legal or beneficial interest therein (except without Beneficiary's consent for (i) certain Transfers of the Accessories expressly permitted in this Deed of Trust, (ii) Transfers that will result in the full payoff of the Loan within ten (10) days after such Transfer, (iii) any Transfer to an affiliate or wholly-owned subsidiary of Grantor, and (iv) Leases which shall be deemed permitted Transfers). The Transfer of more than fifty percent (50.0%) of the voting power in Grantor (whether in one or more transactions during the term of the Loan) shall be deemed to be a prohibited Transfer of the Property, except as otherwise permitted in the Loan Agreement.

Notwithstanding the foregoing, transfers of any interest in Grantor to any person or entity that is directly or indirectly controlled by, controlling or under common control of or with (a) Opus West Corporation, Opus Corporation, Opus, L.L.C., (b) the founder of Opus Corporation, his children, his grandchildren or other members of his family, or (c) the trustee of a trust or trust for the benefit of the founder of Opus Corporation, his children, his grandchildren or other members of his family, shall be permitted hereunder without Beneficiary's consent so long as Opus West Corporation remains fully obligated for the payment of the Obligations.

Section 5.3 Removal, Demolition or Alteration of Accessories and Improvements.

Except to the extent permitted by the following sentence, no Improvements or Accessories shall be removed, demolished or materially altered without the prior written consent of Beneficiary. Notwithstanding the foregoing, Grantor may remove and dispose of, free from the Lien of this Deed of Trust and without Beneficiary's consent, such Accessories as from time to time become worn out or obsolete, provided that, either (a) at the time of, or prior to, such removal, any such Accessories are replaced with other Accessories which are free from Liens other than Permitted Encumbrances and have a value at least equal to that of the replaced Accessories (and by such removal and replacement Grantor shall be deemed to have subjected such Accessories to the Lien of this Deed of Trust), or (b) so long as a prepayment may be made without the imposition of any premium pursuant to the Note, such Accessories are sold at fair market value for cash and the net cash proceeds received from such disposition are paid over promptly to Beneficiary to be applied to the prepayment of the principal of the Loan.

Section 5.4 <u>Additional Improvements</u>.

Unless in accordance with the Design and Construction Documents, Grantor will not construct any Improvements other than those presently on the Land and those described in the Loan Agreement without the prior written consent of Beneficiary. Grantor will complete and pay for, within a reasonable time, any Improvements which Grantor is permitted to construct on the Land. Grantor will construct and erect any permitted Improvements (a) strictly in accordance with all applicable Laws and any private restrictive covenants, (b) entirely on lots or parcels of the Land, (c) so as not to encroach upon any easement or right of way or upon the land of others, and (d) wholly within any building restriction and setback lines applicable to the Land.

Section 5.5 Restrictive Covenants, Zoning, etc.

Unless in accordance with the Design and Construction Documents, without the prior written consent of Beneficiary, Grantor will not initiate, join in, or consent to any change in, any restrictive covenant, easement, zoning ordinance, or other public or private restrictions limiting or defining the uses which may be made of the Property, or consent to or vote in favor of the inclusion of the Property in any community facilities district or any other improvement, assessment or similar district. Grantor (a) will promptly perform and observe, and use commercially reasonably efforts to cause to be performed and observed, all of the terms and conditions of all agreements affecting the Property, and (b) will do or use commercially reasonably efforts to cause to be done all things necessary to preserve intact and unimpaired any

and all easements, appurtenances and other interests and rights in favor of, or constituting any portion of, the Property.

ARTICLE VI EVENTS OF DEFAULT

The occurrence or happening, from time to time, of any one or more of the following shall constitute an Event of Default under this Deed of Trust:

Section 6.1 Payment Obligations.

Grantor fails to pay any of the Obligations when due, whether on the scheduled due date or upon acceleration, maturity or otherwise and such failure is not cured within five (5) days after receipt of Notice from Beneficiary to Grantor; <u>provided</u>, <u>however</u>, no Notice shall be required at maturity.

Section 6.2 Transfers.

Grantor Transfers, or contracts to Transfer, all or any part of the Property or any legal or beneficial interest therein (except for as otherwise permitted in the Loan Agreement and except for Transfers of the Accessories expressly permitted under this Deed of Trust). The Transfer of stock in Grantor (whether in one or more transactions during the term of the Loan) except as otherwise permitted in the Loan Agreement shall be deemed to be a prohibited Transfer of the Property constituting an Event of Default.

Section 6.3 Other Obligations.

Grantor fails to promptly perform or comply with any of the Obligations set forth in this Deed of Trust (other than those expressly described in other Sections of this <u>Article VI</u>), and such failure continues uncured for a period of fifteen (15) days after receipt of Notice from Beneficiary to Grantor.

Section 6.4 Event of Default Under Other Loan Documents.

An Event of Default (as defined therein) occurs under the Note or the Loan Agreement, or Borrower fails to promptly pay, perform, observe or comply with any obligation or agreement contained in any of the other Loan Documents (within any applicable grace or cure period).

Section 6.5 Change in Zoning or Public Restriction.

Any change in any zoning ordinance or regulation or any other public restriction is enacted, adopted or implemented that limits or defines the uses which may be made of the Property such that the present or intended use of the Property, as specified in the Loan Documents, would be in violation of such zoning ordinance or regulation or public restriction, as changed, and such violation will materially and adversely affect Borrower's ability to perform its obligations under the Loan Documents.

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 54 of 64

Section 6.6 <u>Default Under Major Leases.</u>

Grantor fails duly to perform its obligations under any Major Lease, and such failure is not cured within the grace period, if any, provided in the Major Lease.

Section 6.7 <u>Default Under Other Lien Documents.</u>

A default by Grantor occurs under any other mortgage, deed of trust or security agreement covering the Property, including any Permitted Encumbrances and is not cured within the grace period, if any, provided therein.

Section 6.8 <u>Execution</u>; Attachment.

Any execution or attachment is levied against any of the Property, and such execution or attachment is not set aside, discharged or stayed within thirty (30) days after the same is levied.

ARTICLE VII RIGHTS AND REMEDIES

Upon the happening of any Event of Default, Beneficiary, or Trustee at the direction of Beneficiary, shall have the right, in addition to any other rights or remedies available to Beneficiary under any of the Loan Documents or applicable Law, to exercise any one or more of the following rights, powers or remedies:

Section 7.1 Acceleration.

Beneficiary may accelerate all Obligations under the Loan Documents whereupon such Obligations shall become immediately due and payable, without notice of default, notice of acceleration or intention to accelerate, presentment or demand for payment, protest, notice of protest, notice of nonpayment or dishonor, or notices or demands of any kind or character (all of which are hereby waived by Grantor).

Section 7.2 <u>Foreclosure</u>; Power of Sale.

Trustee, if and as directed by Beneficiary, shall have all of the rights and may exercise all of the powers set forth in applicable Law of the State of California, including those powers set forth in Sections 2924 et seq. and Section 2938 of the California Civil Code or any successor provision of Law, Trustee may sell the Property in its entirety or in parcels, and by one or by several sales, as deemed appropriate by Trustee in its sole and absolute discretion. If Trustee chooses to have more than one foreclosure sale, Trustee may cause the foreclosure sales to be held simultaneously or successively, on the same day, or on such different days and at such different times as Trustee may elect. Trustee shall receive and apply the proceeds from the sale of the Property, or any portion thereof, in accordance with Section 2924k of the California Civil Code or any successor provision of Law. Before any foreclosure sale, Beneficiary or Trustee shall give such notice of default and election to sell as may be required by Law. After the lapse of such time as may then be required by Law following the recordation of such notice of default,

and notice of sale having been given as then required by Law, Trustee shall sell the property being sold at a public auction to be held at the time and place specified in the notice of sale. Neither Trustee nor Beneficiary shall have any obligation to make demand on Grantor before any foreclosure sale. From time to time in accordance with then-applicable Law, Trustee may, and in any event at Beneficiary's request shall, postpone any foreclosure sale by public announcement at the time and place noticed for that sale. At any foreclosure sale, Trustee shall sell to the highest bidder at public auction for cash in lawful money of the United States (or cash equivalents acceptable to Trustee to the extent permitted by applicable Law), payable at the time of sale. Trustee shall execute and deliver to the purchaser(s) a deed or deeds conveying the property being sold without any covenant or warranty whatsoever, expressed or implied. The recitals in any such deed of any matters of fact, including any facts bearing upon the regularity or validity of any foreclosure sale, shall be conclusive proof of their truthfulness. Any such deed shall be conclusive against all Persons as to the facts recited therein. Any Person, including Trustee or Beneficiary, may purchase at such sale, and any bid by Beneficiary may be, in whole or in part, in the form of cancellation of all or any part of the Obligations.

Section 7.3 <u>Judicial Action</u>.

Beneficiary and Trustee, if and as directed by Beneficiary, shall have the right to bring an action in any court of competent jurisdiction for foreclosure of this Deed of Trust and a deficiency judgment as provided by Law, or for specific enforcement of any of the covenants or agreements of this Deed of Trust.

Section 7.4 Collection of Rents.

Upon the occurrence of an Event of Default, the license granted to Grantor to collect the Rents shall be automatically and immediately revoked, without further notice to or demand upon Grantor. Beneficiary may, but shall not be obligated to, exercise any or all of the rights and remedies provided in Section 2938 of the California Civil Code and perform any or all obligations of the landlord under any or all of the Leases, and Beneficiary may, but shall not be obligated to, exercise and enforce any or all of Grantor's rights under the Leases. Without limitation to the generality of the foregoing, Beneficiary may notify the tenants under the Leases that all Rents are to be paid to Beneficiary, and following such notice all Rents shall be paid directly to Beneficiary and not to Grantor or any other Person other than as directed by Beneficiary, it being understood that a demand by Beneficiary on any tenant under the Leases for the payment of Rent shall be sufficient to warrant payment by such tenant of Rent to Beneficiary without the necessity of further consent by Grantor. Grantor hereby irrevocably authorizes and directs the tenants under the Lease to pay all Rents to Beneficiary instead of to Grantor, upon receipt of written notice from Beneficiary, without the necessity of any inquiry of Grantor and without the necessity of determining the existence or non-existence of an Event of Default. Grantor hereby appoints Beneficiary as Grantor's attorney-in-fact with full power of substitution, which appointment shall take effect upon the occurrence of an Event of Default and is coupled with an interest and is irrevocable prior to the full and final payment and performance of the Obligations, in Grantor's name or in Beneficiary's name: (a) to endorse all checks and other instruments received in payment of Rents and to deposit the same in any account selected by

Beneficiary; (b) to give receipts and releases in relation thereto; (c) to institute, prosecute and/or settle actions for the recovery of Rents; (d) to modify the terms of any Leases including terms relating to the Rents payable thereunder; (e) to cancel any Leases; (f) to enter into new Leases; and (g) to do all other acts and things with respect to the Leases and Rents which Beneficiary may deem necessary or desirable to protect the security for the Obligations. Any Rents received shall be applied first to pay all Expenses and next in reduction of the other Obligations. Grantor shall pay, on demand, to Beneficiary, the amount of any deficiency between (i) the Rents received by Beneficiary, and (ii) all Expenses incurred together with interest thereon as provided in the Loan Agreement and the other Loan Documents.

Section 7.5 <u>Taking Possession or Control of the Property.</u>

As a matter of right without regard to the adequacy of the security, and to the extent permitted by Law without notice to Grantor, Beneficiary shall be entitled, upon application to a court of competent jurisdiction, to the immediate appointment of a receiver for all or any part of the Property and the Rents, whether such receivership may be incidental to a proposed sale of the Property or otherwise, and Grantor hereby consents to the appointment of such a receiver and agrees that such receiver shall have all of the rights and powers granted to Beneficiary pursuant to Section 7.4. In addition, to the extent permitted by Law, and with or without the appointment of a receiver, or an application therefor, Beneficiary may (a) enter upon, and take possession of (and Grantor shall surrender actual possession of), the Property or any part thereof, without notice to Grantor and without bringing any legal action or proceeding, or, if necessary by force, legal proceedings, ejectment or otherwise, and (b) remove and exclude Grantor and its agents and employees therefrom.

Section 7.6 Management of the Property.

Upon obtaining possession of the Property or upon the appointment of a receiver as described in Section 7.5, Beneficiary, Trustee or the receiver, as the case may be, may, at its sole option, (a) make all necessary or proper repairs and Additions to or upon the Property, (b) operate, maintain, control, make secure and preserve the Property, and (c) complete the construction of any unfinished Improvements on the Property and, in connection therewith, continue any and all outstanding contracts for the erection and completion of such Improvements and make and enter into any further contracts which may be necessary, either in their or its own name or in the name of Grantor (the costs of completing such Improvements shall be Expenses secured by this Deed of Trust and shall accrue interest as provided in the Loan Agreement and the other Loan Documents). Beneficiary, Trustee or such receiver shall be under no liability for, or by reason of, any such taking of possession, entry, holding, removal, maintaining, operation or management, except for gross negligence or willful misconduct. The exercise of the remedies provided in this Section shall not cure or waive any Event of Default, and the enforcement of such remedies, once commenced, shall continue for so long as Beneficiary shall elect, notwithstanding the fact that the exercise of such remedies may have, for a time, cured the original Event of Default.

Section 7.7 <u>Uniform Commercial Code.</u>

Beneficiary may proceed under the Uniform Commercial Code as to all or any part of the Personalty, and in conjunction therewith may exercise all of the rights, remedies and powers of a secured creditor under the Uniform Commercial Code. Upon the occurrence of any Event of Default, Grantor shall assemble all of the Accessories and make the same available within the Improvements. Any notification required by the Uniform Commercial Code shall be deemed reasonably and properly given if sent in accordance with the Notice provisions of this Deed of Trust at least ten (10) days before any sale or other disposition of the Personalty. Disposition of the Personalty shall be deemed commercially reasonable if made pursuant to a public sale advertised at least twice in a newspaper of general circulation in the community where the Property is located. It shall be deemed commercially reasonable for the Trustee to dispose of the Personalty without giving any warranties as to the Personalty and specifically disclaiming all disposition warranties. Alternatively, Beneficiary may choose to dispose of some or all of the Property, in any combination consisting of both Personalty and Real Property, in one sale to be held in accordance with the Law and procedures applicable to real property, as permitted by Article 9 of the Uniform Commercial Code. Grantor agrees that such a sale of Personalty together with Real Property constitutes a commercially reasonable sale of the Personalty.

Section 7.8 Application of Proceeds.

Unless otherwise provided by applicable Law, all proceeds from the sale of the Property or any part thereof pursuant to the rights and remedies set forth in this <u>Article VII</u> and any other proceeds received by Beneficiary from the exercise of any of its other rights and remedies hereunder or under the other Loan Documents shall be applied first to pay all Expenses and next in reduction of the other Obligations, in such manner and order as Beneficiary may elect.

Section 7.9 Other Remedies.

Beneficiary shall have the right from time to time to protect, exercise and enforce any legal or equitable remedy against Grantor provided under the Loan Documents or by applicable Laws.

ARTICLE VIII TRUSTEE

Section 8.1 <u>Liability of Trustee.</u>

Trustee shall have no liability or responsibility for, and make no warranties in connection with, the validity or enforceability of any of the Loan Documents or the description, value or status of title to the Property. Trustee shall be protected in acting upon any notice, request, consent, demand, statement, note or other paper or document believed by Trustee to be genuine and to have been signed by the party or parties purporting to sign the same. Trustee shall not be liable for any error of judgment, nor for any act done or step taken or omitted, nor for any mistakes of law or fact, nor for anything which Trustee may do or refrain from doing in good

faith, nor generally shall Trustee have any accountability hereunder except for its willful misconduct or gross negligence. The powers and duties of Trustee hereunder may be exercised through such attorneys, agents or servants as Trustee may appoint, and Trustee shall have no liability or responsibility for any act, failure to act, negligence or willful misconduct of such attorney, agent or servant, so long as the selection was made with reasonable care. In addition, Trustee may consult with legal counsel selected by Trustee, and Trustee shall have no liability or responsibility by reason of any act or failure to act in accordance with the opinions of such counsel. Trustee may act hereunder and may sell or otherwise dispose of the Property or any part thereof as herein provided, although Trustee has been, may now be or may hereafter be, an attorney, officer, agent or employee of Beneficiary, in respect of any matter or business whatsoever. Trustee, however, shall have no obligation to sell all or any part of the Property following an Event of Default or to take any other action authorized to be taken by Trustee hereunder except upon the demand of Beneficiary.

Section 8.2 <u>Indemnification of Trustee.</u>

Grantor agrees to indemnify Trustee and to hold Trustee harmless for, from and against any and all Claims and Expenses directly or indirectly arising out of or resulting from any transaction, act, omission, event or circumstance in any way connected with the Property or the Loan, including but not limited to any Claim arising out of or resulting from any assertion or allegation that Trustee is liable for any act or omission of Grantor or any other Person in connection with the ownership, development, financing, operation or sale of the Property; provided, however, that Grantor shall not be obligated to indemnify Trustee with respect to any Claim arising solely from the gross negligence or willful misconduct of Trustee. The agreements and indemnifications contained in this Section shall apply to Claims arising both before and after the repayment of the Loan and shall survive the repayment of the Loan, any foreclosure or deed, conveyance or assignment in lieu thereof and any other action by Trustee to enforce the rights and remedies of Beneficiary or Trustee hereunder or under the other Loan Documents for a period of twenty-four (24) months thereafter.

Section 8.3 Substitution of Trustee; Multiple Trustees.

Beneficiary shall have, and is hereby granted with warranty of further assurances, the irrevocable power to appoint a new or replacement or substitute Trustee. Such power may be exercised at any time without notice, without cause and without specifying any reason therefor, by filing for record in the office where this Deed of Trust is recorded a Substitution of Trustee. The power of appointment of a successor Trustee may be exercised as often as and whenever Beneficiary may choose, and the exercise of the power of appointment, no matter how often, shall not be an exhaustion thereof. Upon the recordation of such Notice of Substitution of Trustee, the Trustee so appointed shall thereupon, without any further act or deed of conveyance, become fully vested with identically the same title and estate in and to the Property and with all the rights, powers, trusts and duties of its predecessor in the trust hereunder with like effect as if originally named as Trustee hereunder. Whenever in this Deed of Trust reference is made to Trustee, it shall be construed to mean each Person appointed as Trustee for the time being, whether original or successor in trust. All title, estate, rights, powers, trusts and duties granted to

Trustee shall be in each Person appointed as Trustee so that any action hereunder by any Person appointed as Trustee shall for all purposes be deemed to be, and as effective as, the action of all Trustees.

ARTICLE IX MISCELLANEOUS

Section 9.1 Rights, Powers and Remedies Cumulative.

Each right, power and remedy of Beneficiary or Trustee as provided for in this Deed of Trust, or in any of the other Loan Documents or now or hereafter existing by Law, shall be cumulative and concurrent and shall be in addition to every other right, power or remedy provided for in this Deed of Trust, or in any of the other Loan Documents or now or hereafter existing by Law, and the exercise or beginning of the exercise by Beneficiary or Trustee of any one or more of such rights, powers or remedies shall not preclude the simultaneous or later exercise by Beneficiary or Trustee of any or all such other rights, powers or remedies.

Section 9.2 No Waiver by Beneficiary or Trustee.

No course of dealing or conduct by or among Beneficiary, Trustee and Grantor shall be effective to amend, modify or change any provisions of this Deed of Trust or the other Loan Documents. No failure or delay by Beneficiary or Trustee to insist upon the strict performance of any term, covenant or agreement of this Deed of Trust or of any of the other Loan Documents, or to exercise any right, power or remedy consequent upon a breach thereof, shall constitute a waiver of any such term, covenant or agreement or of any such breach, or preclude Beneficiary or Trustee from exercising any such right, power or remedy at any later time or times. By accepting payment after the due date of any of the Obligations, neither Beneficiary nor Trustee shall be deemed to waive the right either to require prompt payment when due of all other Obligations, or to declare an Event of Default for failure to make prompt payment of any such other Obligations. Neither Grantor nor any other Person now or hereafter obligated for the payment of the whole or any part of the Obligations shall be relieved of such liability by reason of (a) the failure of Beneficiary to comply with any request of Grantor or of any other Person to take action to foreclose this Deed of Trust or otherwise enforce any of the provisions of this Deed of Trust, or (b) any agreement or stipulation between any subsequent owner or owners of the Property and Beneficiary, or (c) Beneficiary's extending the time of payment or modifying the terms of this Deed of Trust or any of the other Loan Documents without first having obtained the consent of Grantor or such other Person. Regardless of consideration, and without the necessity for any notice to or consent by the holder of any subordinate Lien on the Property, Beneficiary may release any Person at any time liable for any of the Obligations or any part of the security for the Obligations and may extend the time of payment or otherwise modify the terms of this Deed of Trust or any of the other Loan Documents without in any way impairing or affecting the Lien of this Deed of Trust or the priority of this Deed of Trust over any subordinate Lien. The holder of any subordinate Lien shall have no right to terminate any Lease regardless of whether or not such Lease is subordinate to this Deed of Trust. Beneficiary may resort to the

security or collateral described in this Deed of Trust or any of the other Loan Documents in such order and manner as Beneficiary may elect in its sole discretion.

Section 9.3 Waivers and Agreements Regarding Remedies.

To the fullest extent Grantor may do so, Grantor hereby:

- (a) agrees that it will not at any time plead, claim or take advantage of any Laws now or hereafter in force providing for any appraisement, valuation, stay, extension or redemption, and waives and releases all rights of redemption, valuation, appraisement, stay of execution, extension and notice of election to accelerate the Obligations;
- (b) waives all rights to a marshalling of the assets of Grantor, including the Property, or to a sale in the inverse order of alienation in the event of a foreclosure of the Property, and agrees not to assert any right under any Law pertaining to the marshalling of assets, the sale in inverse order of alienation, the exemption of homestead, the administration of estates of decedents, or other matters whatsoever to defeat, reduce or affect the right of Beneficiary under the terms of this Deed of Trust to a sale of the Property without any prior or different resort for collection, or the right of Beneficiary to the payment of the Obligations out of the proceeds of sale of the Property in preference to every other claimant whatsoever;
- (c) waives any right to bring or utilize any defense, counterclaim or setoff, other than one which denies the existence or sufficiency of the facts upon which any foreclosure action is grounded. If any defense, counterclaim or setoff, other than one permitted by the preceding clause, is timely raised in a foreclosure action, such defense, counterclaim or setoff shall be dismissed. If such defense, counterclaim or setoff is based on a Claim which could be tried in an action for money damages, such Claim may be brought in a separate action which shall not thereafter be consolidated with the foreclosure action. The bringing of such separate action for money damages shall not be deemed to afford any grounds for staying the foreclosure action; and
- (d) waives and relinquishes any and all rights and remedies which Grantor may have or be able to assert by reason of the provisions of any Laws pertaining to the rights and remedies of sureties.

Section 9.4 Successors and Assigns.

All of the grants, covenants, terms, provisions and conditions of this Deed of Trust shall run with the Land and shall apply to and bind the successors and assigns of Grantor (including any permitted subsequent owner of the Property), and inure to the benefit of Beneficiary, its successors and assigns and to the successors in trust of Trustee.

Section 9.5 No Warranty by Beneficiary or Trustee.

By inspecting the Property or by accepting or approving anything required to be observed, performed or fulfilled by Grantor or to be given to Beneficiary or Trustee pursuant to this Deed of Trust or any of the other Loan Documents, Beneficiary and Trustee shall not be deemed to have warranted or represented the condition, sufficiency, legality, effectiveness or legal effect of the same, and such acceptance or approval shall not constitute any warranty or representation with respect thereto by Beneficiary or Trustee.

Section 9.6 Amendments.

This Deed of Trust may not be modified or amended except by an agreement in writing, signed by the party against whom enforcement of the change is sought.

Section 9.7 Severability.

In the event any one or more of the provisions of this Deed of Trust or any of the other Loan Documents shall for any reason be held to be invalid, illegal or unenforceable, in whole or in part or in any other respect, or in the event any one or more of the provisions of the Loan Documents operates or would prospectively operate to invalidate this Deed of Trust or any of the other Loan Documents, then and in either of those events, at the option of Beneficiary, such provision or provisions only shall be deemed null and void and shall not affect the validity of the remaining Obligations, and the remaining provisions of the Loan Documents shall remain operative and in full force and effect and shall in no way be affected, prejudiced or disturbed thereby.

Section 9.8 Notices.

All Notices required or which any party desires to give hereunder or under any other Loan Document shall be in writing and, unless otherwise specifically provided in such other Loan Document, shall be deemed sufficiently given or furnished if delivered by personal delivery, by nationally recognized overnight courier service or by certified United States mail, postage prepaid, addressed to the party to whom directed at the applicable address specified in the Preamble to this Deed of Trust (unless changed by similar notice in writing given by the particular party whose address is to be changed) or by facsimile. Any Notice shall be deemed to have been given either at the time of personal delivery or, in the case of courier or mail, as of the date of first attempted delivery at the address and in the manner provided herein, or, in the case of facsimile, upon receipt; provided that service of a Notice required by any applicable statute shall be considered complete when the requirements of that statute are met. Notwithstanding the foregoing, no notice of change of address shall be effective except upon actual receipt. This Section shall not be construed in any way to affect or impair any waiver of notice or demand provided in this Deed of Trust or in any other Loan Document or to require giving of notice or demand to or upon any Person in any situation or for any reason.

Case 09-34356-hdh11 Claim 82-1 Part 5 Filed 11/09/09 Desc Attachments to Addendum Part 4 of 7 Page 62 of 64

Section 9.9 Joint and Several Liability.

If Grantor consists of two (2) or more Persons, the term "Grantor" shall also refer to all Persons signing this Deed of Trust as Grantor, and to each of them, and all of them are jointly and severally bound, obligated and liable hereunder. Trustee or Beneficiary may release, compromise, modify or settle with any of Grantor, in whole or in part, without impairing, lessening or affecting the obligations and liabilities of the others of Grantor hereunder or under the Note. Any of the acts mentioned aforesaid may be done without the approval or consent of, or notice to, any of Grantor.

Section 9.10 Rules of Construction.

The words "hereof," "herein," "hereunder," "hereto," and other words of similar import refer to this Deed of Trust in its entirety. The terms "agree" and "agreements" mean and include "covenant" and "covenants." The words "include" and "including" shall be interpreted as if followed by the words "without limitation." The headings of this Deed of Trust are for convenience of reference only and shall not be considered a part hereof and are not in any way intended to define, limit or enlarge the terms hereof. All references (a) made in the neuter, masculine or feminine gender shall be deemed to have been made in all such genders, (b) made in the singular or plural number shall be deemed to have been made, respectively, in the plural or singular number as well, (c) to the Loan Documents are to the same as extended, amended, restated, supplemented or otherwise modified from time to time unless expressly indicated otherwise, (d) to the Land, Improvements, Personalty, Real Property or Property shall mean all or any portion of each of the foregoing, respectively, and (e) to Articles or Sections are to the respective Articles or Sections contained in this Deed of Trust unless expressly indicated otherwise. Any term used or defined in the Uniform Commercial Code of the State, as in effect from time to time, which is not defined in this Deed of Trust shall have the meaning ascribed to that term in the Uniform Commercial Code of the State. If a term is defined in Article 9 of the Uniform Commercial Code of the State differently than in another Article of the Uniform Commercial Code of the State, the term shall have the meaning specified in Article 9.

Section 9.11 Governing Law.

This Deed of Trust shall be construed, governed and enforced in accordance with the Laws in effect from time to time in the State, except to the extent that the state in which the Land is located may require that its laws be applied to the creation and priority of liens, to the perfection of security interests and to any foreclosure, trustee's sale, appointment of receiver or other remedy with respect to the Land. Any procedures provided herein for such remedies shall be modified by and replaced with, where inconsistent with or required by, any procedures or requirements of the laws of the state in which the Land is located.

Section 9.12 Entire Agreement.

The Loan Documents constitute the entire understanding and agreement between Grantor and Beneficiary with respect to the transactions arising in connection with the Loan, and

supersede all prior written or oral understandings and agreements between Grantor and Beneficiary with respect to the matters addressed in the Loan Documents. In particular, and without limitation, the terms of any commitment by Beneficiary to make the Loan are merged into the Loan Documents. Except as incorporated in writing into the Loan Documents, there are no representations, understandings, stipulations, agreements or promises, oral or written, with respect to the matters addressed in the Loan Documents.

Section 9.13 Deed of Trust.

This Deed of Trust secures a loan made by Lender to Grantor, some or all of the proceeds of which may be used for the purpose of constructing improvements on the Real Property. For purposes of Section 3097(j) of the California Civil Code, (a) the name and address of Lender and the name and address of Grantor, the owner of the Real Property, are set forth in the introductory paragraph of this Deed of Trust, and (b) a legal description of the Real Property is set forth in Exhibit "A" attached hereto, and the street address of the Real Property is Chino Hills Community Park, 14069 Peyton Drive, Chino Hills, California 90709.

IN WITNESS WHEREOF, Grantor has caused this Deed of Trust to be executed as of the day and year first written above.

GRANTOR:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

Ву:	Cheerles Vocel
Name:	Charles Vogel
Title:	Vice President
	TIOC I ICOIDCIN

STATE OF Arizona) ss.

County of Manicopan)

Subscribed and sworn to (or affirmed) before me on this 17 day of May, 2007 by ______, personally known to me or proved to me on the basis of satisfactory evidence to be the person(s) who appeared before me.

Seal

Signature



JENNIFER WILLIAMS Notary Public - Arizona Maricopa County Expires 06/01/10

EXHIBIT "A"

LEGAL DESCRIPTION

All that real property situate in San Bernardino County, California, more particularly described as follows:

Exhibit "A"

PARCEL A:

THE NORTH 30 ACRES OF THAT PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1 PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

BEGINNING AT A POINT IN THE WEST LINE OF SECTION 16, SOUTH 00 40' 39" 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION; THENCE SOUTH ALONG THE WEST LINE OF SAID SECTION 16 TO A POINT 330.3 FEET SOUTH OF THE NORTHWEST CORNER OF THE SOUTHWEST ONE-QUARTER OF SAID SECTIONS; THENCE EAST PARALLEL TO THE NORTH LINE OF SECTION 16 TO THE WEST LINE OF LAND CONVEYED TO KATE FOWLER MERLE-SMITH, FORMERLY KATE FOWLER, BY THE CHINO LAND AND WATER COMPANY BY DEED RECORDED APRIL 6,1925 IN BOOK 895 OF DEEDS, PAGE 87; THENCE NORTHERLY ALONG THE WEST LINE OF THE ABOVE DESCRIBED LAND TO A POINT 660.7 FEET SOUTH OF THE NORTH LINE OF SECTION 16; THENCE WEST PARALLEL TO THE NORTH OF SECTION 16 TO THE POINT OF BEGINNING.

EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT A POINT ON THE WEST LINE OF THE ABOVE DESCRIBED LAND, 280 FEET NORTH FROM THE SOUTHWEST CORNER OF SAID NORTH 30 ACRES; THENCE EAST 38 FEET; THENCE NORTH 17 FEET, 4 INCHES; THENCE WEST 38 FEET TO THE WEST LINE OF SAID SECTION 16, AND THENCE SOUTH ALONG SAID LINE 17 FEET, 4 INCHES TO THE POINT OF BEGINNING

ALSO EXCEPTING THEREFROM THE NORTH 30 FEET AND THE WEST 30 FEET OF THE ABOVE DESCRIBED PROPERTY DEEDED FOR COUNTY ROADS.

ALSO EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT THE INTERSECTION OF THE EAST LINE OF THAT CERTAIN PARCEL OF LAND, AS DESCRIBED BY DEED TO LELAND C.

LAUNDER, RECORDED JANUARY 21, 1957 IN BOOK 4135 OF OFFICIAL RECORDS, PAGE 105, WITH THE SOUTHERLY LINE OF PEYTON DRIVE, 60 FEET WIDE, AS DESCRIBED IN DEED RECORDED IN BOOK 359 OF OFFICIAL RECORDS, PAGE 205; THENCE ALONG SAID SOUTHERLY LINE, BEING A NON-TANGENT CURVE CONCAVE NORTHERLY WITH A RADIUS OF 377 FEET, FROM A TANGENT BEARING SOUTH 76 41' 27" WEST, THROUGH AN ANGLE OF 12 51' 28" A DISTANCE OF 84.60 FEET;

THENCE NORTH 89 32' 55" EAST 83.90 FEET TO SAID EAST LINE, DISTANT ALONG SAID EAST LINE SOUTH 0 28' 26" EAST 9.45 FEET FROM THE POINT OF BEGINNING; THENCE NORTH 0 28' 26" WEST 9.45 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THAT PORTION AS CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM THAT PORTION THEREOF LYING EASTERLY, NORTHERLY AND WESTERLY OF THE FOLLOWING DESCRIBED LINE:

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 2 of 32 COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED LAND; THENCE ALONG THE SOUTHERLY LINE OF SAID LAND SOUTH 89 24' 44" WEST 21.61 FEET TO TRUE POINT OF BEGINNING, BEING THE BEGINNING OF A NON TANGENT CURVE CONCAVE SOUTHEASTERLY HAVING A RADIUS OF 388.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS SOUTH 89 14' 57" WEST; THENCE NORTHEASTERLY 93.49 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 13 48' 22"; THENCE NORTH 13 03' 19" EAST 10.00 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 45.00 FEET; THENCE NORTHWESTERLY 60.79 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 77 21' 57" TO THE SOUTH LINE OF THE LAND AS DESCRIBED IN THE QUITCLAIM DEED TO THE CITY OF CHINO HILLS RECORDED AUGUST 11, 2003 AS DOCUMENT NO.2003-0591428, OFFICIAL RECORDS, COUNTY OF SAN BERNARDINO, BEING A POINT ON A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET; THENCE NORTHWESTERLY 660.12 FEET ALONG SAID CURVE AND SOUTHERLY LINE THROUGH A CENTRAL ANGLE OF 26 15' 56"; THENCE SOUTH 89 25' 26" WEST 293.36 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1740.00 FEET; THENCE WESTERLY 153.03 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 10 03' 09" TO THE BEGINNING OF A COMPOUND CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 40.00 FEET: THENCE SOUTHWESTERLY 63.89 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 91 31' 07" TO THE BEGINNING OF A REVERSE CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1740.00 FEET; THENCE SOUTHERLY 146.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 05 43' 08" TO THE NORTHERLY LINE OF THE PROPERTY CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OF OFFICIAL RECORDS; THENCE ALONG SAID NORTHERLY LINE SOUTH 89 25' 26" WEST 15.29 FEET TO THE EAST LINE OF THE LAND DESCRIBED IN THE DECLARATION OF ROAD RIGHT OF WAY, RECORDED MARCH 15, 1988 AS DOCUMENT NO. 88-077222 OFFICIAL RECORDS, SAID POINT BEING ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE AND ALONG SAID EAST LINE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE SOUTH 00 37' 46" EAST 590.81 FEET TO THE SOUTH LINE OF SAID NORTH 30 ACRES.

PARCEL B:

THAT PORTION OF THE NORTH 30 ACRES DESCRIBED IN THAT CERTAIN DEED TO THE COUNTY OF SAN BERNARDINO AS PER DEED RECORDED ON DECEMBER 16, 1983, AS INSTRUMENT NO. 83-296886, OFFICIAL RECORDS, AND BEING A PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1, PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

COMMENCING AT THE CENTERLINE INTERSECTION OF GRAND AVENUE AND PEYTON DRIVE, AS SHOWN ON MAP OF TRACT NO. 12581-12, AS PER MAP RECORDED IN BOOK 188 OF MAPS, PAGES 98 THROUGH 100, INCLUSIVE, RECORDS OF SAID COUNTY, SAID INTERSECTION BEING SOUTH 00 40' 39" EAST, A DISTANCE OF 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION 16 AS SHOWN ON THAT RECORD OF SURVEY MAP RECORDED IN BOOK 54 OF RECORDS OF SURVEY, PAGES 58 AND 59, RECORDS OF SAID COUNTY; THENCE SOUTH 00 37' 46" EAST, ALONG THE CENTERLINE OF SAID PEYTON DRIVE, A DISTANCE OF 987.02 FEET TO THE SOUTH LINE OF AFORESAID NORTH 30 ACRES; THENCE NORTH 89 24' 51" EAST ALONG SAID SOUTH LINE, A DISTANCE OF 60.00 FEET TO A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SAID POINT BEING THE TRUE POINT OF BEGINNING;

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to

Addendum Part 5 of 7 Page 3 of 32 THENCE CONTINUING ALONG SAID SOUTH LINE NORTH 89 24 51" EAST, A DISTANCE OF 1263.78 FEET TO THE SOUTHEAST CORNER OF SAID NORTH 30 ACRES; THENCE ALONG THE EAST LINE OF SAID NORTH 30 ACRES, NORTH 00 35' 16" WEST, DISTANCE OF 624.82 FEET; THENCE AT RIGHT ANGLES TO THE EAST LINE OF SAID NORTH 30 ACRES, SOUTH 89 24' 44" WEST, A DISTANCE OF 66.00 FEET TO A LINE THAT IS PARALLEL WITH AND 66.00 FEET WEST OF THE EAST LINE OF SAID NORTH 30 ACRES; THENCE ALONG SAID PARALLEL LINE, NORTH 00 35' 16" WEST, A DISTANCE OF 20.00 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHEAST AND HAVING A RADIUS OF 166.00 FEET; THENCE NORTHEASTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 28 06' 13", A DISTANCE OF 81.42 FEET; THENCE TANGENT TO SAID LAST CURVE NORTH 27 30' 57" EAST A DISTANCE OF 27.82 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 27.00 FEET; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 92 25' 57" A DISTANCE OF 43.55 FEET TO A POINT OF COMPOUND CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 1440.00 FEET, SAID LAST CURVE BEING THE SOUTHERLY LINE OF GRAND AVENUE; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 25 40' 34", A DISTANCE OF 645.31 FEET TO A POINT OF TANGENCY WITH A LINE THAT IS PARALLEL WITH AND 60.00 FEET SOUTH OF THE CENTERLINE OF PEYTON DRIVE AND THE NORTH LINE OF SAID NORTH 30 ACRES AS SHOWN ON AFORESAID RECORD OF SURVEY MAP; THENCE ALONG SAID PARALLEL LINE AND THE SOUTHERLY LINE OF GRAND AVENUE AS DESCRIBED IN THAT DECLARATION OF ROAD RIGHT-OF-WAY BY THE COUNTY OF SAN BERNARDINO AS RESOLUTION NO. 88-88 AND RECORDED AS INSTRUMENT NO. 88-077222, OFFICIAL RECORDS OF SAID COUNTY, SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET; THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 155.00 FEET; THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 90.0 FEET;

THENCE SOUTH 00 37' 46" WEST, A DISTANCE OF 12.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 92.00 FEET;

THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 91.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 380.00 FEET TO THE BEGINNING OF A NON-TANGENT CURVE CONCAVE TO THE WEST HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING OF SAID CURVE THROUGH SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06", A DISTANCE OF 78.23 FEET; THENCE TANGENT TO SAID LAST

CURVE AND ALONG A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SOUTH 00 37' 46" EAST, A DISTANCE OF 590.81 FEET TO THE TRUE POINT OF BEGINNING.

EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE SOUTHWEST CORNER OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG THE WESTERLY LINE NORTH 00 37' 76" WEST 221.18 FEET TO THE TRUE POINT OF BEGINNING; THENCE NORTH 89 22' 13" EAST 1.99 FEET; THENCE NORTH 00 48' 09" EAST 108.60 FEET; THENCE NORTH 45 24' 04" EAST 35.04 FEET; THENCE NORTH 01 28' 29" EAST 221.44 FEET TO THE NORTHERLY LINE OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 25' 26" WEST 15.29 FEET TO THE WESTERLY LINE OF SAID PARCEL BEING A POINT ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE CONTINUING ALONG SAID WESTERLY LINE SOUTH 00 37' 46" EAST 369.63 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

BEGINNING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE OF SAID PARCEL NORTH 00 35' 16" WEST 26.89 FEET; THENCE SOUTH 41 26' 12" WEST 30.53 FEET; THENCE SOUTH 00 39' 11" EAST 4.21 FEET TO THE SOUTHERLY LINE OF SAID PARCEL; THENCE ALONG SAID SOUTHERLY LINE NORTH 89 24' 51" EAST 20.43 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 4 of 32 COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE NORTH 00 35' 16" WEST 453.32 FEET TO THE TRUE POINT OF BEGINNING, BEING THE BEGINNING OF A TANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 312.00 FEET; THENCE NORTHWESTERLY 77.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 16' 30" TO THE BEGINNING OF A REVERSE CURVE CONCAVE NORTHEASTERLY HAVING A RADIUS OF 388.00 FEET; THENCE NORTHERLY 95.57 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 06' 44" TO A POINT ON THE NORTHERLY LINE OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 24' 44" EAST 21.61 FEET TO THE EASTERLY LINE OF SAID PARCEL; THENCE ALONG SAID EASTERLY LINE SOUTH 00 35' 21" EAST 171.50 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE EASTERLY TERMINUS OF THE COURSE SHOWN AS "SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET" IN THE NORTHERLY LINE OF THE ABOVE PARCEL, SAID POINT BEING THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1440.00; THENCE SOUTHEASTERLY 44.43 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 01 46' 04" TO THE TRUE POINT OF BEGINNING; THENCE NONMTANGENT SOUTH 44 24' 14" EAST 44.95 FEET; THENCE SOUTH 86 16' 02" EAST 64.80 FEET; THENCE NORTH 48 08' 12" EAST 41.99 FEET TO THE NORTHERLY LINE OF SAID PARCEL, BEING A POINT ON A NONTANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS NORTH 06 16' 21" EAST; THENCE NORTHWESTERLY 127.70 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 05 04' 51" TO THE TRUE POINT OF BEGINNING.

PARCEL C:

A NON-EXCLUSIVE EASEMENT FOR THE PURPOSE OF INSTALLING, MAINTAINING, OPERATING, REPAIRING, AND REPLACING A PYLON SIGN AS DESCRIBED IN THAT CERTAIN PYLON SIGN EASEMENT AND MAINTENANCE AGREEMENT RECORDED JUNE 2, 2006 AS INSTRUMENT NO. 06-380081 OF OFFICIAL RECORDS.

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 5 of 32

ATTACHMENT 4

ARCHITECT'S CONSENT AND CERTIFICATE

(The Shoppes at Chino Hills, CA - Construction Loan)

The undersigned ("Architect") has entered into certain agreements (collectively, the "Agreement") with OPUS WEST CONSTRUCTION CORPORATION, a Minnesota corporation ("Opus") on behalf of SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Owner"), for architectural services in connection with the construction of certain improvements (the "Improvements") to be located on the land (the "Land") more particularly described on Exhibit "A" attached hereto and made a part hereof. Pursuant to the Agreement, Architect has prepared and will continue to prepare certain plans and specifications (the "Plans") for the Improvements. Architect acknowledges that Owner will assign to BANK OF AMERICA, N.A., a national banking association as Administrative Agent for itself and other Lenders (collectively, "Lender"), all of Owner's rights (but not Owner's obligations) in and to the Agreement and the Plans as security for the obligations of Owner under a Construction Loan Agreement (the "Loan Agreement") to be entered into between Owner and Lender. Architect consents to and agrees to be bound by that assignment. Architect further certifies to and agrees with Lender as follows:

- 1. The Agreement is in full force and effect, and neither Owner nor Architect is in default under the Agreement.
- 2. The Plans have been prepared and will continue to be prepared in accordance with customary professional standards of architectural practice for projects similar to the Improvements. When finalized, the Plans will be complete and adequate for the construction of the Improvements. To the best of Architect's knowledge, the Improvements, if constructed in accordance with the Plans, will comply with all applicable laws, codes, rules and regulations (including, but not limited to, those relating to access and facilities for handicapped persons), and the structural design of the Improvements and the materials specified by the Plans are and will be adequate and appropriate for the Improvements.
- 3. If Owner defaults in making any payment or in performing any other obligation under the Agreement, or if the Agreement is terminated for any reason, Architect will give Lender written notice of the default or termination. Prior to exercising any remedy available to Architect under the Agreement as a result of a default, Architect will afford to Lender a period of thirty (30) days within which to cure the default (it being acknowledged by Architect that Lender shall have no obligation to cure any default by Owner). If the Agreement is terminated, Architect, at the request of Lender, will enter into a new agreement with Lender upon substantially the same terms and conditions as set forth in the Agreement. Any notice of default or termination will be delivered by personal delivery or by a nationally recognized overnight courier service or will be mailed by certified mail, return receipt requested, to the following address:

Bank of America, N.A.

Commercial Real Estate Banking Group, AZ1-200-22-17
201 East Washington Street, 22nd Floor
Phoenix, Arizona 85004-2343

Attention: N. Alonzo

- 4. In the event that Lender or any other party ("Owner's Successor") shall acquire title to the Property through foreclosure or deed in lieu of foreclosure, Architect, if requested by Owner's Successor, will continue to perform its obligations under the Agreement provided that any past due amounts owed to Architect under the Agreement, other than any such amounts for which Lender has previously made an advance under the Loan Agreement, are paid to Architect promptly following the request by Owner's Successor and provided that Architect is thereafter compensated for its services as provided in the Agreement. If Architect is not requested by Owner's Successor to continue to perform its obligations under the Agreement, Architect will discontinue such performance and will not assert any claim against Lender or Owner's Successor for any amounts owed to Architect under the Agreement. Whether or not Architect is requested to continue to perform its obligations under the Agreement, Owner's Successor shall be entitled to use the Plans for the completion of the Improvements without further cost to Owner's Successor.
- 5. Architect acknowledges that Lender will rely on this Consent and Certificate in making the loan to Owner that is contemplated by the Loan Agreement.

Architect has executed and delivered this Consent and Certificate on the 16th day of May, 2007.

ARCHITECT:

OPUS ARCHITECT'S & ENGINEERS, INC., a Minnesota corporation

Name: Robert Carli

Title: Vice President

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 8 of 32

ATTACHMENT 5

ENGINEER'S CONSENT AND CERTIFICATE

(The Shoppes at Chino Hills, CA - Construction Loan)

The undersigned ("Engineer") has entered into certain agreements (collectively, the "Agreement") with OPUS WEST CONSTRUCTION CORPORATION, a Minnesota corporation ("Opus") on behalf of SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Owner"), for engineering services in connection with the construction of certain improvements (the "Improvements") on the land (the "Land") more particularly described on Exhibit "A" attached hereto and made a part hereof. Pursuant to the Agreement, Engineer has prepared and will continue to prepare certain plans and specifications (the "Plans") for the Improvements. Engineer acknowledges that Owner will assign to BANK OF AMERICA, N.A., a national banking association as Administrative Agent for itself and other Lenders (collectively, "Lender"), all of Owner's rights (but not Owner's obligations) in and to the Agreement and the Plans as security for the obligations of Owner under a Construction Loan Agreement (the "Loan Agreement") to be entered into between Owner and Lender. Engineer consents to and agrees to be bound by that assignment. Engineer further certifies to and agrees with Lender as follows:

- 1. The Agreement is in full force and effect, and neither Owner nor Engineer is in default under the Agreement.
- 2. The Plans have been prepared and will continue to be prepared in accordance with customary professional standards of engineering practice for projects similar to the Improvements. When finalized, the Plans will be complete and adequate for the construction of the Improvements. To the best of Engineer's knowledge, the Improvements, if constructed in accordance with the Plans, will comply with all applicable laws, codes, rules and regulations.
- 3. The Land is zoned Out of the Improvements as depicted in the Plans. To the best of Engineer's professional knowledge, Engineer has examined all relevant laws, regulations and ordinances and has determined that the following restrictions and requirements are applicable to the Property:

Minimum Lot Area:	5.000 SF
Height Limitation:	50'Routhing
Maximum Floor Area Ratio (or other type of bulk restriction):	No Max Floor Robin
Limitation on Number of Dwelling Units, if any:	NA
Front Yard Requirements:	0'
Side and Rear Yard Requirements:	D' .
Parking Requirements:	3.25 space / 1000 SF-7014
Other:	NA

4. If Owner defaults in making any payment or in performing any other obligation under the Agreement, or if the Agreement is terminated for any reason, Engineer will give Lender written notice of the default or termination. Prior to exercising any remedy available to Engineer under the Agreement as a result of a default, Engineer will afford to Lender a period of thirty (30) days within which to cure the default (it being acknowledged by Engineer that Lender shall have no obligation to cure any default by Owner). If the Agreement is terminated, Engineer, at the request of Lender, will enter into a new agreement with Lender upon substantially the same terms and conditions as set forth in the Agreement. Any notice of default or termination will be delivered by personal delivery or by a nationally recognized overnight courier service or will be mailed by certified mail, return receipt requested, to the following address:

Bank of America, N.A.

Commercial Real Estate Banking Group, AZ1-200-22-17
201 East Washington Street, 22nd Floor
Phoenix, Arizona 85004-2343
Attention: N. Alonzo

- 5. In the event that Lender or any other party ("Owner's Successor") shall acquire title to the Property through foreclosure or deed in lieu of foreclosure, Engineer, if requested by Owner's Successor, will continue to perform its obligations under the Agreement provided that any past due amounts owed to Engineer under the Agreement, other than any such amounts for which Lender has previously made an advance under the Loan Agreement, are paid to Engineer promptly following the request by Owner's Successor and provided that Engineer is thereafter compensated for its services as provided in the Agreement. If Engineer is not requested by Owner's Successor to continue to perform its obligations under the Agreement, Engineer will discontinue such performance and will not assert any claim against Lender or Owner's Successor for any amounts owed to Engineer under the Agreement. Whether or not Engineer is requested to continue to perform its obligations under the Agreement, Owner's Successor shall be entitled to use the Plans for the completion of the Improvements without further cost to Owner's Successor.
- 6. Engineer acknowledges that Lender will rely on this Consent and Certificate in making the loan to Owner that is contemplated by the Loan Agreement.

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 11 of 32

Engineer has executed and delivered this Consent and Certificate on the 17 day of MAY, 2007.

ENGINEER

By: TRYFON TRYFONOPOULD

Name:

Title: SEMOR PROJECT MAMBLER

ATTACHMENT 6

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 13 of 32 BANK OF AMERICA

201 East Washington Street, 22nd Floor AZ1-200-22-17 Phoenix, Arizona 85004-2343

March 21, 2008

Opus West Corporation Opus West LP O.W. Commercial, Inc. PC 101, Inc. Shoppes at Chino Hills, Inc. Broadstone Cypress, L.P. Hill Country Apartments, L.P. Arch Road Limited Partnership Broadstone Walker Commons, L.P. Irvine Center Partners III, L.L.C. Hill Country Galleria, L.P. Broadstone Galleria, L.P. 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attention: Senior Vice President, Real Estate Finance and Sales

Re: BANK OF AMERICA, N.A., a national banking association, on behalf of itself and as agent for other Lenders ("Lender"), loans to OPUS WEST CORPORATION, a Minnesota corporation ("OPUS"), and various affiliates

Ladies/Gentlemen:

Lender has entered into and hereby modifies the following loan agreements, guaranty agreements, promissory notes, and certificates of representations, warranties, and covenants with OPUS and various affiliates for assorted financings (collectively, the "Loans"):

- (i) a Loan Agreement by and between Lender and OPUS, as borrower, dated July 28, 2003 (as amended from time to time, the "Opus RLC Agreement");
- (ii) a Construction Loan Agreement by and between Lender and OPUS, as borrower, dated March 13, 2007 (as amended from time to time, the "Commons at Chino Hills Construction Agreement");
- (iii) a Construction Loan Agreement by and between Lender and OPUS and O.W. COMMERCIAL, INC., a Delaware corporation, jointly and severally,

as borrower, dated August 31, 2007 (as amended from time to time, the "Camarillo Ranch Agreement");

- (iv) a Construction Loan Agreement by and between Lender and OPUS WEST LP, a Delaware limited partnership, as borrower, dated November 7, 2005 (as amended from time to time, the "Highland Village Condos Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated November 7, 2005 (as amended from time to time, the "Highland Village Condos Guaranty");
- (v) a Construction Loan Agreement by and between Lender and OPUS WEST LP, a Delaware limited partnership, as borrower, dated November 10, 2006 (as amended from time to time, the "Fort Bend Crossing Agreement") and OPUS as guarantor pursuant to a Guaranty Agreement dated November, 2006 (as amended from time to time, the "Fort Bend Crossing Guaranty");
- (vi) a Construction Loan Agreement by and between Lender and SHOPPES AT CHINO HILLS, INC., a Minnesota corporation, as borrower, dated May 16, 2007 (as amended from time to time, the "Shoppes at Chino Hills Agreement") and OPUS as guarantor pursuant to a Guaranty Agreement dated May 16, 2007 (as amended from time to time, the "Shoppes at Chino Hills Guaranty");
- (vii) a Construction Loan Agreement by and between Lender and O.W. COMMERCIAL, INC., a Delaware corporation, as borrower, dated May 29, 2007 (as amended from time to time, the "Freemont Tech Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated May 29, 2007 (as amended from time to time, the "Freemont Tech Guaranty"):
- (viii) a Construction Loan Agreement by and between Lender and PC 101, INC., a Delaware corporation, as borrower, dated May 31, 2005 (as amended from time to time, the "Pima Center Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated May 31, 2005 (as amended from time to time, the "Pima Center Guaranty");
- (ix) a Construction Loan Agreement by and between Lender and PC 101, INC., a Delaware corporation, as borrower, dated December 6, 2007 (as amended from time to time, the "Pima Center III-C Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated December 6, 2007 (as amended from time to time, the "Pima Center III-C Guaranty");
- (x) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and BROADSTONE CYPRESS, L.P., a Delaware limited partnership, as borrower, dated August 31, 2007 (as amended from time to time,

the "Cypress Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated August 31, 2007 (as amended from time to time, the "Cypress Guaranty");

- (xi) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and HILL COUNTRY APARTMENTS, L.P., a Delaware limited partnership, as borrower, dated March 1, 2007 (as amended from time to time, the "Hill Country Apartments Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated March 1, 2007 (as amended from time to time, the "Hill Country Apartments Construction Guaranty");
- (xii) a Promissory Note by HILL COUNTRY APARTMENTS, L.P., a Delaware limited partnership, payable to the order of Lender, as successor by merger to LaSalle Bank National Association, a national banking association, dated March 1, 2007 (as amended from time to time, the "Hill Country Apartments Land Note") and OPUS as a guarantor pursuant to a Guaranty Agreement dated March 1, 2007 (as amended from time to time, the "Hill Country Apartments Land Guaranty");
- (xiii) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and ARCH ROAD LIMITED PARTNERSHIP, a Delaware limited partnership, as borrower, dated October 4, 2007 (as amended from time to time, the "Arch Road Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated October 4, 2007 (as amended from time to time, the "Arch Road Construction Guaranty");
- (xiv) a Promissory Note by OPUS, payable to the order of Lender, as successor by merger to LaSalle Bank National Association, a national banking association, dated January 4, 2007 (as amended from time to time, the "Westlake Village Note") and a Certificate of Representations, Warranties and Covenants executed by OPUS, dated January 4, 2007 (as amended from time to time, the "Westlake Village Certificate");
- (xv) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and OPUS, as borrower, dated September 27, 2007 (as amended from time to time, the "Scripps Ranch Construction Agreement");
- (xvi) a Promissory Note by OPUS, payable to the order of Lender, as successor by merger to LaSalle Bank National Association, a national banking association, dated July 5, 2006 (as amended from time to time, the "Stanford Ranch Note") and a Certificate of Representations, Warranties and Covenants

executed by OPUS dated July 5, 2006 (as amended from time to time, the "Stanford Ranch Certificate");

- (xvii) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and BROADSTONE WALKER COMMONS, L.P., a Delaware limited partnership, as borrower, dated August 31, 2007 (as amended from time to time, the "Walker Commons Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated August 31, 2007 (as amended from time to time, the "Walker Commons Construction Guaranty");
- (xviii) an Amended and Restated Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and IRVINE CENTER PARTNERS III, L.L.C., a Delaware limited liability company, as borrower, dated October 31, 2006 (as amended from time to time, the "Irvine Center Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated October 11, 2006 (as amended from time to time, the "Irvine Center Construction Guaranty");
- (xix) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and HILL COUNTRY GALLERIA, L.P., a Delaware limited partnership, as borrower, dated July 27, 2006 (as amended from time to time, the "Hill Country Galleria Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated July 27, 2006 (as amended from time to time, the "Hill Country Galleria Construction Guaranty");
- (xx) a Construction Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and BROADSTONE GALLERIA, L.P., a Delaware limited partnership, as borrower, dated July 30, 2006 (as amended from time to time, the "Broadstone Galleria Construction Agreement") and OPUS as a guarantor pursuant to a Guaranty Agreement dated June 30, 2006 (as amended from time to time, the "Broadstone Galleria Construction Guaranty"); and
- (xxi) a Loan Agreement by and between Lender, as successor by merger to LaSalle Bank National Association, a national banking association, and OPUS, as borrower, dated May 29, 2003 (as amended from time to time, the "Opus May RLC Agreement").

The documents evidencing, securing, guaranteeing or executed and delivered in connection with the Loans, including, without limitation, the Loan Agreements executed in connection with the Loans (collectively, as amended from time to time, the "Loan Agreements"), the Guaranties executed in connection with the Loans (collectively, as amended from time to

time, the "Guaranties"), and the Certificates of Representations, Warranties, and Covenants executed in connection with the Loans (collectively, as amended from time to time, the "Certificates") are herein called the "Loan Documents".

The Loan Agreements, Guaranties, and Certificates are hereby amended as follows:

- 1. Sections 2.18, 2.19 and 2.20 of the Opus RLC Agreement are hereby amended in their entirety to read as follows:
 - 2.18 <u>Minimum Liquidity</u>. Borrower shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - 2.19 <u>Tangible Net Worth plus Subordinated Debt</u>. Borrower shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.

"Tangible Net Worth" means the gross book value of Borrower's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles), plus debt subordinated to the Lenders (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Administrative Agent (using Administrative Agent's standard form) less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

2.20 <u>Total Liabilities to Tangible Net Worth Ratio</u>. Borrower shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

"Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

2. Sections 3.21, 3.22 and 3.23 of the Commons at Chino Hills Construction Agreement are hereby amended in their entirety to read as follows:

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 18 of 32

Opus West Corporation March 21, 2008 Page 6

- 3.21 <u>Minimum Liquidity</u>. Borrower shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities of not less than Three Million And No/100 Dollars (\$3,000,000.00).
- 3.22 <u>Tangible Net Worth plus Subordinated Debt.</u> Borrower shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.
- 3.23 <u>Total Liabilities to Tangible Net Worth Ratio</u>. Borrower shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.
- 3. Sections 4.27, 4.28 and 4.29 of the Camarillo Ranch Agreement are hereby amended in their entirety to read as follows:
 - Section 4.27 <u>Minimum Liquidity</u>. Opus shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - Section 4.28 <u>Tangible Net Worth plus Subordinated Debt.</u> Opus shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.

"Tangible Net Worth" means the gross book value of Opus' assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles), plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form) less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

Section 4.29 <u>Total Liabilities to Tangible Net Worth Ratio</u>. Opus shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

"Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

- 4. Sections 2.19, 2.20, and 2.21 of the Highland Village Condos Agreement and Fort Bend Crossing Agreement are hereby amended in their entirety to read as follows:
 - 2.19 <u>Minimum Liquidity</u>. Borrower shall cause Guarantor to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - 2.20 <u>Tangible Net Worth plus Subordinated Debt</u>. Borrower shall cause Guarantor to maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and one Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.

"Tangible Net Worth" means the gross book value of Guarantor's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles), plus debt subordinated to the Loan (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Administrative Agent (using Lender's standard form) less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

2.21 <u>Total Liabilities to Tangible Net Worth Ratio</u>. Borrower shall cause Guarantor to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008 and 5.0 to 1.0 thereafter.

"Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

5. Sections 4.27, 4.28 and 4.29 of the Freemont Tech Agreement, the Pima Center Agreement and the Pima Center III-C Agreement are hereby amended in their entirety to read as follows:

Section 4.27 <u>Minimum Liquidity</u>. Borrower shall cause Guarantor to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).

Section 4.28 <u>Tangible Net Worth plus Subordinated Debt</u>. Borrower shall cause Guarantor to maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.

"Tangible Net Worth" means the gross book value of Guarantor's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles), plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form) less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

Section 4.29 <u>Total Liabilities to Tangible Net Worth Ratio</u>. Borrower shall cause Guarantor to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

- 6. Sections 14(b)(i), (ii), and (iii) of the Highland Village Condos Guaranty, the Fort Bend Crossing Guaranty, the Pima Center Guaranty and the Pima Center III-C Guaranty are hereby amended in their entirety to read as follows:
 - (b) Guarantor hereby agrees as follows:
 - (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 21 of 32

Opus West Corporation March 21, 2008 Page 9

- (ii) to maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Guarantor (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And no/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities. including but not limited to accrued and deferred income taxes. and any reserves against assets; and
- (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).
- 7. Section 13(B) of the Shoppes at Chino Hills Guaranty and the Freemont Tech Guaranty are hereby amended in their entirety to read as follows:

B. Guarantor hereby agrees as follows:

- (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);
- (ii) to maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Guarantor (excluding goodwill, patents, trademarks, trade names,

organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And no/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and

- (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).
- 8. Section 22(a) of the Cypress Guaranty, the Hill Country Apartments Construction Guaranty, the Arch Road Construction Guaranty, and the Walker Commons Construction Guaranty are hereby amended in their entirety to read as follows:
 - (a) Guarantor hereby agrees as follows (the following covenants are collectively called the "Financial Covenants"):
 - (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);
 - (ii) to maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Guarantor (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And no/100 Dollars [\$30,000,000.00]) in a manner acceptable to

Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and

- (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).
- 9. Section 17(a) of the Westlake Village Certificate and the Stanford Ranch Certificate are hereby amended in their entirety to read as follows:
 - (a) Borrower hereby agrees as follows (the following covenants are collectively called the "Financial Covenants"):
 - (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);
 - to maintain at the end of each fiscal quarter on a (ii) consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million AND No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Borrower (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 24 of 32

Opus West Corporation March 21, 2008 Page 12

- (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).
- 10. Section 23(a) of the Hill Country Galleria Construction Guaranty and the Irvine Center Construction Guaranty are hereby amended in their entirety to read as follows:
 - (a) Guarantor hereby agrees as follows (the following covenants are collectively called the "Financial Covenants"):
 - (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);
 - to maintain at the end of each fiscal quarter on a (ii) consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Guarantor (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And no/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and
 - (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender

(limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

- 11. Section 23(a) of the Hill Country Apartments Land Guaranty is hereby amended in its entirety to read as follows:
 - (a) Guarantor hereby agrees as follows (the following covenants are collectively called the "Financial Covenants"):
 - (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00);
 - to maintain at the end of each fiscal quarter on a (ii) consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million And No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter. "Tangible Net Worth" means the gross book value of the assets of Guarantor (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And no/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and
 - (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter. "Total Liabilities" means in accordance with GAAP the sum of current liabilities plus long-term liabilities plus consolidated joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 26 of 32

Opus West Corporation March 21, 2008 Page 14

- (iv) to furnish to Lender certified annual financial statements within one hundred twenty (120) days after the close of each fiscal year; and
- (v) to furnish to Lender certified quarterly financial statements, including compliance certificates, within sixty (60) days of each fiscal quarter and fiscal year end.
- 12. The following definitions are hereby added to Section 1.1 of the Scripps Ranch Construction Agreement:

"Tangible Net Worth" means the gross book value of Borrower's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

- 13. Sections 8.14(a), (b), (c), (d), and (e) of the Scripps Ranch Construction Agreement are hereby amended in their entirety to read as follows:
 - (a) Borrower shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - (b) Borrower shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million AND No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 27 of 32

Opus West Corporation March 21, 2008 Page 15

- (c) Borrower shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.
- 14. The following definitions are hereby added to Section 1.1 of the Irvine Center Construction Agreement:

"Tangible Net Worth" means the gross book value of Guarantor's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

- 15. Sections 8.14(i), (ii), (iii), and (iv) of the Irvine Center Construction Agreement are hereby amended in their entirety to read as follows:
 - (i) Guarantor shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - (ii) Guarantor shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million AND No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.
 - (iii) Guarantor shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 28 of 32

Opus West Corporation March 21, 2008 Page 16

16. The following definitions are hereby added to Section 1.1 of the Broadstone Galleria Construction Agreement:

"Tangible Net Worth" means the gross book value of Opus West's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

- 17. Sections 8.13, 8.14, and 8.15 of the Broadstone Galleria Construction Agreement are hereby amended in their entirety to read as follows:
 - 8.13 Opus West shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - 8.14 Opus West shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million AND No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.
 - 8.15 Opus West shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

Case 09-34356-hdh11 Claim 82-1 Part 6 Filed 11/09/09 Desc Attachments to Addendum Part 5 of 7 Page 29 of 32

Opus West Corporation March 21, 2008 Page 17

- 18. Section 8.16 of the Broadstone Galleria Construction Agreement is hereby deleted in its entirety.
- 19. The following definitions are hereby added to Section 1.1 of the Opus May RLC Agreement:

"Tangible Net Worth" means the gross book value of Borrower's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles) plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form), less total liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets.

- 20. Sections 5.4(b), (c), and (d) of the Opus May RLC Agreement are hereby amended in their entirety to read as follows:
 - (b) Borrower shall maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Three Million And No/100 Dollars (\$3,000,000.00).
 - (c) Borrower shall maintain at the end of each fiscal quarter on a consolidated basis Tangible Net Worth plus subordinated debt equal to at least One Hundred Twenty-Five Million AND No/100 Dollars (\$125,000,000.00) for fiscal year 2008 and One Hundred Million And No/100 Dollars (\$100,000,000.00) thereafter.
 - (d) Borrower shall maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2008, and 5.0 to 1.0 thereafter.

21. Section 5.4(f) of the Opus May RLC Agreement is hereby deleted in its entirety.

Each of the Loan Documents, to the extent applicable, is hereby modified to incorporate the foregoing modifications which shall govern and control in the event of any inconsistency with any other provision of any of the Loan Documents. All other terms and conditions of the Loan Documents shall remain unchanged and in full force and effect. Any property or rights or interest in property granted as security for the Loans shall remain as security for the Loans and the obligations of the borrowers under the Loan Documents.

BANK OF AMERICA, N.A., a national banking association

By: FOCARO

Name: EDGARDO MARTINEZ

Title: ASST. VICE ARESIDENT

March 21, 20	Sorperation – 800		. ** • •	٠.	٠.		
Page 19		•	,			•	,
					:		

Acknowledged and Agreed to as of the Zo day of Wolcows, 2008

OPUS WEST CORPORATION, a Minnesota corporation

By:	Clearly Vor
Name:_	Charles Voge
Title:	Signior Vice Physician
	BORROWER

With respect to the Commons at Chino Hills Construction Agreement:

Acknowledged and Agreed to as of the PD day of Mechan, 2008

OPUS WEST CORPORATION, a Minnesota corporation

By: Villes your Title: Smile Vice Free dant

BORROWER

	h 21	st Corp , 2008	oratio	n			
					. J		

With respect to the Camarillo Ranch Agreement:

O.W. COMMERCIAL, INC., a Delaware corporation

Name: Market Constitution of the Constitution

OPUS WEST CORPORATION, a Minnesota corporation

By: Charles Votes
Title: Senior Vice Pression

BORROWER

With respect to the Highland Village Condos Agreement and the Highland Village Condos Guaranty:

Acknowledged and Agreed to as of the Zoday of M. Colon, 2008

OPUS WEST LP, a Delawate limited partnership, Borrower

BY: OPUS WEST CORRORATION, a
Minnesota corporation, its General Partner

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Name: Spilot Vice President

With respect to the Fort Bend Crossing Agreement and the Fort Bend Crossing Guaranty:

Acknowledged and Agreed to as of the 2008 thurs 2008,

OPUS WEST LP, a Delaware limited partnership, Borrower :

BY: OPUS WEST CORPORATION, a.

Minnesota corporation, Its General Partner

Name: Title:

OPUS WEST CORPORATION, a Minnesota corporation

By: Name: Title:

With respect to the Shoppes at Chino Hills Agreement and the Shoppes at Chino Hills Guaranty:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

By: Charles Voya.
Title:

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Chicles Versel
Title: Senior Vice President

Opus-West Corp	or	ation
March 21, 2008	٠	. :"
Page 24	·	•

With respect to the Freemont Tech Agreement and the Fremont Tech Guaranty:

Acknowledged and Agreed to as of the 2008

O.W. COMMERCIAL, INC., a Delaware corporation

By: Confidence State Title: View President

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:
Name: Charles Vo. 10
Title: Senior fee Charles

With respect to the Pima Center Agreement and the Pima Center Guaranty:

PC 101, INC. a Delaware corporation

By: Charles Variety
Title: Charles Variety

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:______ Name:_____ Title:_____

With respect to the Pima Center III-C Agreement and the Pima Center III-C Guaranty:

Acknowledged and Agreed to as of the Zoday of Williams, 2008

PC 101, INC., a Delaware corporation

By:____ Name:___ Title:

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:_____ Name:__ Title;___

-Opus	r₩e	str	ÇOI	porat	ion-	٠	 <u> </u>	<u>i</u>
Marc	h 21	, 2	008	3 .	•			
Page								
		٠		٠			٠.	

With respect to the Cypress, Agreement and the Cypress Guaranty.

Acknowledged and Agreed to as of the Louis of Manager 2008

BROADSTONE CYPRESS, L.P., a Delaware limited partnership

BY: OWR CYPRESS INC., a Delaware corporation, its General Partner

By: Charles Vogel
Name: Charles Vogel
Title: President

OPUS WEST CORPORATION, a Minnesota corporation

Name: Senior Vice Fresident

With respect to the Hill Country Apartments Construction Agreement and the Hill Country Apartments Construction Guaranty:

Acknowledged and Agreed to as of the Zoos

HILL COUNTRY APARTMENTS, L.P., a Delawate limited partnership

BY: OWR Hill Country, Inc., a Delaware corporation, its General Partner

By:
Name: Charles Vells
Title: Was Dresses

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:____ Name: Title:

With respect to the Hill Country Apartments Land Note and the Hill Country Apartments Land Guaranty:

Acknowledged and Agreed to as of the Zoday of 2008

HILL COUNTRY APARTMENTS, L.P., a Delaware limited partnership

BY: OWR Hill Country, Inc., a Delaware corporation, its General Manager

By: Name: Citation Spiritific Spi

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Claries Vote: Title: Senior Vote: Senior

With respect to the Arch Road Construction Agreement and the Arch Road Construction Guaranty:

ARCH ROAD LIMITED PARTNERSHIP, a Delaware limited partnership

BY: Stockton GP, L.L.C., a Delaware limited liability company, its General Partner

By: 0554
Name: 17 had selected.
Title: 16 had selected.

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Name: Title: Senior Vice S

Programme and the second secon

With repeat of the Westlate Village Note and the

OPER WEST TOTAL TON IN COMPAND corporation

QBACTIVE SHEEPS

Opus Was Consulation
March 21, 2008
Page 50

With respect to a Salinea Resun Goest retion

OPUS WEST GURPTRATION AMERICAN

BORROWER

Various (1988)
Page 41

OPUS WEET CORPORA TON a Mineson.

By Manual Manual

MINE VER

CHACTIVE GEORGE

With respect to the Walker Commons Construction Agreement and the Walker Commons Construction Guaranty:

BROADSTONE WALKER COMMONS, L.P., a Delaware limited partnership

BY: OWR Walker Commons, Inc., a Delaware corporation, its General Partner

By: Charles Vos F Name: Charles Vos F Title: Miss Business

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:____ Name:__ Title:

Charge Vol. A.

With respect to the Invine Center Construction Agreement and the Irvine Center Construction Guaranty:

Acknowledged and Agreed to as of the 2008

IRVINE CENTER PARTNERS III, L.L.C., a Delaware limited liability company

BY: Opus West Corporation, a Minnesota corporation, its Manager

By: Name: Pital State of Title: Was Proches

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Charles Cost

Name: Charles Cost

Title: Senior Vice President

With respect to the Hill Country Galleria Construction Agreement and the Hill Country Galleria Construction Guaranty:

Acknowledged and Agreed to as of the Z day of 2008

HILL COUNTRY GALLERIA, L.P., a Delaware limited partnership

BY: OWR Hill Country, Inc., a Delaware corporation, its General Partner

By: Vocal Name: Title:

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By: Charles Voted

Title: Senior Vision President

With respect to the Broadstone Galleria Construction Agreement and the Broadstone Galleria Construction Guaranty:

Acknowledged and Agreed to as of the 2008

BROADSTONE GALLERIA, L.P., a Delaware limited partnership.

BY: OWR Galleria, Inc., a Delaware corporation, its General Partner

By: Charles Visual
Title: Charles Visual

BORROWER

OPUS WEST CORPORATION, a Minnesota corporation

By:_____

Charles Vag

GUARANTOI

With respect to the Douglast RIC Agreement

At anowied and A greet to see of the Le day of Lease 24 7000

OPUS WEST CORPORATION, a Minuscote corporation

-Opus-West-Gesperation March 21, 2008 Page 39

CONSENT AND AGREEMENT OF GUARANTOR

The undersigned ("Guarantor") executed guarantees in connection with one or more of the Loans described and defined in the foregoing letter agreement. The Guarantor hereby consents and agrees to the modifications and all other matters contained in the letter agreement and agrees that nothing contained in the letter agreement shall in any manner affect or impair any of the obligations or liabilities of the undersigned Guarantor to Lender. Bach guaranty executed by the Guarantor, in connection with one or more of the Loans is continued in full force and effect and remains unaffected and unahanged except to the extent specifically amended by this consent and agreement. All such guaranties are hereby ratified and reaffirmed and Guarantor specifically acknowledges the validity and enforceability thereof.

Dated as of March 20, 2008.

OPUS WEST CORPORATION, a Minnesota comporation

By: Name: Charles Voges of Title: Senior Vice Project

GUARANTOR

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Addendum Part 6 of 7 Page 20 of 32

ATTACHMENT 7

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Addendum Part 6 of 7 Page 21 of 32

					Filing Date Filing Tim State of M Processing	Office: Se	2
UCC FINANCING FOLLOW INSTRUCTIONS (fr	STATEMEN	IT LY			Filed by: s	chdo01	
A. NAME & PHONE OF CONTA	CT AT FILER options	}					
Mila Cruz, 602-22	9-5523						
B. SEND ACKNOWLEDGEME		dress)					
,	. T 1 D		7]			
Quarles & Brady Renaissance One			•				
Two North Cent	ral Avenue			1			
Phoenix, Arizon				Ĭ			
				THE ABOVE SPACE	CE IS FOR FILING	OFFICE USE ONL	Y
Attention: Mila				<u></u>			
I. DEBTOR'S EXACT FUL	L LEGAL NAME - i	sert only one debtor name (In or 1b)	- do not al	obreviate or combine names			
Is ORGANIZATION'S N	IAME						
Shoppes at	Chino Hills, Inc.		FIRST	NAME	MIDDLE NAME		SUFFIX
OR IB. INDIVIDUAL'S LAS	T NAME		1		_		<u>·</u>
			CITY		STATE		COUNTRY
IC MAILING ADDRESS	al. Dood Suite	800]	Phoenix	AZ_	85016-9267	USA
2555 East Camelba	ADD'L INFO RE	Is. TYPE OF ORGANIZATION	it ru	USDICTION OF ORGANIZATION	Is. ORGANIZATI	ONAL ID #. if any	
Id. TAX ID# SSN or EIN	ORGANIZATION	corporation	-1	Minnesota	12K-315	(Minnesota)	NONE
51-0278211	. DEBTOR			and the season of the season o	mes		
		EGAL NAME - insert only one de	pptos same	(2a or 2b) - do not abbreviate or combine na			
2a ORGANIZATION'S	NAME						
OR 25, INDIVIDUAL'S LA	CT NAME		FIRST	NAME.	MIDDLE NAME		SUFFIX
25, INDIVIDUAL'S CA	71 LYME		1		<u> </u>	T 0075	COUNTRY
2c. MAILING ADDRESS			CITY		STATE	POSTAL CODE	COUNTRY
26. MAILLING ADDITION					2 ORGANIZAT	FIONAL ID #, if any	<u> </u>
24. TAX ID# SSN or FIN	ADD'L INFO RE	2e. TYPE OF ORGANIZATION	2r. T.	RISDICTION OF ORGANIZATION	Zg. OKOANIZA	10,000.00.00.00	NONE
51-0278211	ORGANIZATION DEBTOR				!		L_JROKE
3 SECURED PARTY'S	NAME (or NAME of	TOTAL ASSIGNEE of ASSIG	NOR S/P	- insert only one secured party name (3a or	36)		
BANK C)F AMERIĆA,	N.A., a national bank	ing ass	ociation, as Administrative	MIDDLE NAM	<u> </u>	SUFFIX
OR Jb. INDIVIDUAL'S L	AST NAME		FIR	ST NAME	This been street	_	}
			CIT	~ · · · · · · · · · · · · · · · · · · ·	STATE	POSTAL CODE	COUNTRY
3c. MAILING ADDRESS		a 471 000 00	1	•	· I	1	
Commercial Real	Estate Banking	Group, AZ1-200-22-	''	Phoenix	AZ	85004-234	3 USA
201 East Washing	ton Street, 22n	d Floor	1	ruccinx	1	1	<u> </u>
Attn: N. Alonzo					L	_L	مسبيب سيسيب
4 This FINANCING STATE	MENT covers the follow	ving colleteral					
See Schedule	e "I" attached h	ereto and by this refer	ence in	corporated herein.			
Sec Selledan		•					
							· · · · · · · · · · · · · · · · · · ·
5 ALTERNATIVE DESIG	NATION Lif apolicable):			CONSIGNOR BAILEE/BAILOR	SELLER/BUYER	AG. LIEN	NON-UCC FILING
		d (for record) in the REAL	Check to I	LEQUEST SEARCH REPORT(s) on Debtor AL FEE] [optional]	⁽⁹⁾	All Debtors	Debtor 1 Debtor 2
ESTATE RECORD	S. Attach Addendum	if applicable				<u>'</u>	
8. OPTIONAL FILER REF	ERENCE DATA				1		
Minnesota S	Secretary State	for The Shoppes at Ch	ino Hi	lls, California - Constructio	n Loan		
FILING OFFICE COP	Y — UCC FINANC	ING STATEMENT (FORM UC	C1) (RE	V, 05/22/02)			

SCHEDULE "I"

(The Shoppes at Chino Hills, California - Construction Loan)

Financing Statement (continued)

Name of Debtor:

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation

All of the Property (as hereinafter defined) which constitutes personal property or fixtures, all proceeds and products thereof, and all supporting obligations ancillary to or arising in any way in connection therewith, where:

"Accessories" means all fixtures, equipment, systems, machinery, furniture, furnishings, appliances, inventory, goods, building and construction materials, supplies and other articles of personal property, of every kind and character, tangible and intangible (including software embedded therein), now owned or hereafter acquired by Debtor, which are now or hereafter attached to or situated in, on or about the Land or Improvements, or used in or necessary to the complete and proper planning, development, use, occupancy or operation thereof, or acquired (whether delivered to the Land or stored elsewhere) for use or installation in or on the Land or Improvements, and all Additions to the foregoing, all of which are hereby declared to be permanent accessions to the Land.

"Accounts" means all accounts of Debtor within the meaning of the Uniform Commercial Code of the State, derived from or arising out of the use, occupancy or enjoyment of the Property or for services rendered therein or thereon.

"Additions" means any and all alterations, additions, accessions and improvements to property, substitutions therefor, and renewals and replacements thereof.

"Claim" means any liability, suit, action, claim, demand, loss, expense, penalty, fine, judgment or other cost of any kind or nature whatsoever, including fees, costs and expenses of attorneys, consultants, contractors and experts.

"Condemnation" means any taking of title to, use of, or any other interest in the Property under the exercise of the power of condemnation or eminent domain, whether temporarily or permanently, by any Governmental Authority or by any other Person acting under or for the benefit of a Governmental Authority.

"Condemnation Awards" means any and all judgments, awards of damages (including severance and consequential damages), payments, proceeds, settlements, amounts paid for a taking in lieu of Condemnation, or other compensation heretofore or hereafter made, including interest thereon, and the right to receive the same, as a result of, or in connection with, any Condemnation or threatened Condemnation.

"Contract of Sale" means any contract for the sale of all or any part of the Property or any interest therein, whether now in existence or hereafter executed.

"Design and Construction Contracts" means, collectively, (a) all contracts for services to be rendered, work to be performed or materials to be supplied in the development of the Land or the construction or repair of Improvements, including all agreements with architects, engineers or contractors for such services, work or materials; (b) all plans, drawings and specifications for the development of the Land or the construction or repair of Improvements; (c) all permits, licenses, variances and other rights or approvals issued by or obtained from any Governmental Authority or other Person in connection with the development of the Land or the construction or repair of Improvements; and (d) all amendments of or supplements to any of the foregoing.

"Governmental Authority" means any governmental or quasi-governmental entity, including any court, department, commission, board, bureau, agency, administration, service, district or other instrumentality of any governmental entity.

"Improvements" means all buildings, structures and other improvements now or hereafter existing, erected or placed on the Land, together with any on-site improvements and off-site improvements in any way used or to be used in connection with the use, enjoyment, occupancy or operation of the Land.

"Insurance Proceeds" means the insurance claims under and the proceeds of any and all policies of insurance covering the Property or any part thereof, including all returned and unearned premiums with respect to any insurance relating to such Property, in each case whether now or hereafter existing or arising.

"Land" means the real property described in Exhibit "A" attached hereto and made a part hereof.

"Leases" means all leases, license agreements and other occupancy or use agreements (whether oral or written), now or hereafter existing, which cover or relate to the Property or any part thereof, together with all options therefor, amendments thereto and renewals, modifications and guaranties thereof, including any cash or security deposited under the Leases to secure performance by the tenants of their obligations under the Leases, whether such cash or security is to be held until the expiration of the terms of the Leases or applied to one or more of the installments of rent coming due thereunder.

"Loan" means the loan from Beneficiary to Debtor with respect to the Land.

"Person" means an individual, a corporation, a partnership, a joint venture, a limited liability company, a trust, an unincorporated association, any Governmental Authority or any other entity.

"Personalty" means all personal property of any kind or nature whatsoever, whether tangible or intangible and whether now owned or hereafter acquired, in which Debtor now has or hereafter acquires an interest and which is used in the

construction of, or is placed upon, or is derived from or used in connection with the maintenance, use, occupancy or enjoyment of, the Property, including (a) the Accessories; (b) the Accounts; (c) all franchise, license, management or other agreements with respect to the operation of the Real Property or the business conducted therein and all general intangibles (including payment intangibles, trademarks, trade names, goodwill, software and symbols) related to the Real Property or the operation thereof; (d) all sewer and water taps, appurtenant water stock or water rights, any Type 2 nonirrigation grandfathered water rights, contractual rights to water, allocations and agreements for utilities, bonds, letters of credit, permits, certificates, licenses, guaranties, warranties, causes of action, judgments, Claims, profits, security deposits, utility deposits, and all rebates or refunds of fees, Taxes, assessments, charges or deposits paid to any Governmental Authority related to the Real Property or the operation thereof; (e) all of Debtor's rights and interests under all Swap Contracts, including all rights to the payment of money from Beneficiary under any Swap Contract and all accounts, deposit accounts and general intangibles, including payment intangibles, described in any Swap Contract; (f) all insurance policies held by Debtor with respect to the Property or Debtor's operation thereof; and (g) all money, instruments and documents (whether tangible or electronic) arising from or by virtue of any transactions related to the Property, and all deposits and deposit accounts of Debtor with Beneficiary related to the Property, including any such deposit account from which Debtor may from time to time authorize Beneficiary to debit and/or credit payments due with respect to the Loan; together with all Additions to and Proceeds of all of the foregoing.

"Proceeds," when used with respect to any of the Property, means all proceeds of such Property, including all Insurance Proceeds and all other proceeds within the meaning of that term as defined in the Uniform Commercial Code of the State.

"Property" means the Real Property and the Personalty and all other rights, interests and benefits of every kind and character which Debtor now has or hereafter acquires in, to or for the benefit of the Real Property and/or the Personalty and all other property and rights used or useful in connection therewith, including all Leases, all Rents, all Condemnation Awards, all Proceeds, and all of Debtor's right, title and interest in and to all Design and Construction Contracts, all Contracts of Sale and all Refinancing Commitments.

"Real Property" means the Land and Improvements, together with (a) all estates, title interests, title reversion rights, remainders, increases, issues, profits, rights of way or uses, additions, accretions, servitudes, strips, gaps, gores, liberties, privileges, water rights, water courses, alleys, passages, ways, vaults, licenses, tenements, franchises, hereditaments, appurtenances, easements, rights-of-way, rights of ingress or egress, parking rights, timber, crops, mineral interests and other rights, now or hereafter owned by Debtor and belonging or appertaining to the Land or Improvements; (b) all Claims whatsoever of Debtor with respect to the Land or Improvements, either in law or in equity, in possession or in expectancy; (c) all estate, right, title and interest of Debtor in and to all streets, roads and public places,

opened or proposed, now or hereafter adjoining or appertaining to the Land or Improvements; and (d) all options to purchase the Land or Improvements, or any portion thereof or interest therein, and any greater estate in the Land or Improvements, and all Additions to and Proceeds of the foregoing.

"Refinancing Commitment" means any commitment from or other agreement with any Person providing for the financing of the Property, some or all of the proceeds of which are intended to be used for the repayment of all or a portion of the Loan.

"Rents" means all of the rents, royalties, issues, profits, revenues, earnings, income and other benefits of the Property, or arising from the use or enjoyment of the Property, including all such amounts paid under or arising from any of the Leases and all fees, charges, accounts or other payments for the use or occupancy of rooms or other public facilities within the Real Property.

"State" means the State of Arizona.

"Swap Contract" means any agreement, whether or not in writing, relating to any transaction that is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap or option, bond, note or bill option, interest rate option, forward foreign exchange transaction, interest cap, collar or floor transaction, currency swap, cross-currency rate swap, swap option, currency option or any other similar transaction (including any option to enter into the foregoing) or any combination of the foregoing, and, unless the context otherwise clearly requires, any form of master agreement published by the International Swaps and Derivatives Association, Inc., or any other master agreement, entered into between Beneficiary (or its affiliate) and Debtor (or its affiliate) in connection with the Loan, together with any related schedules and confirmations, as amended, supplemented, superseded or replaced from time to time, relating to or governing any or all of the foregoing.

"Taxes" means all taxes and assessments, whether general or special, ordinary or extraordinary, or foreseen or unforeseen, which at any time may be assessed, levied, confirmed or imposed by any Governmental Authority or any community facilities or other private district on Debtor or on any of its properties or assets or any part thereof or in respect of any of its franchises, businesses, income or profits.

The Land is all that real property situate in the County of San Bernardino, State of California, and more particularly described on Exhibit "A" attached hereto.

Exhibit "A "

PARCEL A:

THE NORTH 30 ACRES OF THAT PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1 PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

BEGINNING AT A POINT IN THE WEST LINE OF SECTION 16, SOUTH 00 40' 39" 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION; THENCE SOUTH ALONG THE WEST LINE OF SAID SECTION 16 TO A POINT 330.3 FEET SOUTH OF THE NORTHWEST CORNER OF THE SOUTHWEST ONE-QUARTER OF SAID SECTIONS; THENCE EAST PARALLEL TO THE NORTH LINE OF SECTION 16 TO THE WEST LINE OF LAND CONVEYED TO KATE FOWLER MERLE-SMITH, FORMERLY KATE FOWLER, BY THE CHINO LAND AND WATER COMPANY BY DEED RECORDED APRIL 6,1925 IN BOOK 895 OF DEEDS, PAGE 87; THENCE NORTHERLY ALONG THE WEST LINE OF THE ABOVE DESCRIBED LAND TO A POINT 660.7 FEET SOUTH OF THE NORTH LINE OF SECTION 16; THENCE WEST PARALLEL TO THE NORTH OF SECTION 16 TO THE POINT OF BEGINNING.

EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT A POINT ON THE WEST LINE OF THE ABOVE DESCRIBED LAND, 280 FEET NORTH FROM THE SOUTHWEST CORNER OF SAID NORTH 30 ACRES; THENCE EAST 38 FEET; THENCE NORTH 17 FEET, 4 INCHES; THENCE WEST 38 FEET TO THE WEST LINE OF SAID SECTION 16, AND THENCE SOUTH ALONG SAID LINE 17 FEET, 4 INCHES TO THE POINT OF BEGINNING

ALSO EXCEPTING THEREFROM THE NORTH 30 FEET AND THE WEST 30 FEET OF THE ABOVE DESCRIBED PROPERTY DEEDED FOR COUNTY ROADS.

ALSO EXCEPTING FROM SAID 30 ACRES THE FOLLOWING:

BEGINNING AT THE INTERSECTION OF THE EAST LINE OF THAT CERTAIN PARCEL OF LAND, AS DESCRIBED BY DEED TO LELAND C. LAUNDER, RECORDED JANUARY 21, 1957 IN BOOK 4135 OF OFFICIAL RECORDS, PAGE 105, WITH THE SOUTHERLY LINE OF PEYTON DRIVE, 60 FEET WIDE, AS DESCRIBED IN DEED RECORDED IN BOOK 359 OF OFFICIAL RECORDS, PAGE 205; THENCE ALONG SAID SOUTHERLY LINE, BEING A NON-TANGENT CURVE CONCAVE NORTHERLY WITH A RADIUS OF 377 FEET, FROM A TANGENT BEARING SOUTH 76 41' 27" WEST, THROUGH AN ANGLE OF 12 51' 28" A DISTANCE OF 84.60 FEET; THENCE NORTH 89 32' 55" EAST 83.90 FEET TO SAID EAST LINE, DISTANT ALONG SAID EAST LINE SOUTH 0 28' 26" EAST 9.45 FEET FROM THE POINT OF BEGINNING; THENCE NORTH 0 28' 26" WEST 9.45 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THAT PORTION AS CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OFFICIAL RECORDS.

ALSO EXCEPTING THEREFROM THAT PORTION THEREOF LYING EASTERLY, NORTHERLY AND WESTERLY OF THE FOLLOWING DESCRIBED LINE:

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Addendum Part 6 of 7 Page 27 of 32 Filing NO: 200716885438

COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED LAND; THENCE ALONG THE SOUTHERLY LINE OF SAID LAND SOUTH 89 24' 44" WEST 21.61 FEET TO TRUE POINT OF BEGINNING, BEING THE BEGINNING OF A NON TANGENT CURVE CONCAVE SOUTHEASTERLY HAVING A RADIUS OF 388.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS SOUTH 89 14' 57" WEST; THENCE NORTHEASTERLY 93.49 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 13 48' 22"; THENCE NORTH 13 03' 19" EAST 10.00 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 45.00 FEET; THENCE NORTHWESTERLY 60.79 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 77 21' 57" TO THE SOUTH LINE OF THE LAND AS DESCRIBED IN THE QUITCLAIM DEED TO THE CITY OF CHINO HILLS RECORDED AUGUST 11, 2003 AS DOCUMENT NO.2003-0591428, OFFICIAL RECORDS, COUNTY OF SAN BERNARDINO, BEING A POINT ON A CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET; THENCE NORTHWESTERLY 660.12 FEET ALONG SAID CURVE AND SOUTHERLY LINE THROUGH A CENTRAL ANGLE OF 26 15' 56"; THENCE SOUTH 89 25' 26" WEST 293.36 FEET TO THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1740.00 FEET; THENCE WESTERLY 153.03 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 10 03' 09" TO THE BEGINNING OF A COMPOUND CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 40.00 FEET: THENCE SOUTHWESTERLY 63.89 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 91 31' 07" TO THE BEGINNING OF A REVERSE CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1740.00 FEET; THENCE SOUTHERLY 146.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 05 43' 08" TO THE NORTHERLY LINE OF THE PROPERTY CONVEYED TO THE CITY OF CHINO HILLS BY DEED RECORDED JUNE 12, 1992 AS INSTRUMENT NO. 249424 OF OFFICIAL RECORDS; THENCE ALONG SAID NORTHERLY LINE SOUTH 89 25' 26" WEST 15.29 FEET TO THE EAST LINE OF THE LAND DESCRIBED IN THE DECLARATION OF ROAD RIGHT OF WAY, RECORDED MARCH 15, 1988 AS DOCUMENT NO. 88-077222 OFFICIAL RECORDS, SAID POINT BEING ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE AND ALONG SAID EAST LINE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE SOUTH 00 37' 46" EAST 590.81 FEET TO THE SOUTH LINE OF SAID NORTH 30 ACRES.

PARCEL B:

THAT PORTION OF THE NORTH 30 ACRES DESCRIBED IN THAT CERTAIN DEED TO THE COUNTY OF SAN BERNARDINO AS PER DEED RECORDED ON DECEMBER 16, 1983, AS INSTRUMENT NO. 83-296886, OFFICIAL RECORDS, AND BEING A PORTION OF THE WEST ONE-HALF OF THE WEST ONE-HALF OF SECTION 16, TOWNSHIP 2 SOUTH, RANGE 8 WEST, SAN BERNARDINO MERIDIAN, BEING A PORTION OF LOT 37 OF THE RANCHO SANTA ANA DEL CHINO, AS PER MAP RECORDED IN BOOK 1, PAGE 12 OF MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY, DESCRIBED AS FOLLOWS:

COMMENCING AT THE CENTERLINE INTERSECTION OF GRAND AVENUE AND PEYTON DRIVE, AS SHOWN ON MAP OF TRACT NO. 12581-12, AS PER MAP RECORDED IN BOOK 188 OF MAPS, PAGES 98 THROUGH 100, INCLUSIVE, RECORDS OF SAID COUNTY, SAID INTERSECTION BEING SOUTH 00 40' 39" EAST, A DISTANCE OF 659.40 FEET (SOUTH 660.7 FEET) FROM THE NORTHWEST CORNER OF SAID SECTION 16 AS SHOWN ON THAT RECORD OF SURVEY MAP RECORDED IN BOOK 54 OF RECORDS OF SURVEY, PAGES 58 AND 59, RECORDS OF SAID COUNTY; THENCE SOUTH 00 37' 46" EAST, ALONG THE CENTERLINE OF SAID PEYTON DRIVE, A DISTANCE OF 987.02 FEET TO THE SOUTH LINE OF AFORESAID NORTH 30 ACRES; THENCE NORTH 89 24' 51" EAST ALONG SAID SOUTH LINE, A DISTANCE OF 60.00 FEET TO A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SAID POINT BEING THE TRUE POINT OF BEGINNING;

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Filing NO: 200 denselys Part 6 of 7 Page 28 of 32

THENCE CONTINUING ALONG SAID SOUTH LINE NORTH 89 24' 51" EAST, A DISTANCE OF 1263.78 FEET TO THE SOUTHEAST CORNER OF SAID NORTH 30 ACRES; THENCE ALONG THE EAST LINE OF SAID NORTH 30 ACRES, NORTH 00 35' 16" WEST, DISTANCE OF 624.82 FEET; THENCE AT RIGHT ANGLES TO THE EAST LINE OF SAID NORTH 30 ACRES, SOUTH 89 24' 44" WEST, A DISTANCE OF 66.00 FEET TO A LINE THAT IS PARALLEL WITH AND 66.00 FEET WEST OF THE EAST LINE OF SAID NORTH 30 ACRES; THENCE ALONG SAID PARALLEL LINE, NORTH 00 35' 16" WEST, A DISTANCE OF 20.00 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHEAST AND HAVING A RADIUS OF 166.00 FEET; THENCE NORTHEASTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 28 06' 13", A DISTANCE OF 81.42 FEET; THENCE TANGENT TO SAID LAST CURVE NORTH 27 30' 57" EAST A DISTANCE OF 27.82 FEET TO THE BEGINNING OF A TANGENT CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 27.00 FEET; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 92 25' 57" A DISTANCE OF 43.55 FEET TO A POINT OF COMPOUND CURVE CONCAVE TO THE SOUTHWEST HAVING A RADIUS OF 1440.00 FEET, SAID LAST CURVE BEING THE SOUTHERLY LINE OF GRAND AVENUE; THENCE NORTHWESTERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 25 40' 34", A DISTANCE OF 645.31 FEET TO A POINT OF TANGENCY WITH A LINE THAT IS PARALLEL WITH AND 60.00 FEET SOUTH OF THE CENTERLINE OF PEYTON DRIVE AND THE NORTH LINE OF SAID NORTH 30 ACRES AS SHOWN ON AFORESAID RECORD OF SURVEY MAP; THENCE ALONG SAID PARALLEL LINE AND THE SOUTHERLY LINE OF GRAND AVENUE AS DESCRIBED IN THAT DECLARATION OF ROAD RIGHT-OF-WAY BY THE COUNTY OF SAN BERNARDINO AS RESOLUTION NO. 88-88 AND RECORDED AS INSTRUMENT NO. 88-077222, OFFICIAL RECORDS OF SAID COUNTY, SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET; THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 155.00 FEET; THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 90.0 FEET;

THENCE SOUTH 00 37' 46" WEST, A DISTANCE OF 12.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 92.00 FEET;

THENCE SOUTH 00 37' 46" EAST, A DISTANCE OF 91.00 FEET;

THENCE SOUTH 89 25' 26" WEST, A DISTANCE OF 380.00 FEET TO THE BEGINNING OF A NON-TANGENT CURVE CONCAVE TO THE WEST HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING OF SAID CURVE THROUGH SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06", A DISTANCE OF 78.23 FEET; THENCE TANGENT TO SAID LAST

CURVE AND ALONG A LINE THAT IS PARALLEL WITH AND 60.00 FEET EAST OF THE CENTERLINE OF SAID PEYTON DRIVE, SOUTH 00 37' 46" EAST, A DISTANCE OF 590.81 FEET TO THE TRUE POINT OF BEGINNING.

EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE SOUTHWEST CORNER OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG THE WESTERLY LINE NORTH 00 37' 76" WEST 221.18 FEET TO THE TRUE POINT OF BEGINNING; THENCE NORTH 89 22' 13" EAST 1.99 FEET; THENCE NORTH 00 48' 09" EAST 108.60 FEET; THENCE NORTH 45 24' 04" EAST 35.04 FEET; THENCE NORTH 01 28' 29" EAST 221.44 FEET TO THE NORTHERLY LINE OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 25' 26" WEST 15.29 FEET TO THE WESTERLY LINE OF SAID PARCEL BEING A POINT ON A NONTANGENT CURVE CONCAVE WESTERLY HAVING A RADIUS OF 1453.00 FEET, A RADIAL BEARING TO SAID POINT BEARS NORTH 86 17' 08" EAST; THENCE SOUTHERLY 78.23 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 03 05' 06"; THENCE CONTINUING ALONG SAID WESTERLY LINE SOUTH 00 37' 46" EAST 369.63 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

BEGINNING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE OF SAID PARCEL NORTH 00 35' 16" WEST 26.89 FEET; THENCE SOUTH 41 26' 12" WEST 30.53 FEET; THENCE SOUTH 00 39' 11" EAST 4.21 FEET TO THE SOUTHERLY LINE OF SAID PARCEL; THENCE ALONG SAID SOUTHERLY LINE NORTH 89 24' 51" EAST 20.43 FEET TO THE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Filing NO: 2000 description Part 6 of 7 Page 29 of 32

COMMENCING AT THE SOUTHEAST CORNER OF THE ABOVE DESCRIBED PARCEL; THENCE ALONG THE EASTERLY LINE NORTH 00 35' 16" WEST 453.32 FEET TO THE TRUE POINT OF BEGINNING, BEING THE BEGINNING OF A TANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 312.00 FEET; THENCE NORTHWESTERLY 77.73 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 16' 30" TO THE BEGINNING OF A REVERSE CURVE CONCAVE NORTHEASTERLY HAVING A RADIUS OF 388.00 FEET; THENCE NORTHERLY 95.57 FEET ALONG SAID CURVE THROUGH A CENTRAL ANGLE OF 14 06' 44" TO A POINT ON THE NORTHERLY LINE OF SAID ABOVE DESCRIBED PARCEL; THENCE ALONG SAID NORTHERLY LINE NORTH 89 24' 44" EAST 21.61 FEET TO THE EASTERLY LINE OF SAID PARCEL; THENCE ALONG SAID EASTERLY LINE SOUTH 00 35' 21" EAST 171.50 FEET TO THE TRUE POINT OF BEGINNING.

ALSO EXCEPTING THEREFROM THE FOLLOWING:

COMMENCING AT THE EASTERLY TERMINUS OF THE COURSE SHOWN AS "SOUTH 89 25' 26" WEST, A DISTANCE OF 35.23 FEET" IN THE NORTHERLY LINE OF THE ABOVE PARCEL, SAID POINT BEING THE BEGINNING OF A CURVE CONCAVE SOUTHERLY HAVING A RADIUS OF 1440.00; THENCE SOUTHEASTERLY 44.43 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 01 46' 04" TO THE TRUE POINT OF BEGINNING; THENCE NONMTANGENT SOUTH 44 24' 14" EAST 44.95 FEET; THENCE SOUTH 86 16' 02" EAST 64.80 FEET; THENCE NORTH 48 08' 12" EAST 41.99 FEET TO THE NORTHERLY LINE OF SAID PARCEL, BEING A POINT ON A NONTANGENT CURVE CONCAVE SOUTHWESTERLY HAVING A RADIUS OF 1440.00 FEET, A RADIAL BEARING TO SAID CURVE BEARS NORTH 06 16' 21" EAST; THENCE NORTHWESTERLY 127.70 FEET ALONG SAID CURVE AND SAID NORTHERLY LINE THROUGH A CENTRAL ANGLE OF 05 04' 51" TO THE TRUE POINT OF BEGINNING.

PARCEL C:

A NON-EXCLUSIVE EASEMENT FOR THE PURPOSE OF INSTALLING, MAINTAINING, OPERATING, REPAIRING, AND REPLACING A PYLON SIGN AS DESCRIBED IN THAT CERTAIN PYLON SIGN EASEMENT AND MAINTENANCE AGREEMENT RECORDED JUNE 2, 2006 AS INSTRUMENT NO. 06-380081 OF OFFICIAL RECORDS.

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Addendum Park 25.7 Page 30 of 32

Uniform Commercial Code 60 Empire Drive, Suite 100 Saint Paul, MN 55103 Mark Ritchie Secretary of State

Minnesota Central Filing System

UCC Filing Acknowledgement

May 23, 2007 Page 1 of 1

QUARLES & BRADY STREICH LANG AT LAW 2 N CENTRAL AVE 1 RENAISSANCE SQUARE PHOENIX AZ 85004

The Minnesota Central Filing System has received and filed your document. The information below reflects the data that was indexed in our system. Please review the information for accuracy. If you find a potential error, please notify the appropriate filing office.

Client Account Number: 54046453

Batch Number: 2368231

Filing Type: UCC Financing Stmt

Original Filing Number: 200716885438

Filed Date: 05/22/2007

Filed Time: 5:00 p.m.

Lapse Date: 5/22/2012

Party Type

Party Name and Address

Debtor

SHOPPES AT CHINO HILLS INC

PHOENIX AZ

Secured Party

BANK OF AMERICA NA A NATIONAL BANKING

ASSOCIATION AS ADMINISTRATIVE AGENT

PHOENIX AZ

Filing by the Minnesota Central Filing System is not conclusive proof that all conditions for securing priority have been met. Ensuring that accurate information is on the document to be filed is the responsibility of the filing party. If the filing is challenged, the filing office does not guarantee that the filing is legally sufficient to secure priority under UCC Article 9 and expressly disclaims any liability for failure of the filing party to secure priority resulting from the information contained in the filed document, or the lack of information on the filed document.

User ID: schdo01

County ID: 88

Case 09-34356-hdh11 Claim 82-1 Part 7 Filed 11/09/09 Desc Attachments to Addendum Batter 7 Page 31 of 32

Uniform Commercial Code 60 Empire Drive, Suite 100 Saint Paul, MN 55103 Mark Ritchie Secretary of State

Office of the Secretary of State Packing Slip

May 23, 2007

QUARLES & BRADY STREICH LANG AT LAW MILA F CRUZ 2 N CENTRAL AVE 1 RENAISSANCE SQUARE PHOENIX, AZ 85004 Page 1 of 1

Client Account Number: 54046453

Batch Number: 2368231

Document Number	Document Detail	 Filing Number	Fee
23682310002	UCC1 Standard	 200716885438	20.00

Total Fees \$20.00

Payment Type Received	Payment Reference Number	Amount Paid
Check	131394	20.00
	Total Payments Received	\$20.00

Any overage amount on account with our agency will be refunded after 60 days if not used.



Visit our online services web page to discover timesaving, electronic methods of doing business! www.online.sos.state.mn.us

(651) 296-2803 TTY (800) 627-3529

ATTACHMENT 8

GUARANTY AGREEMENT

(The Shoppes at Chino Hills, CA –Construction Loan)

This Guaranty Agreement (this "Guaranty") is made as of the 16th day of May, 2007 by OPUS WEST CORPORATION, a Minnesota corporation (individually and collectively, jointly and severally, "Guarantor"), in favor of BANK OF AMERICA, N.A., a national banking association, as Administrative Agent for itself and other Lenders (collectively together with its successors and assigns, "Lender").

RECITALS

SHOPPES AT CHINO HILLS, INC., a Minnesota corporation ("Borrower") has requested that Lender make a loan (the "Loan") to Borrower evidenced by one or more Deed of Trust Notes of even date herewith in the aggregate principal face amount of ONE HUNDRED TEN MILLION AND NO/100 (\$110,000,000.00) made by Borrower to the order of Lender (as the same may from time to time be amended, supplemented, restated or otherwise modified, the "Note"). Certain terms and conditions of the Loan are set forth in the Construction Loan Agreement of even date herewith among Borrower and Lender (as the same may from time to time be amended, supplemented, restated or otherwise modified, the "Loan Agreement"). As a condition precedent to making the Loan, Lender has required that Guarantor execute and deliver this Guaranty to Lender. Any capitalized term used and not defined in this Guaranty shall have the meaning given to such term in the Loan Agreement.

<u>AGREEMENTS</u>

For good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, and in order to induce Lender to make the Loan to Borrower, Guarantor hereby guarantees to Lender the prompt and full payment and performance of the indebtedness and obligations described below in this Guaranty (collectively called the "Guaranteed Obligations"), this Guaranty being upon the following terms and conditions:

Section 1. Guaranty of Payment.

Guarantor hereby unconditionally and irrevocably guarantees to Lender the punctual payment when due, whether by lapse of time, by acceleration of maturity, or otherwise, of all principal, interest (including interest accruing after the commencement of any bankruptcy or insolvency proceeding by or against Borrower, whether or not allowed in such proceeding), prepayment premiums, fees, late charges, costs, expenses, indemnification indebtedness, and other sums of money now or hereafter due and owing, or which Borrower is obligated to pay, pursuant to the terms of the Note, the Loan Agreement, the Deed of Trust, the Environmental Agreement, any application, agreement, note or other document executed and delivered in connection with any Letter of Credit, any Swap Contract or any of the other Loan Documents or any Swap Contract, as the same may from time to time be amended, supplemented, restated or otherwise modified (collectively, the "Indebtedness"). The Indebtedness includes all costs and expenses incurred by Lender in seeking to enforce Lender's rights and remedies with respect to the Indebtedness, including court costs, costs of alternative dispute resolution and reasonable attorneys' fees, whether or not suit is filed or other proceedings are initiated thereon. This

Guaranty covers the Indebtedness presently outstanding and the Indebtedness arising subsequent to the date hereof, including all amounts advanced by Lender in stages or installments. The guaranty of Guarantor as set forth in this <u>Section 1</u> is a continuing guaranty of payment and not a guaranty of collection.

Section 2. Guaranty of Performance.

- (a) Guarantor also hereby unconditionally and irrevocably guarantees to Lender the complete performance when due of all other Obligations of Borrower under all of the Loan Documents, including, without limiting the generality of the foregoing all such obligations of Borrower to:
 - (i) Complete the construction of the Improvements in accordance with the requirements of the Loan Agreement;
 - (ii) Make all deposits required under the terms of the Loan Agreement and the other Loan Documents, as and when required;
 - (iii) Promptly pay in full and discharge all Property Assessments (as that term is defined in the Deed of Trust) prior to the day upon which the same shall become delinquent (subject to the terms of the Deed of Trust regarding permitted contests of such Property Assessments);
 - (iv) Pay, at or before the times required by the Loan Documents, the premiums on all policies of insurance required to be maintained under the terms of the Loan Documents; and
 - (v) Duly and punctually perform and observe all other terms, covenants and conditions of the Note, the Loan Agreement, the Deed of Trust, the Environmental Agreement, any Swap Contract and all other Loan Documents.

The foregoing obligations guaranteed under this Section 2(a) are defined as the "Guaranteed Performance Obligations." The Guaranteed Performance Obligations are included as part of the Guaranteed Obligations for all purposes of this Guaranty.

(b) Upon demand by Lender following the occurrence of an Event of Default, Guarantor will cause all Improvements to be completed in accordance with the requirements of the Loan Agreement and will pay all bills in connection therewith. If Lender shall have requested Guarantor to complete or cause the completion of Construction of the Improvements or the renovation or equipping thereof, Guarantor shall be entitled to requisition and draw all of the undisbursed Loan proceeds intended to be used for the Construction of the Improvements pursuant to the Budget (but not in excess of the committed amount of the Loan), together with any deposits in the Borrower's Deposit Account made with respect to the Construction of the Improvements. Lender shall disburse such funds for the purpose of, and to the extent necessary

for, completing the Construction of the Improvements in accordance with the Budget, provided that:

- (i) Guarantor shall be performing the Guaranteed Performance Obligations or causing the performance of the same with due diligence;
- (ii) Guarantor shall have made all required deposits into the Borrower's Deposit Account and all other deposits required under the Loan Agreement;
- (iii) All such disbursements of Loan proceeds to Guarantor shall be secured by the Deed of Trust and any other collateral or security for the Loan with the same priority as all previous advances of Loan proceeds to Borrower;
- (iv) Guarantor shall have cured all monetary and non-monetary Defaults and Events of Default under the Loan, provided that Guarantor shall not be required to cure any non-monetary Default or Event of Default which is personal to Borrower and therefore not susceptible to cure by Guarantor; and
- (v) Guarantor shall otherwise comply with the provisions of the Loan Agreement governing draw requests and disbursement of the Loan.
- (c) The liability and obligations under this Section 2 shall not be limited or restricted by the existence of, or any terms of, the guaranty of payment under Section 1.

Section 3. Primary Liability of Guarantor; Environmental Obligations.

(a) This Guaranty is an absolute, irrevocable and unconditional guaranty of payment and performance, and Guarantor shall be liable for the payment and performance of the Guaranteed Obligations as a primary obligor. This Guaranty shall be effective as a waiver of, and Guarantor hereby expressly waives, any right to which Guarantor may otherwise have been entitled, whether existing under statute, at Law or in equity, to require Lender to take prior recourse or proceedings against any collateral, security or Person. It shall not be necessary for Lender, in order to enforce such payment or performance by Guarantor, first to institute suit or pursue or exhaust any rights or remedies against Borrower or other Person liable on such indebtedness or for such performance, or to enforce any rights against any security given to secure such indebtedness or performance, or to join Borrower or any other Person liable for the payment or performance of the Guaranteed Obligations or any part thereof in any action to enforce this Guaranty, or to resort to any other means of obtaining payment or performance of the Guaranteed Obligations; provided, however, that nothing herein contained shall prevent Lender from foreclosing the Deed of Trust or exercising any other right under the Loan Documents.

- (b) Suit may be brought or demand may be made against Borrower or against any or all parties who have signed this Guaranty or any other guaranty covering all or any part of the Guaranteed Obligations, or against any one or more of them, separately or together, without impairing the rights of Lender against any party hereto.
- (c) The liability of Guarantor or any other Person hereunder for Guaranteed Obligations arising out of or related to the Environmental Agreement shall not be limited or affected in any way by any provision in this Guaranty, the other Loan Documents or applicable Law limiting the liability of Borrower, Guarantor or such other Person, or Lender's recourse or rights to a deficiency judgment.

Section 4. Certain Agreements and Waivers by Guarantor.

- (a) Guarantor agrees that neither Lender's rights or remedies nor Guarantor's obligations under the terms of this Guaranty shall be released, diminished, impaired, reduced or affected by any one or more of the following events, actions, facts, or circumstances, Guarantor waives any rights, claims or defenses arising from any such events, actions, facts, or circumstances, and the liability of Guarantor under this Guaranty shall be absolute, unconditional and irrevocable irrespective of:
 - (i) any limitation on the liability of, or recourse against, any other Person in any Loan Document or arising under any Law;
 - (ii) any claim or defense that this Guaranty was made without consideration or is not supported by adequate consideration or that the obligations of Guarantor hereunder exceed or are more burdensome than those of Borrower under the other Loan Documents:
 - (iii) the taking or accepting of any other security or guaranty for, or right of recourse with respect to, any or all of the Guaranteed Obligations;
 - (iv) the operation of any statutes of limitation or other Laws regarding the limitation of actions, all of which are hereby waived as a defense to any action or proceeding brought by Lender against Guarantor, to the fullest extent permitted by Law;
 - (v) any homestead exemption or any other exemption under applicable Law;
 - (vi) any release, surrender, abandonment, exchange, alteration, sale or other disposition, subordination, deterioration, waste, failure to protect or preserve, impairment, or loss of, or any failure to create or perfect any lien or security interest with respect to, or any other dealings with, any collateral or security at any time existing or purported, believed or expected to exist in connection

with any or all of the Guaranteed Obligations, or any impairment of Guarantor's recourse against any Person or collateral;

- (vii) whether express or by operation of Law, any partial release of the liability of Guarantor hereunder (except to the extent expressly so released) or any complete or partial release of Borrower or any other Person liable, directly or indirectly, for the payment or performance of any or all of the Guaranteed Obligations;
- (viii) the death, insolvency, bankruptcy, disability, dissolution, liquidation, termination, receivership, reorganization, merger, consolidation, change of form, structure or ownership, sale of all assets, or lack of corporate, partnership or other power of Borrower or any other Person at any time liable for the payment or performance of any or all of the Guaranteed Obligations;
- (ix) either with or without notice to or consent of Guarantor, any renewal, extension, modification, supplement, subordination or rearrangement of the terms of any or all of the Guaranteed Obligations and/or any of the Loan Documents, including material alterations of the terms of payment (including changes in maturity date(s) and interest rate(s)) or performance (including changes with respect to the construction of the Improvements) or any other terms thereof, or any waiver, termination, or release of, or consent to departure from, any of the Loan Documents or any other guaranty of any or all of the Guaranteed Obligations, or any adjustment, indulgence, forbearance, or compromise that may be granted from time to time by Lender to Borrower or any other Person at any time liable for the payment or performance of any or all of the Guaranteed Obligations;
- (x) any neglect, lack of diligence, delay, omission, failure, or refusal of Lender to take or prosecute (or in taking or prosecuting) any action for the collection or enforcement of any of the Guaranteed Obligations, or to foreclose or take or prosecute any action to foreclose (or in foreclosing or taking or prosecuting any action to foreclose) upon any security therefor, or to exercise (or in exercising) any other right or power with respect to any security therefor, or to take or prosecute (or in taking or prosecuting) any action in connection with any Loan Document, or any failure to sell or otherwise dispose of in a commercially reasonable manner any collateral securing any or all of the Guaranteed Obligations;

- any failure of Lender to notify Guarantor of any (xi) creation, renewal, extension, rearrangement, modification, supplement, subordination, or assignment of the Guaranteed Obligations or any part thereof, or of any Loan Document, or of any release of or change in any security, or of the occurrence or existence of any Default or Event of Default, or of any other action taken or refrained from being taken by Lender against Borrower or any security or other recourse, or of any new agreement between Lender and Borrower, it being understood that Lender shall not be required to give Guarantor any notice of any kind under any circumstances with respect to or in connection with the Guaranteed Obligations, any and all rights to notice Guarantor may have otherwise had being hereby waived by Guarantor, and Guarantor shall be responsible for obtaining for itself information regarding Borrower, and any collateral, including any changes in the business or financial condition of Borrower or any collateral, and Guarantor acknowledges and agrees that Lender shall have no duty to notify Guarantor of any information which Lender may have concerning Borrower or any collateral;
- (xii) the existence of any claim, counterclaim, set-off or other right that Guarantor may at any time have against Borrower, Lender, or any other Person, whether or not arising in connection with this Guaranty, the Note, the Loan Agreement, the Environmental Agreement or any other Loan Document;
- (xiii) the unenforceability of all or any part of the Guaranteed Obligations against Borrower, whether because the Guaranteed Obligations exceed the amount permitted by Law or violate any usury law, or because the Persons creating the Guaranteed Obligations acted in excess of their authority, or because of a lack of validity or enforceability of or defect or deficiency in any of the Loan Documents, or because Borrower has any valid defense, claim or offset with respect thereto, or because Borrower's obligation ceases to exist by operation of Law, or because of any other reason or circumstance, it being agreed that Guarantor shall remain liable hereon regardless of whether Borrower or any other Person be found not liable on the Guaranteed Obligations, or any part thereof, for any reason (and regardless of any joinder of Borrower or any other party in any action to obtain payment or performance of any or all of the Guaranteed Obligations);
- (xiv) any order, ruling or plan of reorganization emanating from proceedings under Title 11 of the United States Code with respect to Borrower or any other Person, including any extension, reduction, composition, or other alteration of the

Guaranteed Obligations, whether or not consented to by Lender, or any action taken or omitted by Lender in any such proceedings, including any election to have Lender's claim allowed as being secured, partially secured or unsecured, any extension of credit by Lender in any such proceedings or the taking and holding by Lender of any security for any such extension of credit;

- (xv) any other condition, event, omission, action that would in the absence of this paragraph result in the release or discharge of the Guarantor from the performance or observance of any obligation, covenant or agreement contained in this Guaranty or any other agreement;
- (xvi) any early termination of any of the Guaranteed Obligations;
- (xvii) Lender's enforcement or forbearance from enforcement of the Guaranteed Obligations on a net or gross basis; or
- (xviii) any liability, irregularity or unenforceability in whole or in part (including with respect to any netting provision) of any Swap Contract or any confirmation, instrument or agreement required thereunder or related thereto, or any transaction entered into thereunder, or any limitation on the liability of Borrower thereunder or any limitation on the method or terms of payment thereunder which may now or hereafter be caused or imposed in any manner whatsoever.
- (b) In the event any payment by Borrower or any other Person to Lender is held to constitute a preference, fraudulent transfer or other voidable payment under any bankruptcy, insolvency or similar Law, or if for any other reason Lender is required to refund such payment or pay the amount thereof to any other party, such payment by Borrower or any other party to Lender shall not constitute a release of Guarantor from any liability hereunder, and this Guaranty shall continue to be effective or shall be reinstated (notwithstanding any prior release, surrender or discharge by Lender of this Guaranty or of Guarantor), as the case may be, with respect to, and this Guaranty shall apply to, any and all amounts so refunded by Lender or paid by Lender to another Person (which amounts shall constitute part of the Guaranteed Obligations), and any interest paid by Lender and any attorneys' fees, costs and expenses paid or incurred by Lender in connection with any such event.
- (c) It is the intent of Guarantor and Lender that the obligations and liabilities of Guarantor hereunder are absolute, irrevocable and unconditional under any and all circumstances and that until the Guaranteed Obligations are fully and finally paid and performed, and not subject to refund or disgorgement, the obligations and liabilities of Guarantor hereunder shall not be discharged or released, in whole or in part, by any act or occurrence that might, but for the provisions of this Guaranty, be deemed a legal or equitable discharge or release of a guarantor.

- (d) Guarantor's obligations shall not be affected, impaired, lessened or released by loans, credits or other financial accommodations now existing or hereafter advanced by Lender to Borrower in excess of the Guaranteed Obligations. All payments, repayments and prepayments of the Loan, whether voluntary or involuntary, received by Lender from Borrower, any other Person or any other source (other than from Guarantor pursuant to a demand by Lender hereunder), and any amounts realized from any collateral for the Loan, shall be deemed to be applied first to any portion of the Loan which is not covered by this Guaranty, and last to the Guaranteed Obligations, and this Guaranty shall bind Guarantor to the extent of any Guaranteed Obligations that may remain owing to Lender. Lender shall have the right to apply any sums paid by Guarantor to any portion of the Loan in Lender's sole and absolute discretion.
- (e) If acceleration of the time for payment of any amount payable by Borrower under the Note, the Loan Agreement, any other Loan Document or any Swap Contract is stayed or delayed by any Law or tribunal, all such amounts shall nonetheless be payable by Guarantor on demand by Lender.
- Guarantor further waives: (i) any defense to the recovery by Lender against Guarantor of any deficiency or otherwise to the enforcement of this Guaranty or any security for this Guaranty based upon Lender's election of any remedy against Guarantor or Borrower, including the defense to enforcement of this Guaranty (the so-called "Gradsky" defense) which, absent this waiver, Guarantor would have by virtue of an election by Lender to conduct a nonjudicial foreclosure sale (also known as a "trustee's sale") of any real property security for the Indebtedness, it being understood by Guarantor that any such non-judicial foreclosure sale will destroy, by operation of California Code of Civil Procedure ("CCP") Section 580d, all rights of any party to a deficiency judgment against Borrower and, as a consequence, will destroy all rights that Guarantor would otherwise have (including the right of subrogation, the right of reimbursement, and the right of contribution) to proceed against Borrower, (ii) any defense or benefits that may be derived from CCP Sections 580a, 580b, 580d or 726, or comparable provisions of the laws of any other jurisdiction and all other anti-deficiency and one form of action defenses under the laws of California and any other jurisdiction; and (iii) any right to a fair value hearing under CCP Section 580a, or any other similar law, to determine the size of any deficiency owing (for which Guarantor would be liable hereunder) following a non-judicial foreclosure sale. Nothing in this subsection (f) shall operate to change, waive or affect the provisions of Section 19 hereof.
- (g) Without limiting any other provision of this Guaranty, Guarantor waives all rights and defenses that Guarantor may have because the Guaranteed Obligations are secured by real property. This means, among other things:
 - (i) That Lender may collect from Guarantor without first foreclosing on any real or personal property collateral pledged by Borrower; and
 - (ii) If Lender forecloses on any real property collateral pledged by Borrower: (A) the amount of the Guaranteed Obligations may be reduced only by the price for which that collateral is sold at the foreclosure sale, even if the collateral is

worth more than the sale price; and (B) Lender may collect from Guarantor even if Lender, by foreclosing on the real property collateral, has destroyed any right Guarantor may have to collect from Borrower.

This is an unconditional and irrevocable waiver of any rights and defenses that Guarantor may have because the Guaranteed Obligations are secured by real property. These rights and defenses include, but are not limited to, any rights or defenses based upon CCP Sections 580a, 580b, 580d, or 726.

- (h) Guarantor waives all rights and defenses arising out of an election of remedies by Lender, even though that election of remedies, such as a nonjudicial foreclosure with respect to security for the Guaranteed Obligations, has destroyed Guarantor's rights of subrogation and reimbursement against Borrower by operation of CCP Section 580d or otherwise.
- (i) Guarantor waives Guarantor's rights of subrogation and reimbursement, including (i) any defenses Guarantor may have by reason of an election of remedies by Lender, and (ii) any rights or defenses Guarantor may have by reason of protection afforded to Borrower with respect to the Guaranteed Obligations pursuant to the anti-deficiency or other laws of California limiting or discharging Borrower's obligations, including CCP Sections 580a, 580b, 580d or 726.
- (j) Guarantor waives notice of acceptance of this Guaranty, any rights, defenses and benefits that may be derived from Sections 2787 to 2855, inclusive, of the California Civil Code or comparable provisions of the laws of any other jurisdiction, and all other suretyship defenses Guarantor would otherwise have under the laws of California or any other jurisdiction.
- (k) No provision or waiver in this Guaranty shall be construed as limiting the generality of any other provision or waiver contained in this Guaranty. All the waivers contained herein are irrevocable and unconditional and are intentionally and freely made by Guarantor.

Section 5. Subordination.

If, for any reason whatsoever, Borrower is now or hereafter becomes indebted to Guarantor:

- (a) such indebtedness and all interest thereon and all liens, security interests and rights now or hereafter existing with respect to property of Borrower securing such indebtedness shall, at all times, be subordinate in all respects to the Guaranteed Obligations and to all liens, security interests and rights now or hereafter existing to secure the Guaranteed Obligations;
- (b) Guarantor shall not be entitled to enforce or receive payment, directly or indirectly, of any such indebtedness of Borrower to Guarantor until the Guaranteed Obligations have been fully and finally paid and performed; provided, however, that so long as no Default shall have occurred and be continuing, Guarantor shall not be prohibited from receiving such (i) reasonable management fees or reasonable salary from Borrower as Lender may find acceptable from time to time in its sole and absolute discretion, and (ii) distributions from Borrower in

an amount equal to any income taxes imposed on Guarantor which are attributable to Borrower's income from the Property;

- (c) Guarantor hereby assigns and grants to Lender a security interest in all such indebtedness and security therefor, if any, of Borrower to Guarantor now existing or hereafter arising, including any dividends and payments pursuant to debtor relief or insolvency proceedings referred to below. In the event of receivership, bankruptcy, reorganization, arrangement or other debtor relief or insolvency proceedings involving Borrower as debtor, Lender shall have the right to prove its claim in any such proceeding so as to establish its rights hereunder and shall have the right to receive directly from the receiver, trustee or other custodian (whether or not an Event of Default shall have occurred or be continuing under any of the Loan Documents), dividends and payments that are payable upon any obligation of Borrower to Guarantor now existing or hereafter arising, and to have all benefits of any security therefor, until the Guaranteed Obligations have been fully and finally paid and performed. If, notwithstanding the foregoing provisions, Guarantor should receive any payment, claim or distribution that is prohibited as provided above in this Section 5, Guarantor shall pay the same to Lender immediately, Guarantor hereby agreeing that it shall receive the payment, claim or distribution in trust for Lender and shall have absolutely no dominion over the same except to pay it immediately to Lender; and
- (d) Guarantor shall promptly upon request of Lender from time to time execute such documents and perform such acts as Lender may require to evidence and perfect its interest and to permit or facilitate exercise of its rights under this Section 5, including execution and delivery of proofs of claim, further assignments and security agreements, and delivery to Lender of any promissory notes or other instruments evidencing indebtedness of Borrower to Guarantor. All promissory notes, accounts receivable ledgers or other evidences, now or hereafter held by Guarantor, of obligations of Borrower to Guarantor shall contain a specific written notice thereon that the indebtedness evidenced thereby is subordinated under and is subject to the terms of this Guaranty.

Section 6. Other Liability of Guarantor or Borrower.

If Guarantor is or becomes liable, by endorsement or otherwise, for any indebtedness owing by Borrower to Lender other than under this Guaranty, such liability shall not be in any manner impaired or affected hereby, and the rights of Lender hereunder shall be cumulative of any and all other rights that Lender may have against Guarantor. If Borrower is or becomes indebted to Lender for any indebtedness other than or in excess of the Guaranteed Obligations, any payment received or recovery realized upon such other indebtedness of Borrower to Lender may be applied to such other indebtedness. This Guaranty is independent of (and shall not be limited by) any other guaranty now existing or hereafter given. Further, Guarantor's liability under this Guaranty is in addition to any and all other liability Guarantor may have in any other capacity, including, if applicable, its capacity as a general partner.

Section 7. Lender Assigns; Disclosure of Information.

This Guaranty is for the benefit of Lender and Lender's successors and assigns, and in the event of an assignment of the Guaranteed Obligations, or any part thereof, the rights and benefits hereunder, to the extent applicable to the Guaranteed Obligations so assigned, may be transferred with such Guaranteed Obligations. Guarantor waives notice of any transfer or assignment of the Guaranteed Obligations or any part thereof. Lender may sell or offer to sell the Loan or interests therein to one or more assignees or participants. Guarantor shall execute, acknowledge and deliver any and all instruments reasonably requested by Lender in connection therewith, and to the extent, if any, specified in any such assignment or participation, such assignee(s) or participant(s) shall have the same rights and benefits with respect to the Loan Documents as such Person(s) would have if such Person(s) were Lender hereunder. Lender may disclose to any such assignee or participant or prospective assignee or participant, to Lender's affiliates, including Banc of America Securities LLC, to any regulatory body having jurisdiction over Lender and to any other parties as necessary or appropriate in Lender's reasonable judgment, any information Lender now has or hereafter obtains pertaining to the Guaranteed Obligations, this Guaranty, or Guarantor, including information regarding any security for the Guaranteed Obligations or for this Guaranty, and/or credit or other information on Guarantor and/or any other Person liable, directly or indirectly, for any part of the Guaranteed Obligations.

Section 8. Binding Effect; Joint and Several Liability.

This Guaranty is binding not only on Guarantor, but also on Guarantor's heirs, personal representatives, successors and assigns. Upon the death of Guarantor, if Guarantor is a natural person, this Guaranty shall continue against Guarantor's estate as to all of the Guaranteed Obligations, including that portion incurred or arising after the death of Guarantor and shall be provable in full against Guarantor's estate, whether or not the Guaranteed Obligations are then due and payable. If this Guaranty is signed by more than one Person, then all of the obligations of Guarantor arising hereunder shall be jointly and severally binding on each of the undersigned, and their respective heirs, personal representatives, successors and assigns, and the term "Guarantor" shall mean all of such Persons and each of them individually.

Section 9. Governing Law.

The validity, enforcement, and interpretation of this Guaranty, shall for all purposes be governed by and construed in accordance with the laws of the State of Arizona (without regard to its conflicts of law principles) and applicable United States federal law, and is intended to be performed in accordance with, and only to the extent permitted by, such laws. All obligations of Guarantor hereunder are payable and performable at the place or places where the Guaranteed Obligations are payable and performable.

Section 10. Invalidity of Certain Provisions.

If any provision of this Guaranty or the application thereof to any Person or circumstance shall, for any reason and to any extent, be declared to be invalid or unenforceable, neither the remaining provisions of this Guaranty nor the application of such provision to any other Person or circumstance shall be affected thereby, and the remaining provisions of this Guaranty, or the

applicability of such provision to other Persons or circumstances, as applicable, shall remain in effect and be enforceable to the maximum extent permitted by applicable Law.

Section 11. Costs and Expenses of Enforcement.

Guarantor agrees to pay to Lender on demand all costs and expenses incurred by Lender in seeking to enforce Lender's rights and remedies under this Guaranty, including court costs, costs of alternative dispute resolution, and reasonable attorneys' fees, whether or not suit is filed or other proceedings are initiated hereon. All such costs and expenses incurred by Lender shall constitute a portion of the Guaranteed Obligations hereunder, shall be subject to the provisions hereof with respect to the Guaranteed Obligations and shall be payable by Guarantor on demand by Lender.

Section 12. No Usury.

It is not the intention of Lender or Guarantor to obligate Guarantor to pay interest in excess of that lawfully permitted to be paid by Guarantor under applicable Law. Should it be determined that any portion of the Guaranteed Obligations or any other amount payable by Guarantor under this Guaranty constitutes interest in excess of the maximum amount of interest that Guarantor, in Guarantor's capacity as guarantor, may lawfully be required to pay under applicable Law, the obligation of Guarantor to pay such interest shall automatically be limited to the payment thereof in the maximum amount so permitted under applicable Law. The provisions of this Section shall override and control all other provisions of this Guaranty and of any other agreement between Guarantor and Lender.

Section 13. Representations, Warranties, and Covenants of Guarantor.

A. Until the Guaranteed Obligations are paid and performed in full and each and every term, covenant and condition of this Guaranty is fully performed, Guarantor hereby represents, warrants, and covenants that: (a) Guarantor has a financial interest in Borrower and will derive a material and substantial benefit, directly or indirectly, from the making of the Loan to Borrower and from the making of this Guaranty by Guarantor; (b) this Guaranty is duly authorized and valid, and is binding upon and enforceable against Guarantor; (c) Guarantor is not, and the execution, delivery and performance by Guarantor of this Guaranty will not cause Guarantor to be, in violation of or in default with respect to any law or in default (or at risk of acceleration of indebtedness) under any agreement or restriction by which Guarantor is bound or affected; (d) unless Guarantor is a natural person, Guarantor is duly organized, validly existing, and in good standing under the laws of the state of its organization and has full power and authority to enter into and perform this Guaranty; (e) there is no litigation pending or, to the knowledge of Guarantor, threatened by or before any tribunal against or affecting Guarantor; (f) all financial statements and information heretofore furnished to Lender by Guarantor do, and all financial statements and information hereafter furnished to Lender by Guarantor will, fully and accurately present the condition (financial or otherwise) of Guarantor as of their dates and the results of Guarantor's operations for the periods therein specified, and, since the date of the most recent financial statements of Guarantor heretofore furnished to Lender, no material adverse change has occurred in the financial condition of Guarantor, nor, except as heretofore disclosed in writing to Lender, has Guarantor incurred any material liability, direct or indirect, fixed or

contingent; (g) after giving effect to this Guaranty, Guarantor is solvent, is not engaged or about to engage in business or a transaction for which the property of Guarantor is an unreasonably small capital, and does not intend to incur or believe that it will incur debts that will be beyond its ability to pay as such debts mature; (h) Guarantor has read and fully understands the provisions contained in the Note, the Loan Agreement, the Deed of Trust, the Environmental Agreement and the other Loan Documents. Guarantor further represents, warrants and covenants that if any Swap Contract shall at any time be in effect, (x) Guarantor has received and examined copies of each such Swap Contract, the observance and performance of which by Borrower is hereby guaranteed; (y) Guarantor will benefit from Lender's entering into each such Swap Contract and any transaction thereunder with Borrower, and Guarantor has determined that the execution and delivery by Guarantor of this Guaranty are necessary and convenient to the conduct, promotion and attainment of the business of Guarantor; and (z) Lender has no duty to determine whether any Swap Contract, or any other transaction relating to or arising under any Swap Contract, will be or has been entered into by Borrower for purposes of hedging interest rate, currency exchange rate, or other risks arising in its businesses or affairs and not for purposes of speculation, or is otherwise inappropriate for Borrower. Guarantor's representations, warranties and covenants are a material inducement to Lender to enter into the other Loan Documents and any Swap Contract and shall survive the execution hereof and any bankruptcy. foreclosure, transfer of security or other event affecting Borrower, Guarantor, any other party, or any security for all or any part of the Guaranteed Obligations.

B. Guarantor hereby agrees as follows:

- (i) to maintain during the life of the Loan at the end of each fiscal quarter, unpledged cash, cash equivalents as determined by generally accepted accounting principles, and marketable securities ("Liquidity") of not less than Two Million And No/100 Dollars (\$2,000,000.00);
- (ii) to maintain at the end of each fiscal quarter on a consolidated basis a Tangible Net Worth equal to at least One Hundred Million And No/100 Dollars (\$100,000,000.00). "Tangible Net Worth" means the gross book value of Guarantor's assets (excluding goodwill, patents, trademarks, trade names, organization expense, treasury stock, unamortized debt discount and expense, deferred research and development costs, deferred marketing expenses, and other like intangibles), plus debt subordinated to Lender (limited to a maximum of Thirty Million And No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form) less Total Liabilities, including but not limited to accrued and deferred income taxes, and any reserves against assets; and
- (iii) to maintain at the end of each fiscal quarter on a consolidated basis a ratio of Total Liabilities to Tangible Net Worth not exceeding 6.0 to 1.0 through December 31, 2007, and 5.0 to 1.0 thereafter. "Total Liabilities" means the sum of current

liabilities plus long term liabilities plus joint venture debt less debt subordinated to Lender (limited to a maximum of Thirty Million and No/100 Dollars [\$30,000,000.00]) in a manner acceptable to Lender (using Lender's standard form).

Section 14. Notices.

All notices, requests, consents, demands and other communications required or which any party desires to give hereunder or under any other Loan Document shall be in writing and, unless otherwise specifically provided in such other Loan Document, shall be deemed sufficiently given or furnished if delivered by personal delivery, by nationally recognized overnight courier service, or by certified United States mail, postage prepaid, addressed to the party to whom directed at the addresses specified in this Guaranty (unless changed by similar notice in writing given by the particular party whose address is to be changed) or by facsimile. Any such notice or communication shall be deemed to have been given either at the time of personal delivery or, in the case of courier or mail, as of the date of first attempted delivery at the address and in the manner provided herein, or, in the case of facsimile, upon receipt; provided that service of a notice required by any applicable statute shall be considered complete when the requirements of that statute are met. Notwithstanding the foregoing, no notice of change of address shall be effective except upon actual receipt. This Section shall not be construed in any way to affect or impair any waiver of notice or demand provided in this Guaranty or in any other Loan Document or to require giving of notice or demand to or upon any Person in any situation or for any reason.

Section 15. <u>Cumulative Rights</u>.

All of the rights and remedies of Lender under this Guaranty and the other Loan Documents are cumulative of each other and of any and all other rights at law or in equity, and the exercise by Lender of any one or more of such rights and remedies shall not preclude the simultaneous or later exercise by Lender of any or all such other rights and remedies. No single or partial exercise of any right or remedy shall exhaust it or preclude any other or further exercise thereof, and every right and remedy may be exercised at any time and from time to time. No failure by Lender to exercise, nor delay in exercising, any right or remedy shall operate as a waiver of such right or remedy or as a waiver of any Event of Default. No notice to or demand on Guarantor in any case shall of itself entitle Guarantor to any other or further notice or demand in similar or other circumstances. No provision of this Guaranty or any right or remedy of Lender with respect hereto, or any default or breach, can be waived, nor can this Guaranty or Guarantor be released or discharged in any way or to any extent, except specifically in each case by a writing intended for that purpose (and which refers specifically to this Guaranty) executed and delivered by Lender to Guarantor.

Section 16. Term of Guaranty.

This Guaranty shall continue in effect until all the Guaranteed Obligations and all of the obligations of Guarantor to Lender under this Guaranty are fully and finally paid, performed and discharged and are not subject to any bankruptcy preference period or any other disgorgement.

Section 17. Financial Statements.

Guarantor agrees to provide to Lender, as and when required, the financial statements and other financial information required to be delivered to Lender with respect to Guarantor pursuant to the terms of the Loan Agreement and the other Loan Documents, in the form and detail required by the Loan Documents. Guarantor also agrees to provide to Lender such other and further financial information with respect to Guarantor as Lender shall from time to time request.

Section 18. Subrogation.

Guarantor shall not have any equitable, contractual or other right of subrogation under any of the Loan Documents or any right to participate in any security for the Guaranteed Obligations or any right to reimbursement, exoneration, contribution, indemnification or any similar rights, until the Guaranteed Obligations have been fully and finally paid, performed and discharged in accordance with Section 16 above, and Guarantor hereby waives all of such rights.

Section 19. <u>Time of Essence</u>.

Time shall be of the essence in this Guaranty with respect to all of Guarantor's obligations hereunder.

Section 20. Entire Agreement; Counterparts; Construction.

This Guaranty embodies the entire agreement between Lender and Guarantor with respect to the guaranty by Guarantor of the Guaranteed Obligations. This Guaranty supersedes all prior agreements and understandings, if any, with respect to the guaranty by Guarantor of the Guaranteed Obligations. This Guaranty shall be effective upon execution by Guarantor and delivery to Lender. This Guaranty may not be modified, amended or superseded except in a writing signed by Lender and Guarantor referencing this Guaranty by its date and specifically identifying the portions hereof that are to be modified, amended or superseded. This Guaranty has been executed in a number of identical counterparts, each of which shall be deemed an original for all purposes and all of which constitute, collectively, one agreement. As used herein, the words "include" and "including" shall be interpreted as if followed by the words "without limitation."

Section 21. <u>Dispute Resolution</u>.

(a) Arbitration. Except to the extent expressly provided below, any controversy, claim or dispute between or among the parties hereto, including any such controversy, claim or dispute arising out of or relating to (i) this Guaranty, (ii) any other Loan Document, (iii) any related agreements or instruments, or (iv) the transaction contemplated herein or therein (including any claim based on or arising from an alleged personal injury or business tort) collectively, a "Dispute"), shall, upon the request of either party, be determined by binding arbitration in accordance with the Federal Arbitration Act, Title 9, United States Code (or if not applicable, the applicable state law), the then-current rules for arbitration of financial services disputes of the American Arbitration Act ("AAA") and the "Special Rules" set forth below. In the event of any inconsistency, the Special Rules shall control. The filing of a court action is not intended to constitute a waiver of the right of Guarantor or Lender, including the suing party,

thereafter to require submittal of the Dispute to arbitration. Any party to this Guaranty may bring an action, including a summary or expedited proceeding, to compel arbitration of any Dispute in any court having jurisdiction over such action. For the purposes of this Dispute Resolution Section only, the terms "party" and "parties" shall include any parent corporation, subsidiary or affiliate of Lender involved in the servicing, management or administration of any obligation described in or evidenced by this Guaranty, together with the officers, employees, successors and assigns of each of the foregoing.

(b) Special Rules.

- (i) The arbitration shall be conducted in any U.S. state where real or tangible personal property collateral is located, or if there is no such collateral in the City and County where Lender is located pursuant to its address for notice purposes in this Guaranty.
- (ii) The arbitration shall be administered by AAA, who will appoint an arbitrator. If AAA is unwilling or unable to administer or legally precluded from administering the arbitration, or if AAA is unwilling or unable to enforce or legally precluded from enforcing any and all provisions of this Dispute Resolution Section, then any party to this Guaranty may substitute another arbitration organization that has similar procedures to AAA and that will observe and enforce any and all provisions of this Dispute Resolution Section. All Disputes shall be determined by one arbitrator; however, if the amount in controversy in a Dispute exceeds Five Million Dollars (\$5,000,000), upon the request of any party, the Dispute shall be decided by three arbitrators (for purposes of this Guaranty, referred to collectively as the "arbitrator").
- (iii) All arbitration hearings will be commenced within ninety (90) days of the demand for arbitration and completed within ninety (90) days from the date of commencement; provided, however, that upon a showing of good cause, the arbitrator shall be permitted to extend the commencement of such hearing for up to an additional sixty (60) days.
- (iv) The judgment and the award, if any, of the arbitrator shall be issued within thirty (30) days of the close of the hearing. The arbitrator shall provide a concise written statement setting forth the reasons for the judgment and for the award, if any. The arbitration award, if any, may be submitted to any court having jurisdiction to be confirmed and enforced, and such confirmation and enforcement shall not be subject to arbitration.
- (v) The arbitrator will give effect to statutes of limitations and any waivers thereof in determining the disposition

of any Dispute and may dismiss one or more claims in the arbitration on the basis that such claim or claims is or are barred. For purposes of the application of the statute of limitations, the service on AAA under applicable AAA rules of a notice of Dispute is the equivalent of the filing of a lawsuit.

- (vi) Any dispute concerning this arbitration provision, including any such dispute as to the validity or enforceability of this provision, or whether a Dispute is arbitrable, shall be determined by the arbitrator; provided, however, that the arbitrator shall not be permitted to vary the express provisions of these Special Rules or the Reservations of Rights in <u>subsection</u> (d) below.
- (vii) The arbitrator shall have the power to award legal fees and costs pursuant to the terms of this Guaranty.
- (viii) The arbitration will take place on an individual basis without reference to, resort to, or consideration of any form of class or class action.
- (c) <u>Judicial Reference</u>. If the Dispute arises from or relates to an obligation to Lender secured by real property located in the State of California, unless both Guarantor and Lender consent to submission of the Dispute to arbitration to be conducted as provided in subsections (a) and (b) above, the Dispute shall be resolved by judicial reference pursuant to CCP Sections 638 et seq. This provision constitutes a reference agreement between or among the parties as provided in CCP Section 638. The referee(s) shall be chosen by the parties under the auspices of AAA in the same manner as arbitrators are selected in proceedings administered under the AAA rules and procedures for the arbitration of financial services disputes. The referee (or the presiding referee of the panel) must be an active attorney or a retired judge. The award that results from the decision of the referee(s) shall be entered as a judgment in the court that appointed the referee, in accordance with the provisions of CCP Sections 644 and 645.
- (d) Reservations of Rights. Nothing in this Guaranty shall be deemed to (i) limit the applicability of any otherwise applicable statutes of limitation and any waivers contained in this Guaranty, or (ii) apply to or limit the right of Lender (A) to exercise self help remedies such as (but not limited to) setoff, or (B) to foreclose judicially or nonjudicially against any real or personal property collateral, or to exercise judicial or nonjudicial power of sale of rights, (C) to obtain from a court provisional or ancillary remedies such as (but not limited to) injunctive relief, writ of possession, prejudgment attachment, or the appointment of a receiver, or (D) to pursue rights against a party to this Guaranty in a third-party proceeding in any action brought against Lender in a state, federal or international court, tribunal or hearing body (including actions in specialty courts, such as bankruptcy and patent courts). Lender may exercise the rights set forth in clauses (A) through (D), inclusive, before, during or after the pendency of any arbitration or judicial reference proceeding brought pursuant to this Guaranty. Neither the exercise of self help remedies nor the institution or maintenance of an action for foreclosure or provisional or ancillary remedies shall constitute a waiver of the right of any party, including the claimant in

any such action, to arbitrate, or submit to judicial reference, the merits of the Dispute occasioning resort to such remedies. No provision in the Loan Documents regarding submission to jurisdiction and/or venue in any court is intended or shall be construed to be in derogation of the provisions in any Loan Document for arbitration or judicial reference of any Dispute.

- (e) <u>Conflicting Provisions for Dispute Resolution</u>. If there is any conflict between the terms, conditions and provisions of this Section and those of any other provision or agreement for arbitration, judicial reference or dispute resolution, the terms, conditions and provisions of this Section shall prevail as to any Dispute arising out of or relating to (i) this Guaranty, (ii) any other Loan Document, (iii) any related agreements or instruments, or (iv) the transaction contemplated herein or therein (including any claim based on or arising from an alleged personal injury or business tort). In any other situation, if the resolution of a given Dispute is specifically governed by another provision or agreement for arbitration, judicial reference or other dispute resolution, the other provision or agreement shall prevail with respect to said Dispute.
- (f) <u>Jury Trial Waiver in Arbitration</u>. By agreeing to this Section, the parties irrevocably and voluntarily waive any right they may have to a trial by jury in respect of any Dispute.

Section 22. Forum.

Guarantor hereby irrevocably submits generally and unconditionally for itself and in respect of its property to the jurisdiction of any state court or any United States federal court sitting in the State specified in the governing law section of this Guaranty and to the jurisdiction of any state court or any United States federal court sitting in the state in which any of the Property is located, over any Dispute. Guarantor hereby irrevocably waives, to the fullest extent permitted by Law, any objection that Guarantor may now or hereafter have to the laying of venue in any such court and any claim that any such court is an inconvenient forum. Guarantor hereby agrees and consents that, in addition to any methods of service of process provided for under applicable law, all service of process in any such suit, action or proceeding in any state court or any United States federal court sitting in the state specified in the governing law section of this Guaranty may be made by certified or registered mail, return receipt requested, directed to Guarantor at its address for notice set forth in this Guaranty, or at a subsequent address of which Lender received actual notice from Guarantor in accordance with the notice section of this Guaranty, and service so made shall be complete five (5) days after the same shall have been so mailed. Nothing herein shall affect the right of Lender to serve process in any manner permitted by Law or limit the right of Lender to bring proceedings against Guarantor in any other court or jurisdiction.

Section 23. WAIVER OF JURY TRIAL.

WITHOUT INTENDING IN ANY WAY TO LIMIT THE PARTIES' AGREEMENT TO RESOLVE ANY "DISPUTE" (FOR PURPOSES OF THIS SECTION, AS DEFINED ABOVE), BY JUDICIAL REFERENCE OR ARBITRATION AS SET FORTH IN THIS GUARANTY, GUARANTOR AND LENDER WAIVE TRIAL BY JURY IN RESPECT OF ANY SUCH "DISPUTE" AND ANY ACTION ON SUCH

"DISPUTE." THIS WAIVER SHALL APPLY TO THE EXTENT ANY "DISPUTE" IS NOT SUBMITTED TO JUDICIAL REFERENCE OR ARBITRATION. OR IS DEEMED BY THE ARBITRATOR, REFEREE OR ANY COURT WITH JURISDICTION TO BE NOT REQUIRED TO BE DETERMINED BY JUDICIAL REFERENCE OR ARBITRATION, OR NOT SUSCEPTIBLE OF BEING SO DETERMINED. WAIVER IS KNOWINGLY, WILLINGLY AND VOLUNTARILY MADE BY GUARANTOR AND LENDER, AND GUARANTOR AND LENDER HEREBY REPRESENT THAT NO REPRESENTATIONS OF FACT OR OPINION HAVE BEEN MADE BY ANY PERSON OR ENTITY TO INDUCE THIS WAIVER OF TRIAL BY JURY OR TO IN ANY WAY MODIFY OR NULLIFY ITS EFFECT. THIS PROVISION IS A MATERIAL INDUCEMENT FOR THE PARTIES ENTERING INTO THE LOAN DOCUMENTS. GUARANTOR AND LENDER ARE EACH HEREBY AUTHORIZED TO FILE A COPY OF THIS SECTION IN ANY PROCEEDING AS CONCLUSIVE EVIDENCE OF THIS WAIVER OF JURY TRIAL. GUARANTOR FURTHER REPRESENTS AND WARRANTS THAT IT HAS BEEN REPRESENTED IN THE SIGNING OF THIS GUARANTY AND IN THE MAKING OF THIS WAIVER BY INDEPENDENT LEGAL COUNSEL, OR HAS HAD THE OPPORTUNITY TO BE REPRESENTED BY INDEPENDENT LEGAL COUNSEL SELECTED OF ITS OWN FREE WILL, AND THAT IT HAS HAD THE OPPORTUNITY TO DISCUSS THIS WAIVER WITH COUNSEL.

IN WITNESS WHEREOF, Guarantor has duly executed this Guaranty as of the date first written above.

Address of Guarantor:

GUARANTOR:

2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

OPUS WEST CORPORATION, a Minnesota corporation

Name:_ Title:

Charles Vogel

Address of Lender:

Bank of America, N.A.

Commercial Real Estate Banking
Group, AZ1-200-22-17

201 East Washington Street, 22nd Floor
Phoenix, Arizona 85004-2343

Attention: N. Alonzo

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 20 of 33

ATTACHMENT 9

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 21 of 33



One Renaissance Square Two North Central Avenue Phoenix, Arizona 85004-2391 Tel 602.229.5200 Fax 602.229.5690 www.quarles.com Attorneys at Law in: Phoenix and Tucson, Arizona Naples, Florida Chicago, Illinois Milwaukee and Madison, Wisconsin

April 6, 2009

DEFAULT NOTICE

VIA UPS OVERNIGHT

Shoppes At Chino Hills, Inc.
2555 East Camelback, Suite 800
Phoenix, Arizona 85016-9267
Attn: Senior Vice President, Real Estate Finance and Sales

Opus West Corporation 2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

RE: \$110,000,000 Loan (the "Loan") made by Bank of America, N.A., a national banking association, individually as a Lender and as Administrative Agent ("Bank of America"), and certain other financial institutions as Lenders (together with Bank of America, the "Lenders") to Shoppes At Chino Hills, Inc., a Minnesota corporation ("Borrower")

Ladies and Gentlemen:

The purpose of this letter is to provide you with notice of the occurrence of a default (hereinafter, the "Default") under: (i) Section 2.11 of the Construction Loan Agreement dated May 16, 2007 between Borrower and Lenders, as supplemented by two letter agreements dated May 16, 2007, and as amended by that certain letter agreement dated March 21, 2008 (the "Loan Agreement"), and (ii) the terms of certain Construction Deed Of Trust Notes dated as of even date therewith executed by Borrower payable to the order of each Lender in the amount of its respective Commitment (the "Notes"). The Loan Agreement, Notes, and all other agreements, documents and instruments evidencing, securing or otherwise relating to the Loan are sometimes referred to individually and collectively herein as the "Loan Documents". Borrower is in default as the result of Borrower's failure to make the installment payment of interest due and payable on April 1, 2009 as required under the Loan Agreement and the Notes. If payment is not received in good funds on or before April 8, 2009, then such failure will constitute a Default under the Loan Agreement and Notes. Late charges (as described in and calculated according to Section 2.10 of the Loan Agreement) will apply to the defaulted amounts. Interest at the Past Due Rate (provided in and calculated according to Section 2.8 of the Loan Agreement) commenced accruing from and after April 2, 2009 on any amounts outstanding as of such date.

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 22 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation April 6, 2009 Page 2

Notwithstanding the existence of the above-described Default, or any additional defaults that may arise under the Loan Documents, Bank of America, on behalf of itself and the other Lenders, reserves the right to elect whether or not to fund any presently pending or future draw requests during the existence of any default or Event of Default. To the extent Bank of America or any of the other Lenders elects to fund any such draw request, neither such election nor the actual funding thereof does or shall: constitute a commitment to lend or to modify the Loan; constitute an offer to extend credit or modify the existing Loan terms; constitute a waiver by any Lender of the Default or performance or cure of the Default by Borrower; constitute an amendment of any term or condition of any Loan Document; create an obligation of any Lender to agree to, or to negotiate or consider an agreement to, any waiver of any obligation or default by Borrower under any Loan; or directly or indirectly impair or otherwise affect any of the Lenders' rights, interests or remedies with respect to the Loan.

Bank of America, on behalf of itself and the other Lenders, hereby reserves the right to accept payments made after the date of this letter even though such payments are not offered as, or are insufficient to, cure the Default described herein or reinstate the Loan pursuant to the provisions of the law governing the jurisdiction in which the project is located. Each of the Lenders may apply such payments in the order provided in the Loan Agreement or other Loan Documents.

In the event the foregoing Default is not cured on or before April 11, 2009 by payment in full of all amounts due and owing under the Loan as of April 1, 2009, plus any and all accrued Past Due Rate interest, late charges, costs and fees (including attorneys' fees): (i) Bank of America, on behalf of itself and each of the Lenders, is entitled to the exercise of the Lenders' rights and remedies under the Loan Documents, at law and in equity, including, without limitation, acceleration of the indebtedness due under the Loan Documents and the imposition of Past Due Rate interest, late charges and other charges recoverable as a consequence of the Default (including attorneys' fees and costs); and (ii) Bank of America, on behalf of itself and the other Lenders, is entitled to enforce the Lenders' rights and remedies relating to the collateral, including, without limitation, appointment of receiver and foreclosure. Please contact me for payment information relating to the Loan.

Nothing in this letter or in Lenders' application of such payments is or shall be deemed to be a waiver, election or estoppel of any rights, remedies, defenses and objections available to the Lenders and their affiliated entities under the Loan Agreement and related Loan Documents and under applicable laws and in equity in connection with any default by Borrower, and Bank of America, on behalf of itself and the other Lenders, expressly reserves all such rights, remedies, defenses and objections, including with respect to any defaults now or hereafter in existence, whether known or unknown, to any of the Lenders, that are not referenced in this letter.

Opus West Corporation ("Guarantor") guaranteed the payment and performance of the Loan and Borrower's obligations under the Loan Documents under that certain Guaranty Agreement (The Shoppes at Chino Hills, CA -Construction Loan) dated May 16, 2007 (the

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 23 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation April 6, 2009 Page 3

"Guaranty"). Demand is hereby made on Guarantor for prompt payment or performance of Borrower's obligations under the Default described herein.

Please give this matter your immediate attention.

Very truly yours,

QUARLES & BRADY LLP

Lori L. Winkelman

cc: Casey Carpenter, Bank of America
Daniel Denton, Esq., Bank of America
Brian Sirower, Esq.

Via UPS Overnight

Opus West Corporation 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department

Via UPS Overnight

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 24 of 33



One Renaissance Square Two North Central Avenue Phoenix, Arizona 85004-2391 Tel 602.229.5200 Fax 602.229.5690 www.quarles.com Attorneys at Law in: Phoenix and Tucson, Arizona Naples, Florida Chicago, Illinois Mihvaukee and Madison, Wisconsin

April 9, 2009

DEFAULT NOTICE

VIA UPS OVERNIGHT

Shoppes At Chino Hills, Inc. 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Senior Vice President, Real Estate Finance and Sales

Opus West Corporation 2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

RE: \$110,000,000 Loan (the "Loan") made by Bank of America, N.A., a national banking association, individually as a Lender and as Administrative Agent ("Bank of America"), and certain other financial institutions as Lenders (together with Bank of America, the "Lenders") to Shoppes At Chino Hills, Inc., a Minnesota corporation ("Borrower")

Ladies and Gentlemen:

The purpose of this letter is to provide you with notice of the occurrence of an Event of Default (hereinafter, the "Event of Default") under Section 5.1(c) of the Construction Loan Agreement dated May 16, 2007 between Borrower and Lenders (as amended, supplemented, or modified from time to time, the "Loan Agreement"), and under Section 3(a)(ii) of certain Construction Deed Of Trust Notes dated as of even date therewith, made by Borrower and delivered to each Lender (as amended, supplemented, or modified from time to time, the "Notes"). The Loan Agreement, Notes, and all other agreements, documents and instruments evidencing, securing or otherwise relating to the Loan are sometimes referred to individually and collectively herein as the "Loan Documents". Borrower has informed Bank of America that, for the fiscal quarter ended December 31, 2008, Opus West Corporation ("Guarantor") will fail to satisfy the Total Liabilities to Tangible Net Worth requirement of not exceeding 6.0 to 1.0 on a consolidated basis as required under Section 13(B)(iii) of that certain Guaranty Agreement (The Shoppes at Chino Hills, CA-Construction Loan) dated May 16, 2007 (as amended, supplemented, or modified from time to time, the "Guaranty"). Guarantor's failure to meet the requirements of Section 13(B)(iii) of the Guaranty constitutes an immediate Event of Default under Section 5.1(c) of the Loan Agreement and Section 3(a)(iii) of the Notes.

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 25 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation April 9, 2009 Page 2

Notwithstanding the existence of the above-described Event of Default, or any additional defaults that may arise under the Loan Documents, Bank of America, on behalf of itself and the other Lenders, reserves the right to elect whether or not to fund any presently pending or future draw requests during the existence of any default or Event of Default. To the extent Bank of America or any of the other Lenders elects to fund any such draw request, neither such election nor the actual funding thereof does or shall: constitute a commitment to lend or to modify the Loan; constitute an offer to extend credit or modify the existing Loan terms; constitute a waiver by any Lender of the Event of Default or performance or cure of the Event of Default by Borrower; constitute an amendment of any term or condition of any Loan Document; create an obligation of any Lender to agree to, or to negotiate or consider an agreement to, any waiver of any obligation or default by Borrower under any Loan; or directly or indirectly impair or otherwise affect any of the Lenders' rights, interests or remedies with respect to the Loan.

Bank of America, on behalf of itself and the other Lenders, hereby reserves the right to accept payments made after the date of this letter even though such payments are not offered as, or are insufficient to, cure the Event of Default described herein or reinstate the Loan pursuant to the provisions of the law governing the jurisdiction in which the project is located. Each of the Lenders may apply such payments in the order provided in the Loan Agreement or other Loan Documents.

As a result of the Event of Default: (i) Bank of America, on behalf of itself and each of the Lenders, is entitled to the exercise of the Lenders' rights and remedies under the Loan Documents, at law and in equity, including, without limitation, acceleration of the indebtedness due under the Loan Documents and the imposition charges recoverable as a consequence of the Event of Default (including attorneys' fees and costs); and (ii) Bank of America, on behalf of itself and the other Lenders, is entitled to enforce the Lenders' rights and remedies relating to the collateral, including, without limitation, appointment of receiver and foreclosure.

Nothing in this letter or in Lenders' application of such payments is or shall be deemed to be a waiver, election or estoppel of any rights, remedies, defenses and objections available to the Lenders and their affiliated entities under the Loan Agreement and related Loan Documents and under applicable laws and in equity in connection with any default by Borrower, and Bank of America, on behalf of itself and the other Lenders, expressly reserves all such rights, remedies, defenses and objections, including with respect to any defaults now or hereafter in existence, whether known or unknown, to any of the Lenders, that are not referenced in this letter.

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 26 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation April 9, 2009 Page 3

Please give this matter your immediate attention.

Very truly yours,

QUARLES & BRADY LLP

Lori L. Winkelman

cc: Casey Carpenter, Bank of America Daniel Denton, Esq., Bank of America Brian Sirower, Esq.

Via UPS Overnight

Opus West Corporation 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department

Via UPS Overnight

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 27 of 33



One Renaissance Square Two North Central Avenue Phoenix, Arizona 85004-2391 Tel 602.229.5200 Fax 602.229.5690 www.quarles.com Attorneys at Law in: Phoenix and Tucson, Arizona Naples, Florida Chicago, Illinois Mihvaukee and Madison, Wisconsin

May 7, 2009

DEFAULT NOTICE

VIA UPS OVERNIGHT

Shoppes At Chino Hills, Inc. 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267

Attn: Senior Vice President, Real Estate Finance and Sales

Opus West Corporation 2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

RE: \$110,000,000 Loan (the "Loan") made by Bank of America, N.A., a national banking association, individually as a Lender and as Administrative Agent ("Bank of America"), and certain other financial institutions as Lenders (together with Bank of America, the "Lenders") to Shoppes At Chino Hills, Inc., a Minnesota corporation ("Borrower")

Ladies and Gentlemen:

The purpose of this letter is to provide you with notice of the occurrence of a Default (hereinafter, the "Specified Default") under: (i) Section 5.1(c) of the Construction Loan Agreement dated May 16, 2007 between Borrower and Lenders, as supplemented by two letter agreements dated May 16, 2007, and as amended by that certain letter agreement dated March 21, 2008 (the "Loan Agreement"), and (ii) under Section 3(a) of each of the Construction Deed Of Trust Notes dated as of even date therewith executed by Borrower payable to the order of each Lender in the amount of its respective Commitment (the "Notes"). The Loan Agreement, Notes, and all other agreements, documents and instruments evidencing, securing or otherwise relating to the Loan are sometimes referred to individually and collectively herein as the "Loan Documents". The Default occurred as the result of Borrower's failure to make the installment payment of interest due and payable on May 1, 2009 as required under the Loan Agreement and the Notes. Late charges (as described in and calculated according to Section 2.10 of the Loan Agreement) will apply to the defaulted amounts. Interest at the Past Due Rate (provided in and calculated according to Section 2.8 of the Loan Agreement) commenced accruing from and after May 2, 2009 on any amounts outstanding as of such date.

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 28 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation May 7, 2009 Page 2

Notwithstanding the existence of the above-described Specified Default, or any additional defaults that may arise under the Loan Documents, Bank of America, on behalf of itself and the other Lenders, reserves the right to elect whether or not to fund any presently pending or future draw requests during the existence of any default or Event of Default. To the extent Bank of America or any of the other Lenders elects to fund any such draw request, neither such election nor the actual funding thereof does or shall: constitute a commitment to lend or to modify the Loan; constitute an offer to extend credit or modify the existing Loan terms; constitute a waiver by any Lender of the Specified Default or performance or cure of the Specified Default by Borrower; constitute an amendment of any term or condition of any Loan Document; create an obligation of any Lender to agree to, or to negotiate or consider an agreement to, any waiver of any obligation or default by Borrower under any Loan; or directly or indirectly impair or otherwise affect any of the Lenders' rights, interests or remedies with respect to the Loan.

Bank of America, on behalf of itself and the other Lenders, hereby reserves the right to accept payments made after the date of this letter even though such payments are not offered as, or are insufficient to, cure the Specified Default described herein or reinstate the Loan pursuant to the provisions of the law governing the jurisdiction in which the project is located. Each of the Lenders may apply such payments in the order provided in the Loan Agreement or other Loan Documents.

As a result of the Specified Default: (i) Bank of America, on behalf of itself and each of the Lenders, is entitled to the exercise of the Lenders' rights and remedies under the Loan Documents, at law and in equity, including, without limitation, acceleration of the indebtedness due under the Loan Documents and the imposition of Past Due Rate interest, late charges and other charges recoverable as a consequence of the Default (including attorneys' fees and costs); and (ii) Bank of America, on behalf of itself and the other Lenders, is entitled to enforce the Lenders' rights and remedies relating to the collateral, including, without limitation, appointment of receiver and foreclosure. Please contact me for payment information relating to the Loan.

Nothing in this letter or in Lenders' application of such payments is or shall be deemed to be a waiver, election or estoppel of any rights, remedies, defenses and objections available to the Lenders and their affiliated entities under the Loan Agreement and related Loan Documents and under applicable laws and in equity in connection with any default by Borrower, and Bank of America, on behalf of itself and the other Lenders, expressly reserves all such rights, remedies, defenses and objections, including with respect to any defaults now or hereafter in existence, whether known or unknown, to any of the Lenders, that are not referenced in this letter.

Opus West Corporation ("Guarantor") guaranteed the payment and performance of the Loan and Borrower's obligations under the Loan Documents under that certain Guaranty Agreement (The Shoppes at Chino Hills, CA -Construction Loan) dated May 16, 2007 (the

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 29 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation May 7, 2009 Page 3

"Guaranty"). Demand is hereby made on Guarantor for prompt payment or performance of Borrower's obligations under the Default described herein.

Please give this matter your immediate attention.

Very truly yours,

QUARLES & BRADY LLP

Lori L. Winkelman

cc: Casey Carpenter, Bank of America
Daniel Denton, Esq., Bank of America
Brian Sirower, Esq.

Via UPS Overnight

Opus West Corporation 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department

Via UPS Overnight

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 30 of 33



One Renaissance Square Two North Central Avenue Phoenix, Arizona 85004-2391 Tel 602.229.5200 Fax 602.229.5690 www.quarles.com

Attorneys at Law in: Phoenix and Tucson, Arizona Naples, Florida Chicago, Illinois Mikuaukee and Madison, Wisconsin

June 16, 2009

DEFAULT NOTICE

VIA UPS OVERNIGHT

Shoppes At Chino Hills, Inc. 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Senior Vice President, Real Estate Finance and Sales

Opus West Corporation 2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

RE: \$110,000,000 Loan (the "Loan") made by Bank of America, N.A., a national banking association, individually as a Lender and as Administrative Agent ("Bank of America"), and certain other financial institutions as Lenders (together with Bank of America, the "Lenders") to Shoppes At Chino Hills, Inc., a Minnesota corporation ("Borrower")

Ladies and Gentlemen:

The purpose of this letter is to provide you with notice of the occurrence of a Default (hereinafter, the "Specified Default") under: (i) Section 5.1(a) of the Construction Loan Agreement dated May 16, 2007 between Borrower and Lenders, as supplemented by two letter agreements dated May 16, 2007, and as amended by that certain letter agreement dated March 21, 2008 (the "Loan Agreement"), and (ii) under Section 3(a) of each of the Construction Deed Of Trust Notes dated as of even date therewith executed by Borrower payable to the order of each Lender in the amount of its respective Commitment (the "Notes"). The Loan Agreement, Notes, and all other agreements, documents and instruments evidencing, securing or otherwise relating to the Loan are sometimes referred to individually and collectively herein as the "Loan Documents". The Specified Default occurred as the result of Borrower's failure to pay the full amount of principal and interest due and owing on the Maturity Date (as defined in the Loan Agreement), and the conditions for extending the Maturity Date set forth in Section 2.11(b) of the Loan Agreement have not been satisfied. Late charges (as described in and calculated according to Section 2.10 of the Loan Agreement) will apply to the defaulted amounts. Interest at the Past Due Rate (provided in and calculated according to Section 2.8 of the Loan Agreement) will continue to accrue on any amounts outstanding.

Shoppes At Chino Hills, Inc. Opus West Corporation June 16, 2009 Page 2

Notwithstanding the existence of the above-described Specified Default, or any additional defaults that may arise under the Loan Documents, Bank of America, on behalf of itself and the other Lenders, reserves the right to elect whether or not to fund any presently pending or future draw requests during the existence of any default or Event of Default. To the extent Bank of America or any of the other Lenders elects to fund any such draw request, neither such election nor the actual funding thereof does or shall: constitute a commitment to lend or to modify the Loan; constitute an offer to extend credit or modify the existing Loan terms; constitute a waiver by any Lender of the Specified Default or performance or cure of the Specified Default by Borrower; constitute an amendment of any term or condition of any Loan Document; create an obligation of any Lender to agree to, or to negotiate or consider an agreement to, any waiver of any obligation or default by Borrower under any Loan; or directly or indirectly impair or otherwise affect any of the Lenders' rights, interests or remedies with respect to the Loan.

Bank of America, on behalf of itself and the other Lenders, hereby reserves the right to accept payments made after the date of this letter even though such payments are not offered as, or are insufficient to, cure the Specified Default described herein or reinstate the Loan pursuant to the provisions of the law governing the jurisdiction in which the project is located. Each of the Lenders may apply such payments in the order provided in the Loan Agreement or other Loan Documents.

As a result of the Specified Default: (i) Bank of America, on behalf of itself and each of the Lenders, is entitled to the exercise of the Lenders' rights and remedies under the Loan Documents, at law and in equity, including, without limitation, acceleration of the indebtedness due under the Loan Documents and the imposition of Past Due Rate interest, late charges and other charges recoverable as a consequence of the Specified Default (including attorneys' fees and costs); and (ii) Bank of America, on behalf of itself and the other Lenders, is entitled to enforce the Lenders' rights and remedies relating to the collateral, including, without limitation, appointment of receiver and foreclosure. Please contact me for payment information relating to the Loan.

Nothing in this letter or in Lenders' application of such payments is or shall be deemed to be a waiver, election or estoppel of any rights, remedies, defenses and objections available to the Lenders and their affiliated entities under the Loan Agreement and related Loan Documents and under applicable laws and in equity in connection with any default by Borrower, and Bank of America, on behalf of itself and the other Lenders, expressly reserves all such rights, remedies, defenses and objections, including with respect to any defaults now or hereafter in existence, whether known or unknown, to any of the Lenders, that are not referenced in this letter.

Opus West Corporation ("Guarantor") guaranteed the payment and performance of the Loan and Borrower's obligations under the Loan Documents under that certain Guaranty Agreement (The Shoppes at Chino Hills, CA -Construction Loan) dated May 16, 2007 (the

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 32 of 33

Shoppes At Chino Hills, Inc. Opus West Corporation June 16, 2009 Page 3

"Guaranty"). Demand is hereby made on Guarantor for prompt payment or performance of Borrower's obligations under the Specified Default described herein.

Please give this matter your immediate attention.

Very truly yours,

QUARLES & BRADY LLP

Lori L. Winkelman

cc: Casey Carpenter, Bank of America
Daniel Denton, Esq., Bank of America
Brian Sirower, Esq.

Via UPS Overnight

Opus West Corporation 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Legal Department

Via UPS Overnight

Case 09-34356-hdh11 Claim 82-1 Part 8 Filed 11/09/09 Desc Attachments to Addendum Part 7 of 7 Page 33 of 33



One Renaissance Square Two North Central Avenue Phoenix, Arizona 85004-2391 Tel 602.229.5200 Fax 602.229.5690 www.quarles.com Attorneys at Law in: Phoenix and Tucson, Arizona Naples, Florida Chicago, Illinois Milwaukee and Madison, Wisconsin

May 7, 2009

DEFAULT NOTICE

VIA UPS OVERNIGHT

Shoppes At Chino Hills, Inc. 2555 East Camelback, Suite 800 Phoenix, Arizona 85016-9267 Attn: Senior Vice President, Real Estate Finance and Sales

Opus West Corporation 2555 East Camelback Road, Suite 800 Phoenix, Arizona 85016-9267

RE: \$110,000,000 Loan (the "Loan") made by Bank of America, N.A., a national banking association, individually as a Lender and as Administrative Agent ("Bank of America"), and certain other financial institutions as Lenders (together with Bank of America, the "Lenders") to Shoppes At Chino Hills, Inc., a Minnesota corporation ("Borrower")

Ladies and Gentlemen:

The purpose of this letter is to provide you with notice of the occurrence of a Default (hereinafter, the "Specified Default") under: (i) Section 5.1(c) of the Construction Loan Agreement dated May 16, 2007 between Borrower and Lenders, as supplemented by two letter agreements dated May 16, 2007, and as amended by that certain letter agreement dated March 21, 2008 (the "Loan Agreement"), and (ii) under Section 3(a) of each of the Construction Deed Of Trust Notes dated as of even date therewith executed by Borrower payable to the order of each Lender in the amount of its respective Commitment (the "Notes"). The Loan Agreement, Notes, and all other agreements, documents and instruments evidencing, securing or otherwise relating to the Loan are sometimes referred to individually and collectively herein as the "Loan Documents". The Default occurred as the result of Borrower's failure to make the installment payment of interest due and payable on May 1, 2009 as required under the Loan Agreement and the Notes. Late charges (as described in and calculated according to Section 2.10 of the Loan Agreement) will apply to the defaulted amounts. Interest at the Past Due Rate (provided in and calculated according to Section 2.8 of the Loan Agreement) commenced accruing from and after May 2, 2009 on any amounts outstanding as of such date.

Northern District of Texas Claims Register

09-34356-hdh11 Opus West Corporation

Judge: Harlin DeWayne Hale

Chapter: 11

Office: Dallas

Last Date to file claims: 11/09/2009

Trustee:

Last Date to file (Govt):

Creditor: (12791347)BANK OF AMERICA, N.A. c/o Casey Carpenter

Original Filed Date: 11/09/2009 Original Entered Date: 11/09/2009

Claim No: 82

Status: Filed by: CR

Vice President Real Estate Special Assets 333 S. Hope St., 11th Floor Los Angeles, CA 90071-

Entered by: Richter, Katharine

Modified:

1406

Unsecured claimed: \$106807380.80 claimed: \$106807380.80

History:

Total

Details

82-1 11/09/2009 Claim #82 filed by BANK OF AMERICA, N.A., total amount claimed:

\$106807380.8 (Richter, Katharine)

Description: (82-1) Shoppes at Chino Hills

Remarks: (82-1) But see Addendum for possible secured status

Claims Register Summary